

### Inquiry into powers of attorney

# Final report of the Victorian Parliament Law Reform Committee

### August 2010

ORDERED TO BE PRINTED

by Authority Victorian Government Printer

Parliamentary Paper

No. 352, Session 2006-2010

### Parliament of Victoria, Australia

Inquiry into powers of attorney

**Law Reform Committee** 

ISBN - 978-0-9807201-2-9

ISBN - 978-0-9807201-3-6 (electronic)

Printed on recycled paper

This report is also available at www.parliament.vic.gov.au/lawreform

### **Committee members**

Mr Johan Scheffer MLC Chair

Mr Robert Clark MP

Deputy Chair

Mr Colin Brooks MP

Mr Luke Donnellan MP

Mr Martin Foley MP

Mrs Jan Kronberg MLC

Mrs Heidi Victoria MP Member from 4 June 2009

Mr Edward O'Donohue MLC Member until 2 June 2009

### **Staff**

Ms Kerryn Riseley *Executive Officer* 

Ms Kerry Harrison Research Officer

Ms Liana Levin Research Officer from 19 April 2010 to 30 June 2010

> Ms Vathani Shivanandan Research Officer from 30 June 2010

> > Ms Helen Ross-Soden Administration Officer

Ms Yardena Lankri Legal Policy Intern

### **Functions of the Law Reform Committee**

The functions of the Law Reform Committee are set out in section 12 of the *Parliamentary Committees Act 2003* (Vic). That section states:

- 1) The functions of the Law Reform Committee are, if so required or permitted under this Act, to inquire into, consider and report to the Parliament on any proposal, matter or thing concerned with
  - a) legal, constitutional or parliamentary reform
  - b) the administration of justice
  - c) law reform.

### Terms of reference

The following reference was made by the Legislative Assembly on 4 December 2008:

To the Law Reform Committee – for inquiry, consideration and report no later than 31 December 2009\* on law reforms aimed at streamlining and simplifying power of attorney documents to enable more Victorians to plan for the future financial, lifestyle and healthcare needs – specifically the Committee is asked to:

- a) consider the differing formality requirements and terminology, and coverage of the power of attorney documents, governed by the *Instruments Act 1958* (Vic) and the *Guardianship and Administration Act 1986* (Vic)
- b) establish whether the donor of a power of attorney has capacity to create a legally enforceable document and differing execution requirements and the different tests that apply
- c) clarify the powers granted by the donor when making a power of attorney
- d) examine ways of minimising abuse in relation to the execution of and exercise of powers under powers of attorney documents
- e) consider the issue of legal capacity in the context of when an enduring power of attorney is executed and activated
- f) advise on the need for adopting potential safeguards such as the registration of documents (voluntary or mandatory).

<sup>\*</sup> The reporting date was extended to 31 August 2010 by resolution of the Legislative Assembly on 13 August 2009.

### **Contents**

List o	of figures		xiii
List o	of case stu	idies	xiv
List o	of recomn	nendations	XV
Gloss	sary		XXXV
Chai	r's forewo	ord	xli
Exec	utive sum	mary	.xliii
		roduction	
1.1		pe of this Inquiry	
	1.1.1	Terms of reference	
	1.1.2	What types of powers of attorney is the Committee reviewing?	2
	1.1.3	What is 'capacity'?	4
1.2		text for this Inquiry	
	1.2.1	Social and demographic changes	
	1.2.2	A rights-based framework	
	1.2.3	An area in transition	
1.3		t of the Inquiry	
1.4	Outline	of this report	8
Chap	oter 2: Po	wers of attorney in Victoria – The current landscape	11
2.1		e powers of attorney?	
	2.1.1	The history of powers of attorney	
	2.1.2	Powers of attorney in Victoria	
		General (non-enduring) powers of attorney	
		Enduring powers of attorney (financial) Enduring powers of attorney (guardianship)	
		The role of the Victorian Civil and Administrative Tribunal.	
		The role of the Office of the Public Advocate	
2.2	What ha	appens if a person does not have enduring powers of attorney?	
	2.2.1	Appointment of guardians and administrators	
	2.2.2	Semi-formal arrangements	
	2.2.3	Informal arrangements	19
2.3	The use	of powers of attorney	
	2.3.1	How widely are powers of attorney used?	
	2.3.2	The benefits of powers of attorney	
	2.3.3	Barriers to making powers of attorney	
		Lack of awareness and understanding  Complexity	
		Psychological barriers	
		Cost	
		Family dynamics	
		Fear of abuse	
2.4	The abu	se of powers of attorney	26
	2.4.1	The level of abuse of powers of attorney	26
	2.4.2	How are powers of attorney abused?	
	2.4.3	Who are the victims of abuse of powers of attorney?	
2.5	2.4.4	Who are the perpetrators of abuse of powers of attorney?	
2.5	w eignii	ng up the benefits and risks of powers of attorney	31

Chapt	er 3: A leg	gislative framework for powers of attorney	33
3.1	What sho	uld the legislative framework look like?	33
	3.1.1	Consolidating powers of attorney legislation	
	3.1.2	A national approach to powers of attorney?	
3.2	Guiding p	principles for powers of attorney legislation	38
	3.2.1	A principles-based approach	
	3.2.2	Principles articulating rights in the International Convention on	
		the Rights of Persons with Disabilities	41
3.3	Simplifyi	ng terminology	42
	3.3.1	Naming the documents and powers	42
	3.3.2	Naming the parties to a power of attorney	44
3.4	Keeping t	the legislation relevant and effective into the future	47
Chapt	er 4: Crea	ating a power of attorney	51
4.1		ng formal requirements while protecting against abuse	
4.2		creating powers of attorney	
7.2	4.2.1	Issues with multiple power of attorney forms	
	4.2.2	Consolidating power of attorney forms	
	1.2.2	One form for creating all enduring powers of attorney	
		A separate general (non-enduring) power of attorney form?	
		A single form for creating enduring powers of attorney	
		guardianship and medical treatment	56
		Consolidating power of attorney forms – The Committee's	
		view	56
	4.2.3	Making power of attorney forms user-friendly	58
		Length	
		Language	59
		Format	59
		Privacy	60
		Developing new power of attorney forms	
		Making power of attorney forms user-friendly – The	
	4.0.4	Committee's view	
	4.2.4	Should it be mandatory to use the standard forms?	
	4.2.5	Where should the prescribed forms be located?	
	4.2.6	Making power of attorney forms widely available	
4.3		ng a power of attorney document	
	4.3.1	Consistent witnessing requirements or a graduated system?	
	4.3.2	What is the role of witnesses?	66
		What is the witnesses' role in assessing a principal's	67
		capacity?	07
		How do witnesses determine whether a principal signs a power of attorney freely and voluntarily?	71
		The role of witnesses – The Committee's view	
	4.3.3	Who should witness power of attorney documents?	
	4.3.3	The number and qualifications of witnesses	
		Who should not be able to witness power of attorney	13
		documents?	77
		Can a representative's employees witness power of	, ,
		attorney documents?	79
		Proving the witness's authority	

	4.3.4	Supporting witnesses	
		Information and education for witnesses	
		Access to interpreters  Protecting witnesses	
4.4	Informir		
4.4	4.4.1	ng and supporting principals  Information and educational resources for principals	
	4.4.2	Access to legal advice	
4.5		nce of appointment by representatives	
4.6		appens if a power of attorney document is not correctly executed?	
4.7		nould an enduring power of attorney commence?	
4.8	Revokin	g powers of attorney	93
	4.8.1	Should a principal be able to revoke a power of attorney	y
	4.8.2	orally?	
		attorney?	
	4.8.3	Should a principal have to inform representatives and third	
		parties that a power of attorney has been revoked?	
	4.8.4	What happens if VCAT appoints a guardian or administrator?	
4.9		powers of attorney up to date	
4.10	•	ng copies of power of attorney documents	
4.11	Making	sure power of attorney documents can be located when needed	102
4.12	The port	tability of powers of attorney – Mutual recognition	104
Chap	ter 5: Caj	pacity	107
5.1	Why is o	capacity important to this Inquiry?	107
5.2	A consis	stent approach to capacity	108
5.3	Starting	from a presumption of capacity	109
5.4	Defining	g capacity	110
5.5		ng rights when assessing capacity - Principles to guide capacity	
		ents	113
	5.5.1 5.5.2	Capacity is decision-specific	
	5.5.3	Capacity should not be assumed based on a person's	
	0.0.0	appearance	
	5.5.4	Capacity assessments should focus on a person's ability to	
		make a decision not the outcome of a decision	
	5.5.5	People should be supported to make their own decisions where	
	5.5.6	possible	
	3.3.0	view	
5.6	Who sho	ould assess capacity?	
	5.6.1	When is a medical assessment of capacity required?	
		Medical assessment of capacity to create a power of	
		attorney	121
		A medical assessment of capacity at time of activation?	
	5.6.2	Medical assessment of capacity – The Committee's view Who should conduct medical assessments of capacity?	
57			
5.7	vv nat ev	idence is required to activate an enduring power of attorney?	120

5.8	_	g guidance and support to those assessing capacity
	5.8.1	Resources to assist with capacity assessments
		New South Wales Capacity toolkit
		Law Society of New South Wales – Guide for solicitors 128
		United Kingdom Mental Capacity Act 2005 Code of
		practice
		Resources to assist with capacity assessments – The
		Committee's view
	5.8.2	Educating capacity assessors
	5.8.3	Tools for assessing capacity
		Six step capacity assessment process
		Mini mental state examination
		Other tools
		Tools for assessing capacity – The Committee's view 134
5.9	A gap in	the law: people who cannot make enduring powers of attorney
	and canno	ot obtain a guardianship order
Chapt	er 6: The	role of representatives
6.1		ing the contribution of representatives
6.2	Appointin	ng appropriate representatives
	6.2.1	What skills do representatives need?
	6.2.2	Helping principals appoint appropriate representatives
	6.2.3	Should there be restrictions on who can be appointed as a
		representative?
	6.2.4	Should there be restrictions on the number of principals for
		whom an individual representative can act?
6.3		ng multiple and alternative representatives
	6.3.1	How many representatives should a principal be able to
		appoint?
	6.3.2	How can multiple representatives make decisions?
	6.3.3	What happens if multiple representatives cannot agree?
	6.3.4	Avoiding uncertainty where there are multiple representatives 148
		What happens if the principal does not specify how
		multiple representatives are appointed?
		What happens if the power of one joint representative
	6.3.5	ends?
	0.5.5	able to appoint?
	6.3.6	When should alternative representatives be able to act?
6.4		ers of representatives
0.4	6.4.1	What powers do representatives have?
	6.4.2	Defining representatives' powers
	6.4.3	Clarifying and extending representatives' powers
	0.4.5	Power to restrict a principal's visitors
		Power to access a principal's will
		Power to access personal and health information
		Power to act as a trustee
		Power to lodge a Queen's caveat
		Power to refuse medical treatment
	6.4.4	Principals' ability to limit representatives' powers
	6.4.5	Which representative's decisions take precedence? 164

6.5	The dutie 6.5.1	s of representatives	
	6.5.2	Providing clarity about representatives' duties	
	6.5.3	How should representatives make decisions?	
		The role of principals' wishes and 'best interests' in	
		decision making	
		Principles to guide decision making	172
6.6	Increasing	g representatives' understanding of their role, powers and duties.	175
	6.6.1	How well do representatives understand their role, powers and duties?	
	6.6.2	Providing information and support for representatives	
		Information about role, powers and duties on powers of attorney forms	
		Educational resources for representatives	177
		Educating representatives when they accept their appointment	
		Educating and supporting representatives – The Committee's view	
6.7	Should re	presentatives be entitled to be paid?	180
Chapt		ecting principals from abuse	
7.1		reporting and investigating abuse of powers of attorney	
7.1	7.1.1	Why is abuse difficult to detect?	
	7.1.2	Why is abuse not reported?	
	7.1.3	Increasing the capacity of service providers to detect and	
		report abuse	
		Mandatory reporting	
		Protecting whistleblowers	
	7.1.4	Educating the general community about abuse	
	7.1.5	Clarifying where to report abuse	
	7.1.6	Investigating abuse	
7.2		to prevent abuse of validly executed powers of attorney	
	7.2.1 7.2.2	Balancing protection and utility	
	7.2.2	Reducing abuse – The evidence gaps	
	1.2.3	Annual statements	
		Auditing	
		Reporting and auditing requirements – The Committee's	190
		view	197
	7.2.4	Personal monitors.	
	7.2.5	Regulating gifts and other benefits	
	7.2.6	Regulating dealings with major assets	
	7.2.7	Protecting gifts made under a will	
	7.2.8	Notifying activation of an enduring power of attorney	
		document	
	7.2.9	Performance bonds	205
7.3	Criminal	offences	
	7.3.1	Specific offences in the Crimes Act	
	7.3.2	Specific offences in the powers of attorney legislation	208

7.4		g redress for victims of abuse	
7.5	VCAT's 7.5.1	powers	•
	7.5.2	When should VCAT be able to revoke a representative's appointment?	
	7.5.3 7.5.4	Who should be able to apply to VCAT?	217
	7.5.5	powers of attorney that have been revoked?	
7.6		applications about enduring powers of attorney be able to be in other courts?	
7.7	Alternati	ive dispute resolution	220
Chap	ter 8: A r	egister of power of attorney documents?	225
8.1		efits and risks of a register of power of attorney documents	
	8.1.1	The benefits of registration	
		Will a register protect principals from abuse?	
	8.1.2	The risks of registration	
	8.1.3	Weighing the benefits and risks: Should a registration system	
		be implemented?	
8.2		ould a registration system look like?	
	8.2.1	Should registration be national or state-based?	
	8.2.2	Should registration be voluntary or mandatory?	
	8.2.3	When should a power of atterney do agree the registered?	
	8.2.4 8.2.5	When should a power of attorney document be registered?	
	0.2.3	•	238
	8.2.6	Should a representative have to notify the registration body	
	0.2.0	when an enduring power of attorney is activated?	
	8.2.7	Accessing the register	
		What information should be included in the register?	
		Who should have access to the register?	
		How should the register be searchable?	
		Accessing the register – The Committee's view	245
	8.2.8	Registration costs	
	8.2.9	Where should the register be located?	
	8.2.10	Should a registration body play a role in checking that	
	0.2.11	documents are correctly executed?	
	8.2.11	Who should be entitled to notice of registration?	
	8.2.12	Should the registration body play a role in educating	
		representatives and monitoring their activities?	
		representatives	
		The role of the registration body in monitoring	
		representatives' activities	
	8.2.13	What is the effect of registration in another jurisdiction?	
8.3	Promotin	ng the registration system	

Chapt	ter 9: Pro	moting powers of attorney	257
9.1	Current a	activities promoting powers of attorney	257
	9.1.1	The Office of the Public Advocate	
	9.1.2	Council on the Ageing	258
	9.1.3	Seniors Rights Victoria	
	9.1.4	Other organisations providing education about powers of	
		attorney	259
9.2	A coordi	nated approach to promoting powers of attorney	260
9.3	Increasin	g community awareness and understanding of powers of	
	attorney.		261
	9.3.1	General community education	
		What are the benefits of general community education	
		about powers of attorney?	262
	0.2.2	What kind of community education campaign is needed?	
	9.3.2	Educating groups with low levels of awareness, understanding	
		and use	
		People from culturally and linguistically diverse	
		backgrounds	
		Young people	
		Members of the Aboriginal community	
		Other groups	
9.4	Educating	g lawyers about powers of attorney	273
9.5	Educating	g the health and community sectors	277
9.6	Increasin	g the recognition and acceptance of powers of attorney	280
	9.6.1	Government agencies	280
	9.6.2	Financial institutions	283
9.7	Educating	g about changes to the law	284
Chapt	ter 10: Co	nclusion	287
Apper	ndix A: Li	ist of submissions	289
Apper	ndix B: Li	st of witnesses	293
Apper	ndix C: Li	ist of events attended	301
Apper	ndix D: Pe	ersons who may witness affidavits and statutory declarations	.303
Biblio	graphy		307
			321

# List of figures

Figure 1	General principles: Queensland Powers of Attorney Act	40
Figure 2	The six step capacity assessment process	132
Figure 3	Defining representatives' powers: ACT Powers of Attorney Act	156
Figure 4	Specific responsibilities of representatives	166
Figure 5	General principles to guide decision making by representatives	169

### List of case studies

Case study 1	She 'was unable to explain what type of power of attorney document she had executed or when'
Case study 2	'He was aware that she did not know what she was signing' 46
Case study 3	Unsuitable representative did not act in principal's best interests 48
Case study 4	The enduring power of attorney was signed 'well after he was deemed not to have capacity'
Case study 5	'He did not know what he was signing'
Case study 6	Brian was persuaded into signing an enduring power of attorney70
Case study 7	'He believes that Serge has been pressured into signing the powers of attorney'
Case study 8	The power of attorney was 'worthless'
Case study 9	He was unable to return home as his house had been emptied out
Case study 10	'Her cognitive impairment was not readily apparent' 123
Case study 11	Twelve months after her stroke she had regained capacity to appoint a representative
Case study 12	The representative was a stranger to the principal
Case study 13	The representative did not have the power to restrict visitors 158
Case study 14	While the principal was ill his son arranged for him to sign a power of attorney
Case study 15	The son used his mother's money for his household expenses 181
Case study 16	'A significant proportion of her assets had been transferred by the attorney to himself'
Case study 17	Internet scam victim stole from parents
Case study 18	Solicitor uses power of attorney to steal from 'vulnerable' client 222
Case study 19	Administrator to investigate money withdrawn by representative
Case study 20	The representatives had not protected her interests
Case study 21	'The solicitor should have advised them to create an enduring power of attorney'
Case study 22	A bank's failure to recognise a general power of attorney 282

## List of recommendations

Recommendation 1: A Powers of Attorney Act
The Committee recommends the Victorian Government draft a Powers of Attorney Act which contains all laws about general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship).
Recommendation 2: National harmonisation of power of attorney laws 38
The Committee recommends the Victorian Government, through the Standing Committee of Attorneys-General, actively promote and support national harmonisation of power of attorney laws.
Recommendation 3: Founding principles of the Powers of Attorney Act 42
The Committee recommends the Powers of Attorney Act contain a statement of principles that must be applied by all those exercising powers or functions under the Act in relation to a person with impaired decision-making capacity. The principles should include that those persons must exercise their powers and functions in relation to a principal with impaired decision-making capacity:
<ul> <li>in a way that is as least restrictive of the principal's freedom of decision and action as is possible in the circumstances</li> </ul>
• so that the principal is provided with appropriate support to allow him or her to exercise his or her legal capacity to the maximum extent possible.
Recommendation 4: Consistent names for documents and powers 44
The Committee recommends the Powers of Attorney Act call the power of attorney documents and the powers created under them 'general (non-enduring) power of attorney', 'enduring power of attorney (financial)' and 'enduring power of attorney (guardianship)'.
Recommendation 5: Consistent names for parties to a power of attorney 47
The Committee recommends the Powers of Attorney Act:
a) call a person who creates all types of powers of attorney a 'principal'
b) call a person appointed to exercise powers under all types of powers of attorney a 'representative'.
Recommendation 6: Reviewing the Powers of Attorney Act 47
The Committee recommends the Victorian Government commission an evaluation of the new Powers of Attorney Act after it has been in operation for five years, to determine whether it has been effective in meeting its objectives

community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design  b) widely test the draft power of attorney forms  c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.  Recommendation 10: Producing power of attorney forms and accompanying information in community languages	document powers	mmittee recommends the Victorian Government develop a consolidated of the comprising a single form or package of forms for creating enduring of attorney (financial) and enduring powers of attorney (guardianship). This llow different representatives to be appointed for different powers.
creating general (non-enduring) powers of attorney. This form should highlight that the power ceases to operate when a principal has impaired decision-making capacity.  Recommendation 9: Developing new power of attorney forms	Recomr	mendation 8: General (non-enduring) power of attorney form57
The Committee recommends the Victorian Government:  a) in consultation with a wide range of stakeholders, including members of the community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design  b) widely test the draft power of attorney forms  c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.  Recommendation 10: Producing power of attorney forms and accompanying information in community languages	creating	general (non-enduring) powers of attorney. This form should highlight that
<ul> <li>a) in consultation with a wide range of stakeholders, including members of the community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design</li> <li>b) widely test the draft power of attorney forms</li> <li>c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.</li> <li>Recommendation 10: Producing power of attorney forms and accompanying information in community languages</li></ul>	Recomr	nendation 9: Developing new power of attorney forms61
community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design  b) widely test the draft power of attorney forms  c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.  Recommendation 10: Producing power of attorney forms and accompanying information in community languages	The Cor	nmittee recommends the Victorian Government:
c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.  Recommendation 10: Producing power of attorney forms and accompanying information in community languages	a)	in consultation with a wide range of stakeholders, including members of the community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design
implemented, and every five years thereafter.  Recommendation 10: Producing power of attorney forms and accompanying information in community languages	b)	widely test the draft power of attorney forms
The Committee recommends the Victorian Government:  a) make the new power of attorney forms available in a range of community languages. The translated forms should be in a bilingual format, including both English and each community language  b) make all information and educational materials about powers of attorney available in a range of community languages.  Recommendation 11: Powers of attorney to be in the prescribed form	c)	review the new power of attorney forms two years after they are implemented, and every five years thereafter.
<ul> <li>a) make the new power of attorney forms available in a range of community languages. The translated forms should be in a bilingual format, including both English and each community language</li> <li>b) make all information and educational materials about powers of attorney available in a range of community languages.</li> <li>Recommendation 11: Powers of attorney to be in the prescribed form</li></ul>		
languages. The translated forms should be in a bilingual format, including both English and each community language  b) make all information and educational materials about powers of attorney available in a range of community languages.  Recommendation 11: Powers of attorney to be in the prescribed form	The Cor	nmittee recommends the Victorian Government:
available in a range of community languages.  Recommendation 11: Powers of attorney to be in the prescribed form	a)	make the new power of attorney forms available in a range of community languages. The translated forms should be in a bilingual format, including both English and each community language
The Committee recommends the Powers of Attorney Act provide that all powers of attorney made under the Act must be in the prescribed form.  Recommendation 12: Location of prescribed power of attorney forms	b)	make all information and educational materials about powers of attorney available in a range of community languages.
attorney made under the Act must be in the prescribed form.  Recommendation 12: Location of prescribed power of attorney forms	Recomr	nendation 11: Powers of attorney to be in the prescribed form62
The Committee recommends the Powers of Attorney Act set out the forms for creating general (non-enduring) powers of attorney, enduring powers of attorney		• • •
creating general (non-enduring) powers of attorney, enduring powers of attorney	Recomr	nendation 12: Location of prescribed power of attorney forms63
	creating	general (non-enduring) powers of attorney, enduring powers of attorney

Recommendation 7: Consolidated enduring powers of attorney document ......57

Recommendation 13: Making power of attorney forms accessible 65
The Committee recommends the Victorian Government make the forms for creating all powers of attorney widely available both online and in hard copy.
Recommendation 14: The role of witnesses of power of attorney documents
The Committee recommends the Powers of Attorney Act provide that each witness to a power of attorney must certify that:
a) the principal has signed the power of attorney freely and voluntarily in the witness's presence
b) the principal appears to have capacity and, in particular, appears to understand the nature and effect of the power of attorney.
Recommendation 15: Who may witness a power of attorney document? 77
The Committee recommends the Powers of Attorney Act require all power of attorney documents to be witnessed by two witnesses, one of whom is authorised to witness affidavits or is a medical practitioner.
Recommendation 16: Excluding parties and their relatives from witnessing power of attorney documents
The Committee recommends:
a) the Powers of Attorney Act prohibit the following persons from witnessing a power of attorney document:
- a party to the document
- any person who could benefit from the document
b) the Powers of Attorney Act define a person who could benefit from the power of attorney document. This definition should be broad and include a relative of a party to the document
c) the forms for creating all powers of attorney require each witness to declare that he or she is not a party to the document and is not related to any party to the document.
Recommendation 17: Witnessing by a representative's employees
The Committee recommends the Powers of Attorney Act clarify that a person is no excluded from witnessing a power of attorney document merely because he or she is an employee of the representative who witnesses the document while acting in the ordinary course of employment.

	mendation 18: Authorised witness to state the basis of his or her ation to witness the document80
	mmittee recommends the forms for creating all powers of attorney require the ed witness to state the basis of his or her qualification to witness the nt.
	mendation 19: Power of attorney forms to emphasise the role and ons of witnesses82
	mmittee recommends that the forms for creating powers of attorney include tion about the role and obligations of witnesses.
Recomi	mendation 20: Education, training and resources for witnesses82
	mmittee recommends the Victorian Government, in conjunction with relevant onal organisations, develop and provide:
a)	practical resources to help witnesses fulfil their role, including checklists to assess a principal's understanding of the nature and effect of the document and information about how to identify evidence of duress and evidence that may displace the presumption of capacity
b)	education and training for authorised witnesses about their role.
Recomi	mendation 21: Resources for principals85
understa	mmittee recommends the Victorian Government provide simple, easy-to- and information and educational materials for people making and plating making powers of attorney.
	mendation 22: Legal support and advice for people wishing to make of attorney87
support	mmittee recommends the Victorian Government implement a scheme to and enable members of the community to access legal advice and assistance powers of attorney.
Recomi	mendation 23: Acceptance of appointment by representatives89
The Cor	nmittee recommends the Powers of Attorney Act:
a)	require all representatives to formally accept their appointment
b)	require representatives' acceptance of appointment to be witnessed by a person authorised to witness affidavits. This witness does not need to be the same person who witnessed the creation of the power of attorney document by the principal.

Recommendation 24: Dispensing with formal execution requirements ............ 90

The Committee recommends the Powers of Attorney Act provide that VCAT may declare valid an enduring power of attorney (financial) or enduring power of attorney (guardianship) document that does not meet the formal requirements, if satisfied that the principal intended the document to be such a power of attorney, that the principal had the necessary capacity to make a power of attorney and that the principal signed the document freely and voluntarily.

### Recommendation 25: Time of activation of an enduring power of attorney...... 93

The Committee recommends the Powers of Attorney Act provide:

- a) a principal may elect to make an enduring power of attorney (financial) or enduring power of attorney (guardianship) effective immediately or upon another date, event or occasion
- b) if a principal does not specify when an enduring power commences, it commences immediately.

### 

The Committee recommends the Powers of Attorney Act:

- a) provide that the revocation of all powers of attorney made under the Act must be in writing in the approved form
- b) set out an approved form for revoking a power of attorney
- c) provide that the revocation of a power of attorney must be witnessed in the same way as a document creating a power of attorney.

### 

#### The Committee recommends:

- a) the Powers of Attorney Act provide that, unless a principal specifies otherwise, a later enduring power of attorney (financial) will revoke an earlier enduring power of attorney (financial) and a later enduring power of attorney (guardianship) will revoke an earlier enduring power of attorney (guardianship)
- b) the forms for creating enduring powers of attorney allow a principal to specify any previous enduring powers of attorney that the principal does not wish to revoke.

	nendation 28: Informing representatives that a power of attorney n revoked99
revokes	nmittee recommends the Powers of Attorney Act provide that if a principal a power of attorney, the principal must take reasonable steps to inform all tatives that the power has been revoked.
Recomn	nendation 29: Effect of guardianship or administration order99
The Con	nmittee recommends the Powers of Attorney Act provide:
a)	if VCAT makes an administration order in relation to a principal, the representative may exercise power under an enduring power of attorney (financial) only to the extent authorised by the Tribunal
b)	if VCAT makes a guardianship order in relation to a principal, the representative may exercise power under an enduring power of attorney (guardianship) only to the extent authorised by the Tribunal.
Recomn	nendation 30: Certified copies of power of attorney documents102
attorney and com regulated any othe prescribe	nmittee recommends the Powers of Attorney Act provide that a power of document can be proved by a copy of the document that is certified as a true uplete copy of the original by a legal practitioner, financial services licensee, diprincipal under the Corporations Act, justice of the peace, public notary or er officer authorised by law to administer an oath or by a person of a ed class.  nendation 31: Making sure powers of attorney can be located when
	103
promotion the exist	mmittee recommends the Victorian Government produce and distribute on all materials such as wallet cards that can be used to inform people about stence of a power of attorney and provide basic information such as tatives' names and contact details.
Recomn	nendation 32: Mutual recognition of powers of attorney105
The Con	nmittee recommends:
a)	the Victorian Government, through the Standing Committee of Attorneys-General, actively promote and support the implementation of effective mutual recognition provisions for all enduring powers of attorney
b)	the Powers of Attorney Act provide, to the maximum extent possible, for recognition in Victoria of all power of attorney documents validly executed in another Australian state or territory.
Recomn	nendation 33: Presumption of capacity110
	nmittee recommends the Powers of Attorney Act provide that a person is d to have capacity to make his or her own decisions.

	endation 34: Defining capacity and impaired decision-making113
The Com	mittee recommends the Powers of Attorney Act:
a)	state that a person has capacity to make a decision if he or she has:
	- the ability to understand the information relevant to making the decision
	- the ability to retain the relevant information
	- the ability to weigh up the relevant information
	AND
	- the ability to communicate the decision in some way.
	state that a person has impaired decision-making capacity if, in relation to a decision, he or she does not have:
	- the ability to understand the information relevant to making the decision
	- the ability to retain the relevant information
	- the ability to weigh up the relevant information
	OR
	- the ability to communicate the decision in some way.
Recomm	endation 35: Principles to guide capacity assessments 120
Attorney	amittee recommends the statement of principles underpinning the Powers of Act recommended in recommendation 3 include principles to guide assessments. These principles should include:
•	capacity is specific to each decision to be made
•	impaired decision-making capacity may be temporary or permanent
•	capacity should not be assumed based on a person's appearance
	a person must not be presumed to have impaired decision-making capacity merely because he or she makes a decision that is, in the opinion of others, unwise

a person should not be treated as unable to make a decision if it is possible for him or her to make that decision with appropriate support.

Recommendation 36: Capacity resource130	
The Committee recommends the Victorian Government, in consultation with a wide range of stakeholders, including representatives from the legal and health care sectors, develop a comprehensive resource about capacity and capacity assessments, based on the New South Wales <i>Capacity toolkit</i> .	
Recommendation 37: Providing guidance about appropriate representatives140	
The Committee recommends the Victorian Government provide simple, easy-to- understand information and educational materials for principals, and persons likely to advise principals, which includes information about the sorts of skills representatives require and guidance about appropriate appointments.	
Recommendation 38: Excluding unsuitable representatives143	
The Committee recommends the Powers of Attorney Act provide:	
a) a person is not eligible to be appointed as a representative under an enduring power of attorney (financial) if he or she has previously been convicted of an offence involving dishonesty	
b) a principal is entitled to apply to VCAT for approval to appoint as a representative under an enduring power of attorney (financial) a person who has previously been convicted of an offence involving dishonesty	
c) when accepting an appointment as a representative, a person must declare that he or she is eligible to be appointed as a representative	
d) a person who accepts an appointment as a representative when he or she is not eligible, is guilty of an offence.	
Recommendation 39: Appointing multiple representatives146	
The Committee recommends the Powers of Attorney Act provide that a principal may appoint one or more representatives.	
Recommendation 40: Decision making by multiple representatives147	
The Committee recommends the Powers of Attorney Act provide that a principal can appoint multiple representatives to act jointly, jointly and severally, or in any combination, for example as a majority. This provision should apply to powers of attorney made prior to the commencement of this provision.	
Recommendation 41: VCAT directions when representatives cannot agree148	
The Committee recommends the Powers of Attorney Act provide that representatives can apply to VCAT for directions when they cannot agree about a decision to be made in relation to the exercise of their power under an enduring power of attorney (financial) or an enduring power of attorney (guardianship).	

Recommendation 42: Presumption that representatives are appointed jointly
The Committee recommends the Powers of Attorney Act provide that if a principal does not specify how two or more representatives are appointed, they are presumed to be appointed jointly.
Recommendation 43: Effect of one joint representative's power ending 149
The Committee recommends the Powers of Attorney Act provide that, unless the power of attorney document states otherwise, when a joint representative's power ends, any remaining representative or representatives may continue to exercise power under the power of attorney document.
Recommendation 44: Appointing multiple alternative representatives 150
The Committee recommends the Powers of Attorney Act provide that a principal may appoint one or more alternative representatives. This provision should apply to powers of attorney made prior to the commencement of this provision.
Recommendation 45: Relinquishing power to an alternative representative 151
The Committee recommends the Powers of Attorney Act provide that a representative can relinquish all powers under an enduring power of attorney to an alternative representative nominated in the enduring power of attorney document by providing a signed notice stating that the representative is not willing to act in that role to the alternative representative and to the body responsible for registering enduring power of attorney documents.
Recommendation 46: Legislated clarity about representatives' powers 157
The Committee recommends the Powers of Attorney Act:

- a) provide that representatives appointed under an enduring power of attorney (financial) have powers in relation to financial and legal matters
- b) provide that representatives appointed under an enduring power of attorney (guardianship) have powers in relation to guardianship matters
- c) define 'financial and legal matters' and 'guardianship matters' and give examples of the type of matters that a representative may deal with under each power.

# Recommendation 47: Decision of representative with guardianship powers to prevail.......165

The Committee recommends the Powers of Attorney Act provide that if there is conflict between a representative with powers in relation to guardianship matters (whether appointed by VCAT or a principal) and a representative with powers in relation to financial and legal matters, then the decision of the representative with powers in relation to guardianship matters prevails, and the representative with powers in relation to financial and legal matters must take such available steps as may be necessary to allow the decision of the representative with powers in relation to guardianship matters to be implemented.

#### Recommendation 48: The duties of representatives......169

The Committee recommends the Powers of Attorney Act:

- a) provide that representatives have duties to act honestly and in good faith, to put the principal's interests ahead of the representative's interests, to keep accurate records and to act within their powers
- b) require representatives to undertake to act in accordance with the legislated duties when accepting an appointment as a representative.

#### 

The Committee recommends the statement of principles underpinning the Powers of Attorney Act recommended in recommendation 3 include principles to guide decision making by representatives. These principles should include:

- the starting point for any decision making should be the principal's wishes
- principals should be encouraged to participate in decision making, even when they have impaired decision-making capacity
- representatives must act in a way that promotes the personal and social wellbeing of the principal. This should be defined as including matters such as recognising the principal's role as a valued member of society, taking into account the principal's existing supportive relationships, values and cultural and linguistic environment and recognising the principal's right to confidentiality of information.

# Recommendation 50: Representatives' undertaking to act in accordance with the statement of principles......174

The Committee recommends the Powers of Attorney Act require representatives when accepting an appointment to undertake to act in accordance with the statement of principles.

The Con	nmittee recommends:
a)	the forms for creating powers of attorney and accepting an appointment as a representative set out in the Powers of Attorney Act provide a summary of representatives' powers and duties
b)	the Victorian Government produce simple, easy-to-understand information and educational materials for representatives
c)	the Victorian Government provide advice and ongoing support to representatives, including through a telephone advice service and information sessions for new representatives.
Recomn	nendation 52: Remuneration of representatives181
represen	ammittee recommends the Powers of Attorney Act provide that a tative is not entitled to be paid unless payment is specifically authorised by er of attorney document.
Recomn	nendation 53: Increasing detection and reporting of abuse 187
The Con	nmittee recommends the Victorian Government:
a)	develop and implement protocols and training for professionals in the health, aged care and community sectors about detecting and reporting abuse of powers of attorney
b)	work in conjunction with the Australian Government, other jurisdictions and the banking and finance sector to develop and implement a national system for detecting and reporting suspected financial abuse and a training program for banking staff.
Recomn	nendation 54: Protecting whistleblowers188
	nmittee recommends the Powers of Attorney Act provide protection from criminal liability for people who, in good faith, report abuse of powers of
Recomn	nendation 55: Educating the general community about abuse 189
	nmittee recommends the Victorian Government develop and implement an n campaign to increase general community awareness about abuse of powers ey.

Recommendation 51: Educating and supporting representatives ...... 180

	endation 56: Office of the Public Advocate to receive reports of190
	mittee recommends the Victorian Government empower the Office of the dvocate to receive reports of suspected abuse of powers of attorney.
Recomm	endation 57: Personal monitors200
The Com	mittee recommends:
,	the Powers of Attorney Act provide that a principal may appoint one or more personal monitors to oversee the operation of an enduring power of attorney (financial) or an enduring power of attorney (guardianship)
	the Victorian Government produce simple, easy-to-understand information and educational materials for personal monitors.
Recomm	endation 58: Regulating gifts202
representa	nmittee recommends the Powers of Attorney Act provide that a ative can make a gift of the principal's property, including to the ative, only if:
	the gift is reasonable in the circumstances, particularly in view of the principal's financial situation AND
•	the gift:
	- is to a relative or close friend of the principal and is of a seasonal nature or for a special event OR
	- the gift is a type of donation that the principal made when he or she had capacity or might reasonably be expected to make.
Recomm	endation 59: Protecting gifts made in the principal's will204
about wl	mittee recommends the Victorian Government conduct further consultation hether the Powers of Attorney Act should protect the interests of ries under a principal's will.
	endation 60: Notification of activation of an enduring power of205

The Committee recommends the Powers of Attorney Act provide that prior to commencing to act under an enduring power of attorney (financial) or enduring power of attorney (guardianship) that states that it commences when a principal has impaired decision-making capacity, the representative must give notice of his or her intention to activate that enduring power of attorney to the personal monitor or monitors, any other persons designated in the power of attorney document and the body responsible for registering enduring power of attorney documents in Victoria.

Recommendation 61: Criminal offences for abuse of powers of attorney ...... 211

The Committee recommends the Powers of Attorney Act provide the following offences:

- procuring a power of attorney by threat or deception
- not acting honestly and with reasonable diligence to protect the principal's interests, having regard to the principal's expressed wishes
- knowingly exercising powers under a revoked power of attorney
- failing to keep accurate records.

#### 

The Committee recommends the Powers of Attorney Act provide that where VCAT finds that an enduring power of attorney has been abused, VCAT may order a representative to compensate a principal for any loss.

### 

The Committee recommends the Powers of Attorney Act provide that VCAT may revoke a representative's appointment under an enduring power of attorney (financial) or an enduring power of attorney (guardianship) if satisfied that the representative has not acted in the best interests of the principal.

#### Recommendation 64: Clarifying who has standing to apply to VCAT...... 218

The Committee recommends the Powers of Attorney Act provide:

- a) the following people may apply to VCAT for an order about an enduring power of attorney (financial) or an enduring power of attorney (guardianship): the Public Advocate, the principal, the representative, a close relative of the principal, the personal monitor or monitors appointed by the principal, or any other person who VCAT is satisfied has a special interest in the principal's affairs.
- b) the following people must be given notice of an application to VCAT and any hearing of an application in relation to an enduring power of attorney (financial) or an enduring power of attorney (guardianship): the Public Advocate, the principal, the representative, a close relative of the principal, the personal monitor or monitors appointed by the principal, or any other person who VCAT is satisfied has a special interest in the principal's affairs.

Recomn	nendation 65: Referral by VCAT to Director of Public Prosecutions219
refer cas	nmittee recommends the Powers of Attorney Act provide that VCAT may see of suspected abuse of enduring powers of attorney to the Director of rosecutions.
Recomn	nendation 66: A register for power of attorney documents233
The Con	nmittee recommends the Victorian Government:
a)	develop and implement a register for power of attorney documents
b)	through the Standing Committee of Attorneys-General, actively promote and support the development and implementation of a national register for power of attorney documents.
	nendation 67: Mandatory registration of enduring powers of
creating	mmittee recommends the Powers of Attorney Act require all documents and revoking enduring powers of attorney (financial) and enduring powers ey (guardianship) to be registered.
	nendation 68: Voluntary registration of general (non-enduring) of attorney236
	nmittee recommends the Powers of Attorney Act permit documents creating king general (non-enduring) powers of attorney to be registered.
Recomn	nendation 69: The time of registration238
	mmittee recommends the Powers of Attorney Act require all documents to be registered under the Act to be registered at the time they are created.
Recomn	nendation 70: The effect of non-registration239
The Con	nmittee recommends the Powers of Attorney Act provide:
a)	any act performed under an enduring power of attorney (financial) or enduring power of attorney (guardianship) has no legal effect unless the document is registered
b)	VCAT may extend the time for a document creating or revoking an enduring power of attorney (financial) or enduring power of attorney (guardianship) to be registered if it believes the document is valid.

Recom	mendation 71: Accessing information on the register245
registra	ommittee recommends the Victorian Government, in implementing the tion system for power of attorney documents, ensure that the privacy of als' information is protected by providing:
a)	a document verification system for approved service providers such as banks
b)	greater access to information on the register for health care providers
c)	access to information on the register only to members of the community who are able to clearly demonstrate that they have an interest in that information.
Recom	mendation 72: Registration fees246
registra kept to	ommittee recommends the Victorian Government, in implementing the tion system for power of attorney documents, ensure that registration fees are a minimum, with concession rates or fee waivers available.  Implementing the tion system for power of attorney documents, ensure that registration fees are a minimum, with concession rates or fee waivers available.  248
The Co	mmittee recommends the Powers of Attorney Act provide that the Registry of Deaths and Marriages maintain the register of power of attorney documents.
Recom docum	mendation 74: Registration body to check power of attorney ents249
body to	ommittee recommends the Powers of Attorney Act require the registration of conduct a basic check to ensure that a document creating or revoking a of attorney meets formal requirements before registering the document.
Recom	mendation 75: Notice of registration250
The Cobody to	ommittee recommends the Powers of Attorney Act require the registration
a)	notify any personal monitors nominated in an enduring power of attorney document of an application for registration
b)	notify the principal that an application for registration has been made if the application is made by a person other than the principal.

Recomm	nendation 76: Objections to registration250
The Cor	nmittee recommends the Powers of Attorney Act:
a)	empower a principal, any personal monitors nominated in an enduring power of attorney document or any other person with a special interest in the affairs of the principal, to object to the registration of an enduring power of attorney document
b)	empower VCAT to hear and determine objections to registrations of enduring power of attorney documents.
Recomr	mendation 77: Effect of registration in another jurisdiction253
The Cor	nmittee recommends:
a)	the Victorian Government, through the Standing Committee of Attorneys-General, actively support and promote an arrangement whereby a certified copy of an enduring power of attorney registered in one Australian jurisdiction is sufficient to satisfy the requirement to register an enduring power of attorney in another jurisdiction in order to deal with land in that jurisdiction
b)	the Powers of Attorney Act provide, to the maximum extent possible, for recognition of an enduring power of attorney document registered in another Australian state or territory.
Recom	mendation 78: Promoting the registration system254
The Cor	nmittee recommends the Victorian Government:
a)	provide information about registration requirements on the forms for creating enduring powers of attorney (financial) and enduring powers of attorney (guardianship)
b)	conduct a public education campaign to inform members of the community who have already made an enduring power of attorney (financial) or enduring power of attorney (guardianship) about the option of registering the document.
	mendation 79: A coordinated approach to providing information and on about powers of attorney261
	mmittee recommends the Victorian Government adopt a coordinated, whole- rnment approach to providing information and education about powers of .

The Committee recommends the Victorian Government develop and implement an ongoing state-wide community education campaign to increase awareness and understanding of powers of attorney. The campaign should:

- a) provide simple easy-to-understand information about powers of attorney in plain English and in a variety of community languages
- b) use a wide variety of media, including websites, DVDs, TV, radio, newspapers, newsletters, pamphlets, fact sheets and posters
- c) include a community engagement component, with information sessions provided through a range of existing community forums
- d) be supported by advice and support mechanisms, including a telephone advice service.

### Recommendation 81: Educating seniors about powers of attorney ................ 267

The Committee recommends the Victorian Government, in consultation with seniors and seniors' organisations:

- a) develop targeted information and resources about powers of attorney for seniors in a range of formats and disseminate these widely through appropriate media
- b) provide information sessions about powers of attorney for seniors through a range of existing community forums.

### 

The Committee recommends the Victorian Government:

- a) develop targeted information and resources about powers of attorney for people from CALD backgrounds. This information should be developed in consultation with CALD organisations and members of CALD communities, be available in a wide range of formats and be disseminated through appropriate media
- b) provide information sessions for members of CALD communities through a range of existing community forums.

# Recommendation 83: Education about powers of attorney for young people .......271

The Committee recommends the Victorian Government develop targeted information and resources about powers of attorney for young people. This information should be developed in consultation with young people and youth organisations and disseminated through youth-focused media.

# Recommendation 84: Research about use of powers of attorney by members of the Aboriginal community......272

The Committee recommends the Victorian Government, in consultation with Aboriginal people and organisations, conduct research into the level of use of enduring powers of attorney by members of the Aboriginal community. This research should identify ways in which the principles underpinning powers of attorney arrangements may be applied to better support members of the Aboriginal community in the event of impaired decision-making capacity.

### Recommendation 85: Research about who makes powers of attorney ......273

The Committee recommends the Victorian Government conduct a study of the demographic profile of people making powers of attorney in Victoria to inform the development of information and education about powers of attorney.

#### Recommendation 86: Educating lawyers about powers of attorney......276

The Committee recommends the Victorian Government:

- a) develop targeted information and resources about powers of attorney for lawyers. This information should be developed in consultation with legal organisations and disseminated through appropriate media
- b) encourage law schools to incorporate substitute decision making into their curriculum, including education about powers of attorney
- c) encourage the Law Institute of Victoria to develop a training program on powers of attorney for lawyers and encourage lawyers to participate in this training as part of their continuing professional development.

	mendation 87: Educating the health and community sectors about of attorney
The Cor	nmittee recommends the Victorian Government:
a)	develop targeted information and resources about powers of attorney for those working in the health and community sectors. This information should be developed in consultation with community and health sector organisations and disseminated through appropriate media
b)	develop a training program on powers of attorney for workers in the community and health sectors in conjunction with relevant professional associations and encourage workers in these sectors to participate in this training
c)	encourage tertiary institutions to incorporate information about powers of attorney into relevant courses for the health and community sectors.
	mendation 88: Recognition of powers of attorney by Australian ment agencies281
Commit impleme	mmittee recommends the Victorian Government, through the Standing tee of Attorneys-General, advocate that all Australian Government agencies ent policies and practices to recognise powers of attorney that are validly under state or territory law.
Recomr financia	mendation 89: Enhancing recognition of powers of attorney by
finance	mmittee recommends the Victorian Government encourage the banking and sector to develop policies and procedures for accepting powers of attorney vide appropriate training for staff.
	nendation 90: Educating key professionals about the new Powers ney Act285
profession about the who are	mmittee recommends the Victorian Government, in conjunction with relevant onal bodies, conduct an education campaign to inform key professionals e new Powers of Attorney Act. This should be targeted at those professionals e likely to help people make powers of attorney or provide advice about of attorney.

### **Glossary**

activation coming into effect; commencing to operate

administration order an order made by VCAT under the Guardianship and

Administration Act 1986 (Vic) appointing an

administrator to make financial and legal decisions for a person with impaired decision-making capacity

administrator a person appointed by VCAT under the Guardianship

and Administration Act to make financial and legal decisions for a person with impaired decision-making

capacity

advance statement a document which sets out a person's wishes for

future treatment and care in the event that the person becomes unable to make these decisions for himself

or herself. Also called advance directives

agent a representative appointed by a principal under an

enduring power of attorney (medical treatment). See

representative

alternative representative a person nominated by the principal to stand in place

of the representative if the representative is unable to

fulfil the role of representative

appointee see representative

appointer see principal

attorney see representative

authorised witness a witness who meets certain criteria and is eligible to

witness a document. For example, a person authorised

to witness an affidavit or a person authorised to

witness a statutory declaration

CALD culturally and linguistically diverse

capacity for the purposes of this report, a person has capacity

to make a decision if he or she has:

- the ability to understand the information relevant

to making the decision

- the ability to retain the relevant information

- the ability to weigh up the relevant information

**AND** 

- the ability to communicate the decision in some

way

capacity assessment a test or process used to determine whether a person

has capacity to make a decision

caveat a notice to stop some action pending the decision of a

court

certified copy a copy of a document authenticated so that the copy is

treated as a true and accurate copy of the original

certify to vouch in writing that formal requirements are met

competence see capacity

composite representatives where a principal appoints multiple representatives

under a power of attorney and some of them can act

together, for instance as a two-thirds majority

donee see representative

donor see principal

duress any pressure or coercion or threat of pressure or

coercion used to compel a person to perform an act

against his or her will

elder abuse an act, or lack of appropriate action, occurring within

a relationship of trust which causes harm or distress to

an older person. It can include physical,

psychological, emotional, sexual and financial harm

and neglect

enduring guardian a representative appointed by a principal under an

enduring power of attorney (guardianship). See

representative

enduring power of attorney a power of attorney which lasts or 'endures' when the

principal has impaired decision-making capacity. In this report 'enduring powers of attorney' or 'enduring powers' is used to mean both enduring power of attorney (financial) and enduring power of attorney

(guardianship)

enduring power of attorney

(financial)

a power of attorney made under part XIA of the *Instruments Act 1958* (Vic). This power lasts or 'endures' when the principal has impaired decision-making capacity. A representative's powers under an enduring power of attorney (financial) are generally

characterised as financial and legal powers

enduring power of attorney (guardianship)

a power of attorney made under division 5A of the Guardianship and Administration Act. This power commences when the principal has impaired decision-making capacity. A representative's powers under an enduring power of attorney (guardianship) relate to lifestyle and some health care matters

enduring power of attorney (medical treatment)

a power of attorney made under section 5A of the *Medical Treatment Act 1988* (Vic) which allows the representative to make decisions about the principal's medical treatment

enduring power of guardianship

see enduring power of attorney (guardianship)

EPA or EPOA enduring power of attorney

EPG or EPOG enduring power of attorney (guardianship)

execution formal creation of a legal document

execution requirements things that must be done (such as having the

document signed by a witness) to formally create a

legal document

fiduciary duties common law obligations that one person ('the

fiduciary') owes to another person because of a

relationship of trust and confidence

formal

requirements/formality

requirements

see execution requirements

general (non-enduring) power of attorney

a power of attorney made under part XI of the Instruments Act. A representative's powers under a general (non-enduring) power of attorney are generally characterised as financial and legal powers. The representative's powers automatically end when the principal has impaired decision-making capacity

general power of attorney see general (non-enduring) power of attorney

guardian a person appointed by VCAT under the Guardianship

and Administration Act to make lifestyle and some health care decisions for a person with impaired

decision-making capacity

Guardianship List

a part of VCAT that has a mandate to protect adults who are unable to make decisions for themselves because of a disability. The Guardianship List can deal with matters relating to enduring powers of attorney

guardianship order

an order made by VCAT appointing a guardian to make lifestyle and some health care decisions for a person with impaired decision-making capacity

impaired decision-making capacity

for the purposes of this report, a person has impaired decision-making capacity if, in relation to a decision, he or she does not have:

- the ability to understand the information relevant to making the decision
- the ability to retain the relevant information
- the ability to weigh up the relevant information

OR

- the ability to communicate the decision in some way

instrument

a legal document

joint and several representatives

multiple representatives who can act together ('jointly') or separately ('severally')

joint representatives

multiple representatives who can only act together

mutual recognition

the recognition in one state or territory of a legal document made in another state or territory

nominee s

see representative

non-enduring power of

attorney

see general (non-enduring) power of attorney

OPA

the Office of the Public Advocate

person authorised to witness an affidavit

a person authorised under section 107A of the *Evidence (Miscellaneous Provisions) Act 1958* (Vic) to witness affidavits; see appendix D for a full list of

people who can witness an affidavit

person authorised to witness a statutory declaration

a person authorised under section 123C of the Evidence (Miscellaneous Provisions) Act to witness statutory declarations; see appendix D for a full list of people who can witness a statutory declaration

personal monitor for the purposes of this report, a person appointed by

a principal to oversee the operation of an enduring

power of attorney

POA power of attorney

power of attorney document a legal document creating a power of attorney

arrangement

power of attorney an arrangement by which a principal appoints a

representative to make decisions on his or her behalf

principal a person who makes a power of attorney document

appointing a representative to make decisions on his

or her behalf

representative a person appointed by a principal under a power of

attorney to make decisions on the principal's behalf

revoke to cancel

SCAG the Standing Committee of Attorneys-General. SCAG

comprises the Attorneys-General of the

Commonwealth, states and territories, and the Minister of Justice of New Zealand. It provides a forum for discussing matters of mutual interest about

justice policy, justice services and programs

seniors older members of the community; generally

considered to be people over the age of 60

standing the right to bring an action or take part in proceedings

before a court or tribunal

substitute decision making making decisions on behalf of another person

superior court any of the higher courts of the legal system, for

example, the Supreme Court of Victoria

supported decision making an approach to decision making that involves

empowering a person to make decisions for himself or

herself to the maximum extent possible

Take control kit Take control: a kit for making powers of attorney and

guardianship. This kit is produced by the Office of the Public Advocate and Victoria Legal Aid. It comprises a booklet and a DVD that provide information about powers of attorney in Victoria undue influence any form of pressure that influences a person to do or

refrain from doing anything without the benefit of

free will

VCAT the Victorian Civil and Administrative Tribunal

witness a person who observes the signing of a document and

certifies that certain requirements are met, for example, that the principal understands a power of

attorney

## Chair's foreword

Powers of attorney are valuable tools that empower people to plan for their future. They allow a principal to choose another person to make financial, health and lifestyle decisions on his or her behalf. Such arrangements are particularly beneficial when the principal is unable to make a decision for himself or herself, for example in the event of an accident or illness.

The evidence gathered by the Law Reform Committee in this extensive Inquiry makes it clear that powers of attorney have the potential to benefit many Victorians. These documents enhance principals' rights, empowering them to take control of their own future; they provide a flexible and low cost way of planning for the future; they provide certainty for third parties such as aged care facilities, about who can make decisions on behalf of a principal.

However, the Committee's work has revealed that powers of attorney are not as widely understood, used and recognised as they could be. Many Victorians do not know what a power of attorney is. Even those who know about powers of attorney, may not understand the difference between the various types of documents or know how to go about making one. The Committee also heard that even when a person has made a power of attorney, there is no guarantee that it will be accepted by a service provider, such as a bank or government agency.

Of particular concern were the instances of abuse of power of attorney documents that the Committee came across during this Inquiry. Abuse of these documents is often perpetrated by those the principal trusts the most. Evidence suggests that abuse of powers of attorney is rarely detected or reported.

However, the evidence presented to the Committee clearly demonstrates that the benefits of powers of attorney vastly outweigh any risk of abuse. Therefore the recommendations in this report aim to raise awareness of powers of attorney, to make them easier to use and to increase recognition of these documents throughout both Victoria and Australia. The Committee has also endeavoured to strike a balance between providing better safeguards against abuse and ensuring that people are not deterred from entering into these arrangements.

The Committee has recommended a multi-faceted approach to simplifying and streamlining power of attorney arrangements. A new Powers of Attorney Act is proposed to provide a simple and consistent framework for powers of attorney in Victoria. This legislation will be underpinned by a statement of principles to protect and promote the rights of people with impaired decision-making capacity.

The Committee's strategy to protect principals from abuse includes strengthening witnessing requirements, allowing the appointment of personal monitors to oversee the use of powers by a representative, clarifying the powers and duties of representatives and creating criminal offences for abuse of these documents. More support and resources for principals and representatives and for those involved in the assessment of a principal's capacity, is also recommended.

The vast majority of participants in this Inquiry called for a registration system to assist with the ready location, verification and validation of power of attorney

documents. The Committee has proposed mandatory registration of all documents creating and revoking enduring powers of attorney. In doing so, the Committee acknowledges the issues of cost and privacy that registration raises. However, the Committee believes that these issues can be addressed through the careful design of the registration system.

The final component of the Committee's approach is for the Victorian Government to conduct a comprehensive state-wide information and education campaign to increase awareness, understanding and acceptance of power of attorney documents, as well as encourage greater detection and reporting of abuse.

This Inquiry has taken place over a 12 month period, involving extensive research and consultation. On behalf of members of the Law Reform Committee, I would like to sincerely thank all those individuals and organisations who made written submissions and appeared before the Committee at its public hearings. In particular, the Committee would like to express appreciation to everyone who attended the Committee's Seniors' Forum and Culturally and Linguistically Diverse Communities' Forum which were co-sponsored by the Council of the Ageing Victoria and the Ethnic Communities' Council of Victoria respectively.

I offer special thanks to those individuals who shared their personal experiences of using powers of attorney with the Committee. Some of their stories appear as case studies in this report.

The report is a cooperative effort and I would like to thank my fellow Committee members for their enthusiasm and thoughtful contributions, in particular, the Deputy Chair, Mr Robert Clark MLA.

I also acknowledge the hard work of the Committee secretariat, led by Ms Kerryn Riseley and comprising Ms Kerry Harrison, Ms Liana Levin, Ms Vathani Shivanandan and Ms Helen Ross-Soden. I also recognise the contribution of Victorian Law Foundation intern, Ms Yardena Lankri, who worked hard to research and write many of the case studies that appear throughout this report.

I believe that the Committee's recommendations will provide a framework to make powers of attorney widely available to all Victorians, so they can safely and confidently plan for their future.

Johan Scheffer MLC Chair

## **Executive summary**

In this report the Committee presents a simplified and streamlined framework for powers of attorney in Victoria.

Powers of attorney are legal documents that allow a person (in this report called a 'principal') to choose another person (representative) to make decisions on his or her behalf. This report examines three types of powers of attorney:

- general (non-enduring) powers of attorney
- enduring powers of attorney (financial)
- enduring powers of attorney (guardianship).

The terms of reference for this Inquiry did not include powers of attorney (medical treatment) and thus the Committee does not consider these in this report.

While non-enduring powers of attorney are valid only while a principal has capacity, enduring powers of attorney may start or continue even when a principal does not have the ability to make decisions for himself or herself. Therefore, enduring powers are extremely useful tools for helping people plan for the possibility of an accident or illness in the future.

The Committee has found that powers of attorney have many benefits for principals, their family and friends, as well for the general community. These include:

- providing a simple, flexible and low cost way of planning for the future
- enhancing principals' rights by empowering them to make arrangements for when they are unable to make decisions themselves
- avoiding the need for a guardianship and/or administration order which may be stressful for family and friends, as well as expensive for the state
- providing certainty to third parties such as health services and banks about who can make a decision on a principal's behalf.

Australia's ageing population and growing rates of dementia and other forms of disability mean that powers of attorney can potentially benefit many more people in the future. Thus the recommendations in this report aim to encourage the increased use of these documents while, at the same time ensuring that principals are protected from abuse.

## Encouraging use of powers of attorney

An estimated 11% of Australians have made an enduring power of attorney, with powers that relate to financial matters much more widely used than those that relate to lifestyle matters (guardianship). The complexity of the current regime and lack of general community awareness and understanding of powers of attorney are major barriers to more Victorians making these arrangements.

To encourage more people to create powers of attorney, the Committee recommends simplified forms for making these documents, providing more information for principals and those contemplating making a power of attorney, as well as providing greater access to legal advice. The Committee also proposes a state-wide community education campaign to make sure that as many Victorians as possible are aware of powers of attorney and the benefits they offer. As part of this campaign information will be specifically targeted to groups to which these documents offer particular advantages, such as seniors, as well as groups with low levels of use, including young people and members of culturally and linguistically diverse communities.

## A new Powers of Attorney Act

The three types of powers of attorney under review in this Inquiry are currently governed by two separate pieces of legislation: the *Instruments Act 1958* (Vic) and the *Guardianship and Administration Act 1986* (Vic). This legislative framework provides a confusing and often inconsistent approach to powers of attorney.

The Committee recommends the introduction of a consolidated, standalone Powers of Attorney Act, covering general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). This will provide a simple, consistent framework for the operation of these powers in Victoria.

A key component of the new legislation will be a statement of principles which will apply to all acts and decisions made under the statute in relation to a principal with impaired decision-making capacity. The proposed statement of principles has three elements:

- an articulation of human rights based on international law
- a statement of principles relevant to making decisions about a person's capacity
- a statement of principles to guide the actions of those making decisions on behalf of a person with impaired decision-making capacity.

Many participants in this Inquiry expressed frustration about inconsistent state-based power of attorney laws in an increasingly mobile and borderless society. There is currently no guarantee that a power of attorney document created in Victoria will be accepted in another Australian state or territory. In this report the Committee supports work towards national harmonisation of power of attorney laws, as well as efforts to increase mutual recognition of these documents.

## Protecting against abuse

Power of attorney documents can be extremely empowering for principals, however the Committee also heard that they are susceptible to abuse. While the extent of abuse is not known, the Committee was told that abuse, particularly of enduring powers of attorney (financial), is 'not uncommon'. Abuse is often perpetrated by a close family member. Abuse can occur both at the time a power of attorney document is created and when the power is being used by the representative, although abuse of validly executed documents appears to be more common. Overall, the Committee considers that the benefits that powers of attorney offer greatly outweigh the risk of abuse.

While participants in this Inquiry called for more safeguards against abuse, the Committee was cautioned against imposing requirements that are too onerous. In particular, there was concern that such requirements may deter principals from making powers of attorney or discourage people from agreeing to act as representatives. In this report the Committee has endeavoured to strike a balance between protecting principals and ensuring that the flexibility and useability of these arrangements is not unduly compromised.

The Committee heard that sometimes a principal may be pressured into signing a power of attorney and may not fully understand the implications of the document. To protect against abuse when a power of attorney document is created, the Committee recommends restricting the classes of persons who may witness these documents, making the witnesses' role clearer and providing more education and support for witnesses.

At present the use of powers by a representative is almost entirely unregulated. To reduce abuse at this stage the Committee recommends that principals be empowered to appoint a friend or independent person or organisation as a 'personal monitor' who can oversee the use of powers by the representative. While it will generally be up to the principal to specify key decisions and actions which must be notified to the monitor, the Committee believes that representatives should be required to notify the monitor when an enduring power of attorney document is registered or upon activation of an enduring power of attorney document that is stated to commence when the principal has impaired decision-making capacity. The Committee also recommends clearer rules around when a representative is able to make a gift using the principal's funds.

Sometimes misuse of a power of attorney may be unintentional: some representatives simply do not understand the limits of their authority or the obligations imposed on them by the role. The Committee recommends providing legislative clarity about the powers that a representative has under each type of power of attorney and the duties the role imposes. In addition, the Committee recommends the creation of four offences that relate specifically to abuse of powers of attorney and empowering the Victorian Civil and Administrative Tribunal (VCAT) to refer findings of abuse to the Director of Public Prosecutions and to award compensation where it finds abuse.

The Committee also makes a number of recommendations aimed at increasing the detection and reporting of abuse of powers of attorney, including educating parties who may be in a position to detect abuse, such as bank staff.

## Supporting representatives

Acting as a representative for another person can be very demanding. Representatives are usually family members of the principal and are essentially volunteers.

Currently representatives receive minimal guidance in performing their role. As noted above, the framework proposed by the Committee supports representatives by providing more clarity about representatives' powers, duties and how they should approach decision making. In addition, the Committee recommends that representatives be supported by information and education, including written materials, information sessions, and a telephone advice service.

## Defining, assessing and enhancing capacity

The concept of capacity is central to powers of attorney. A principal must have capacity to create all types of powers of attorney. General (non-enduring) powers of attorney are automatically revoked when a principal has impaired decision-making capacity. In addition, some enduring powers of attorney (financial) and all enduring powers of attorney (guardianship) are activated when a principal has impaired decision-making capacity.

This Inquiry revealed widespread debate about what is capacity and how and by whom it should be assessed.

The Committee recommends that the new Powers of Attorney Act contain simple definitions of both capacity and impaired decision-making capacity. This will provide greater certainty for all people assessing capacity and affected by capacity assessments. The definitions proposed by the Committee recognises that capacity may fluctuate over time, and is relevant to each decision. The Committee also recommends that a presumption of capacity underpins the legislation, making it clear that all capacity assessments must start from an assumption that a person has capacity to make decisions for himself or herself.

As assessing capacity can be extremely complex, the Committee suggests the development of a capacity resource to assist all those involved in or affected by decisions about capacity. This resource should provide helpful advice about what is capacity, indicators of lack of capacity and when an expert assessment of capacity is required. In particular, it will have an emphasis on providing strategies to empower people to make their own decisions wherever possible.

## A registration system

An overwhelming number of participants in this Inquiry called for the introduction of a registration system for powers of attorney. While there is no evidence that registration of power of attorney documents has any impact on abuse, the Committee has concluded that a registration system should be introduced in Victoria. The Committee has identified that registration will have three main benefits:

- making it easy to establish the existence of an enduring power of attorney document
- making it easy to locate an enduring power of attorney document
- facilitating the validation of enduring power of attorney documents.

The system proposed by the Committee involves compulsory registration of all documents creating and revoking enduring powers of attorney. The Committee acknowledges concerns about the cost and privacy implications of registration, but believes that these issues can be addressed through the careful design of the registration system. In particular, to protect the principal's privacy the Committee recommends restricting both who may access the register and the information that may be accessed.

## Promoting powers of attorney

At present there is an ad hoc approach to providing information about powers of attorney in Victoria. The Committee recommends that the Victorian Government implement a coordinated whole-of-government approach to providing information and education about powers of attorney. This should include developing and providing information to principals, representatives, personal monitors, the general community and key professionals such as lawyers and the people working in the finance, aged care, health and community sectors.

The focus on education in this report will ensure greater acceptance and wider and more appropriate use of these valuable future-planning tools.

## **Chapter 1: Introduction**

On 4 December 2008 the Victorian Parliament's Legislative Assembly gave the Law Reform Committee terms of reference to conduct an Inquiry into powers of attorney. The terms of reference ask the Committee to consider how the existing power of attorney laws can be streamlined and simplified to enable more Victorians to plan for their future.

Powers of attorney are legal documents that help people plan for their financial, health and lifestyle needs. They allow a person (in this report called a 'principal') to choose another person ('representative') to make decisions on his or her behalf. Powers of attorney are a useful way to prepare for the future if a person becomes unable to make decisions for himself or herself, for example, because of an accident or illness.

Powers of attorney have the potential to benefit many members of the community but they are not being used as widely as they could be. Many Victorians do not know what a power of attorney is. Even those who are aware of powers of attorney may be confused by the multiple documents they need to sign to create the various powers and perplexed by the different powers conferred by different documents. People who have powers of attorney may be frustrated when a service provider such as a bank refuses to recognise a valid document.

The recommendations in this report aim to make more people aware of these arrangements and the benefits they offer. They aim to ensure that it is easy for all members of the community to create a power of attorney and that these documents are recognised as widely as possible, both in Victoria and throughout Australia.

The Committee's Inquiry has taken place in an environment of increased focus on human rights. As a tool that enables people to take control of their future, powers of attorney can potentially enhance human rights. Yet, at the same time, the Committee heard that all too often the wishes and needs of principals are ignored and powers of attorney are used to perpetrate abuse against principals, particularly financial abuse. Throughout this Inquiry the Committee has encountered significant tension between the need to make powers of attorney simpler and easier to use and the need to provide appropriate checks and balances to ensure that principals are protected from abuse.

Australia's ageing population and increasing rates of disability, in particular dementia, means that powers of attorney will potentially benefit many more members of the community in the future. The Committee's recommendations provide a framework to make powers of attorney widely available to all Victorians to help them plan for their future with ease and confidence.

## 1.1 The scope of this Inquiry

#### 1.1.1 Terms of reference

The terms of reference for this Inquiry require the Committee to consider ways to streamline and simplify power of attorney documents. The terms of reference ask the Committee to consider a number of specific issues:

- The formality requirements for making powers of attorney.
- How to establish whether a principal has capacity to create a power of attorney document, and when capacity is lost in relation to the activation of an enduring power of attorney.
- The powers that are granted by powers of attorney.
- How to minimise abuse of powers of attorney.
- Whether a registration system for power of attorney documents should be introduced.

## 1.1.2 What types of powers of attorney is the Committee reviewing?

There are several different types of powers of attorney in Victoria. The Committee's terms of reference ask it to review powers of attorney made under the *Instruments Act 1958* (Vic) and the *Guardianship and Administration Act 1986* (Vic).

The Instruments Act provides for two types of powers of attorney:

- General powers of attorney (in this report called 'general (non-enduring) powers of attorney') are used where the principal is still capable of making decisions but wants someone else to make decisions for him or her, for example, when the principal is on holiday and needs someone to take care of his or her banking. General (non-enduring) powers of attorney usually operate for a limited time. The representative's powers stop automatically if the principal loses the capacity to make these decisions for himself or herself.
- Enduring powers of attorney (in this report called 'enduring powers of attorney (financial)') allow the representative to make financial and some legal decisions such as banking or selling property. The representative's powers may start or continue (endure) after the principal loses the capacity to make these decisions for himself or herself.

The powers that a representative has under each of these powers of attorney are generally characterised as financial and some legal powers.<sup>1</sup> However, there is some debate about the scope of these powers and this issue is explored in detail in chapter six of this report.

The Guardianship and Administration Act allows a principal to make an **enduring power of guardianship** (in this report called an 'enduring power of attorney (guardianship)'). This allows the representative to make a range of decisions for the principal, including lifestyle and some health care decisions. The representative's powers start when the principal loses capacity to make these decisions for himself or herself.

A fourth type of power of attorney is provided by the *Medical Treatment Act 1988* (Vic). An enduring power of attorney (medical treatment) allows a representative to make certain decisions about the principal's medical treatment, including refusing treatment, when the principal does not have capacity to make these decisions for himself or herself.

Many participants in this Inquiry urged the Committee to also consider enduring powers of attorney (medical treatment) as part of this review. In particular, there was a strong view that the goal of streamlining power of attorney laws will be frustrated if one key type of power of attorney is excluded. While the Committee acknowledges these concerns, in keeping with its terms of reference it has not considered enduring powers of attorney (medical treatment) in detail as part of this Inquiry. However, throughout this report the Committee identifies opportunities to create more synergies between enduring powers of attorney (medical treatment) and the powers of attorney under review.

Some participants suggested that the Committee should also consider advance statements as part of this Inquiry.<sup>3</sup> Advance statements (also called advance directives) are statements setting out a person's wishes for future treatment and care if the person becomes unable to make these decisions for himself or herself.<sup>4</sup> As a future-planning tool, advance statements have some overlap with the arrangements the Committee is considering in this report. However, advance statements are distinguishable from powers of attorney in that they do not involve the appointment of another person to act on the principal's behalf. The Committee has not examined advance statements as part of this Inquiry.

Victoria Legal Aid and Office of the Public Advocate, *Take control: A kit for making powers of attorney and guardianship* 10th edition (2007), 28.

Australian Medical Association Victoria, Submission 69, 1; Mental Health Legal Centre Inc, Submission 59, 3; Law Institute of Victoria, Submission 41, 4; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 1; Royal College of Nursing Australia, Submission 26, 2; George Madden, Submission 34, 1; Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 2; Laura Helm, Policy Adviser, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 3; Belinda Evans, Senior Advocate, Elder Rights Advocacy, Transcript of evidence, Melbourne, 22 October 2009, 5.

Mental Health Legal Centre Inc, *Submission 59*, 3; Alzheimer's Australia Vic, *Submission 32*, 2; Australian Medical Association Victoria, *Submission 69*, 1, 3; Royal College of Nursing Australia, *Submission 26*, 2.

Department of Human Services, Victoria, Review of the Mental Health Act 1986: Government response to the Community consultation report - July 2009 (2009), 5.

## 1.1.3 What is 'capacity'?

The Committee's terms of reference ask it to consider the issue of a principal's capacity in the context of when an enduring power of attorney is created and activated.

A power of attorney is only valid if the principal has 'capacity' at the time the document is created. This means the principal must understand the document he or she is creating and the effects it will have. In addition, some powers of attorney may come into effect when a principal loses capacity, that is, the ability to make a decision for himself or herself.

Capacity is a complex area and the Committee devotes an entire chapter of this report, chapter five, to considering how capacity should be defined and assessed.

In this report the Committee uses the term 'impaired decision-making capacity' rather than 'loss of capacity' or 'incapacity'. In the Committee's view, this term reflects contemporary understandings that capacity is decision specific and may fluctuate over time.

## 1.2 The context for this Inquiry

## 1.2.1 Social and demographic changes

Changes in the demographic make up of the population will make powers of attorney even more useful to a greater number of Victorians in coming years.

Australia's population is ageing, with both the number and proportion of older Australians increasing rapidly, as a consequence of low fertility rates and increased longevity. According to the Australian Bureau of Statistics, 13.2% of the Australian population was aged 65 years and over in 2008 and this will increase to 20.8% by 2036 and to 22.9% by 2056. In the same period, there will be a significant increase in the proportion of the so-called 'old old' in the population. In 2008 1.7% of all Australians were aged 85 years and older and it is projected that this will rise to 4.9% in 2056. There are similar demographic trends in Victoria.

An ageing population is associated with a high prevalence of severe disability and dementia. It is estimated that 40.6% of Australians aged 65-69 years and 92.1% of those aged 90 and over have a disability. The incidence of dementia is projected to increase from 1% of the Victorian population in 2005 to 2.8% in 2050. While the older a person is, the more likely they are to have dementia, dementia is not a disease

<sup>&</sup>lt;sup>5</sup> Australian Bureau of Statistics, 1301.0 - Year Book Australia, 2009-10 (2010), 204, 206.

<sup>&</sup>lt;sup>6</sup> Australian Bureau of Statistics, 4102.0, Australian Social Trends, Data Cube - Population Table 1: Population, National Summary, 1998-2008 (2009).

Australian Bureau of Statistics, 4102.0, Australian Social Trends, Data Cube - Population Table 2.2: Population, Vic. Summary, 1998-2008 (2009).

Australian Bureau of Statistics, 4430.0.55.001 Disability, ageing and carers, Australia: Disability and long term health conditions Table 1: All persons, disability rates by age and sex (2004).

Access Economics Pty Limited, Dementia estimates and projections: Australian states and territories, report for Alzheimer's Australia (2005), 11.

of ageing alone and is increasing among the younger population as well.<sup>10</sup> Rising dementia rates are particularly relevant to this Inquiry, as a person with dementia may not be able to make some decisions for himself or herself.

Other societal changes may mean that older people are at greater risk of abuse, especially financial abuse. In particular, the last few decades have seen significant changes to the way wealth is distributed among the Australian population. There is a growing concentration of wealth and assets among seniors, particularly through home ownership. Approximately 80% of people aged 65 years and older own their own home outright, compared with 25% of those aged under 65. Older Australians are less likely to have debts and may also receive a regular income through a pension or investments. At the same time, there is an increasing emphasis on user charges for services such as aged care, which may conflict with expectations of younger generations about inheritance.

#### 1.2.2 A rights-based framework

Powers of attorney are potentially important mechanisms to promote the human rights and self-determination of principals. For example, the submission of Mr Julian Gardner, the former Victorian Public Advocate, stated:

The use of enduring powers promotes autonomy and respect for the inherent dignity of the individual. This occurs by enabling a competent adult to make decisions that can be implemented and respected after they cease to be competent (should this occur) and therefore become unable to exercise their autonomy. <sup>13</sup>

On the other hand, powers of attorney can be used to perpetrate abuse against principals. This may take many forms such as the misuse of a principal's finances, or a representative taking a paternalistic attitude to decision making that does not respect the principal's wishes or needs.

Victoria's *Charter of Human Rights and Responsibilities Act 2006* (Vic) enshrines a range of human rights derived from the *International Covenant on Civil and Political Rights* as part of the law in this state. The Charter sets out a number of rights that are particularly important to powers of attorney, including:

- recognition and equality before the law<sup>14</sup>
- freedom of movement<sup>15</sup>

<sup>&</sup>lt;sup>10</sup> Ibid, 5.

Australian Bureau of Statistics, 4102.0 - Australian Social Trends, 2005 Housing arrangements: Housing for older Australians (2005); Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people Trends & issues in crime and criminal justice, no. 139, Australian Institute of Criminology (1999), 5-6.

Australian Bureau of Statistics, 4102.0 - Australian Social Trends, 2009 Household debt (2009), 31-34.

Julian Gardner, Submission 21, 1. See also Victorian Equal Opportunity & Human Rights Commission, Submission 52, 2.

Charter of Human Rights and Responsibilities Act 2006 (Vic) s 8. See also International Covenant on Civil and Political Rights, opened for signature 16 December 1966, 999 UNTS 171, art 26 (entered into force 23 March 1976).

- privacy and reputation <sup>16</sup>
- protection of families <sup>17</sup>
- property rights. 18

At an international level, the *International Convention on the Rights of Persons with Disabilities* which Australia ratified in 2008, emphasises the equal rights of people with disabilities. In particular, it highlights that people with disabilities should have the necessary support and assistance to ensure they can make decisions themselves where possible and instils a principle of least restriction on a person's autonomy. <sup>19</sup> The Convention has laid the foundation for an international shift in emphasis from making decisions for a person (substitute decision making), to empowering the person to make decisions for himself or herself to the maximum extent possible (supported decision making). <sup>20</sup>

Protecting and enhancing the human rights of principals in line with the Charter and international conventions is at the core of the Committee's approach to powers of attorney in this report.

#### 1.2.3 An area in transition

Powers of attorney and the broader field of substitute decision making have attracted considerable attention in recent years. A number of significant reviews have been completed, advocating widespread reforms.

Of particular note is the House of Representatives Standing Committee on Legal and Constitutional Affairs' 2007 report *Older people and the law*. That expansive report considered the legal needs of older Australians, recommending a national approach to powers of attorney including uniform legislation, a national register and a nationally consistent approach to the assessment of capacity. Participants in this Inquiry urged the Committee to take the findings of that review into account. <sup>22</sup>

A number of other Australian jurisdictions have also recently reviewed their power of attorney laws.<sup>23</sup> The Committee draws upon the previous work in this area extensively throughout this report.

Charter of Human Rights and Responsibilities Act 2006 (Vic) s 12. See also International Covenant on Civil and Political Rights, 999 UNTS 171, art 12.

Charter of Human Rights and Responsibilities Act 2006 (Vic) s 13. See also International Covenant on Civil and Political Rights, 999 UNTS 171, art 17.

<sup>17</sup> Charter of Human Rights and Responsibilities Act 2006 (Vic) s 17; International Covenant on Civil and Political Rights, 999 UNTS 171, art 23.

<sup>&</sup>lt;sup>18</sup> Charter of Human Rights and Responsibilities Act 2006 (Vic) s 20.

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12 (entered into force 3 May 2008).

See generally Victorian Equal Opportunity & Human Rights Commission, Submission 52, 4-6.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), ch 3.

Alzheimer's Australia Vic, Submission 32, 2; Seniors Rights Victoria, Submission 38, 16-17.

Advance Directives Review Committee, Planning ahead: Your health, your money, your life. First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy (2008); Advance

In conducting this Inquiry the Committee has also been mindful of a number of relevant contemporaneous projects at both the national and state level. These include:

- The Victorian Law Reform Commission is reviewing the Guardianship and Administration Act. The Commission is considering enduring powers of attorney (guardianship) as part of its review. The Commission is also exploring a number of concepts such as capacity and supported decision making. The Commission's final report is due by 30 June 2011.
- The Standing Committee of Attorneys-General (SCAG) has agreed to a project to harmonise power of attorney laws nationally, although work on this project has not yet commenced. <sup>24</sup> It is likely that SCAG's work will be confined to arrangements for mutual recognition of power of attorney documents.
- The Victorian Department of Health recently reviewed the *Mental Health Act 1986* (Vic). A Mental Health Bill is currently being drafted and is likely to include provisions relating to capacity, supported decision making and advance statements.<sup>25</sup>
- The Health Ministers' Working Group on Advance Care Planning is currently establishing national standards for advance care planning. The National Framework for Advance Care Directives is presently in draft form and is expected to be finalised by the end of 2010 following a targeted national consultation process. <sup>26</sup>

## 1.3 Conduct of the Inquiry

The Committee called for public submissions to the Inquiry in June 2009. The call for submissions was advertised in *The Age*, the *Herald Sun* and the *Weekly Times*. The Chair of the Committee also wrote directly to over 240 key stakeholders, including government agencies, service providers and organisations representing the legal, health, disability, mental health, financial and community sectors.

The Committee also placed articles about the Inquiry and invited submissions in the Ethnic Communities' Council of Victoria's *Golden Years* newsletter in June 2009,

Directives Review Committee, Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008); Land and Property Management Authority, New South Wales, Review of the Powers of Attorney Act 2003, Undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009); Standing Committee on Social Issues, Substitute decision-making for people lacking capacity Report 43, New South Wales Legislative Council (2010); Standing Committee on Health and Community Care, Elder abuse in the ACT Report number 11, Legislative Assembly for the Australian Capital Territory (2001).

Standing Committee of Attorneys-General, *Communiqué - November 2008* (2008).

Department of Health, Victoria, Information sheet: Mental Health Act 1986 Review - February 2010 (2010). See also Department of Human Services, Victoria, above n 4.

Email from Lynton Huxley, Health, Community and Disability Services Ministerial Council Secretariat, to Administration Officer, Victorian Parliament Law Reform Committee, 22 July 2010.

the Council on the Ageing Victoria's *COTA news* in July 2009 and the Law Institute of Victoria's *Law Institute Journal* in July 2009.

The Committee received 75 written submissions to the Inquiry. These are listed in appendix A.

The Committee held five public hearings on 1 and 22 October 2009, 14 and 17 December 2009, and 30 March 2010. In addition, on 30 March 2010 the Committee hosted two forums to obtain input from senior Victorians and members of culturally and linguistically diverse communities. The Seniors' Forum was organised in partnership with the Council on the Ageing Victoria and the Culturally and Linguistically Diverse Communities' Forum was arranged in conjunction with the Ethnic Communities' Council of Victoria. Appendix B lists the people who gave evidence to the Committee at these hearings and forums.

The Committee also sought information about the systems for registering power of attorney documents in other Australian states and territories and the United Kingdom by writing to the registration body and, where appropriate, the public advocate in each of those jurisdictions. The Committee wrote directly to relevant government agencies and law societies in some other jurisdictions requesting information about the development of key resources such as resources to assist with capacity assessments.

The Committee's secretariat conducted an extensive literature review on powers of attorney in Australian states and territories and relevant overseas jurisdictions. The bibliography at the end of this report sets out the results of this work.

The Committee researched cases of abuse of power of attorney documents in Victoria. This project was completed by an intern recruited through the Victoria Law Foundation who examined the print media and law reports. The results of this research are set out in case studies throughout this report.

Finally, Committee representatives attended a number of conferences, forums and other events relevant to this Inquiry. The events attended are set out in appendix C.

## 1.4 Outline of this report

This report is divided into ten chapters:

- This chapter, chapter one, has provided an overview of the Inquiry, including its scope, context and key definitions.
- Chapter two sets out the current arrangements for powers of attorney, as well as other forms of supported and substituted decision making in Victoria.
- Chapter three considers the existing legislative framework for powers of attorney in Victoria and how it can be streamlined.

• Chapter four explores the formal requirements for creating and revoking a power of attorney document and makes suggestions for simplifying these processes, while safeguarding against abuse.

- Chapter five examines capacity, both at the time a power of attorney document is created and when it is activated, and identifies strategies for providing more certainty and transparency about decisions about capacity.
- Chapter six looks at the role of representatives appointed by a power of attorney and, in particular, explores the powers and duties of representatives and the support representatives need to effectively perform their role.
- Chapter seven considers the abuse of validly executed powers of attorney and identifies a range of possible mechanisms to protect against abuse.
- Chapter eight explores the possible benefits and problems associated with a registration system for power of attorney documents and sets out the key issues that need to be considered in designing a registration system.
- Chapter nine examines awareness and understanding of powers of attorney among the general community and key groups and identifies strategies for increasing knowledge about powers of attorney.
- Chapter ten contains a brief conclusion to the Inquiry.

# Chapter 2: Powers of attorney in Victoria – The current landscape

This chapter provides an overview of the current arrangements for powers of attorney in Victoria. It also considers other decision-making mechanisms that can be used if a person does not have a power of attorney. In this chapter the Committee examines the level of use of powers of attorney, the benefits these arrangements offer and the barriers that prevent more people making powers of attorney. The Committee also explores the incidence and nature of abuse of these documents. While there is a danger that powers of attorney, particularly financial powers, may be abused, the Committee concludes that the benefits these arrangements offer outweigh the risk of abuse.

## 2.1 What are powers of attorney?

## 2.1.1 The history of powers of attorney

General (non-enduring) powers of attorney have been used for centuries to allow a person to appoint another person to act on his or her behalf. Historically these arrangements were primarily used by wealthy merchants to assist with business transactions.

A general (non-enduring) power of attorney was traditionally created by a lengthy written document, executed under seal, which set out the powers delegated in detail.<sup>27</sup> At common law a general (non-enduring) power of attorney terminates automatically when a principal has impaired decision-making capacity.<sup>28</sup>

As the middle class grew and life expectancy increased, there was growing demand for an easy and cost-effective method to allow more people to appoint others to make decisions on their behalf, particularly when they were unable to make these decisions for themselves.<sup>29</sup> In the 1970s and 1980s most Australian states and territories passed legislation for enduring powers of attorney. These powers continue (endure) even when the principal has impaired decision-making capacity.<sup>30</sup>

In Victoria amendments to the *Instruments Act 1958* (Vic) introduced a statutory short form general (non-enduring) power of attorney in 1980<sup>31</sup> and an enduring power of attorney (financial) in 1982.<sup>32</sup>

The Victorian Bar, Submission 40, 5; Berna Collier and Shannon Lindsay, Powers of attorney in Australia and New Zealand (1992) The Federation Press, 8.

Berna Collier and Shannon Lindsay, above n 27, 223.

NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales, Review of the *Powers of Attorney Act 2003* (2009), 3-4 as attachment to letter from Imelda Dodds, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010.

<sup>30</sup> Ibid 4

Instruments (Powers of Attorney) Act 1980 (Vic).

Instruments (Enduring Powers of Attorney) Act 1981 (Vic). See generally The Victorian Bar, Submission 40, 5-8.

There is considerable debate about whether a power of attorney empowers a representative to make decisions about personal matters, as opposed to financial or legal matters. <sup>33</sup> Many jurisdictions have clarified this by enacting specific legislation permitting a representative appointed under a power of attorney to make decisions about personal and lifestyle matters. In Victoria enduring powers of attorney (guardianship) were introduced in 2000 through changes to the *Guardianship and Administration Act* 1986 (Vic). <sup>34</sup>

## 2.1.2 Powers of attorney in Victoria

This section provides a brief overview of the three types of powers of attorney under review in this Inquiry. The intricacies of each type of power of attorney are explored in more detail in later chapters of this report. This section also introduces the Office of the Public Advocate (OPA) and the Victorian Civil and Administrative Tribunal (VCAT), two key agencies involved in overseeing the operation of, and providing information and advice about, powers of attorney in Victoria.

## General (non-enduring) powers of attorney

Since 1980 the Instruments Act has provided the framework for general (non-enduring) powers of attorney in Victoria. Under this Act, a principal can authorise one or more representatives to do on his or her behalf anything 'which he can lawfully do by an attorney'. Actions that cannot lawfully be done by a representative are those that involve the exercise of 'personal skill and discretion' such as making a will or swearing an affidavit. The legislation identifies two specific actions that involve the exercise of personal skill and discretion that a representative cannot perform: acting as a trustee and delegating the powers given under the power of attorney. The second statement of the power of attorney.

The powers that a representative has under a general (non-enduring) power of attorney are usually characterised as 'financial and legal' powers. These include the power to buy or sell property and undertake banking transactions.<sup>38</sup>

The Instruments Act sets out a standard form for creating a general (non-enduring) power of attorney, but it is not mandatory to use the statutory form.<sup>39</sup> The principal's signing of a general (non-enduring) power of attorney document does not need to be witnessed.

A general (non-enduring) power of attorney commences either when the document is signed or on the date specified in the document. The powers are often specified to be only for a particular purpose or for a fixed period of time. In addition, a principal can

\_

Berna Collier and Shannon Lindsay, above n 27, 154; Robin Creyke, 'Enduring powers of attorney: Cinderella story of the 80s' (1991) 21 Western Australian Law Review 122, 142-143.

<sup>&</sup>lt;sup>34</sup> Guardianship and Administration (Amendment) Act 1999 (Vic) s 12.

<sup>35</sup> *Instruments Act 1958* (Vic) s 107(1).

Berna Collier and Shannon Lindsay, above n 27, 39-41.

Instruments Act 1958 (Vic) s 107.

Victoria Legal Aid and Office of the Public Advocate, *Take control: A kit for making powers of attorney and guardianship* 10th edition (2007), 6, 28.

Instruments Act 1958 (Vic) s 107, sch 12.

revoke the power at any time. The representative's powers automatically end if the principal dies or has impaired decision-making capacity. The representative's powers also stop if he or she becomes bankrupt, resigns or has impaired decision-making capacity. 40

VCAT does not have any jurisdiction in relation to general (non-enduring) powers of attorney.<sup>41</sup> Any matters relating to these powers must be litigated in the Supreme Court of Victoria.

#### **Enduring powers of attorney (financial)**

The Instruments Act also provides for enduring powers of attorney (financial). These allow a principal to appoint one or more representatives 'to do anything on behalf of the donor that the donor can lawfully authorise an attorney to do'. The principal can place restrictions on the representative's powers. The types of powers that may be granted are similar to those under a general (non-enduring) power of attorney; however, the representative's powers continue even when the principal has impaired decision-making capacity and is unable to make decisions for himself or herself.

The principal can specify a date, event or occasion when the representative's powers commence. If the principal does not specify any particular commencement date or occasion, the powers start when the document is executed.<sup>45</sup>

The Secretary of the Department of Justice has approved a form for creating an enduring power of attorney (financial) and use of this form is compulsory. <sup>46</sup> The form must be signed by the principal and two adult witnesses. <sup>47</sup> One witness must be authorised to witness statutory declarations and only one witness is permitted to be a relative of either the principal or representative. <sup>48</sup>

Both of the witnesses to an enduring power of attorney (financial) must sign a certificate stating that the principal signed the document 'freely and voluntarily in the presence of the witness' and appeared to have the capacity necessary to make the document. A principal has capacity to make an enduring power of attorney (financial) if he or she understands the nature and effect of the document. This includes understanding matters such as when the power is exercisable, that the power continues even after he or she has impaired decision-making capacity and that the principal may revoke the document.

13

Victoria Legal Aid and Office of the Public Advocate, above n 38, 6, 21; Berna Collier and Shannon Lindsay, above n 27, ch 11.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Transcript of evidence, Melbourne, 1 October 2009, 3.

<sup>42</sup> Instruments Act 1958 (Vic) s 115(1).

Instruments Act 1958 (Vic) s 115(1); Victoria Legal Aid and Office of the Public Advocate, above n 38, 9.

<sup>&</sup>lt;sup>44</sup> Instruments Act 1958 (Vic) s 115(2).

<sup>45</sup> Instruments Act 1958 (Vic) s 117.

<sup>46</sup> Instruments Act 1958 (Vic) ss 123, 125ZL; Victorian Government, 'The Instruments (Enduring Powers of Attorney) Act 2003 - Approved forms' (2004) G9 Victoria Government Gazette 437.

Instruments Act 1958 (Vic) s 123.

Instruments Act 1958 (Vic) s 125. See appendix D for a list of people authorised to witness statutory declarations

<sup>&</sup>lt;sup>49</sup> *Instruments Act 1958* (Vic) s 125A(1).

Instruments Act 1958 (Vic) s 118.

Only people who are aged 18 years or over and who are not insolvent are eligible to be appointed as representatives.<sup>51</sup> All representatives must accept their appointment by signing a statement of acceptance which includes an undertaking to perform the role of representative with reasonable diligence, to avoid conflicts of interests and to exercise the powers in accordance with the Act.<sup>52</sup>

Representatives are required to keep accurate records and accounts of all dealings and transactions made using an enduring power of attorney (financial).<sup>53</sup>

A representative may resign while a principal has decision-making capacity. However if the principal has impaired decision-making capacity, a representative may only resign with the permission of a court or VCAT.<sup>54</sup>

A principal who has decision-making capacity can revoke an enduring power of attorney (financial) at any time.<sup>55</sup> These powers are also automatically revoked in a number of other circumstances including if the representative has impaired decision-making capacity, becomes insolvent or dies.<sup>56</sup>

VCAT has a number of powers in relation to enduring powers of attorney (financial), which are considered in more detail later in this chapter.

#### **Enduring powers of attorney (guardianship)**

Since 2000 the Guardianship and Administration Act has provided for enduring powers of attorney (guardianship) in Victoria. Through an enduring power of attorney (guardianship) a principal can confer on a single representative specified powers or 'all the powers and duties which the guardian would have if he or she were a parent and the appointer his or her child ...'<sup>57</sup> This includes the power to make decisions about matters such as where the person lives and the health care the person receives.<sup>58</sup> The representative's powers only come into effect when the principal has impaired decision-making capacity and is unable to make a decision for himself or herself.<sup>59</sup>

A representative appointed under an enduring power of attorney (guardianship) must be at least 18 years old. <sup>60</sup> The principal cannot appoint a representative who is responsible for providing care, treatment or accommodation to the principal in a professional or administrative capacity. <sup>61</sup>

Instruments Act 1958 (Vic) ss 119(4), 121.

Instruments Act 1958 (Vic) s 125B.

Instruments Act 1958 (Vic) s 125D.

<sup>&</sup>lt;sup>54</sup> Instruments Act 1958 (Vic) s 125M.

<sup>55</sup> Instruments Act 1958 (Vic) s 125I.

Instruments Act 1958 (Vic) ss 125N-125P.
 Guardianship and Administration Act 1986 (Vic) s 35B.

Guardianship and Administration Act 1986 (Vic) s 24(2). Note this does not include all health care decisions, for example, the removal of tissue for transplant or the ability to refuse treatment: see Guardianship and Administration Act 1986 (Vic) s 24(2)(d), Part 4A; Medical Treatment Act 1988 (Vic) ss 5C, 5D.

<sup>&</sup>lt;sup>59</sup> Guardianship and Administration Act 1986 (Vic) s 35B.

<sup>&</sup>lt;sup>60</sup> Guardianship and Administration Act 1986 (Vic) s 35A(3).

<sup>&</sup>lt;sup>61</sup> Guardianship and Administration Act 1986 (Vic) s 35A(4).

The legislation requires representatives to act in the best interests of the principal. This includes acting as an advocate for the principal, encouraging the principal's participation in community life, protecting the principal from neglect, abuse or exploitation, and consulting the principal and taking his or her wishes into account as far as possible. <sup>62</sup>

An enduring power of attorney (guardianship) must be in the form set out in the Guardianship and Administration Act.<sup>63</sup> The principal must sign the document before two witnesses, one of whom is authorised to witness statutory declarations. Neither witness is permitted to be a party to the document or a relative of a party to the document.<sup>64</sup> Each witness must certify that the principal has signed the document freely and voluntarily in his or her presence and appeared to understand the effect of the instrument.<sup>65</sup>

A representative is required to formally accept the appointment. This is done by signing an acceptance of appointment, which must be witnessed in the same manner as the principal's signature. <sup>66</sup> The representative must undertake to exercise the powers honestly and in accordance with the Act. <sup>67</sup>

A principal who has capacity may revoke an enduring power of attorney (guardianship) at any time using the statutory form. <sup>68</sup>

VCAT also has a number of powers in relation to enduring powers of attorney (guardianship) including the ability to revoke these documents. <sup>69</sup> VCAT's powers are discussed in the next section.

#### The role of the Victorian Civil and Administrative Tribunal

The Guardianship List of VCAT has a mandate to protect adults who are unable to make decisions for themselves because of a disability. Most matters in the Guardianship List relate to the appointment of guardians and administrators under the Guardianship and Administration Act (discussed later in this chapter). However, the Guardianship List also has powers in relation to both enduring powers of attorney (financial) and enduring powers of attorney (guardianship).

<sup>&</sup>lt;sup>62</sup> Guardianship and Administration Act 1986 (Vic) ss 28, 35B(5).

Guardianship and Administration Act 1986 (Vic) s 35A(2), sch 4 form 1.

Guardianship and Administration Act 1986 (Vic) s 35A.

<sup>&</sup>lt;sup>65</sup> Guardianship and Administration Act 1986 (Vic) sch 4 form 1.

Guardianship and Administration Act 1986 (Vic) s 35A(2).

Guardianship and Administration Act 1986 (Vic) sch 4 form 1.

Guardianship and Administration Act 1986 (Vic) s 35C, sch 4 form 2.

<sup>&</sup>lt;sup>69</sup> Guardianship and Administration Act 1986 (Vic) s 35D.

Victorian Civil and Administrative Tribunal (VCAT), *Guardianship and Admin*, <a href="http://www.vcat.vic.gov.au/CA256DBB0022825D/page/Guardianship+and+Admin?OpenDocument&1=6">http://www.vcat.vic.gov.au/CA256DBB0022825D/page/Guardianship+and+Admin?OpenDocument&1=6</a> 0-Guardianship+and+Admin~&2=~&3=~>, viewed 31 May 2010; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Submission 37*, 1.

John Billings, Submission 37, above n 70, 1; Victorian Civil and Administrative Tribunal (VCAT), Annual report 2008/2009 (2009), 24.

In relation to an enduring power of attorney (financial) document, VCAT can:

- declare a document to be invalid if the principal had impaired decision-making capacity at the time the document was made, the document does not comply with formal requirements, or the document is invalid for another reason, for example, the principal was induced to make the document through dishonesty or undue influence<sup>72</sup>
- revoke the appointment of a representative if satisfied that it is in the best interests of the principal to do so. VCAT can only exercise this power if satisfied that a principal has impaired decision-making capacity<sup>73</sup>
- give permission to a representative to resign. If a principal has impaired decision-making capacity, a representative can only resign with the permission of VCAT or a court<sup>74</sup>
- order a representative to lodge accounts and other documents with VCAT or that accounts be examined by a person appointed by VCAT<sup>75</sup>
- give declarations, orders, directions or recommendations about the scope of a representative's powers or the exercise of the representative's powers<sup>76</sup>
- give advisory opinions on any matter<sup>77</sup>
- vary or suspend the document. 78

VCAT usually exercises these powers when an application is made by the Public Advocate, the principal, the representative or a person with a special interest in the principal's affairs. However, VCAT can exercise most of these powers on its own initiative.<sup>79</sup>

In 2008-2009 there were approximately 400 applications in the Guardianship List concerning enduring powers of attorney (financial). This was approximately 4% of all applications to the List.<sup>80</sup>

In relation to enduring powers of attorney (guardianship), VCAT has the power to revoke the appointment of a representative if the representative requests it, the representative is not willing or able to act, or the representative has not acted in the best interests of the principal or has acted in an incompetent or negligent manner.<sup>81</sup> Applications for these orders may be made by the Public Advocate, the representative

\_

<sup>&</sup>lt;sup>72</sup> Instruments Act 1958 (Vic) s 125Y.

<sup>&</sup>lt;sup>73</sup> Instruments Act 1958 (Vic) s 125X.

<sup>&</sup>lt;sup>74</sup> Instruments Act 1958 (Vic) s 125M(2).

<sup>&</sup>lt;sup>75</sup> Instruments Act 1958 (Vic) s 125ZB(1).

<sup>&</sup>lt;sup>76</sup> Instruments Act 1958 (Vic) s 125V(1).

Instruments Act 1958 (Vic) s 125ZA.

Instruments Act 1958 (Vic) s 125Z(1).
 Instruments Act 1958 (Vic) s 125V(2), 125Z.

Wictorian Civil and Administrative Tribunal (VCAT), above n 71, 25; John Billings, *Transcript of evidence*, above n 41, 3

Guardianship and Administration Act 1986 (Vic) s 35D(1).

or any person with an interest in the principal or the estate of the principal. 82 VCAT may also give an advisory opinion to a representative on its own initiative. 83

In addition, a representative under an enduring power of attorney (guardianship) may apply to VCAT for an advisory opinion or directions on any matter or question about the scope of a representative's appointment or the exercise of a power by the representative. VCAT may provide an advisory opinion, directions, vary or suspend the document or make any other orders it thinks necessary.<sup>84</sup>

VCAT also has the power to refer any matter relating to a proceeding about an enduring power of attorney (guardianship) to the Public Advocate, a government department, public authority, service provider, guardian or administrator for investigation and report. 85

#### The role of the Office of the Public Advocate

The Public Advocate is an independent statutory officer appointed under the Guardianship and Administration Act. 86 OPA plays an important advocacy, investigation and guardianship role for people with cognitive disabilities. 87

The Public Advocate may be appointed as a representative under an enduring power of attorney (guardianship) for a principal who has no close family members or friends. While this is only in a small number of cases, OPA's submission to the Inquiry stated, 'This is a significant role for the Public Advocate to fulfill for vulnerable citizens ...'88

In the absence of any agency with formal responsibility for powers of attorney in Victoria, OPA has become 'a de facto expert' on powers of attorney in this state. <sup>89</sup> In particular, OPA has played an active role in providing information and education about powers of attorney. In conjunction with Victoria Legal Aid, OPA has produced *Take control: A kit for making powers of attorney and guardianship.* This kit includes a comprehensive booklet about powers of attorney and a DVD. OPA also provides information on powers of attorney through public presentations and its website, as well as providing information and advice through a telephone advice service. <sup>90</sup>

In addition, OPA investigates, reports and makes recommendations to the Attorney-General about matters under the Guardianship and Administration Act. <sup>91</sup> In 2008-2009 OPA conducted 680 investigations, 53 of which related to enduring powers of attorney (financial). <sup>92</sup>

<sup>&</sup>lt;sup>82</sup> Guardianship and Administration Act 1986 (Vic) s 35D(2).

<sup>&</sup>lt;sup>83</sup> Guardianship and Administration Act 1986 (Vic) s 35E(3).

Guardianship and Administration Act 1986 (Vic) s 35E.

Victorian Civil and Administrative Tribunal Act 1998 (Vic) sch 1 cl 35.

<sup>&</sup>lt;sup>86</sup> Guardianship and Administration Act 1986 (Vic) s 14.

Office of the Public Advocate, Submission 9, 5; Guardianship and Administration Act 1986 (Vic) ss 15-16.

Office of the Public Advocate, *Submission 9*, 6.

John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 2.

Office of the Public Advocate, Submission 9, 5; ibid.

<sup>&</sup>lt;sup>91</sup> Guardianship and Administration Act 1986 (Vic) s 15.

Office of the Public Advocate, Annual report 2008-09 (2009), 16.

## 2.2 What happens if a person does not have enduring powers of attorney?

Many members of the community do not have enduring powers of attorney in place. There are a variety of other mechanisms frequently used to support and assist people when they are unable to make decisions themselves. These range from formal arrangements under the Guardianship and Administration Act to semi-formal and informal arrangements.

## 2.2.1 Appointment of guardians and administrators

When an adult with impaired decision-making capacity does not have an enduring power of attorney any person can make an application to VCAT for the appointment of a guardian and/or an administrator to make decisions on behalf of that person. <sup>93</sup> An administrator makes decisions about legal or financial matters. The powers of a guardian are the same as a representative appointed under an enduring power of attorney (guardianship) and relate to personal matters such as where the person lives. <sup>94</sup> VCAT may give guardians or administrators all such powers or, more commonly, limited, specified powers. <sup>95</sup>

VCAT can make an order appointing a guardian or administrator if satisfied that a person:

- has an intellectual impairment, mental disorder, brain injury, physical disability or dementia
- is unable to make reasonable judgments in respect of his or her person or circumstances (guardianship) or estate (administration)
- is in need of a guardian or administrator. 96

In determining whether a person needs a guardian or administrator, VCAT must take into account matters such as the person's wishes and whether the person's needs could be met by other, less restrictive, means. Further, VCAT can only make these orders if satisfied that it is in the person's best interests.

Guardians and administrators must be at least 18 years old, must not be in a position that conflicts with the interests of the person and must consent to the appointment. Family and friends are often appointed as guardians and administrators. Lawyers, financial advisors and trustee companies such as State Trustees are also frequently appointed as administrators, and the Public Advocate may be appointed as a guardian

<sup>&</sup>lt;sup>93</sup> Guardianship and Administration Act 1986 (Vic) ss 19, 43.

Guardianship and Administration Act 1986 (Vic) ss 24(2), 58B.

Guardianship and Administration Act 1986 (Vic) ss 24, 25, 48.

Guardianship and Administration Act 1986 (Vic) ss 22, 46.

<sup>&</sup>lt;sup>97</sup> Guardianship and Administration Act 1986 (Vic) ss 22, 46.

<sup>&</sup>lt;sup>98</sup> Guardianship and Administration Act 1986 (Vic) ss 23(1), 47(1).

if no other suitable person is available.  $^{99}$  About 60% of all guardianship orders involve the appointment of the Public Advocate as guardian.  $^{100}$ 

If a matter is urgent VCAT may make a temporary order to appoint a guardian or administrator for a period of up to 21 days. <sup>101</sup>

VCAT conducts regular reassessments of all guardianship and administration orders. 102

As with representatives appointed under an enduring power of attorney (guardianship), guardians and administrators are required to act in the best interests of the principal. <sup>103</sup> In addition, administrators are required to submit accounts of the administration of the estate to VCAT on an annual basis. <sup>104</sup> Guardians and administrators may also apply to VCAT for advice about the scope of the orders or the exercise of power under it. VCAT may approve or disapprove any proposed actions or give any advice or orders it considers necessary. <sup>105</sup>

## 2.2.2 Semi-formal arrangements

The use of semi-formal arrangements to assist some members of the community, especially older people, is even more prevalent than the use of formal arrangements such as enduring powers of attorney and guardianship and administration orders. Semi-formal arrangements include making another person a joint signatory to a bank account or a nominee for an agency such as Centrelink. <sup>106</sup>

One study identified that while 15.4% of people aged over 65 years had an enduring power of attorney and 1.4% had a guardianship or administration order, 18.7% used semi-formal arrangements. Formal arrangements were found to be more commonly used to assist those aged over 80 years old.

## 2.2.3 Informal arrangements

Informal arrangements are the most popular way of providing help to people who need assistance with performing tasks and making decisions. In particular, many older people prefer to use informal arrangements based on trust. <sup>108</sup> Informal arrangements involve a person such as a family member or friend making decisions or transactions with, or even for, another person without formal legal authority.

Guardianship and Administration Act 1986 (Vic) s 16.

Office of the Public Advocate, *Community guardianship manual*, 12.

Guardianship and Administration Act 1986 (Vic) ss 32, 33, 59, 60.

Guardianship and Administration Act 1986 (Vic) s 61.

Guardianship and Administration Act 1986 (Vic) ss 28, 49.

Guardianship and Administration Act 1986 (Vic) s 58.

Guardianship and Administration Act 1986 (Vic) ss 55, 30.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 101-104.

Cheryl Tilse, Jill Wilson et al, 'Older people's assets: A contested site' (2005) 24(s1) *Australasian Journal on Ageing* s51, s54; Cheryl Tilse, Deborah Setterlund et al, 'Minding the money: A growing responsibility for informal carers' (2005) 25 *Ageing & Society* 215, 222.

Cheryl Tilse, Jill Wilson et al, above n 107, s54; Cheryl Tilse, Deborah Setterlund et al, above n 107, 223.
See also Rosemarie Draper, Aged Services Development Officer (Equity & Access), New Hope Foundation, Submission 72, 1.

One study found that 9.8% of people aged over 65 had given their ATM Personal Identification Numbers (PIN) to another person, 10.5% had had someone else make an electronic payment on their behalf and 17% had filled in a cheque or withdrawal slip for another person to transact. In addition 49.7% of people assisting an older person had made payments with their own money and then been reimbursed or reimbursed themselves from the person's assets. <sup>109</sup>

Evidence to this Inquiry also suggested that the use of informal arrangements is common. Ms Diane Tate, Director of Financial Services, Corporations, Community Policy with the Australian Bankers' Association, described informal arrangements as 'more prevalent than we would like to accept'. <sup>110</sup> Mr John Hogan, a participant in the Committee's Seniors' Forum, told the Committee:

Anecdotally, amongst people I know, it is very common for them to give details of their PIN numbers or passwords on electronic-based accounts to members of family and you can virtually transact everything these days without a signature except for, say, real estate. You can transfer shares or pay huge bills. In these circumstances a lot of people are seeing that getting a financial power of attorney is too much effort when financial transactions can all be done electronically. <sup>111</sup>

Informal arrangements are relatively easy to abuse, as there are no record keeping, monitoring and accountability requirements associated with these arrangements. Older people are often reluctant to question transactions made using informal arrangements as they prioritise maintaining family relationships over accountability. 112

The Australian Bankers' Association told the Committee that bank customers who use informal arrangements such as giving their PIN to others breach their contract with the bank and forfeit any consumer protection. Mr Ian Gilbert, Director of Retail Policy with the Association, told the Committee 'if you voluntarily disclose your access code, or PIN, to anybody and there is an unauthorised transaction, then you are liable'. 113

## 2.3 The use of powers of attorney

## 2.3.1 How widely are powers of attorney used?

Powers of attorney are essentially private agreements between individuals and therefore there is very limited information about the extent to which they are used.<sup>114</sup>

Evidence to the House of Representatives Standing Committee on Legal and Constitutional Affairs inquiry into Older people and the law suggested that

<sup>&</sup>lt;sup>09</sup> Cheryl Tilse, Deborah Setterlund et al, above n 107, 222-223; Cheryl Tilse, Jill Wilson et al, above n 107, s54.

Diane Tate, Director, Financial Services, Corporations, Community Policy, Australian Bankers' Association Inc, *Transcript of evidence*, Melbourne, 17 December 2009, 3.

John Hogan, *Transcript of evidence*, Melbourne, 30 March 2010, 6.

Cheryl Tilse, Jill Wilson et al, above n 107, s54; Cheryl Tilse, Deborah Setterlund et al, above n 107, 223.

Ian Gilbert, Director, Retail Policy, Australian Bankers' Association Inc, *Transcript of evidence*, Melbourne,
 17 December 2009, 2. See also Australian Bankers' Association Inc, *Submission* 55, 2.

Office of the Public Advocate, *Submission 9*, 7. See also John Chesterman, *Transcript of evidence*, above n 89, 2; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 106, 71; Tony Fitzgerald, Managing Director, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 3-4.

approximately 11% of Australians have an enduring power of attorney. Of those people:

- 8% were under 35 years old
- 45.5% were between 35 and 64 years old
- 45.5% were aged 65 years or older. 115

That report also identified that women, people with secondary or tertiary education and those with higher incomes are more likely to have made an enduring power of attorney. Other evidence suggests that those with disabilities and members of the culturally and linguistically diverse and Aboriginal communities are less likely to have enduring powers of attorney. Here

There is no data available about the level of use of powers of attorney in Victoria. OPA suggested that enduring powers of attorney 'appear to be widely used', based on the number of phone calls to its advice line and requests for the *Take control* kit. State Trustees informed the Committee that it wrote 200 enduring powers of attorney (financial) for clients in 2008-2009 and that it currently administers about 700 of these powers. 119

The Committee heard that enduring powers of attorney (financial) are much more widely used than enduring powers of attorney (guardianship). State Trustees estimated that it writes 20 enduring powers of attorney (financial) for every one enduring power of attorney (guardianship), while the Trustee Corporations Association of Australia put the ratio even higher, at 30 to one. 120

The Committee explores strategies for increasing the uptake of powers of attorney, including in groups with particularly low levels of use, in chapter nine of this report.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 106, 71 citing Office of the Public Advocate, Queensland Government, submission 76 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006), 7.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 106, 71 citing Office of the Public Advocate, Queensland Government, above n 115, 7; Susan Edwards and Antonia Fontana, Legal information needs of older people, Law and Justice Foundation of New South Wales (2004), 310; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation: Limits to informed choice' (2002) 21(3) Australasian Journal on Ageing 128, 130-131; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people Trends & issues in crime and criminal justice, no. 139, Australian Institute of Criminology (1999), 5.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 116, 131; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 116, 5.

Office of the Public Advocate, *Submission 9*, 7. See also John Chesterman, *Transcript of evidence*, above n 89, 2.

Angela Burton, General Manager, Personal Financial Solutions, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 4; Tony Fitzgerald, *Transcript of evidence*, above n 114, 3.

Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 2; Claire Hausler, Manager, TCL Legal Services (Vic) Pty Ltd, Trustee Corporations Association of Australia, *Transcript of evidence*, Melbourne, 22 October 2009, 5. See also Nikki Isaks, Senior Social Worker, Royal District Nursing Service, *Transcript of evidence*, Melbourne, 17 December 2009, 3; Margaret Brown and Suzanne Jarrad, 'Putting "the powers" in place: Barriers for people with memory loss in planning for the future' (2008) 15(4) *Journal of Law and Medicine* 530, 535.

## 2.3.2 The benefits of powers of attorney

Participants in this Inquiry told the Committee that powers of attorney, especially enduring powers, offer a number of benefits to principals, their family and friends, as well as to the community in general.

The Committee heard that powers of attorney provide 'a flexible, low cost, personal and relatively simple' process which allows principals to plan for the future. <sup>121</sup> Many participants in the Inquiry emphasised that powers of attorney promote principals' rights, autonomy and dignity by empowering them to make arrangements for when they are unable to make decisions for themselves. The Mental Health Legal Centre told the Committee that enduring powers of attorney 'exist as a key mechanism by which Victorians with a mental illness can continue to exercise some control over their affairs through the appointment of an attorney to act in their interests'. <sup>122</sup> Ms Laura Helm, Policy Adviser with the Law Institute of Victoria, described powers of attorney as 'an important expression of autonomy'. <sup>123</sup>

Research has revealed that many members of the community, particularly older people, rely on powers of attorney not because they have impaired decision-making capacity, but rather because they are frail, immobile or lack confidence. <sup>124</sup> This was also reflected in evidence to this Inquiry. For instance, Ms Sarnia Birch, Acting Deputy Manager, Regional Offices at Victoria Legal Aid, stated:

the Committee should also be aware of the convenience of having a power of attorney when someone is perhaps physically frail but mentally quite well and who is not able to go and do things like the banking. It is a good idea for them to have someone who can do those sorts of things for them simply with a power of attorney. 125

Powers of attorney also provide substantial benefits to those close to or caring for a principal. Former Victorian Public Advocate, Mr Julian Gardner, said that powers of attorney give principals an opportunity to discuss their wishes with their representative and cited research suggesting that knowing that a person's wishes are respected has significant health benefits for the family and friends of a principal. 126

State Trustees Limited, Submission 58, 3. See also Terry Carney, Professor of Law, The University of Sydney. Transcript of evidence, Melbourne, 1 October 2009, 2.

Mental Health Legal Centre Inc, Submission 59, 2. See also Ministerial Advisory Council of Senior Victorians, Submission 48, 1; Julian Gardner, Submission 21, 1; Law Institute of Victoria, Submission 41, 5; John Chesterman, Transcript of evidence, above n 89, 2; Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 2; Office of the Public Advocate, Submission 9, 7.

Laura Helm, Policy Adviser, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 3.

Cheryl Tilse, Deborah Setterlund et al, above n 107, 220-221. See also Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 116, 130-132; Cheryl Tilse, Jill Wilson et al, above n 107, s55; Julian Gardner, *Submission 21*, 2.

Sarnia Birch, Acting Deputy Manager, Regional Offices, Victoria Legal Aid, *Transcript of evidence*, Melbourne, 17 December 2009, 7. See also Julian Gardner, *Transcript of evidence*, Melbourne, 1 October 2009, 2.

Julian Gardner, Submission 21, 2. See also Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 5.

If a person does not have an enduring power of attorney, an application may have to be made to VCAT for a guardianship and/or administration order. Participants in this Inquiry saw that process as highly stressful for all involved, especially those close to a principal. 127

Finally, powers of attorney also provide substantial benefits to the community as a whole. Powers of attorney reduce the cost to the community by avoiding reliance on VCAT and public administration and guardianship services. 128 They also reduce the cost to both government and non-government agencies such as health services and banks by providing certainty about who can make decisions on a principal's behalf. 129

#### 2.3.3 Barriers to making powers of attorney

Evidence to the Committee suggests that powers of attorney are not as widely used in Victoria as they could be. 130 The take-up rate is influenced by a variety of factors including lack of knowledge and understanding, the complexity of the documents, a reluctance to face death and disability, cost, family dynamics and fear of abuse.

#### Lack of awareness and understanding

Participants in the Inquiry emphasised that many members of the community are not aware of powers of attorney. Dr John Chesterman, Manager of Policy and Education at OPA, told the Committee:

The barriers of knowledge are just that they exist. Often the first time a person hears about the need for an enduring power of attorney is when their parent, for example, is on the cusp of not being able to make their own decisions and we have a disproportionate number of situations where we experience that ... There is the problem of just general knowledge about them ... We regularly come across situations where people, had they thought about it and had they been able to get their affairs in order, would have signed an enduring power of attorney.<sup>131</sup>

Research has identified that some groups in the community, such as members of culturally and linguistically diverse communities, people with disabilities and those on low incomes, have particularly low levels of awareness and understanding of powers of attorney. 132

The Committee also heard that even when members of the community are aware of powers of attorney, they might not understand the benefits of having these

Palliative Care Victoria, Submission 70, 2; Office of the Public Advocate, Submission 9, 7; Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 4.

Julian Gardner, Submission 21, 1; Office of the Public Advocate, Submission 9, 7.

Julian Gardner, Submission 21, 1.

John Billings, Submission 37, above n 70, 2; Terry Carney, Transcript of evidence, above n 121, 2.

John Chesterman, Transcript of evidence, above n 89, 5-6. See also Julian Gardner, Transcript of evidence, above n 125, 3; Catherine Leslie, Transcript of evidence, above n 122, 2-3.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 116, 3; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 116, 130.

arrangements in place or know how to go about making them. <sup>133</sup> People on lower incomes and those who have few assets may see limited benefits in making formal legal arrangements for managing their affairs in the event of impaired decision-making capacity. In particular, they may not understand that these arrangements can authorise decisions about personal matters as well as legal and financial matters. <sup>134</sup>

The Committee considers strategies for raising awareness and understanding about powers of attorney in chapter nine of this report.

#### Complexity

Evidence to the Inquiry also suggested that the complexity of the current arrangements for making powers of attorney in Victoria discourages uptake. Factors such as the number of different forms, the differing formal requirements and the use of legal terminology on the forms can be confusing for many people. <sup>135</sup>

Strategies for making powers of attorney easier to create are explored in chapter four of this report.

## **Psychological barriers**

Participants in the Inquiry also identified a number of psychological barriers that prevent members of the community making powers of attorney. Some people may fear dealing with lawyers or legal documents. Many members of the community find contemplating future disability and death extremely confronting. Younger people in particular may find it difficult to deal with morbidity and mortality issues. 138

Mr Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, described people as 'fatalistic'. He informed the Committee that when he delivers education sessions:

we normally talk about the concept that life is unpredictable. A person will be found to make decisions for you one way or another. You can take control, fill out the forms now, or if something happens to you, VCAT will make that decision. The

Julian Gardner, *Transcript of evidence*, above n 125, 6; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 1; Jenny Chapman, *Transcript of evidence*, above n 127, 4; Julian Gardner, *Submission 21*, 3.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, *Substitute decision making and older people*, above n 116, 3-5; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 116, 131.

John Billings, Transcript of evidence, above n 41, 2; Julian Gardner, Submission 21, 3; Julian Gardner, Transcript of evidence, above n 125, 3; Nikki Isaks, Transcript of evidence, above n 120, 5; Dale Reddick, Transcript of evidence, above n 126, 10; Terry Carney, Transcript of evidence, above n 121, 2; Alzheimer's Australia Vic, Submission 32, 1; Ministerial Advisory Council of Senior Victorians, Submission 48, 1; Jenny Chapman, Transcript of evidence, above n 127, 4; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 116, 5; State Government of Victoria, submission 121 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2007), 24.

Julian Gardner, Submission 21, 3.

Julian Gardner, Submission 21, 3; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation' above n 116, 132; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 116, 3-4.

Jenny Chapman, *Transcript of evidence*, above n 127, 7.

problem with the VCAT decision is that you will have little or no input into that. ... They would rather go with the flow ... [and let VCAT make an appointment if necessary]<sup>139</sup>

#### Cost

The cost of making a power of attorney or obtaining professional advice is also a barrier that stops some people making powers of attorney. While people are able to make these documents themselves using the template documents in *Take control*, evidence suggests that many people find making these documents difficult without professional assistance. 141

The Committee examines strategies for making legal advice more widely available to people wishing to make a power of attorney in chapter four.

### **Family dynamics**

Some people do not make powers of attorney because they trust that their family members will manage their affairs informally in their best interests. Some members of the community, particularly some cultural groups, have different notions of decision making where it is assumed that members of the younger generation will take on this responsibility. 143

Other members of the community may be reluctant to make powers of attorney because of uncertainty about the involvement of an outsider in decision making in the event that a son or daughter marries. <sup>144</sup> There may also be a reluctance to talk about wills and other decisions about personal affairs because it may create 'hurt, tension and change of family roles'. <sup>145</sup>

#### Fear of abuse

People may also be reluctant to make powers of attorney because they fear losing control of their affairs and are concerned about the potential for abuse. 146

Mr Reddick told the Committee that, when informed of the additional accountability requirements for guardians and administrators, many of his clients elect to not appoint representatives, preferring to rely on a possible future VCAT appointment. 147

Dale Reddick, *Transcript of evidence*, above n 126, 5. See also Jenny Chapman, *Transcript of evidence*, above n 127, 7-8.

Hume Riverina Community Legal Service, *Submission 36*, 3; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 5; Latrobe Regional Hospital, *Submission 44*, 2.

Phillip Hamilton, Submission 7, 1-3; Law Institute of Victoria, Submission 41, 13; Southport Community Legal Service Inc, Submission 43, 1.

Cheryl Tilse, Deborah Setterlund et al, above n 107, 223.

Eva Wakim, Victorian Arabic Social Services, Transcript of evidence, Melbourne, 30 March 2010, 4; Virginia Landrito-Balanon, Community Aged-Care Packages Case Manager, Filipino Community Council of Victoria Inc, Transcript of evidence, Melbourne, 30 March 2010, 5-6. See also Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 116, 3.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, *Substitute decision making and older people*, above n 116, 3.

<sup>&</sup>lt;sup>146</sup> Ibid; Kris Spark, Manager, Seniors Information Victoria, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 7.

Throughout this report the Committee identifies strategies for reducing the abuse and potential abuse of powers of attorney, particularly enduring powers. The Committee hopes that the implementation of its recommendations will give more members of the community peace of mind that their rights will be protected if they choose to make powers of attorney.

## 2.4 The abuse of powers of attorney

## 2.4.1 The level of abuse of powers of attorney

There is very little data available about the extent of abuse of power of attorney documents. OPA told the Committee the fact that these are private documents, with no accountability requirements makes them relatively easy to abuse and the level of abuse difficult to assess. 148

One often-quoted article from the United States claimed, 'Powers of attorney may be the single most abused legal documents in our judicial system', although did not provide data to support this assertion. Similarly, a police officer interviewed as part of a study of financial issues for older people in central Victoria viewed 'abuse of powers of attorney as the biggest fraud problem facing older people in our community at the present time'.

The Master of the Court of Protection in the United Kingdom estimated that between 10% and 15% of registered powers of attorney in that jurisdiction were subject to financial abuse. One New Zealand study found even higher rates of abuse – up to 24%. 152

There has not been any comprehensive research in Australia about the abuse of powers of attorney. However, several Australian studies have concluded on the basis of anecdotal evidence that most powers of attorney work well and are not subject to abuse. <sup>153</sup> This view was also generally supported by Inquiry participants. <sup>154</sup>

Dale Reddick, *Transcript of evidence*, above n 126, 10.

Office of the Public Advocate, Submission 9, 7. See also Jo Wainer, Peteris Darzins and Kei Owada, Prevalence of financial elder abuse in Victoria. Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2010), 29; Lillian Jeter, Executive Director, Elder Abuse Prevention Association, Transcript of evidence, Melbourne. 30 March 2010, 4-5.

Mathis E McRae, 'Policing the guardians: Combating guardianship and power of attorney fraud' (1994) 63(2) *The FBI Law Enforcement Bulletin* 1, 2.

Frances Gibson, Financial & consumer credit issues for older consumers in central Victoria, report for Loddon Campaspe Community Legal Centre (2008), 24.

Denzil Lush, 'Taking liberties: Enduring powers of attorney and financial abuse' (1998) Solicitors Journal 808, 808

Age Concern New Zealand, Elder abuse and enduring power of attorney: A special report from the Age Concern New Zealand Elder Abuse and Neglect Database covering the period 1 July 2002 to 31 December 2003 (2004), 28.

Land and Property Management Authority, New South Wales, *Review of the Powers of Attorney Act 2003*, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009), 5; Georgia Lowndes et al, *Financial abuse of elders: A review of the evidence*, Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2009), 4.

Law Institute of Victoria, Submission 41, 15; Office of the Public Advocate, Submission 9, 9; Mental Health Legal Centre Inc, Submission 59, 6; John Chesterman, Transcript of evidence, above n 89, 2.

Nevertheless, the Committee heard that abuse of enduring powers of attorney was 'not uncommon'. The Murray Mallee Community Legal Service advised that in the past five years 'a significant number of clients sought advice in relation to an attorney having abused the power given to them'. OPA stated that it 'regularly' finds incidences of abuse, with most of these coming to light through its telephone advice service and investigations. Ms Dahni Houseman, Seniors Rights Victoria's Policy Officer, told the Committee that between April 2008 and October 2009 her organisation had come across 113 financial abuse cases which involved powers of attorney or guardianship and administration. 158

As noted earlier in this chapter, VCAT receives about 400 applications related to enduring powers of attorney (financial) each year; although most relate to cases of abuse it is not clear exactly how many. State Trustees indicated that 'not infrequently' VCAT appoints it as an administrator in cases where a representative 'has, whether intentionally or otherwise, caused a significant depletion of the donor's estate'. Seniors Rights Victoria argued, 'The number of cases brought before VCAT in which issues of record keeping or abuse of powers are raised is highly disproportionate to the anecdotal and statistical evidence of the number of cases of abuse. 161

Research suggests that older people who have a power of attorney are no less likely to be the victims of abuse, particularly financial abuse, than people who rely on the informal arrangements discussed earlier in this chapter. <sup>162</sup>

## 2.4.2 How are powers of attorney abused?

While there is no research on how abuse of powers of attorney is perpetrated, the Committee heard that enduring powers are the most susceptible to abuse. In addition, the abuse of a valid power of attorney appears to be more common than pressure to sign a power of attorney document, or the use of a power of attorney that has been revoked. <sup>163</sup>

Evidence to the Inquiry suggested enduring powers of attorney (financial) are the most frequently abused type of powers of attorney. OPA's submission explained why this is the case:

Seniors Rights Victoria, Submission 38, 34; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2-3. See also Julian Gardner, Submission 21, 4; Phillip Hamilton, Submission 7, 3; Jeni Lee, Submission 57, 1; State Trustees Limited, Submission 58, 12; Combined Pensioners & Superannuants Association of Victoria Inc, Submission 10, 1.

<sup>&</sup>lt;sup>156</sup> Murray Mallee Community Legal Service, Submission 22, 3.

Office of the Public Advocate, Submission 9, 7.

Dahni Houseman, Policy Officer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 2.

John Billings, *Transcript of evidence*, above n 41, 3.

State Trustees Limited, Submission 58, 11.

Seniors Rights Victoria, Submission 38, 36.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 106, 19, 81; Cheryl Tilse, Jill Wilson and Deborah Setterlund, 'The mismanagement of the assets of older people: The concerns and actions of aged care practitioners in Queensland' (2003) 22(1) *Australasian Journal on Ageing* 9, 13.

Office of the Public Advocate, Submission 9, 7; John Chesterman, Transcript of evidence, above n 89, 4, 8.

Julian Gardner, Submission 21, 4; Royal District Nursing Service, Submission 25, 4; Seniors Rights Victoria, Submission 38, 11, 13; Lillian Jeter, Transcript of evidence, above n 148, 5.

currently de facto 'point of use' safeguards exist in the operation of EPGs [enduring powers of attorney (guardianship)] and medical treatment EPAs [enduring powers of attorney (medical treatment)]. The exercise of these EPAs are constrained to some extent in ways that financial EPAs are not, largely because the number of situations in which they can be used is much smaller than exists for financial EPAs. Hospital staff will often readily be able to see that a medical treatment EPA is being abused, and an EPG's use in relation to health care or accommodation is similarly only feasibly used with a small number of service providers. In addition, an EPG and EPA (Medical Treatment) are far less able than an EPA (Financial) to be abused to the material benefit of an attorney. <sup>165</sup>

Participants in this Inquiry identified a range of examples of financial abuse. Common types of abuse highlighted in evidence to the Committee include:

- transferring the principal's house or other property to the representative
- mortgaging the principal's house or other property
- paying the representative's household expenses with the principal's funds
- making 'gifts' to the representative using the principal's funds
- not spending money on care the principal needs. 166

Another frequently cited type of abuse was the representative acting outside the scope of his or her powers by using an enduring power of attorney (financial) to make lifestyle decisions. <sup>167</sup> This is discussed in detail in chapter six.

The case studies set out throughout this report contain many real life examples of abuse of powers of attorney.

Not all abuse is easily labelled as such. Mr Tony Fitzgerald, the Managing Director of State Trustees, told the Committee:

What is the definition of abuse? There might be a situation where a daughter or a son is looking after a particular parent. You know, they do a lot of running around for the client in terms of collecting income, paying bills, which are your responsibilities under the power of attorney. They might take \$10 or \$20 out of Mum's money to pay for petrol. Is that abuse of the financial power of attorney, because they did not specifically ask the parent for it, or not? ... if somebody has transferred Mum's house into the name of the daughter, because the daughter has the attorney, and they sell the house and use the money for their own ends, that is clearly abuse. But there are some grey areas where you could say that there is

Office of the Public Advocate, *Submission* 9, 17.

For example, Royal District Nursing Service, Submission 25, 4; State Trustees Limited, Submission 58, 12-13; Council on the Ageing Victoria, Submission 39, 2-3; Seniors Rights Victoria, Submission 38, 31-32; Office of the Public Advocate, Submission 9, 9; Murray Mallee Community Legal Service, Submission 22, 3; Jeni Lee, Elder Law Committee and Disability Law Committee, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 7; Elizabeth Samra, Lawyer, Seniors Rights Victoria, Transcript of evidence, Melbourne, 22 October 2009, 3. See also Cheryl Tilse, Jill Wilson and Deborah Setterlund, above n 162, 11.

Seniors Rights Victoria, Submission 38, 35; Jeni Lee, Submission 57, 2; Stephen Taffe, Legal Counsel, Alfred Health, Transcript of evidence, Melbourne, 14 December 2009, 2; Sue Connelly, Committee member, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 6.

justification around spending a bit of petrol money and that kind of thing for Mum, because I am running around doing things for Mum  $\dots$  In most cases most people would say that taking a bit of petrol money might be fine, but in the strict sense of the law you are actually taking it without permission.  $^{168}$ 

Several Inquiry participants observed that not all abuse is intentional. State Trustees' submission highlighted:

The attorney may believe they are acting in accordance with the donor's wishes, or that their actions would be seen as "fair" in the context of the surrounding circumstances; they often believe the enduring power of attorney (financial) is intended to give them the authority to do these things. 169

The Committee considers strategies for educating representatives, service providers and other members of the community about the scope of representatives' powers and duties in chapters six and seven respectively.

## 2.4.3 Who are the victims of abuse of powers of attorney?

Most victims of abuse of powers of attorney are older people. 170

There is increasing awareness and concern about the abuse of older people in the community. It is estimated that between one and five per cent of older Victorians are victims of elder abuse. <sup>171</sup> The abuse of older people may take many forms including psychological and physical, but financial abuse is the most common. Many participants in this Inquiry highlighted that powers of attorney are used to perpetrate elder abuse.

The Council on the Ageing Victoria's submission highlighted that there are two groups particularly vulnerable to financial abuse:

- women over 75 years of age who live by themselves. Many of these women are unfamiliar with financial management as they have relied on their partner or husband to deal with these matters
- people with dementia. People with cognitive impairments may not understand the details and implications of particular financial decisions. 172

Changes to the demographic makeup of the community noted in the last chapter, in particular increasing longevity and rates of dementia, mean there will be even larger numbers of people in both of these groups in the future.

Tony Fitzgerald, *Transcript of evidence*, above n 114, 7-8. See also Angela Burton, *Transcript of evidence*, above n 119, 7.

State Trustees Limited, Submission 58, 13-14. See also Seniors Rights Victoria, Submission 38, 12, 32; Sally Costar, Manager, Patient and Family Services, Caulfield Hospital, Alfred Health, Transcript of evidence, Melbourne, 14 December 2009, 3.

Office of the Public Advocate, *Submission 9*, 7; Seniors Rights Victoria, *Submission 38*, 11; Council on the Ageing Victoria, *Submission 39*, 2-3; Age Concern New Zealand, above n 152, 11-12.

Office of Senior Victorians, *Rights. Respect. Trust.* Victorian Government Elder Abuse Prevention Strategy (2009), 3-4; Office of Senior Victorians, *Strengthening Victoria's response to elder abuse* Report of the Elder Abuse Prevention Project (2005), 12-14.

<sup>172</sup> Council on the Ageing Victoria, Submission 39, 3. See also Alzheimer's Australia Vic, Submission 32, 2; Jo Wainer, Peteris Darzins and Kei Owada, above n 148, 6, 29.

Other risk factors for the financial abuse of elders include social isolation and being dependent on other people. <sup>173</sup>

However, not all victims of abuse of powers of attorney are older people. The Committee received very limited information about the extent to which younger people are victims of abuse. The Victorian Coalition of Acquired Brain Injury (ABI) Service Providers stated that abuse of powers of attorney in relation to people with ABI, many of whom are young, is 'not uncommon'. 174

The abuse of powers of attorney can have a severe negative impact on a principal's quality of life. Professor Adam Graycar and Ms Marianne James from the Australian Institute of Criminology have observed:

The impact of economic crime has a devastating effect on older people. Not only can a comfortable lifestyle collapse, but they do not have the time or opportunity for financial recovery, and a blow to financial security is often a permanent and life threatening setback, characterised by fear, lack of trust and the onset, often of acute and chronic anxiety. <sup>176</sup>

## 2.4.4 Who are the perpetrators of abuse of powers of attorney?

Research both in Australia and internationally has identified that elder abuse, in particular abuse using a power of attorney, is most commonly perpetrated by a close family member, usually a son or a daughter. 177

Participants in this Inquiry also identified family members as the most frequent abusers of powers of attorney. OPA's submission suggested that most abuse occurs when representatives 'have a sense of entitlement to a parent's assets. Where an attorney is acting for a parent, the misguided justification can be made that the money spent by an attorney on themselves would eventually be theirs anyway, through inheritance'. 178

A recent Victorian study found that it is not clear from the currently available data whether people who perpetrate financial abuse against older people do so deliberately or whether the abuse is accidental, resulting from 'a culmination of a series of small improper decisions or sloppy practices'. <sup>179</sup>

<sup>173</sup> Georgia Lowndes et al, above n 153, 6; Age Concern New Zealand, above n 152, 22.

Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2-3.

State Trustees Limited, Submission 58, 11.

Adam Graycar and Marianne James, 'Crime and older Australians: Understanding and responding to crime and older people' (Paper presented at the Family Futures: Issues in Research and Policy - 7th Australian Institute of Family Studies Conference, Sydney, 24-26 July 2000), 7. See also Georgia Lowndes et al, above n 153, 12.

National Center on Elder Abuse, The American Public Human Services Association, *The National Elder Abuse Incidence Study* Final report, report for U.S. Department of Health and Human Services (1998), 7, 4-29; Office of Senior Victorians, *Strengthening Victoria's response to elder abuse*, above n 171, 13; Age Concern New Zealand, above n 152, 18; Anne-Louise McCawley et al, 'Access to assets: Older people with impaired capacity and financial abuse' (2006) 8(1) *The Journal of Adult Protection* 20, 28; Georgia Lowndes et al, above n 153, 15; Jo Wainer, Peteris Darzins and Kei Owada, above n 148, 29.

Office of the Public Advocate, *Submission 9*, 9. See also Seniors Rights Victoria, *Submission 38*, 12; Murray Mallee Community Legal Service, *Submission 22*, 3; Mental Health Legal Centre Inc, *Submission 59*, 6.

Jo Wainer, Peteris Darzins and Kei Owada, above n 148, 30.

# 2.5 Weighing up the benefits and risks of powers of attorney

Several participants in the Inquiry noted the tension between the potential benefits that powers of attorney offer and the danger that the arrangements will facilitate the abuse of the principal. However, all those who commented on this issue came to the conclusion that the overall value of powers of attorney outweighs the risk of abuse. For example, the Law Institute of Victoria stated:

It appears that there are a large number of enduring powers of attorney in place in society and in the main these work well and without problems or abuse and provide great benefits to a large number of people. By comparison it seems that the numbers of enduring powers of attorney suffering abuse would be a very small percentage of the overall number of enduring powers of attorney in use. 180

Similarly, OPA wrote 'the value of EPAs [enduring powers of attorney] outweighs the dangers presented by them for abuse'. 181

The Committee concurs with the view that the potential benefits that powers of attorney offer all Victorians significantly outweigh the risk of abuse. In particular, powers of attorney promote the rights of principals, providing a mechanism by which they can take control of their future.

The remainder of this report focuses on promoting the use of powers of attorney by increasing awareness about these documents and making them easy to use, while at the same time providing protections to ensure that principals are not vulnerable to abuse.

Office of the Public Advocate, *Submission 9*, 9. See also Mental Health Legal Centre Inc, *Submission 59*, 6; John Chesterman, *Transcript of evidence*, above n 89, 2.

Law Institute of Victoria, Submission 41, 15.

## Case study 1: She 'was unable to explain what type of power of attorney document she had executed or when' 182

'Mrs Richards is a 94 year old widow who lives alone in a small country town three hours from Melbourne. She has relatively good health, but has limited communication skills due to poor hearing and ageing. Mrs Richards has lived in the town for over 65 years and there are no longer any close relatives nearby. Mrs Richards was assisted by her son Geoff who had regularly commuted from Melbourne to deal with household chores and manage financial matters. She also has two daughters one in Melbourne and the other interstate. A general power of attorney had been given appointing Geoff and his sister, Laura and they assisted intermittently with matters that Mrs Richards could not deal with particularly as her ability to communicate worsened.

Sometime in 2005, a distant niece, Jan, suddenly increased her level of contact with Mrs Richards and assisted her from time to time with attending appointments and purchasing groceries. Jan suggested that she could purchase the neighbouring block of land owned by Mrs Richards to help care for Mrs Richards. Despite objections from her children, Mrs Richards transferred her block of land for minimal consideration to Jan. Mrs Richards felt sorry for Jan as Jan had told her she had a difficult life, bad marriage and financial problems. Jan then built a comfortable and modern home on the block of land next to Mrs Richards. Jan also used the arrangement to obtain carer's benefits from Centrelink on the basis that she was a carer of Mrs Richards, when in fact she merely dropped off milk to her neighbour and occasionally drove her to a neighbouring town for appointments.

Geoff discovered that at some point his mother had executed a power of attorney document appointing Jan as the Donee [representative]. And that Jan therefore had access to Mrs Richards' banking and financial information. All of the family were not aware that this meant that Geoff and his sister no longer had the power to act in their mother's best interests and assist her. Even Mrs Richards was not aware of the impact this action had on pre-existing documents as she continued to want her son to assist her. Mrs Richards was unable to explain what type of power of attorney document she had executed or when. The family were concerned that it could have been an enduring power of attorney and if so that it would be dangerous if Mrs Richards lost capacity. The family have been unable to find out exactly what power has been given to Jan and are concerned that Jan is not acting in Mrs Richards's best interests and is acting to benefit herself financially. There was also mention by Mrs Richards that she had been taken by her niece to change her will.'

-

<sup>&</sup>lt;sup>82</sup> Case study provided to the Committee by a person who requested that his/her name be withheld.

# Chapter 3: A legislative framework for powers of attorney

In this chapter the Committee considers the appropriate legislative regime for powers of attorney. Inquiry participants described the current Victorian laws as fragmented and confusing. The Committee's recommendations aim to create a simple, consolidated legislative framework which makes powers of attorney easy to use and accessible and promotes the rights of people making these documents.

## 3.1 What should the legislative framework look like?

Two separate pieces of legislation currently govern powers of attorney in Victoria: the *Instruments Act 1958* (Vic) and the *Guardianship and Administration Act 1986* (Vic). A third piece of legislation, the *Medical Treatment Act 1988* (Vic), provides for enduring powers of attorney (medical treatment) which the Committee is not considering in this Inquiry.

Many participants in the Inquiry called the current legislative arrangements confusing and urged the adoption of either a consolidated Victorian or national legislative approach to powers of attorney.

## 3.1.1 Consolidating powers of attorney legislation

The need to consolidate powers of attorney legislation in Victoria was a common theme in the submissions and evidence to the Committee. For example, Palliative Care Victoria's submission stated, 'Currently there are four different types of POA [powers of attorney], all governed by separate Acts of Parliament ... This causes confusion, distress and inconvenience and restricts uptake of this essential advance care planning process.' 183

Queensland and the Australian Capital Territory (ACT) have enacted consolidated powers of attorney legislation, covering the appointment of representatives in relation to both financial and personal matters. All other Australian jurisdictions, with the exception of Victoria and Western Australia, have introduced stand-alone powers of attorney legislation covering general (non-enduring) powers of attorney and enduring powers of attorney (financial). 185

Palliative Care Victoria, Submission 70, 1. See also Alfred Health, Submission 66, 2; Law Institute of Victoria, Submission 41, 5; Health Services Commissioner, Submission 46, 2; Elder Rights Advocacy, Submission 63, 2; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3; Office of the Public Advocate, Submission 9, 10; Murray Mallee Community Legal Service, Submission 22, 1; Seniors Rights Victoria, Submission 38, 18; Mental Health Legal Centre Inc, Submission 59, 4.

Powers of Attorney Act 2006 (ACT); Powers of Attorney Act 1998 (Old).

Powers of Attorney and Agency Act 1984 (SA); Powers of Attorney Act 2000 (Tas); Powers of Attorney Act 1980 (NT).

Inquiry participants commended aspects of legislation in the United Kingdom (UK), <sup>186</sup> the ACT<sup>187</sup> and Queensland. <sup>188</sup> However, the evidence favoured selecting certain features of the legislation in those jurisdictions, rather than using any single piece of legislation as a model. Mr John Billings, Deputy President of the Guardianship List at the Victorian Civil and Administrative Tribunal (VCAT) told the Committee, 'Frankly I think the ideal model piece of legislation is yet to be drafted ...' <sup>189</sup>

Participants identified a number of benefits of amalgamating all laws relating to powers of attorney into a single act. Ms Toni Higgins, Specialist Mental Health, Human Rights and Civil Law at Victoria Legal Aid, stated:

it makes it easier for our clientele, who normally have some sort of a disability or language problem, if it is easily accessible in the one piece of legislation, and probably easier to translate, if that is necessary, into different languages, and probably economically more efficient if it is in the one piece of legislation. <sup>190</sup>

Professor Terry Carney, Professor of Law at the University of Sydney, told the Committee that one of the key benefits of consolidated legislation is that it can form a platform for educating key stakeholders:

it is much easier to run an education campaign for the professionals and others who need to know what the criteria and arrangements are, if you can say there is just this one piece of legislation and we will take you through it today. If you tell people that the law is to be found in three or four different places, that does not facilitate them having confidence in it, or learning what they need to know about how to utilise it. <sup>191</sup>

While most participants argued for a stand-alone powers of attorney act, Professor Carney expressed the view that the Guardianship and Administration Act is the most appropriate location for all laws about enduring powers of attorney. 192

Law Institute of Victoria, *Submission 41*, 9-10; Laura Helm, Policy Adviser, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 2; Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, *Transcript of evidence*, Melbourne, 17 December 2009, 2.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Transcript of evidence, Melbourne, 1 October 2009, 5; John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 9-10; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, 9.

Debra Parnell, Policy Officer, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 2-3; Elizabeth Samra, Lawyer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 3; John Billings, *Transcript of evidence*, above n 187, 5; John Chesterman, *Transcript of evidence*, above n 187, 9-10; Robert Shepherd, Barrister, The Victorian Bar, *Transcript of evidence*, Melbourne, 22 October 2009, 4; John Billings, *Submission 37*, above n 187, 9; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 2.

John Billings, Transcript of evidence, above n 187, 5. See also Margaret Brown, Adjunct Research Fellow, Hawke Research Institute, University of South Australia, Alzheimer's Australia Vic, Transcript of evidence, Melbourne, 22 October 2009, 3.

Toni Higgins, Specialist Mental Health, Human Rights and Civil Law, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 2.

Terry Carney, Professor of Law, The University of Sydney, *Transcript of evidence*, Melbourne, 1 October 2009, 4.

<sup>&</sup>lt;sup>192</sup> Ibid.

The Law Institute of Victoria suggested there should be a comprehensive review, with a view to implementing a single legislative regime for all substitute decision making. <sup>193</sup> Such legislation has been introduced in the UK. <sup>194</sup>

There was also strong support among participants for consolidated powers of attorney legislation to also include enduring powers of attorney (medical treatment) and, potentially, advance care directives. <sup>195</sup> Legislation to this effect has been enacted in Queensland and has recently been recommended in South Australia. <sup>196</sup>

The Committee agrees that a single statutory framework for powers of attorney in Victoria would make it easier for members of the community, as well as businesses and professionals dealing with these documents, to understand and use powers of attorney. It recommends that one piece of legislation should cover general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). The Committee notes suggestions that enduring powers of attorney (medical treatment), which the Committee is not considering in this review, might also naturally fit within the scope of a consolidated powers of attorney act.

The Committee also recognises that powers of attorney are just one area falling under the umbrella of substitute decision making. There may be merit in incorporating power of attorney laws within a broader, capacity-based legislative framework, as in the UK. This is outside the scope of the present Inquiry, but the Committee encourages the Victorian Government to consider this further, subject to the outcomes of the current Victorian Law Reform Commission review of the Guardianship and Administration Act.

#### **Recommendation 1: A Powers of Attorney Act**

The Committee recommends the Victorian Government draft a Powers of Attorney Act which contains all laws about general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship).

## 3.1.2 A national approach to powers of attorney?

Many Inquiry participants advocated for a national approach to powers of attorney. Australia's population is increasingly mobile and inconsistent state-based power of attorney laws may be confusing and inconvenient for both individuals and businesses.

Alfred Health, Submission 66, 2; Victoria Legal Aid, Submission 42, 2-3; Royal College of Nursing Australia, Submission 26, 2; Mental Health Legal Centre Inc, Submission 59, 4; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3.

Law Institute of Victoria, *Submission 41*, 9-10. See also Moreland Community Legal Centre Inc, *Submission 51*, 4.

Mental Capacity Act 2005 (UK) c 9.

Advance Directives Review Committee, Planning ahead: Your health, your money, your life. First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy (2008), 26-27; Powers of Attorney Act 1998 (Qld) s 35, sch 2 part 2.

The Association of Independent Retirees emphasised that national consistency is particularly important to many senior Victorians:

interstate travel is a major pastime in retirement. Trauma resulting in loss of legal capacity can result from car accidents and some of the pastimes we enjoy while on holidays  $\dots$  Documentation should be instantly recognizable and enforceable no matter where the need arises. <sup>197</sup>

Mr Richard Fielding, Principal Examiner, Powers of Attorney/General Law at the Tasmanian Land Titles Office, informed the Committee that state-based legislation results in business uncertainty, for example, property settlements may be delayed if a power of attorney document does not meet the registration requirements in another state. Further, he commented:

Nursing homes, hospitals, doctors, welfare agents and people dealing with estates of the elderly quite often enquire as to their duty in dealing with foreign [non-Tasmanian] Powers of Attorney forms. At times execution under these forms is not considered valid by a third party and a complete new form is required to be registered which by that late stage the donor may not be able to execute because of health reasons. <sup>198</sup>

The Hume Riverina Community Legal Service's submission highlighted that people living in border areas are particularly confused about inconsistent state-based laws and concerned that a power they have created in one state may not be valid in another.<sup>199</sup>

The Law Institute of Victoria told the Committee that a uniform national system of powers of attorney could be accomplished by either a cooperative state-based scheme, whereby all states adopt template legislation, or through the introduction of Commonwealth legislation. The Law Institute's view is that the Australian Government already has the Constitutional power to enact national powers of attorney legislation. Alternatively, states could agree to transfer their powers in this area to the Commonwealth.

This Inquiry found support for both a uniform state-based regime  $^{201}$  as well as national legislation.  $^{202}$ 

Letter from Richard Fielding, Principal Examiner, Powers of Attorney/General Law, Land Titles Office, Tasmania, to Executive Officer, Victorian Parliament Law Reform Committee, 19 August 2009, 6.

Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3. See also National Seniors Australia, Submission 54, 1; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2.

Hume Riverina Community Legal Service, Submission 36, 3. See also Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 3.

Law Institute of Victoria, Submission 41, 15-16 referring to s 51(xxxiiiA) of the Commonwealth of Australia Constitution Act 1900 which permits the Australian Government to make laws about 'the provision of maternity allowances, widows' pensions, child endowment, unemployment, pharmaceutical, sickness and hospital benefits, medical and dental services ... benefits to students and family allowances'. Cf Berna Collier and Shannon Lindsay, Powers of attorney in Australia and New Zealand (1992) The Federation Press, 13.

National Seniors Australia, Submission 54, 1; MF Spottiswood, Submission 45, 3-4; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 3; Hume Riverina Community Legal Service, Submission 36, 3; Seniors Rights Victoria, Submission 38, 17; Council on the Ageing Victoria, Submission

The Australian Guardianship and Administration Council, a national body made up of public advocates, public and adult guardians, guardianship boards and tribunals and public trustees, has proposed to both the Federal and Victorian Attorneys General that there be national legislation for a single power of attorney document which covers financial, personal and healthcare decisions. Mr Billings advised, 'National legislation could exist side by side with State and Territory legislation ... Eventually, if the national document were sufficiently popular, it could overtake existing State and Territory enduring powers.' <sup>204</sup>

The House of Representatives Standing Committee on Legal and Constitutional Affair's 2007 report on *Older people and the law* recommended that the Australian Government and the Standing Committee of Attorneys-General work towards the implementation of uniform legislation on powers of attorney across the states and territories.<sup>205</sup> In the interim it suggested that the effectiveness of mutual recognition provisions should be monitored and enhanced.

National Seniors Australia cautioned that, in reforming its state laws, 'Victoria should not unduly divert from Australia-wide principles relating to Powers of Attorney, as this may lead to future complications and even less portability of a Power of Attorney executed or exercised in Victoria'. On the other hand, while supporting national uniformity, State Trustees warned, 'Uniformity should not be at the expense of sensible and appropriate laws.'

The Office of the Public Advocate noted that one of the benefits of federalism is that jurisdictions can trial different initiatives. Other submissions suggested that Victoria has the opportunity to provide leadership by formulating legislation that could act as a template for uniform national legislation. <sup>209</sup>

The Committee recognises the efficiency and benefits that a national approach to powers of attorney will provide. It believes Victoria should play a leading role in advocating for uniform power of attorney laws through the Standing Committee of Attorneys-General.

However, the Committee recognises that the national harmonisation of power of attorney laws is likely to be a protracted process. Therefore, through the recommendations in this report, the Committee aims to make recommendations whose implementation will lead to a best practice model for powers of attorney legislation, which could provide a template for legislation throughout Australia. In

<sup>39, 5;</sup> Office of the Public Advocate, *Submission 9*, 23; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Submission 29*, 2.

Financial Planning Association of Australia Ltd, Submission 49, 1; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3; Alzheimer's Australia Vic, Submission 32, 6.

John Billings, Submission 37, above n 187, 3.

<sup>&</sup>lt;sup>204</sup> Ibid. See also John Billings, *Transcript of evidence*, above n 187, 2; Grant Sturgeon, *Submission 35*, 1; Carers Victoria, *Submission 65*, 9.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 79-80.

National Seniors Australia, Submission 54, 1.

State Trustees Limited, Submission 58, 3.

Office of the Public Advocate, Submission 9, 23.

John Billings, Submission 37, above n 187, 4; State Trustees Limited, Submission 58, 3.

addition, in the next chapter the Committee discusses the importance of effective mutual recognition provisions to ensure that powers of attorney are as portable as possible in Australia while other longer-term strategies to promote national consistency are pursued.

#### Recommendation 2: National harmonisation of power of attorney laws

The Committee recommends the Victorian Government, through the Standing Committee of Attorneys-General, actively promote and support national harmonisation of power of attorney laws.

# 3.2 Guiding principles for powers of attorney legislation

### 3.2.1 A principles-based approach

The Instruments Act is a wide-ranging piece of legislation which provides the framework for a number of different legal documents in Victoria, including general (non-enduring) powers of attorney and enduring powers of attorney (financial). However, the Act does not give any guidance about how the provisions relating to powers of attorney are to be implemented.

The Guardianship and Administration Act is a more modern piece of legislation. It contains a statement of the objects of the Act. It also sets out three general principles to be applied in the exercise of functions and powers under the Act so that:

- the means least restrictive of a person's freedom of decision and action as is possible in the circumstances is adopted
- the best interests of a person with a disability are promoted
- the wishes of a person with a disability are given effect to, wherever possible. 210

The Victorian Law Reform Commission is considering whether these principles remain relevant as part of its current review of that Act. 211

Participants in the Inquiry drew the Committee's attention to principles underpinning powers of attorney legislation in some other jurisdictions, particularly the UK and Oueensland.<sup>212</sup>

Guardianship and Administration Act 1986 (Vic) s 4(2). Note 'best interests' are defined in s 28(2).

Victorian Law Reform Commission, *Guardianship* Information paper (2010), 48-51.

Laura Helm, *Transcript of evidence*, above n 186, 2; Debra Parnell, *Transcript of evidence*, above n 188, 2-3; Elizabeth Samra, *Transcript of evidence*, above n 188, 3; John Chesterman, *Transcript of evidence*, above n 187, 9-10.

The Mental Capacity Act 2005 (UK) is based on six fundamental principles:

- a person must be assumed to have capacity unless it is established that he or she lacks capacity
- a person is not to be treated as unable to make a decision unless all practicable steps to help him or her to do so have been taken without success
- a person is not to be treated as unable to make a decision merely because he or she makes an unwise decision
- all acts performed or decisions made under the Act on behalf of a person who lacks capacity must be made in his or her best interests
- before an act is performed or a decision made, it must be considered whether it can be achieved in a way that is less restrictive of the person's rights and freedom of action.<sup>213</sup>

The legislation provides that the powers of both representatives and the court that hears applications in relation to the Act are subject to these principles.<sup>214</sup> A code of practice which has been developed to support the UK legislation provides advice and examples about how these principles can be applied.<sup>215</sup>

In Queensland any person who performs a function or exercises a power under the powers of attorney legislation or under an enduring power of attorney is required to comply with general principles which are set out in a schedule to the Act. These are summarised in figure 1. The Queensland Law Reform Commission is examining the ongoing relevance of these principles as part of its review of guardianship laws in that state. <sup>217</sup>

The ACT's powers of attorney legislation sets out general principles similar to those in Queensland. <sup>218</sup>

The statements of principles in the UK, Queensland and the ACT legislation lay the foundation for a rights-based approach to powers of attorney. The principles in these jurisdictions have three common elements:

- an articulation of human rights contained in international conventions
- an emphasis on ensuring that all decisions about a person's capacity uphold his or her rights
- a focus on ensuring that all decisions and actions under powers of attorney promote principals' interests and wellbeing.

39

Mental Capacity Act 2005 (UK) c 9 s 1. Note 'best interests' are defined in s 4.

<sup>&</sup>lt;sup>214</sup> Mental Capacity Act 2005 (UK) c 9 ss 9(4), 16(3).

Department for Constitutional Affairs, United Kingdom, Mental Capacity Act 2005 Code of practice, The Stationery Office (2007), ch 2.

Powers of Attorney Act 1998 (Qld) s 76, sch 1.

Queensland Law Reform Commission, Shaping Queensland's guardianship legislation: Principles and capacity Discussion paper (2008), ch 4.

Powers of Attorney Act 2006 (ACT) s 44, sch 1.

## Figure 1: General principles: Queensland Powers of Attorney Act<sup>219</sup>

#### 1 Presumption of capacity

An adult is presumed to have capacity for a matter.

#### 2 Same human rights

- (1) The right of all adults to the same basic human rights regardless of a particular adult's capacity must be recognised and taken into account.
- (2) The importance of empowering an adult to exercise the adult's basic human rights must also be recognised and taken into account.

#### 3 Individual value

An adult's right to respect for his or her human worth and dignity as an individual must be recognised and taken into account.

#### 4 Valued role as member of society

- (1) An adult's right to be a valued member of society must be recognised and taken into account.
- (2) Accordingly, the importance of encouraging and supporting an adult to perform social roles valued in society must be taken into account.

#### 5 Participation in community life

The importance of encouraging and supporting an adult to live a life in the general community, and to take part in activities enjoyed by the general community, must be taken into account.

#### **6** Encouragement of self-reliance

The importance of encouraging and supporting an adult to achieve the adult's maximum physical, social, emotional and intellectual potential, and to become as self-reliant as practicable, must be taken into account.

#### 7 Maximum participation, minimal limitations and substituted judgment

- (1) An adult's right to participate, to the greatest extent practicable, in decisions affecting the adult's life, including the development of policies, programs and services for people with impaired capacity for a matter, must be recognised and taken into account.
- (2) Also, the importance of preserving, to the greatest extent practicable, an adult's right to make his or her own decisions must be taken into account.
- (3) So, for example
  - (a) the adult must be given any necessary support, and access to information, to enable the adult to participate in decisions affecting the adult's life; and
  - (b) to the greatest extent practicable, for exercising power for a matter for the adult, the adult's views and wishes are to be sought and taken into account; and
  - (c) a person or other entity in performing a function or exercising a power under this Act must do so in the way least restrictive of the adult's rights ...

#### 8 Maintenance of existing supportive relationships

The importance of maintaining an adult's existing supportive relationships must be taken into account.

#### 9 Maintenance of environment and values

- (1) The importance of maintaining an adult's cultural and linguistic environment, and set of values (including any religious beliefs), must be taken into account.
- (2) For an adult who is a member of an Aboriginal community or a Torres Strait Islander, this means the importance of maintaining the adult's Aboriginal or Torres Strait Islander cultural and linguistic environment, and set of values (including Aboriginal tradition or Island custom), must be taken into account.

#### 10 Appropriate to circumstances

Power for a matter should be exercised by an attorney for an adult in a way that is appropriate to the adult's characteristics and needs.

#### 11 Confidentiality

An adult's right to confidentiality of information about the adult must be recognised and taken into account ...

Powers of Attorney Act 1998 (Qld) sch 1.

The Committee acknowledges that the exercise of powers and functions in relation to enduring powers of attorney potentially have a significant impact on a principal's rights. The Committee believes the proposed Powers of Attorney Act should contain a statement of general principles to guide all acts and decisions made under the Act in relation to principals with impaired decision-making capacity. This will help ensure that all people and organisations exercising power under the Act promote and protect principals' rights.

The Committee considers that the proposed statement of general principles should have three components: firstly, an articulation of rights based on relevant international law, secondly, a statement of principles relevant to making decisions about a person's capacity, and finally a statement of principles to guide the actions of those making decisions on behalf of a person with impaired decision-making capacity.

While the Committee believes these three elements should be contained in a single statement of principles in the Act, the Committee discusses them separately in this report in order to provide a clear context for each set of principles. The principles relating to rights based on international law are considered in the next section. The principles relating to making decisions about a person's capacity are discussed in chapter five, Capacity, while the principles to guide the actions of people making decisions on behalf of a person with impaired decision-making capacity are discussed in chapter six, The role of the representative.

## 3.2.2 Principles articulating rights in the *International Convention on the Rights of Persons with Disabilities*

Participants in this Inquiry drew the Committee's attention to the *International Convention on the Rights of Persons with Disabilities* which Australia ratified in 2008, arguing that the principles espoused by the Convention should form the foundation of power of attorney laws. <sup>220</sup> As noted in chapter one, the Convention emphasises the equal rights of people with disabilities and, in particular, instills two important principles:

- least restriction: any measures that restrict an individual's exercise of legal capacity should be 'proportional and tailored to the person's circumstances, [and] apply for the shortest time possible' 221
- supported decision making: individuals should be provided with access to support to allow them to exercise their legal capacity. 222

These rights are complementary to those set out in the *Charter of Human Rights and Responsibilities Act 2006* (Vic), particularly section 8 which provides for recognition

<sup>222</sup> Ibid, art 12(3).

\_

Laura Helm, *Transcript of evidence*, above n 186, 2-3; Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, *Transcript of evidence*, Melbourne, 14 December 2009, 2; Victorian Equal Opportunity & Human Rights Commission, *Submission* 52, 2-6.

<sup>&</sup>lt;sup>221</sup> Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12(4) (entered into force 3 May 2008).

and equality before the law, including recognising the capacity of all persons and ensuring protection from discrimination. <sup>223</sup>

The Committee believes that the principles espoused in the International Convention on the Rights of Persons with Disabilities should be at the core of powers of attorney legislation in Victoria. Having the principles of least restriction and supported decision making as fundamental tenets of the legislation will promote approaches that enhance the rights of people with impaired decision-making capacity. The Committee believes that all persons exercising powers or functions in relation to a principal with impaired decision-making capacity under the Powers of Attorney Act should be required to take these principles into account.

As noted above, the Committee recognises that there are other important principles that should guide other rights-affecting decisions, such as decisions about an individual's capacity and when and how a decision should be made for a person with impaired decision-making capacity. These additional principles are discussed in chapters five and six respectively.

#### Recommendation 3: Founding principles of the Powers of Attorney Act

The Committee recommends the Powers of Attorney Act contain a statement of principles that must be applied by all those exercising powers or functions under the Act in relation to a person with impaired decision-making capacity. The principles should include that those persons must exercise their powers and functions in relation to a principal with impaired decision-making capacity:

- in a way that is as least restrictive of the principal's freedom of decision and action as is possible in the circumstances
- so that the principal is provided with appropriate support to allow him or her to exercise his or her legal capacity to the maximum extent possible.

## 3.3 Simplifying terminology

Participants in the Inquiry stressed that the new powers of attorney legislation should be written in plain English and use simple and consistent terminology. <sup>224</sup> In particular, there was support for simplifying the terminology for the various documents and powers and the parties to them. In this section the Committee discusses appropriate terminology and selects language that is used in the remainder of this report.

## 3.3.1 Naming the documents and powers

At present, the title of the document creating each power of attorney does not reflect the powers given by that document. In addition, the documents and powers have

Charter of Human Rights and Responsibilities Act 2006 (Vic) s 8.

Murray Mallee Community Legal Service, Submission 22, 1; Australian Association of Gerontology Inc, Submission 15, 1.

different names in different locations. The Instruments Act refers to a 'general power of attorney' and an 'enduring power of attorney'. However, the approved form for creating an 'enduring power of attorney' is headed 'enduring power of attorney (financial)'. The Guardianship and Administration Act provides for the 'appointment of enduring guardian', while the template form in the Office of the Public Advocate and Legal Aid's *Take control* kit is entitled 'enduring power of guardianship'. 227

Evidence to the Committee suggested that the lack of uniform terminology for the different power of attorney documents and the powers conferred by them causes confusion and potentially facilitates misuse. For example, the Mental Health Legal Centre's submission pointed out that the 'differing terminologies give rise to much confusion and uncertainty as to the scope and effect of each type of EPA [enduring power of attorney], together with their relationship to each other'.<sup>228</sup>

In particular, it was suggested that the general power of attorney should be renamed to make it clear that it does not confer unlimited power. The submission of the Office of the Public Advocate (OPA) stated:

Confusion particularly is caused by the title of 'General Power of Attorney', which one of OPA's telephone advice service providers notes encourages some citizens to believe they are a generic form of power of attorney that does the job of all powers of attorney.<sup>229</sup>

The Trustee Corporations Association of Australia suggested 'general power of attorney (financial)', while OPA preferred 'general (limited) power of attorney'. Other suggestions for the title of the general power of attorney included 'temporary power of attorney', 'limited power of attorney', and 'interim power of attorney'. 233 and 'interim power of attorney'.

OPA recommended that enduring powers be called 'enduring power of attorney (financial)' and 'enduring power of attorney (guardianship)'. <sup>234</sup> State Trustees agreed with the use of 'enduring power of attorney (financial)' but recommended 'enduring power of guardianship'. <sup>235</sup> Other participants supported harmonised terminology, but did not make specific suggestions. <sup>236</sup>

<sup>&</sup>lt;sup>225</sup> Instruments Act 1958 (Vic) ss 107, 115.

Victorian Government, 'The Instruments (Enduring Powers of Attorney) Act 2003 - Approved forms' (2004) G9 Victoria Government Gazette 437, 437.

Guardianship and Administration Act 1986 (Vic) s 35A, sch 4 form 1; Victoria Legal Aid and Office of the Public Advocate, Take control: A kit for making powers of attorney and guardianship 10th edition (2007), 49.

Noted Health Legal Control Legal Contro

Mental Health Legal Centre Inc, *Submission 59*, 4.

Office of the Public Advocate, *Submission 9*, 10. See also Mental Health Legal Centre Inc, *Submission 59*, 4.

Trustee Corporations Association of Australia, Submission 27, 3; Office of the Public Advocate, Submission
 9 10

Julian Gardner, Submission 21, 6.

<sup>&</sup>lt;sup>232</sup> Ibid.

<sup>&</sup>lt;sup>233</sup> Carers Victoria, *Submission 65*, 10.

Office of the Public Advocate, *Submission 9*, 10. See also Trustee Corporations Association of Australia, *Submission 27*, 3.

State Trustees Limited, Submission 58, 22.

John Billings, Submission 37, above n 187, 4-5; Council on the Ageing Victoria, Submission 39, 3; Mental Health Legal Centre Inc, Submission 59, 4; Alzheimer's Australia Vic, Submission 32, 4.

A recent review in South Australia found that many members of the community did not understand the name 'power of attorney' and suggested the use of 'advance directives' instead.<sup>237</sup> The review recommended that 'advance directives' include financial, health and life management matters.<sup>238</sup> No participants in the current Inquiry argued that the term 'power of attorney' should not continue to be used in Victoria.

The Committee believes that the power of attorney documents and the powers they confer should be named consistently and in a way that informs users about the types of powers granted. The Committee adopts the names recommended by OPA in relation to enduring powers, namely 'enduring power of attorney (financial)' and 'enduring power of attorney (guardianship)'. This is also consistent with 'enduring power of attorney (medical treatment)' under the Medical Treatment Act which many people make at the same time they make other types of powers of attorney.<sup>239</sup>

The Committee agrees that general powers of attorney should be renamed to reflect the fact that they are limited in nature. The Committee believes that the term 'general (non-enduring) power of attorney' is appropriate as it highlights the key distinction between these and powers that endure when a principal has impaired decisionmaking capacity.

The recommended terminology should be used consistently throughout the legislation, documents and all information and educational material about powers of attorney. This will promote recognition and understanding of the documents and powers. The Committee considers the powers that are granted under each type of power of attorney in chapter six of this report.

#### Recommendation 4: Consistent names for documents and powers

The Committee recommends the Powers of Attorney Act call the power of attorney documents and the powers created under them 'general (non-enduring) power of attorney', 'enduring power of attorney (financial)' and 'enduring power of attorney (guardianship)'.

## 3.3.2 Naming the parties to a power of attorney

Many participants in the Inquiry also highlighted that the parties under different types of powers of attorney have different names and this facilitates misunderstandings. The Instruments Act calls the person who creates a power of attorney a 'donor', and the person appointed to act under the document, an 'attorney'. The Guardianship and Administration Act uses the terminology

Advance Directives Review Committee, above n 196, 28-29. Note, as discussed in chapter one, 'advance directive' has different connotations in Victoria and is quite distinct from a power of attorney.

<sup>&</sup>lt;sup>238</sup> Ibid, 27.

Medical Treatment Act 1988 (Vic) s 5A(2), sch 2.

<sup>&</sup>lt;sup>240</sup> Instruments Act 1958 (Vic) ss 104, 106(1), 114(1), 115(1).

'appointor' and 'enduring guardian'. <sup>241</sup> The Medical Treatment Act uses different terminology again: 'donor' and 'agent'. <sup>242</sup>

Again there was strong support from participants for the introduction of consistent terminology, including for parties to enduring powers of attorney (medical treatment). In particular, many participants were worried that members of the community confuse 'donor' with 'organ donor' or someone who gives to charities. 244

Options proposed include 'donor', <sup>245</sup> 'principal', <sup>246</sup> or 'appointer', <sup>247</sup> for all persons who appoint another person to act for them under a power of attorney. OPA's submission explained why it prefers 'principal':

EPAs [enduring powers of attorney] should be seen more tightly as instances where someone acts on behalf of another, rather than as situations where a power is ceded to another. In view of this, the term 'principal', which is used in other jurisdictions including New South Wales and Queensland, ought to be preferred to donor or appointer. <sup>248</sup>

Suggestions for a name for those appointed to act for another under a power of attorney were 'attorney', 'nominee', or 'representative'. One participant expressed concern that 'attorney' could be confused with a lawyer.

Some participants thought it would be useful to maintain the distinction between a 'guardian' and an 'attorney' in order to highlight their different roles. <sup>253</sup> Dr John Chesterman, OPA's Manager of Policy and Education, told the Committee that this would also retain synergies with guardians appointed by VCAT who have similar powers to those appointed under enduring powers of attorney (guardianship). <sup>254</sup>

The Committee agrees that there should be simple and consistent terminology for the parties to a power of attorney document. This will reduce confusion and make the use of these documents and powers less intimidating for lay people. The Committee believes that the person making a power of attorney should be called a 'principal'. This avoids the connotations associated with 'donor' and emphasises that the person is appointing someone else to act on their behalf.

<sup>&</sup>lt;sup>241</sup> Guardianship and Administration Act 1986 (Vic) s 3.

Medical Treatment Act 1988 (Vic) ss 3, 5A(4).

For example, State Trustees Limited, *Submission 58*, 22; John Chesterman, *Transcript of evidence*, above n 187, 2-3.

Ministerial Advisory Council of Senior Victorians, Submission 48, 2; Australian Association of Gerontology Inc, Submission 15, 1; Palliative Care Victoria, Submission 70, 1; St Vincent's Hospital, Submission 19, 2.

<sup>&</sup>lt;sup>245</sup> Murray Mallee Community Legal Service, Submission 22, 2.

Office of the Public Advocate, Submission 9, 10.

<sup>&</sup>lt;sup>247</sup> Ministerial Advisory Council of Senior Victorians, *Submission 48*, 2.

Office of the Public Advocate, Submission 9, 10. See also John Chesterman, Transcript of evidence, above n 187, 2-3; Powers of Attorney Act 2003 (NSW) s 3; Powers of Attorney Act 1998 (Qld) sch 3.

<sup>&</sup>lt;sup>249</sup> Ministerial Advisory Council of Senior Victorians, *Submission 48*, 2.

<sup>&</sup>lt;sup>250</sup> Phillip Hamilton, *Submission* 7, 2.

<sup>&</sup>lt;sup>251</sup> Ibid.

<sup>&</sup>lt;sup>252</sup> Ibid.

<sup>&</sup>lt;sup>253</sup> Murray Mallee Community Legal Service, *Submission 22*, 2; Office of the Public Advocate, *Submission 9*, 10.

John Chesterman, *Transcript of evidence*, above n 187, 9.

## Case study 2: 'He was aware that she did not know what she was signing, 255

The principal was an 83 year old woman. Medical assessments described her as having 'dementia with severe cognitive impairment' and 'multiple medical conditions'. Her son, A, had cared for her at home for many years, with additional support services supplied by Baptcare, a not-for-profit organisation. Baptcare applied to VCAT for a guardianship order because A had restricted Baptcare officers' access to the principal. A applied to be appointed as both his mother's guardian and administrator. A's sisters, B and C, claimed A had prevented them from visiting their mother at home, 'opposed his appointment', and sought to be appointed joint guardians.

Eight months prior to the hearing the principal had signed an enduring power of attorney (financial), an enduring power of attorney (medical treatment) and an enduring power of attorney (guardianship) all appointing A to act on her behalf. A's solicitors informed VCAT that they considered these documents invalid, as the principal did not have capacity at the time of signing. The enduring power of attorney (medical treatment) included a note by A's GP that the principal was 'unable to make decisions about her medical treatment and that she was of insufficiently sound mind to understand the importance of the document'.

VCAT found that A was 'unclear about the nature and purpose' of the enduring powers of attorney (financial) and enduring powers of attorney (guardianship), but that he did initially believe they were 'valid' and 'gave him the authority to reject Baptcare's involvement'.

In the application for guardianship and administration VCAT was not asked to make a declaration on the enduring powers. However the Tribunal felt it was essential to make a declaration because the conflict between the siblings meant it was 'important that there be no doubt in future about the status of the enduring powers'. VCAT held that all three enduring powers were invalid as the principal 'lacked capacity to make them'.

VCAT found that A was unsuitable to be appointed as a guardian for a range of reasons, including that he made the principal sign the enduring powers of attorney and other financial documents when 'he was aware that she did not know what she was signing'. VCAT appointed the Public Advocate as guardian and State Trustees as administrator.

The Committee notes arguments that there are historical and practical reasons for maintaining the distinction between those appointed under an enduring power of attorney (financial) and an enduring power of attorney (guardianship). However, the Committee believes that simple and consistent terminology will promote wider understanding of powers of attorney. Therefore, the Committee recommends that all people appointed to act for another person under a power of attorney document should be called 'representatives'. This is a plain English word which emphasises that the person is representing the principal's interests.

<sup>&</sup>lt;sup>255</sup> DM (Guardianship) [2009] VCAT 1705.

Again this terminology should be used consistently in the legislation and documents, as well as in all information and educational materials about powers of attorney.

The Committee draws the Government's attention to suggestions by Inquiry participants that the names of parties to enduring powers of attorney (medical treatment) should be consistent with those of parties to non-enduring and other types of enduring powers of attorney.

#### Recommendation 5: Consistent names for parties to a power of attorney

The Committee recommends the Powers of Attorney Act:

- a) call a person who creates all types of powers of attorney a 'principal'
- b) call a person appointed to exercise powers under all types of powers of attorney a 'representative'.

## 3.4 Keeping the legislation relevant and effective into the future

In this chapter the Committee has recommended the introduction of a new standalone Powers of Attorney Act in Victoria to provide a single legislative framework for all general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). Throughout the rest of this report the Committee makes specific recommendations about the contents of that legislation and about how it should be implemented and supported.

The Committee considers there are two factors that ensure the successful introduction and ongoing operation of any new legislation. The first is that the new statute must be well promoted, especially to those who are particularly impacted by the changes. This is discussed in chapter nine. Secondly, it is important that the legislation is monitored and evaluated to ensure that it is operating effectively. The Committee suggests that the Powers of Attorney Act should be regularly reviewed to ensure it is meeting its aims and providing an effective mechanism to enable Victorians to plan for their future financial, lifestyle and health care needs.

#### Recommendation 6: Reviewing the Powers of Attorney Act

The Committee recommends the Victorian Government commission an evaluation of the new Powers of Attorney Act after it has been in operation for five years, to determine whether it has been effective in meeting its objectives.

## Case study 3: Unsuitable representative did not act in principal's best interests 256

The principal was an 81 year old man with dementia, who had two daughters, A and B. A lived with the principal for a number of years and helped manage his financial affairs. In September 2006 the principal had a stroke and was admitted to hospital where doctors found that he could not 'make reasonable decisions'. He had signed an enduring power of attorney (financial) but it could not be found, so a hospital social worker applied for the appointment of an administrator. VCAT appointed A as her father's administrator.

As administrator A was required to lodge a Financial Statement and Plan with VCAT. A lodged two plans, both of which VCAT found to be 'deficient'. In March 2007 VCAT revoked A's appointment as administrator and appointed State Trustees instead.

State Trustees requested A hand over the financial information it needed to run her father's affairs. A refused to cooperate and instead threatened State Trustees with legal action. State Trustees found that A had withdrawn over \$1 200 000 from her father's investment accounts and 'at least some' of that money had been paid into her superannuation fund.

In July 2007 A and B sought a rehearing of the order appointing State Trustees as administrator and B also applied for a guardianship order. A provided VCAT with three enduring powers of attorney her father had made appointing her as his representative: an enduring power of attorney (financial) and an enduring power of attorney (guardianship) dated 6 June 2006, and an enduring power of attorney (financial) dated 20 June 2006. It was not clear why the principal had created two similar enduring power of attorney (financial) documents within a short space of time. State Trustees gave VCAT a copy of an enduring power of attorney (financial) made in 1994 in which the principal appointed – in descending order – his wife, or his other daughter B, or State Trustees.

VCAT found the 20 June enduring power of attorney (financial) invalid as it did not contain the witnesses' certificate stating that the principal signed the document freely and voluntarily and that the principal appeared to have the required capacity to create an enduring power of attorney (financial). VCAT found that the enduring power of attorney (financial) and enduring power of attorney (guardianship) made on 6 June met the formal requirements. A 'could not explain why there would be an apparently invalid EPA [enduring power of attorney] made after an apparently valid one'. VCAT also considered the principal's capacity to create a power of attorney. VCAT found that the principal had underlying dementia that left him without capacity by late 2006, and this 'cast some doubt' on the 6 June document. As it was not certain that the principal lacked capacity by mid-2006, VCAT found the documents to be valid.

<sup>&</sup>lt;sup>256</sup> TQ (Guardianship) [2007] VCAT 1300.

However, VCAT revoked both the enduring power of attorney (financial) and the enduring power of attorney (guardianship) made on the 6 June 2006 as A had not acted in the principal's best interests. VCAT found that A's conduct, including her deficient Financial Statement and Plan documents and her dealings with State Trustees, showed that she 'fundamentally misunderstood the role of someone authorised to manage another's financial affairs' and that she was 'unable' to satisfactorily act as her father's representative.

VCAT found that the enduring power of attorney made in 1994 was valid and that B was entitled to act as a representative, given that the principal's wife was in an aged care facility and not capable of doing this. VCAT found that 'reinstating the 1994 EPA ... was less restrictive than making an administration order'. VCAT also appointed B to act as the principal's guardian.

## **Chapter 4: Creating a power of attorney**

A power of attorney is created by the execution of a formal written document. The Committee heard that the current requirements for making a power of attorney are complicated and may discourage people from using these arrangements. This chapter considers the formal requirements for creating powers of attorney and examines how these can be streamlined to make powers of attorney simpler and more accessible to members of the community, while at the same time providing sufficient protection against abuse.

# 4.1 Simplifying formal requirements while protecting against abuse

Many participants in this Inquiry told the Committee that making a power of attorney should be as straightforward as possible. Professor Terry Carney, Professor of Law at the University of Sydney, told the Committee, 'Ease of access is certainly critical. When you are thinking about ease of access it is the most straightforward and least number of formalities, or rather 'perceived barriers', which is critical.' 257

The Committee heard that lengthy, complicated forms and the need to sign more than one document causes confusion for many people. Many of the recommendations in this chapter focus on streamlining and simplifying the current forms and witnessing requirements, as well as making more information available to ensure that the process is as user-friendly as possible.

However, while emphasising the need to make powers of attorney easy to use, many participants also advocated for stronger safeguards to make sure that a principal understands the implications of the documents he or she is signing, as well as to ensure that the principal is signing the document voluntarily.

While there is little research about the nature of abuse of powers of attorney, evidence to the Committee suggested that abuse of validly executed documents is more prevalent than forgery or undue pressure at the time a document is created. The submission of the Office of the Public Advocate (OPA) stated:

OPA's experience is that EPA [enduring power of attorney] abuse far more often involves abuse of a validly executed EPA than it involves either the fraudulent execution of an EPA or the knowing misuse of a superseded EPA, though these forms of abuse do occur. <sup>258</sup>

Nevertheless, abuse does sometimes happen at the time powers of attorney are created, as illustrated by the case studies set out in this chapter. While participants generally supported strengthening execution requirements to protect against abuse,

25

Terry Carney, Professor of Law, The University of Sydney, Transcript of evidence, Melbourne, 1 October 2009, 2.

Office of the Public Advocate, Submission 9, 7. See also John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 8.

many also acknowledged that improving formality requirements may be a disincentive for people to make these documents.<sup>259</sup>

In this chapter the Committee aims to achieve a balance between ensuring that members of the community can create powers of attorney relatively easily and providing appropriate safeguards against abuse at the time a power of attorney document is made. Mechanisms to address abuse of documents after they have been executed are discussed in chapter seven.

## 4.2 Forms for creating powers of attorney

## 4.2.1 Issues with multiple power of attorney forms

At present there are three separate forms for creating general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). A separate form again is required to create an enduring power of attorney (medical treatment). The *Take control* kit produced by OPA and Victoria Legal Aid provides information about the different types of powers of attorney and sets out forms for creating all four types of powers.

The Committee heard that the number of separate forms is confusing for members of the community, particularly as many people create the different types of powers of attorney at the same time. Commercial law firm Hunt & Hunt observed that even more sophisticated clients struggle with the concept of having to sign multiple power of attorney documents and commented:

Quite often the client is overwhelmed by the need to complete 3 separate documents to achieve their future financial, lifestyle and healthcare needs  $\dots$  The process is supposed to simplify their financial, lifestyle and healthcare needs for the future, not confuse them.  $^{260}$ 

Ms Rosemarie Draper, Aged Services Development Officer (Equity and Access) with the New Hope Foundation, told the Committee about her experience assisting her parents to make powers of attorney:

Even doing powers of attorney with my parents was so complicated. Even though I know all about it, when I actually sat down and did the practical thing in doing the documents — they are different; one document needs to have a statutory sort of witness and others do not — it was just a nightmare. I can remember thinking why is it not just one form with all the different powers of attorney so that you can tick the ones you want to assign to different people? It was just crazy. <sup>261</sup>

Several submissions highlighted that many members of the community do not understand the difference between the various documents and the powers they

John Chesterman, *Transcript of evidence*, above n 258, 7; Terry Carney, *Transcript of evidence*, above n 257, 2.

Hunt & Hunt, Submission 24, 1. See also Alzheimer's Australia Vic, Submission 32, 3; Jeni Lee, Submission 57, 1; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2; Seniors Rights Victoria, Submission 38, 19; State Trustees Limited, Submission 58, 4-5.

Rosemarie Draper, Aged Services Development Officer (Equity & Access), New Hope Foundation, *Transcript of evidence*, Melbourne, 30 March 2010, 9.

confer, thinking that 'if they have one, they have them all'.<sup>262</sup> Other participants suggested the current arrangements discourage some people from making powers of attorney and that more streamlined documentation may encourage increased use of powers of attorney.<sup>263</sup>

The Committee also received evidence that having multiple differing power of attorney forms may increase the risk that the documents are incorrectly drafted. The Respecting Patient Choices – Advance Care Planning Program at Northern Health wrote that it observes 'frequent errors' in enduring powers of attorney (medical treatment) documents, with lawyers often modelling these on the enduring power of attorney (financial). <sup>264</sup>

## 4.2.2 Consolidating power of attorney forms

Many Inquiry participants argued that power of attorney forms should be consolidated and suggested a number of possible models.

### One form for creating all enduring powers of attorney

The most widely supported option was to combine all enduring powers of attorney into one document. A number of jurisdictions, including Queensland and the Australian Capital Territory (ACT) have such a combined form. Queensland has a long form to permit the appointment of different representatives with different powers and a short form where the same representative or representatives are given powers in relation to both financial and personal/health matters. The 'long form' is made up of two forms, one for appointing a representative in relation to personal and health matters, and the other for appointing an attorney for financial matters.

\_

Ministerial Advisory Council of Senior Victorians, Submission 48, 1. See also Julian Gardner, Submission 21, 3; Sue Field, NSW Trustee and Guardian Fellow in Elder Law, University of Western Sydney, Submission 61, 1.

Julian Gardner, Submission 21, 3; Alzheimer's Australia Vic, Submission 32, 3; Law Institute of Victoria, Submission 41, 5.

Respecting Patient Choices – Advance Care Planning, Submission 13, 1. See also Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 1-2.

Law Institute of Victoria, Submission 41, 6; Elder Rights Advocacy, Submission 63, 2; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2; Office of the Public Advocate, Submission 9, 11; State Trustees Limited, Submission 58, 5; Mental Health Legal Centre Inc, Submission 59, 3-4; Robert C Bolch, JP, Submission 3, 4; Ronald T Jones, JP, Submission 2, 1; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3; Julian Gardner, Submission 21, 5; Murray Mallee Community Legal Service, Submission 22, 1-2; Hunt & Hunt, Submission 24, 2, 4; Royal District Nursing Service, Submission 25, 2; St Vincent's Hospital, Submission 19, 5; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2; National Seniors Australia, Submission 54, 1-2; Ministerial Advisory Council of Senior Victorians, Submission 48, 2; Latrobe Regional Hospital, Submission 44, 3; Carers Victoria, Submission 65, 13; Peter MacCallum Cancer Centre, Submission 20, 1; Seniors Rights Victoria, Submission 38, 20; Rosemarie Draper, Aged Services Development Officer (Equity & Access), New Hope Foundation, Submission 72, 1; Australian Greek Welfare Society, Submission 74, 2.

Department of Justice and Attorney-General, Queensland Government, Enduring power of attorney: Long form ('Long form') Form 2, Queensland Powers of Attorney Act 1998 (Section 44(1)) (2002); Department of Justice and Attorney-General, Queensland Government, Enduring power of attorney: Short form ('Short form') Form 3, Queensland Powers of Attorney Act 1998 (Section 44(1)) (2003); ACT Government, Enduring power of attorney under the Powers of Attorney Act 2006 (2009).

National Seniors Australia drew the Committee's attention to the combined form in the ACT commenting, 'The powers of the attorney or attorneys are all recorded in the one document, which appears to result in less confusion for the donor.' <sup>267</sup>

Aged & Community Care Victoria submitted that a consolidated document would highlight that there are different types of powers that can be entrusted to different representatives, thus reinforcing to principals, representatives and third parties that there are limits on the representatives' powers. The former Victorian Public Advocate, Mr Julian Gardner, observed that consolidating enduring powers of attorney into one form would ensure that principals 'did not mistakenly believe that they had made all necessary provisions by signing only one of the existing documents'. 269

Most participants who advocated for a consolidated enduring power of attorney form, supported including enduring powers of attorney (medical treatment) in this form. <sup>270</sup>

However, there was not universal support for combining enduring power of attorney documents. MF Spottiswood, a solicitor specialising in estate planning, favoured keeping the current separate documents. He argued that the consolidated documents in other jurisdictions, for example Queensland, are too long and complex and that it is far easier for his clients, particularly elderly people, to understand stand-alone documents. <sup>271</sup>

While supporting the introduction of a consolidated document comprising a single form or package of forms for creating enduring powers, State Trustees asserted that a stand-alone enduring power of attorney (financial) document should be retained as that is the document most commonly requested by its clients. State Trustees' Senior Corporate Lawyer, Mr Alistair Craig, told the Committee, 'To compel people to combine a guardianship form with the financial form is probably adding, in most cases, more material to the document than would be necessary.' 272

Some other participants who supported a combined enduring power of attorney document suggested that it might be too complex to have a single form when the

National Seniors Australia, Submission 54, 2. See also Jill Linklater, Policy and Advocacy Officer, Alzheimer's Australia Vic, Transcript of evidence, Melbourne, 22 October 2009, 2; Mental Health Legal Centre Inc, Submission 59, 3-4.

Aged & Community Care Victoria, *Submission 53*, 6. See also Paul Zanatta, Manager, Community Care, Policy and Small Rural Health, Aged & Community Care Victoria, *Transcript of evidence*, Melbourne, 14 December 2009, 2; Julian Gardner, *Transcript of evidence*, Melbourne, 1 October 2009, 7.

Julian Gardner, Submission 21, 5.

Law Institute of Victoria, Submission 41, 6; Elder Rights Advocacy, Submission 63, 2; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2; Office of the Public Advocate, Submission 9, 11; State Trustees Limited, Submission 58, 5; Robert C Bolch, JP, Submission 3, 4; Ronald T Jones, JP, Submission 2, 1; Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 3; Julian Gardner, Submission 21, 5; Murray Mallee Community Legal Service, Submission 22, 2; Hunt & Hunt, Submission 24, 4; Royal District Nursing Service, Submission 25, 2; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2; National Seniors Australia, Submission 54, 1-2; Ministerial Advisory Council of Senior Victorians, Submission 48, 2; Latrobe Regional Hospital, Submission 44, 3; Peter MacCallum Cancer Centre, Submission 20, 1.

MF Spottiswood, Submission 45, 2.

Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 2. See also State Trustees Limited, *Submission 58*, 5-6.

principal wishes to appoint different representatives for different powers and recommended separate forms in those circumstances.<sup>273</sup>

## A separate general (non-enduring) power of attorney form?

Some submissions argued that members of the community do not understand the distinction between general (non-enduring) powers of attorney and enduring powers of attorney (financial). Alfred Health highlighted this confusion:

There is often a lack of understanding by patients and appointees regarding the different roles of the POA [general (non-enduring) power of attorney] and EPOA [enduring power of attorney]. Patients and appointees often fail to realise that when a patient has lost capacity, a POA no longer applies and an EPOA is required to make decisions. Patients and appointees often express dissatisfaction with the reaction of hospital staff in not acting in accordance with the powers expressed in the documents, even after staff explain that the documents do not apply. <sup>274</sup>

Participants suggested that general (non-enduring) powers of attorney could be combined with the enduring power of attorney (financial) document<sup>275</sup> or the combined enduring power of attorney document discussed in the previous section.<sup>276</sup> Others argued for the maintenance of a separate general (non-enduring) power of attorney document.

The Victorian Bar suggested that a combined form for creating general (non-enduring) and enduring powers of attorney (financial) 'may to a large extent ameliorate the difficulties that arise in ascertaining the point at which a person loses capacity in the context of activation of an Enduring Power of Attorney ...' The Bar proposed that a separate general (non-enduring) power of attorney document could be retained as an alternative choice for people not wishing to use the combined document. <sup>278</sup>

However, some participants strongly argued that a separate general (non-enduring) power of attorney document needs to be retained. The Federation of Community Legal Centres' submission stated, 'Overzealous 'streamlining' of formerly distinct powers may increase confusion about the effects of the various instruments.' The Federation's Policy Officer, Ms Lucie O'Brien, told the Committee, 'We think the degree of difference in the legal ramifications between an enduring power and a general power justifies there being separate forms to try to make sure that people understand the significant difference between those two instruments.' The

John Chesterman, Transcript of evidence, above n 258, 7; Paula Chatfield, Legal Counsel, St Vincent's Hospital, Transcript of evidence, Melbourne, 14 December 2009, 4-5; Latrobe Regional Hospital, Submission 44, 3.

Alfred Health, *Submission 66*, 2. See also Alzheimer's Australia Vic, *Submission 32*, 4; Victoria Legal Aid, *Submission 42*, 3; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 4.

Law Institute of Victoria, Submission 41, 6.

Peter MacCallum Cancer Centre, Submission 20, 1; Alzheimer's Australia Vic, Submission 32, 4; Alfred Health, Submission 66, 3; email from Lauren Adamson, Manager and Principal Solicitor, Seniors Rights Legal Clinic, Seniors Rights Victoria, to Executive Officer, Victorian Parliament Law Reform Committee, 20 May 2010.

The Victorian Bar, Submission 40, 8.

<sup>&</sup>lt;sup>278</sup> Ibid, 10.

Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2.

Lucie O'Brien, Policy Officer, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 2. See also Jeni Lee, Elder Law Committee and Disability Law

Federation's preferred approach is one form for creating a general (non-enduring) power of attorney and another form for creating all types of enduring powers of attorney.

In New South Wales there is a single form to create both a general (non-enduring) power of attorney and an enduring power of attorney (financial). If the principal wishes the power to operate as a non-enduring power he or she must cross out a clause stating that the power will continue to be effective when the principal has impaired decision-making capacity. A recent review in that state identified widespread confusion about the combined document and recommended separate forms for creating general (non-enduring) powers of attorney and enduring powers of attorney (financial). <sup>282</sup>

The Association of Independent Retirees' submission suggested that the general (non-enduring) power of attorney form should highlight that these powers cease to operate if a principal has impaired decision-making capacity: 'A positive acknowledgment of this fact on the document would encourage the donor to consider the consequences of a loss of capacity. This may help ensure that proper Enduring Powers of Attorney are also executed as a precautionary measure.' 283

## A single form for creating enduring powers of attorney guardianship and medical treatment

Three participants argued for consolidating the forms for creating enduring powers of attorney (guardianship) and enduring power of attorney (medical treatment), given the overlapping powers of representatives appointed under these documents.<sup>284</sup>

## Consolidating power of attorney forms – The Committee's view

The Committee believes that the multiplicity of power of attorney forms causes unnecessary confusion for members of the community who wish to make these arrangements. This is particularly an issue because many people create these powers at the same time. Consolidating and simplifying these documents may encourage their wider use in Victoria.

The evidence presented to the Committee demonstrates there is a need to continue to distinguish between powers in relation to financial and guardianship matters, as principals often want to give different types of powers to different people. The Committee's view is that there should be one consolidated document comprising a

Committee, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 11; Laura Helm, Policy Adviser, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 11; Law Institute of Victoria, *Submission 41*, 6; Elder Rights Advocacy, *Submission 63*, 2; Murray Mallee Community Legal Service, *Submission 22*, 1.

Department of Lands, New South Wales, *Powers of attorney in New South Wales* (2005), cl 2.

Land and Property Management Authority, New South Wales, *Review of the Powers of Attorney Act 2003*, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009), 9, 16. See also Standing Committee on Social Issues, *Substitute decision-making for people lacking capacity* Report 43, New South Wales Legislative Council (2010), 118.

Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 4.

Hume Riverina Community Legal Service, *Submission 36*, 2; The Institute of Legal Executives (Victoria), *Submission 33*, 1; Southport Community Legal Service Inc, *Submission 43*, 2-3.

single form or package of forms for creating enduring powers in relation to financial and guardianship matters. This document should clearly set out the different types of powers that may be granted and be flexible enough to enable principals to appoint different representatives for each type of power. It should also allow principals to make individual arrangements if they wish to, for example, delegating only limited specific powers to a representative. The ability of principals to limit or specify representatives' powers is discussed in detail in chapter six.

A consolidated document will help increase the awareness of principals, representatives and third parties about the divisions in powers and alert principals to the need to make the different appointments to ensure that all their future needs are met. The consolidated document could be designed either as a single form, or as a package comprising separate forms, such as the Queensland long form.

The Committee recognises that developing a consolidated document that is simple but flexible will be difficult. There was some concern that a consolidated document may become overly complicated when different representatives are appointed for different powers. However, the Committee believes that the creation of a single enduring document is important for the simplification of enduring powers of attorney in Victoria. Issues such as document complexity should be addressed by the form design. This is discussed in the next section.

Many participants identified the need for the enduring power of attorney (medical treatment) to be included in a consolidated enduring power of attorney document. This is outside the terms of reference for this Inquiry, but the Committee draws this issue to the Government's attention.

The Committee notes there were calls to abolish the general (non-enduring) power of attorney document or incorporate it within a form for creating enduring powers. However, the Committee believes general (non-enduring) powers of attorney have a use for people wanting to make specific and short-term arrangements, for example, those who wish to appoint someone to do their banking or complete a property settlement on their behalf while they are travelling. Therefore, the Committee recommends the general (non-enduring) power of attorney should be retained as a separate document. The form for creating the general (non-enduring) power should clearly identify the limitations of this power.

#### Recommendation 7: Consolidated enduring powers of attorney document

The Committee recommends the Victorian Government develop a consolidated document comprising a single form or package of forms for creating enduring powers of attorney (financial) and enduring powers of attorney (guardianship). This should allow different representatives to be appointed for different powers.

#### Recommendation 8: General (non-enduring) power of attorney form

The Committee recommends the Victorian Government develop a new form for creating general (non-enduring) powers of attorney. This form should highlight that the power ceases to operate when a principal has impaired decision-making capacity.

## 4.2.3 Making power of attorney forms user-friendly

Participants in the Inquiry warned the Committee that creating easy-to-use power of attorney forms, particularly a consolidated document, will be challenging. They identified a number of problems with both the current Victorian documents and the consolidated documents used in other jurisdictions. In this section the Committee identifies a number of specific matters that should be considered in the design of the new forms and recommends a process for developing the new documents.

### Length

Some participants expressed the view that the current forms for creating powers of attorney in Victoria, which are up to five pages in length, are too long. Others were critical of the length of consolidated forms that have been introduced in other jurisdictions. For example, the Queensland enduring power of attorney document has a long form, which is 24 pages in length, and a short form, which is 20 pages. 286

State Trustees' submission stated that long forms may deter people from creating powers of attorney because:

in our experience, many elderly clients find it distressing and confusing to be stepped through lengthy documents, no matter how well presented the document is. Lengthy documents also place additional burdens on attorneys and bodies such as financial institutions that have to deal with attorneys.<sup>287</sup>

State Trustees submitted that the form should be no longer than four pages, while Southport Community Legal Centre opined it should not be more than two pages. <sup>288</sup>

A recent comprehensive review of powers of attorney in South Australia suggested that the Queensland model of long and short forms could be useful, with long forms used when a principal wishes to include detailed instructions or wishes. However, this approach was not supported by participants in this Inquiry. Ms Paula Chatfield, St Vincent's Hospital's Legal Counsel, expressed concern about using long and short forms, pointing out that it requires a person 'to work out which one was the correct form to use. I would think, automatically, a member of the public is going to opt for the short form, because it is easier to understand possibly'. <sup>290</sup>

Another approach has been taken in the UK. New forms introduced in that jurisdiction following an extensive consultation process provide continuation sheets which can be added to the form to allow additional information to be inserted.<sup>291</sup>

Southport Community Legal Service Inc, Submission 43, 2.

58

Southport Community Legal Service Inc, Submission 43, 1-2; Phillip Hamilton, Submission 7, 1.

State Trustees Limited, Submission 58, 6.

<sup>&</sup>lt;sup>287</sup> Ibid.

Advance Directives Review Committee, Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008), 28-29.

Paula Chatfield, *Transcript of evidence*, above n 273, 4.

The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian (Amendment) Regulations 2009 (UK) sch 1; Ministry of Justice, United Kingdom, Reviewing the Mental Capacity Act 2005: Forms, supervision and fees Response to consultation (2009), 23-24.

#### Language

Many members of the community complete power of attorney forms without the help of a professional, such as a lawyer. Submissions noted that forms should be in plain English, use simple, consistent language and be easy to understand. <sup>292</sup>

At present the standard forms for creating powers of attorney are available in English only. Several submissions suggested that these documents, and accompanying information and educational materials, should also be available in community languages. <sup>293</sup>

#### **Format**

Victoria Legal Aid stressed that the forms should comply with Australian standards to maximise readability and ease of use. <sup>294</sup>

Other Inquiry participants highlighted that all power of attorney documents and accompanying information and education materials should be in a single format to help promote awareness and recognition. For instance, Ms Diane Tate, the Director of Financial Services, Corporations, Community Policy at the Australian Bankers' Association, told the Committee it was important for documents to have the same 'look and feel' because it would help bank staff recognise and interpret them.<sup>295</sup>

Some participants also emphasised that the forms should be easy to fill in, with guidance material to help the principal complete the form. However, in the interests of keeping the forms to a minimal length, State Trustees suggested providing more detailed information in a separate guide. Recent consultation about proposed new forms in the UK found having guidance material in the margin of the forms was helpful for people completing the forms. That consultation also found significant tension between ensuring that appropriate information is contained in the forms and keeping the forms short and simple.

The current Victorian forms require principals to cross out sections that do not apply, for example, if they do not wish to impose conditions or limitations on the power or to revoke previous power of attorney documents. Some participants preferred a 'tick box' format. Ms Tate stated that the Australian Bankers' Association's preference was for a form that requires principals to opt in rather than opt out as, 'It focuses

Victoria Legal Aid, Submission 42, 3; Financial Planning Association of Australia Ltd, Submission 49, 1; Alzheimer's Australia Vic, Submission 32, 4. See also Land and Property Management Authority, New South Wales, above n 282, 16.

Paul Zanatta, Transcript of evidence, above n 268, 5; Action on Disability in Ethnic Communities, Submission 14, 1; Phillip Hamilton, Submission 7, 2; Rosemarie Draper, Submission 72, above n 265, 1.

Victoria Legal Aid, Submission 42, 3. See also Standards Australia, Electronic data interchange for administration, commerce and transport (EDIFACT) - Forms design - Basic layout AS 3805-1991.

Diane Tate, Director, Financial Services, Corporations, Community Policy, Australian Bankers' Association Inc, Transcript of evidence, Melbourne, 17 December 2009, 7. See also Alzheimer's Australia Vic, Submission 32, 4; Palliative Care Victoria, Submission 70, 1; Advance Directives Review Committee, above n 289, 28.

<sup>&</sup>lt;sup>296</sup> Phillip Hamilton, *Submission* 7, 2.

State Trustees Limited, Submission 58, 6-7.

<sup>&</sup>lt;sup>298</sup> Ministry of Justice, United Kingdom, above n 291, 19.

<sup>&</sup>lt;sup>299</sup> Ibid, 8, 55.

people's minds on what powers are actually in place and what authority that attorney actually has.'300

### **Privacy**

The South Australian review identified privacy concerns with consolidated power of attorney documents. In particular, if a principal has recorded detailed instructions or wishes in a power of attorney document, he or she would not like these shared unnecessarily – for example a principal would not like information about lifestyle choices to be made available to bank staff. <sup>301</sup> The report suggested that these issues could be addressed by sensible form structure, for example having a cover page with all the formal details, with specific instructions provided on separate pages. No participants in the current Inquiry raised this issue.

### Developing new power of attorney forms

The experiences in other jurisdictions indicate that the design of power of attorney forms can be extremely challenging.

The UK Ministry of Justice conducted an extensive consultation process, including testing the proposed new forms, prior to the introduction of new power of attorney forms in late 2009. The South Australian review identified the need to adopt a collaborative approach to designing new power of attorney forms, involving representatives of the health and legal sectors, as well as seniors and culturally and linguistically diverse communities, to ensure that the forms meet the needs of a range of users. It also suggested that the proposed power of attorney documents be tested prior to implementation and evaluated after they have been in use for between one and two years. <sup>303</sup>

## Making power of attorney forms user-friendly – The Committee's view

Simplifying power of attorney documents may encourage people to create powers of attorney and may reduce errors in the drafting and execution of these documents. The Committee believes it is important for forms to be short and simple, written in plain English and provide the information needed to complete the forms. The forms should also be drafted in a way that protects the principal's privacy, where he or she has made specific instructions or outlined wishes.

The forms and all other information and educational materials should be designed with a consistent look to promote recognition of these documents. In the interests of

60

Diane Tate, Transcript of evidence, above n 295, 7. See also Ian Gilbert, Director, Retail Policy, Australian Bankers' Association Inc, Transcript of evidence, Melbourne, 17 December 2009, 7; Hunt & Hunt, Submission 24, 4.

Advance Directives Review Committee, above n 289, 30.

Ministry of Justice, United Kingdom, above n 291, 55-56; Explanatory memorandum to The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian (Amendment) Regulations 2009 UK No 1884, 3.

Advance Directives Review Committee, above n 289, 19. See also Glenn Dickson, 'The enduring power of attorney form: Should it be changed?' (Paper presented at the Queensland Law Society Symposium 2007, Brisbane, 2-3 March 2007), 10.

keeping the forms simple and easy to use, the Committee believes that information not necessary for the completion of the form should be contained in a separate guide for principals. This is discussed later in the chapter.

The task of developing new power of attorney documents will be extremely challenging. The Committee believes that a collaborative approach should be taken, involving people from the full range of sectors that have participated in this Inquiry. The draft forms should be widely tested before they are introduced. This process will help ensure that the new forms meet the needs of users and that any potential problems are identified at an early stage.

In the previous chapter the Committee recommended the evaluation of the new Powers of Attorney Act after it has been in operation for five years. However, the Committee believes that the proposed new standard form documents need to be evaluated after only two years, given that they are pivotal to the success of the new regime. Following this initial evaluation, the forms should be the subject of ongoing review.

The Committee believes it is also important to ensure that the standard form documents are accessible to all members of the community. The approved forms and all accompanying information and educational materials should be available in a range of community languages. To promote widespread acceptance by service providers, the translated forms should be in a bilingual format, including both English and the particular community language.

### Recommendation 9: Developing new power of attorney forms

The Committee recommends the Victorian Government:

- a) in consultation with a wide range of stakeholders, including members of the community and representatives from the legal profession, the health and community sectors, seniors organisations and culturally and linguistically diverse communities, draft new power of attorney forms that are short, simple, written in plain English and provide appropriate information about how to complete the forms. The forms should have a consistent design
- b) widely test the draft power of attorney forms
- c) review the new power of attorney forms two years after they are implemented, and every five years thereafter.

## Recommendation 10: Producing power of attorney forms and accompanying information in community languages

The Committee recommends the Victorian Government:

- a) make the new power of attorney forms available in a range of community languages. The translated forms should be in a bilingual format, including both English and each community language
- b) make all information and educational materials about powers of attorney available in a range of community languages.

### 4.2.4 Should it be mandatory to use the standard forms?

The Committee heard that some legal practitioners prefer to draft their own forms, rather than use the prescribed forms. For example, law firm Hunt & Hunt submitted that the new regime should:

Allow practitioners to draft the enduring power of attorney document to accommodate the client's wishes rather than being limited to a prescribed form. There may be specific needs that the client requires which may not be considered in the prescribed form. <sup>304</sup>

State Trustees suggested that those drafting powers of attorney should be able to omit text that is not required, in order to keep the documents short and simple. 305

Under the current legislation an enduring power of attorney (financial) is required to be in the approved form, while it is sufficient if the general (non-enduring) power of attorney and the enduring power of attorney (guardianship) are 'to the effect' of the scheduled form. <sup>306</sup>

Evidence suggested that allowing forms to be individualised may increase the risk of drafting errors. For example, the Australian & New Zealand Society for Geriatric Medicine informed the Committee 'we often come across documents, which appear to be drafted "ad hoc" with regard to wording and format'. <sup>307</sup>

Some jurisdictions require all powers of attorney to be in the approved form, for example, Queensland and the UK. The South Australian review found, 'The value of a consistent recognisable form outweighs the convenience of individual versions.'

The Committee believes that it is important to adopt a single regime for creating powers of attorney to promote maximum awareness of the documents. It is of the view that the forms should be developed to be as flexible as possible so that they can be personalised for each principal. However, the Committee believes that all powers of attorney should be in the prescribed form.

### Recommendation 11: Powers of attorney to be in the prescribed form

The Committee recommends the Powers of Attorney Act provide that all powers of attorney made under the Act must be in the prescribed form.

## 4.2.5 Where should the prescribed forms be located?

The official forms for creating different types of powers of attorney are currently in separate locations. The general (non-enduring) power of attorney form is set out in a

Advance Directives Review Committee, above n 289, 27.

\_

Hunt & Hunt, Submission 24, 4. See also Financial Planning Association of Australia Ltd, Submission 49, 1-3.
State Trustees Limited, Submission 58, 6.

Instruments Act 1958 (Vic) ss 107(1), 123(1); Guardianship and Administration Act 1986 (Vic) s 35A(2).

Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Submission* 29, 1. See also Respecting Patient Choices – Advance Care Planning, *Submission* 13, 1.

<sup>&</sup>lt;sup>308</sup> Powers of Attorney Act 1998 (Qld) ss 11, 44(1); Mental Capacity Act 2005 (UK) c 9 sch 1 cl 1(1).

schedule of the *Instruments Act 1958* (Vic), the enduring power of attorney (financial) form is approved by the Secretary of the Department of Justice and published in the Government Gazette, and the form for creating an enduring power of attorney (guardianship) is in a schedule of the *Guardianship and Administration Act 1986* (Vic). Copies of all three forms, plus the form for creating an enduring power of attorney (medical treatment) are set out in the *Take control* kit. 311

Some participants suggested that the official documents should all be in one central location. In particular, participants commented that the approved enduring power of attorney (financial) document is difficult to locate. Southport Community Legal Service's submission observed that the approved form is not available on the Department of Justice's website, and there is no indication that the template form published on OPA's website is the official document. The Committee located the approved form by searching the Government Gazette.

All participants who commented on this issue agreed that the forms should be located in legislation. Southport Community Legal Service expressed concern that a gazetted form might be changed without their lawyers being aware of it, whereas 'we are much more aware of changes made to the Act'. The Service noted there is nothing in the form that would require amendment on a regular basis.

The Committee agrees that the official forms for all of the powers of attorney under review should be in a central, easy-to-find location. The Committee believes that legislation is the best place for these forms. This would make the forms easy to find, reducing the risk of people using incorrect or outdated forms. The introduction of a consolidated Powers of Attorney Act, as recommended in chapter three, will allow these forms to be housed within the statutory framework that governs their use.

The Committee hopes that the location of the forms in legislation will reduce the risk of drafting errors and ensure that regular users such as legal services are alerted when there are changes to the form. The Committee notes that the form for creating an enduring power of attorney (medical treatment) will continue to be separately located in the *Medical Treatment Act 1988* (Vic). The Committee draws the Government's attention to the fact that this may confuse members of the community.

### Recommendation 12: Location of prescribed power of attorney forms

The Committee recommends the Powers of Attorney Act set out the forms for creating general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship).

63

See Instruments Act 1958 (Vic) s 125ZL, sch 12; Victorian Government, 'The Instruments (Enduring Powers of Attorney) Act 2003 - Approved forms' (2004) G9 Victoria Government Gazette 437, 437-441; Guardianship and Administration Act 1986 (Vic) sch 4 form 1.

Victoria Legal Aid and Office of the Public Advocate, *Take control: A kit for making powers of attorney and guardianship* 10th edition (2007), 34-52. Note the form for creating an enduring power of attorney (medical treatment) is found in the *Medical Treatment Act 1988* (Vic) sch 2.

Southport Community Legal Service Inc, Submission 43, 2.

Ibid. See also Paul Zanatta, *Transcript of evidence*, above n 268, 5.

### 4.2.6 Making power of attorney forms widely available

The Committee was not able to obtain data about the extent to which powers of attorney are currently created without legal assistance in Victoria. Evidence from South Australia suggests most powers of attorney are signed in lawyers' offices. 314

Participants in the Inquiry told the Committee that creating power of attorney documents is not easy and mistakes are not uncommon. Mr Gardner told the Committee, 'People find these sorts of documents difficult.' Professor Phillip Hamilton, a notary and specialist lawyer, submitted:

The prescribed forms of Enduring Power of Attorney – Financial and the Appointment of Enduring Guardian challenge the comprehension of both the ordinary citizen and those who must witness their execution. If clients do not attend the solicitor's office to sign these documents, execution is invariably muffed by even the most highly qualified people. 316

Law firm Hunt & Hunt suggested that the widespread availability of do-it-yourself power of attorney kits may facilitate abuse:

Recent current affairs programs have pointed out the amount of abuse and intimidation in our society against the elderly by younger generations enticing them to sign powers of attorney. The fact that these documents are readily available at newsagencies allows the misuse of them by unscrupulous members of our society.<sup>317</sup>

However, an alterative view is that making power of attorney forms widely available saves people money and time, by enabling them to create these documents themselves. Ms Belinda Evans, Elder Rights Advocacy's Senior Advocate, told the Committee 'we need to recognise there are people – whether or not they be older people – who want to be able to make their powers of attorney without necessarily seeing a solicitor'. 318

Mr Gardner told the Committee that whether a person needs help creating a power of attorney document will depend on the individual. He expressed the view that someone with relatively simple affairs may not need assistance. However, he stated that it can be useful to involve a third party, not necessarily a lawyer or a doctor, to make sure principals understand the implications of what they are doing. <sup>319</sup>

The Committee acknowledges that some members of the community may find it difficult to create powers of attorney without professional help. However, it believes that people should have the opportunity to make these documents themselves if they

<sup>314</sup> Advance Directives Review Committee, Planning ahead: Your health, your money, your life. First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy (2008), 39.

Julian Gardner, Transcript of evidence, above n 268, 8. See also Margaret Brown and Suzanne Jarrad, 'Putting "the powers" in place: Barriers for people with memory loss in planning for the future' (2008) 15(4) Journal of Law and Medicine 530, 535.

Phillip Hamilton, *Submission 7*, 1-2.

Hunt & Hunt, *Submission 24*, 3. See also House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 86.

Belinda Evans, Senior Advocate, Elder Rights Advocacy, *Transcript of evidence*, Melbourne, 22 October 2009, 5.

Julian Gardner, *Transcript of evidence*, above n 268, 8.

wish to do so. To facilitate this, the forms for creating powers of attorney, including translated versions, should be widely available, both online and in hard copy. In particular, the forms should be available for download from the websites of the Department of Justice and OPA, and available in hard copy through an extensive network of government agencies and community organisations. The Committee discusses strategies for widely disseminating these forms and other resources about powers of attorney in chapter nine.

The Committee believes that simplifying the power of attorney forms (discussed above), and providing high quality information and guidance materials for principals (discussed below) will assist people to make these documents themselves if they so wish.

The Committee also recognises that some people will find it difficult to make power of attorney documents themselves and makes recommendations later in this chapter for increasing access to legal advice.

### Recommendation 13: Making power of attorney forms accessible

The Committee recommends the Victorian Government make the forms for creating all powers of attorney widely available both online and in hard copy.

## 4.3 Witnessing a power of attorney document

All enduring power of attorney documents in Victoria are required to be witnessed by two adult witnesses. Witnessing serves a number of functions: it highlights the important nature of the document, helps ensure that the principal understands the document and its implications, and provides a check that the principal is signing the document voluntarily.

Participants in this Inquiry raised a number of issues with the current witnessing arrangements. These are discussed in this section.

# 4.3.1 Consistent witnessing requirements or a graduated system?

The witnessing requirements for the different powers of attorney under review differ markedly, both in terms of who may witness the documents and the role that the witness is required to play.

A general (non-enduring) power of attorney is not required to be witnessed, unless it is signed on the principal's direction rather than signed by the principal himself or herself. In that case the document must be signed in the presence of the principal and witnessed by two people. 320

-

<sup>&</sup>lt;sup>320</sup> *Instruments Act 1958* (Vic) s 106(2).

Both the enduring power of attorney (financial) and the enduring power of attorney (guardianship) documents must be witnessed by two adult witnesses, one of whom must be authorised to witness statutory declarations (see appendix D). 321

Witnesses to an enduring power of attorney (financial) must certify that the principal signed the document freely and voluntarily in the witness's presence and appeared to have capacity. Witnesses to an enduring power of attorney (guardianship) are required to certify that the principal has signed the document freely and voluntarily and appears to understand the effect of the document. 323

Many participants called for the witnessing requirements to be the same for all power of attorney documents.<sup>324</sup> These documents are often created at the same time, so having different witnessing requirements in terms of the number and qualification of witnesses and what they must certify, causes confusion.

Credit Union Australia informed the Committee it assumes that the witnesses to enduring documents have correctly performed their role, so that the document is valid. It argued that the fact that general (non-enduring) powers of attorney are not witnessed creates business uncertainty because, 'If the Principal is not present at the Financial Institution for staff to ascertain capacity, there is no current way of checking the validity of the General Power of Attorney.' 325

An alternative view was that there should be a graduated system, with more stringent witnessing requirements for enduring powers and less stringent requirements for general (non-enduring) powers. It was argued that this would emphasise the importance of enduring powers of attorney and their vulnerability to abuse. 326

The Committee finds that the current diverse witnessing requirements for power of attorney documents are confusing. The Committee believes the witnessing requirements should be the same for all powers of attorney, including general (non-enduring) powers of attorney and this is reflected in the recommendations in the remainder of this section. In considering the role, number and qualifications of witnesses, the Committee has endeavoured to strike a balance between providing protection for principals and ensuring that the execution of these documents is not too onerous.

### 4.3.2 What is the role of witnesses?

As noted above, a witness to an enduring power of attorney (financial) must sign a certificate stating that the principal signed the document freely and voluntarily in the

66

Instruments Act 1958 (Vic) ss 123(3), 125(3); Guardianship and Administration Act 1986 (Vic) s 35A(2).

<sup>&</sup>lt;sup>322</sup> Instruments Act 1958 (Vic) s 125A(1).

<sup>&</sup>lt;sup>323</sup> Guardianship and Administration Act 1986 (Vic) sch 4 form 1; Victoria Legal Aid and Office of the Public Advocate, above n 311, 48.

Office of the Public Advocate, *Submission 9*, 20; Alzheimer's Australia Vic, *Submission 32*, 4; Trustee Corporations Association of Australia, *Submission 27*, 2; Credit Union Australia Ltd, *Submission 18*, 2; Murray Mallee Community Legal Service, *Submission 22*, 2; Sue Field, *Submission 61*, above n 262, 1.

<sup>&</sup>lt;sup>325</sup> Credit Union Australia Ltd, Submission 18, 2.

National Seniors Australia, *Submission 54*, 2; Jeni Lee, *Transcript of evidence*, above n 280, 11; Terry Carney, *Transcript of evidence*, above n 257, 3; Royal Victorian Association of Honorary Justices, *Submission 28*, 1; The Victorian Bar, *Submission 40*, 6.

witness's presence and appeared to have the necessary capacity to make an enduring power of attorney. Having capacity to make an enduring power of attorney (financial) is stated to mean understanding the nature and effect of the document, which is defined to include understanding matters such as the principal's ability to limit the power, when the power is exercisable and that the principal may revoke the power at any time. 328

The Guardianship and Administration Act does not provide any information about the witnesses' role, however the scheduled form requires the witnesses to certify that the principal has signed the document freely and voluntarily and appears to understand the effect of the document. The information for witnesses in the *Take control* kit suggests that in determining whether a principal understands the nature and effect of the document, witnesses should consider matters similar to those that must be considered in relation to an enduring power of attorney (financial). 329

Participants in the Inquiry informed the Committee that some witnesses do not understand their role. In particular, many witnesses appear to believe that they are witnessing only that they saw the principal sign the document, and do not consider whether a principal has capacity to create a power of attorney.<sup>330</sup>

# Case study 4: The enduring power of attorney was signed 'well after he was deemed not to have capacity', 331

'A son was operating his father's financial affairs using an enduring power of attorney (EPOA) but also managing his pension under a nominee arrangement. To collect more money he failed to notify Centrelink that his father lived with him and that he was renting the father's house for considerable profit (retained by the son). Centrelink discovered the situation and raised a \$12,000 overpayment against the father.

As the son knew he would not be responsible for any debt under Centrelink legislation, he dropped the father off at his sister's house, emaciated and with only with the clothes he was wearing. Before the administrator could get involved in the retrieval of the rent money and protecting the remaining assets the son had sold the father's house and moved interstate. Both the nominee form and EPOA were signed by the father well after he was deemed not to have capacity by the family doctor.'

## What is the witnesses' role in assessing a principal's capacity?

Many participants suggested the Victorian legislation should set out a presumption of capacity. <sup>332</sup> This would clarify for both witnesses and others that a principal must be

<sup>&</sup>lt;sup>327</sup> Instruments Act 1958 (Vic) s 125A(1).

Instruments Act 1958 (Vic) s 118. Note the definition of nature and effect in s 118 is based on the test set out by the High Court of Australia in *Gibbons v Wright* (1954) 91 CLR 423, 438.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 48.

Office of the Public Advocate, Submission 9, 21.

Seniors Rights Victoria, Submission 38, 33.

Peteris Darzins, Professor of Geriatric Medicine, Monash University, and Director of Geriatric Medicine, Eastern Health, *Transcript of evidence*, Melbourne, 30 March 2010, 3; Law Institute of Victoria, *Submission* 41, 9-10; Mental Health Legal Centre Inc, *Submission* 59, 4; Carers Victoria, *Submission* 65, 12; Victorian

presumed to have capacity unless evidence exists to the contrary. 333 The presumption of capacity is discussed further in the next chapter, where the Committee recommends that it should be a founding principle of Victorian power of attorney laws.

#### No evidence that the presumption of capacity is displaced

Professor Peteris Darzins, Professor of Geriatric Medicine at Monash University and Director of Geriatric Medicine at Eastern Health, told the Committee that if a presumption of capacity is introduced, then the witnesses' role has a fundamentally different emphasis:

Strictly speaking there is no need to prove capacity, there is only a need to disprove it. It is a potentially rebuttable proposition. In the same way as you would rebut the presumption of innocence if there is sufficient evidence, that thinking applies to the presumption of capacity — that you rebut that when there is sufficient evidence of incapacity ... The way the power of attorney forms are written at the moment is that the person who is attesting to the signature is actually attesting to the person's capacity rather than attesting to the absence of evidence of incapacity.<sup>334</sup>

He provided some examples of what might constitute evidence of impaired decisionmaking capacity:

If people do not know the issues they face, that could be evidence of incapacity. It could also be that they just have not been told what the problem is, but if they have been told and still do not know the issues they face, that is evidence of incapacity; or they might not know the possible approaches for dealing with their problems; or they might not appreciate the reasonably foreseeable consequences of choices; or their decisions might be based on delusional constructs. 335

Professor Darzins suggested that if there is evidence of impaired decision-making capacity, a professional capacity assessment should be sought, for example, from a medical practitioner or neuropsychologist.

The Mental Health Legal Centre advocated a similar approach. The Centre's Legal and Policy Officer, Ms Catherine Leslie, told the Committee that there should not be a formal assessment of a principal's capacity 'unless there is ambiguity or real evidence that the person is unable to understand and retain the information in the process of weighing up the information and coming to a decision; unless there is a question of the person lacking capacity ... '336 She asserted that witnesses should be provided with guidance about the types of questions they might ask a principal to ensure that the presumption of capacity is not displaced.

Coalition of Acquired Brain Injury Service Providers, Submission 71, 2; Seniors Rights Victoria, Submission 38, 27; Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 6; Kristen Pearson, President, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 2.

Office of the Public Advocate, Submission 9, 12 quoting Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12(2) (entered into force 3 May 2008).

Peteris Darzins, Transcript of evidence, above n 332, 3.

Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 5.

### Case study 5: 'He did not know what he was signing' 337

'Arthur was diagnosed with mild dementia a couple of years ago and was eligible for a community aged-care package; Arthur was living on a farm with his long-term companion. One day Arthur had a stroke and went to hospital. His companion at the time was in the country. Arthur also had one son who he only saw a few times a year, and they did not have a particularly good relationship. Arthur had no other family, and the son came around to the hospital with a person who was authorised to witness stat decs and executed an enduring power of attorney in favour of the son, George. Then George purported to use that power to place Arthur in a residential aged-care facility some 2 hours away from his farm and then sought to evict Arthur's companion from the house. He advised the aged-care facility that no visitors would be able to access or see Arthur until he settled in because Arthur was not happy being there. Arthur advised us [Seniors Rights Victoria] when we got in touch with him that he did not know what he was signing. It was only a couple of days after his stroke that the document was executed, and Arthur had no access to any bank accounts. His son had taken all his bank accounts, his cards and personal effects. We got involved at that point ...

That case went to VCAT, and a determination was made that Arthur was incapable of making decisions about his finances. The power of attorney was revoked on the grounds that it was void because the donor at the time was not capable of making that decision.'

### Understanding the nature and effect of the document

Other participants in the Inquiry supported the current arrangements under the Instruments Act whereby witnesses must certify that the principal appears to have the necessary capacity, that is, understands the nature and effect of the document.<sup>338</sup>

Mr Gardner described how he would go about assessing a principal's understanding of the nature and effect of a power of attorney:

A sensible lawyer will keep a whole lot of diary notes about what questions they asked and what responses they got and how they made the assessment that a person was competent ... If you were signing one, I would say to you, 'Tell me about this document. What powers are you giving? What is the nature of those powers you are giving? What do you do if you change your mind?'. I would expect you to be able to tell me precisely the extent of the powers you were giving and that you could revoke it. <sup>339</sup>

However, Southport Community Legal Service felt the current test of understanding the nature and effect that is set out in the Instruments Act is too onerous. The Service commented, 'In our experience there are quite a few lawyers who would struggle to meet this requirement – in fact there are a number of technical issues relating to

<sup>&</sup>lt;sup>337</sup> Elizabeth Samra, Lawyer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 3, 9.

Instruments Act 1958 (Vic) ss 125A, 118; Office of the Public Advocate, Submission 9, 13; Law Institute of Victoria, Submission 41, 10-11; Murray Mallee Community Legal Service, Submission 22, 2; Australian Bankers' Association Inc, Submission 55, 5.

Julian Gardner, Transcript of evidence, above n 268, 8-9. See also Julian Gardner, Submission 21, 5.

powers of attorney that we suspect would result in split decisions of the High Court.' It suggested witnesses should be trained to ensure they can explain the document to the principal or, alternatively, softening the requirements so that a principal needs only to demonstrate a general understanding of the document. 341

In New South Wales, the witness to an enduring power of attorney (financial) is required to certify that he or she explained the effect of the instrument to the principal and the principal appeared to understand the effect of the document. State Trustees suggested that a similar requirement could be introduced in Victoria. Mr Gardner proposed that witnesses could be required to ask questions to assess the principal's understanding of matters such as the nature of the document, its effect and the principal's power to change it.

Another approach is taken in Queensland where the principal is required to sign a statement that he or she understands the power of attorney document. This includes matters such as understanding the types of powers the representative will have and understanding the principal's ability to limit the representative's powers or revoke the document. The witness is then required to certify that the principal appeared to understand those matters. The Ministerial Advisory Council of Senior Victorians suggested that a similar approach could be introduced in Victoria. 346

### Case study 6: Brian was persuaded into signing an enduring power of attorney<sup>347</sup>

'Brian was persuaded into signing an EPOA [enduring power of attorney] (financial) and an EPOG [enduring power of attorney (guardianship)], appointing his niece Kate to be his attorney and guardian respectively. Brian had mild cognitive decline at the time and was quite vulnerable. Kate decided to exercise her power not long after and approached a GP, unbeknown to Brian, in order to obtain a medical report stating that Brian could not manage his financial affairs. Kate then advised Brian that he had to move into her granny flat which was out the back of her house. Kate took Brian's bankcards and all identification. Kate then allowed her son to reside in Brian's house free of charge. Kate withdrew a few hundred dollars from Brian's account every fortnight and did not keep receipts of payments she made. Kate also withdrew \$80,000 from Brian's bank account to loan to her son. Kate claims Brian agreed to loan the money to Kate's son. Brian has no recollection of the agreement. The case was referred to VCAT. VCAT found that Kate was not acting in Brian's best interests, had not kept accurate records of all transactions, and Kate's interests were in conflict with the interests of Brian. The EPOA (financial) and EPOG were revoked. However, VCAT did not have the power to make an order to recover the money taken from Brian's account. No other action was taken against Kate by VCAT.'

\_

Southport Community Legal Service Inc, Submission 43, 4.

<sup>&</sup>lt;sup>341</sup> Ibid, 5

<sup>&</sup>lt;sup>342</sup> Powers of Attorney Act 2003 (NSW) s 19(1)(c).

State Trustees Limited, *Submission 58*, 8. See also Lachlan Wraith, Senior Manager, Trusts & Estates, Equity Trustees Ltd, Trustee Corporations Association of Australia, *Transcript of evidence*, Melbourne, 22 October 2009, 3; Australian Bankers' Association Inc, *Submission 55*, 5.

Julian Gardner, Submission 21, 5.

<sup>&</sup>lt;sup>345</sup> Department of Justice and Attorney-General, Queensland Government, Long form, above n 266, 11, 15.

Ministerial Advisory Council of Senior Victorians, Submission 48, 3.

Seniors Rights Victoria, Submission 38, 42.

#### A medical assessment of capacity?

Some participants argued that a medical assessment of capacity should be conducted in all cases before a principal can sign a power of attorney document.<sup>348</sup> Others highlighted that this would add to the complexity of creating a power of attorney as well as increasing the cost, potentially discouraging uptake, and should only be sought where there is doubt about a principal's capacity. <sup>349</sup> The Committee explores these issues more fully in the next chapter, but concludes that a medical assessment of capacity should only be sought where there is doubt about a principal's capacity to create a power of attorney.

## How do witnesses determine whether a principal signs a power of attorney freely and voluntarily?

The Committee received evidence that sometimes a principal may be pressured to sign a power of attorney. The Combined Pensioners & Superannuants Association of Victoria submitted that 'too many relatives anxious to get control of their parent's estate, browbeat, coerce and even con the said parents into giving them the "Power of Attorney". 350

The information for witnesses in the *Take control* kit tells witnesses that they should refuse to sign the power of attorney document if the principal 'appears to be signing it under duress, undue influence or pressure from another person'. 351 However, the kit does not offer any guidance for witnesses about identifying signs of duress.

Resources that have been developed in some other jurisdictions provide helpful information about determining whether a principal is signing a document freely and voluntarily. For example, a guide for New South Wales lawyers suggests they interview the principal separately from the representative. <sup>352</sup>

State Trustees proposed that representatives could be prohibited from being present when the document is signed, to ensure that they are not pressuring the principal.<sup>353</sup> Two recent reviews in Canada have made recommendations to that effect. 354 The witnesses would be required to certify that the representative was not physically present when the principal signed the document. No other participants raised this issue.

Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 4; Audrey Cooke, Submission 5, 1; Murray Mallee Community Legal Service, Submission 22, 2.

Office of the Public Advocate, Submission 9, 12; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Transcript of evidence, Melbourne, 1 October 2009, 4; Mental Health Legal Centre Inc, Submission 59, 5.

Combined Pensioners & Superannuants Association of Victoria Inc, Submission 10, 1.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 36, 48.

The Law Society of New South Wales, When a client's capacity is in doubt: A practical guide for solicitors (2009), 18-19.

State Trustees Limited, Submission 58, 8.

Western Canada Law Reform Agencies, Enduring powers of attorney: Areas for reform (2008), 23; Alberta Law Reform Institute, Enduring powers of attorney: Safeguards against abuse Final report no. 88 (2003), 7, 9.

## Case study 7: 'He believes that Serge has been pressured into signing the powers of attorney' 355

'Serge and Viola had been married for 15 years and more recently had been living together in a home that had been left to Viola by her first husband. Serge became quite unwell and was admitted to hospital. His daughter from a previous marriage flew down to visit and once Serge was released from hospital took him back to Queensland with her. John, Viola's son from her previous marriage, rang SRV [Seniors Rights Victoria] as Serge's daughter had been in contact with Viola regarding accessing her father's finances. Serge had signed both a general and financial power of attorney and his daughter was now demanding that Viola sell the house so that she could access Serge's share of the profits from the sale. John is very concerned both for his mother and Serge as Serge's daughter has not been a part of Serge's life for many years and John has had sole responsibility for looking after Serge and his mother's finances. He believes that Serge has been pressured into signing the powers of attorney.'

### The role of witnesses - The Committee's view

The Committee believes that witnesses play a vital role in ensuring that powers of attorney are validly signed. To reduce confusion, there should be consistent witnessing requirements for all power of attorney documents, both enduring and non-enduring.

The Committee believes that witnesses play an important role in determining whether a principal has capacity to create a power of attorney document. In the next chapter the Committee proposes that the new Powers of Attorney Act provide a definition of capacity which has four elements: the ability to understand, retain and weigh up information relevant to making a decision and the ability to communicate a decision. The Committee believes that the legislation should clarify that a principal has capacity to create a power of attorney if he or she appears to generally understand the nature and effect of the document.

Each witness should be required to certify that the principal signed the document freely and voluntarily in the witness's presence and the principal appears to generally understand the nature and effect of the document. A formal capacity assessment will only be required where there is evidence of impaired decision-making capacity which suggests that the presumption of capacity is displaced.

The Committee notes that one submission suggested that representatives should be prohibited from being present when a power of attorney document is signed. The Committee notes that the representatives are often a principal's closest family members and their presence may support the principal, rather than exert pressure.

The Committee believes witnesses should be provided with information and support to help them undertake their role effectively. This material should provide information about matters such as how to assess the principal's understanding of the

Seniors Rights Victoria, Submission 38, 37.

nature and effect of the document and how to identify signs of duress. The Committee believes this is a more appropriate approach than requiring witnesses to explain the document to the principal. Support for witnesses, including information and educational materials for witnesses, is discussed later in this chapter.

### Recommendation 14: The role of witnesses of power of attorney documents

The Committee recommends the Powers of Attorney Act provide that each witness to a power of attorney must certify that:

- a) the principal has signed the power of attorney freely and voluntarily in the witness's presence
- b) the principal appears to have capacity and, in particular, appears to understand the nature and effect of the power of attorney.

### 4.3.3 Who should witness power of attorney documents?

The documents creating enduring powers of attorney (financial) and enduring powers of attorney (guardianship) must be witnessed by two adult witnesses in the presence of the principal and each other. One of the witnesses must be authorised to witness statutory declarations. Persons authorised to witness statutory declarations include lawyers, court personnel, members of parliament, local councillors, dentists, pharmacists, school principals and bank managers (see appendix D).

Inquiry participants identified a number of concerns with the current witnessing arrangements. These are discussed in this section.

### The number and qualifications of witnesses

As discussed in the previous section, witnesses to a power of attorney document have an important role that requires them to consider the voluntariness of the principal's signing and assess the principal's understanding of the nature and effect of the document. Some Inquiry participants felt that the people currently eligible to witness powers of attorney are not necessarily appropriate for the role. Alfred Health's submission stated:

The witnessing procedure requires verification through questioning of the person giving the EPOA [enduring power of attorney] that they understand the intent of the document. Many professions included in the current list appear not to have had adequate professional training in this area and may not have the skills to correctly assess competence.<sup>359</sup>

The Committee heard evidence that some principals 'shop around' until they find an eligible person who is willing to witness the documents. Ms Jeni Lee, a solicitor who

Instruments Act 1958 (Vic) s 123(3); Guardianship and Administration Act 1986 (Vic) s 35A(2).

<sup>357</sup> Instruments Act 1958 (Vic) s 125(3); Guardianship and Administration Act 1986 (Vic) s 35A(2).

Evidence (Miscellaneous Provisions) Act 1958 (Vic) s 107A.

Alfred Health, Submission 66, 4. See also Australian Medical Association Victoria, Submission 69, 2; Lachlan Wraith, Transcript of evidence, above n 343, 2-3.

gave evidence on behalf of the Law Institute of Victoria, told the Committee that in her experience if a solicitor has 'concern about the relationship between the donor and the donee, the parties will disappear out of your office and go to the chemist or go to the dentist or go somewhere else to have it signed, where there may not be the same attention paid to the capacity issues of the donor'. <sup>360</sup>

Some participants expressed concern that the current broad class of authorised witnesses may make powers of attorney seem less important. For example, OPA's submission stated, 'Anecdotal evidence suggests that the formulaic witnessing of documents by pharmacists, for instance, places EPAs [enduring powers of attorney] alongside other less significant requests.' OPA suggested that reducing the pool of eligible witnesses 'would help distinguish EPAs from other witnessed documents'.

The Committee was presented with a number of options for strengthening the class of persons who can witness powers of attorney.

### Witnessing by persons authorised to witness affidavits

Several participants argued that a power of attorney document should be witnessed by a person who is authorised to witness affidavits. This would exclude many of the seemingly less qualified witnesses such as dentists, pharmacists and vets (see appendix D).

State Trustees argued that this approach is preferable because it would ensure that witnesses are more appropriately qualified, while still being broad enough to ensure that it is relatively easy to locate a witness.<sup>363</sup>

Although medical practitioners can witness statutory declarations, they are not authorised to witness affidavits. Some participants suggested that a person authorised to witness affidavits *or* a medical practitioner should be eligible to witness powers of attorney. Mr Lachlan Wraith, who represented the Trustee Corporations Association of Australia, told the Committee that many power of attorney documents are signed in hospital:

Clearly a time when powers of attorney are executed is when people are in hospitals, and it would be beneficial if medical practitioners could be included in the list of authorised witnesses ... An unexpected event occurs in their life and they find themselves in hospital potentially about to undergo an operation as a result of which their future will be uncertain. Under those circumstances they want to make rapid arrangements to ensure their estate is administered appropriately afterwards. Obviously it is not ideal for people to be engaging in financial planning decisions in that situation, but it is perhaps preferable that they have the option to do that rather than for there to be no facility for them to make plans which means the default

Jeni Lee, Transcript of evidence, above n 280, 5. See also Seniors Rights Victoria, Submission 38, 27.

Office of the Public Advocate, Submission 9, 21. See also John Chesterman, Transcript of evidence, above n 258, 4-5; Lachlan Wraith, Transcript of evidence, above n 343, 3; John Hogan, Transcript of evidence, Melbourne, 30 March 2010, 18.

State Trustees Limited, Submission 58, 9; Royal Victorian Association of Honorary Justices, Submission 28, 1; Law Institute of Victoria, Submission 41, 11.

State Trustees Limited, Submission 58, 9.

position applies and everyone is off to VCAT afterwards to have an administrator appointed.  $^{364}$ 

#### Witnessing by a narrower class of witnesses with legal expertise

In New South Wales only a single witness is required for an enduring power of attorney (financial), but that witness must be a lawyer, court registrar, conveyancer or an appropriately qualified employee of a trustee company.<sup>365</sup>

OPA supported the introduction of similar requirements in Victoria:

While OPA recognises that even people in this group are not immune from ignorance of current witnessing requirements, the creation of a small group of authorised witnesses would both indicate the elevated status of EPAs [enduring powers of attorney] as against other witnessed documents, and would enable a more targeted approach to witness education. 366

Seniors Rights Victoria also supported the New South Wales model. Ms Elizabeth Samra, a lawyer with that organisation, stated, 'We consider that the assessment of whether a person understands the legal nature and effect of the document is a legal test ... it is ultimately up to the legal profession to ensure that the donor understands the legal nature and effect of the document.' 367

The Trustee Corporations Association of Australia told the Committee that adopting the New South Wales system would 'lead to a higher standard of authenticity of documentation'. <sup>368</sup>

However, those who supported the New South Wales model were divided about the number of witnesses needed. OPA suggested there should be two witnesses, with only one of them required to be qualified, while Seniors Rights Victoria and the Trustee Corporations Association of Australia were comfortable with only a single witness. 369

Not all participants agreed that a person with legal expertise is necessarily an appropriate witness. Ms Sarnia Birch, Acting Deputy Manager, Regional Offices with Victoria Legal Aid, expressed the view that sometimes lawyers too struggle with witnessing these documents:

Probably a few years ago more people would have had a relationship with a lawyer but with most of our clients they are people who do not have lawyers at all, unless they come to a point where they decide they need one. But to then have someone walk in with the documents, probably with their son or daughter or a family friend,

Lachlan Wraith, *Transcript of evidence*, above n 343, 3, 6. See also Catherine Leslie, *Transcript of evidence*, above n 336, 4.

Powers of Attorney Act 2003 (NSW) s 19. Note evidence to a recent review by the NSW Standing Committee on Social Issues suggested that clerks of local courts may not be appropriately qualified to act as witnesses: Standing Committee on Social Issues, above n 282, 123.

Office of the Public Advocate, Submission 9, 21. See also Trustee Corporations Association of Australia, Submission 27, 2-3; Seniors Rights Victoria, Submission 38, 27; Sue Field, Submission 61, above n 262, 1.

Elizabeth Samra, Transcript of evidence, above n 337, 7. See also Royal College of Nursing Australia, Submission 26, 4; Australian Bankers' Association Inc, Submission 55, 5.

Trustee Corporations Association of Australia, Submission 27, 3.

Seniors Rights Victoria, Submission 38, 27; Office of the Public Advocate, Submission 9, 21.

and say, 'Please witness these documents', officially verifying that this person has the capacity to make the power, it does not work that way. We generally say to people, 'Look, we think probably your doctor is the best person to be the witness in this case, because they know you a little bit better than we do'.<sup>370</sup>

### Similarly, St Kilda Legal Service's submission stated:

The first contact for many clients of the Legal Service is with the Night Service ... Night Service volunteer lawyers have expressed their concern that it is not appropriate for them to make an assessment of a donor's capacity to make a power of attorney ... It is the view of the Legal Service that powers of attorney documents should be witnessed by the client's local doctor who is in the best position to assess client's capacity. <sup>371</sup>

Research in Queensland has also identified that people with legal expertise sometimes struggle with performing the function of witness. The study found that, of matters before the Guardianship and Administration Tribunal where a principal's capacity to make an enduring power of attorney was in doubt, 50% of the documents had been witnessed by a lawyer.<sup>372</sup>

State Trustees expressed concern that adopting the New South Wales approach may make it difficult to find a witness when required. However, the New South Wales Trustee and Guardian's Office has observed that the witnessing arrangements in that state:

work well in practice, being sufficiently wide to ensure that an eligible witness is readily available. It assists in ensuring a witness is sufficiently knowledgeable to explain the nature and effect of the enduring power of attorney, without adding expense or complexity to the process of making a power of attorney.<sup>374</sup>

### Other witnessing options

In some overseas jurisdictions, there is a requirement for a power of attorney document to be signed by both a solicitor and a doctor (for example, Ireland), or a solicitor and doctor together at the same time (for example, Hong Kong). A recent review in Hong Kong recommended that the requirement to sign before a medical practitioner should be abolished, as it deters people from making powers of attorney. It suggested instead that lawyers be given additional guidance about assessing capacity. 376

The recent South Australian review has recommended that witnesses to powers of attorneys should be 'registered professionals' who have completed an accredited

Sarnia Birch, Acting Deputy Manager, Regional Offices, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 3-4.

St Kilda Legal Service Co-Op Ltd, Submission 50, 2. See also John Myers, Submission 56, 27; James Roughley, Submission 60, 1.

<sup>372</sup> Lindy Willmott and Liza Windle, 'Witnessing EPAs: Empirical research' (2007) 27 Queensland Lawyer 238, 242.

State Trustees Limited, *Submission* 58, 9.

Letter from Imelda Dodds, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010, 2.

Enduring Powers of Attorney Ordinance (Hong Kong) LN 365 of 1997 s 5(2)(a); Powers of Attorney Act 1996 (Ireland) s 5(2)(d).

The Law Reform Commission of Hong Kong, *Enduring powers of attorney* (2008), 32-33.

training course.<sup>377</sup> A 'registered professional' is a person who belongs to a profession that requires registration to practice and includes lawyers, police officers, registered nurses, pharmacists and teachers. No participants in this Inquiry suggested such an approach.

### The number and qualifications of witnesses - The Committee's view

The Committee believes it is important to ensure that the authorised witness is qualified to perform the role, in particular to assess the principal's understanding of the document and to identify any evidence of duress. The Committee believes that the current class of persons who can witness power of attorney documents is too wide.

The Committee recommends that all power of attorney documents should be signed by two witnesses, one of whom is a person who can take affidavits or is a medical practitioner. This is stricter than the current requirements, but is still broad enough to ensure that a qualified witness can be easily accessed when needed. Requiring two witnesses allows for one witness who knows the principal. This person will be in a good position to assess the principal's understanding of the document and their overall capacity.

#### Recommendation 15: Who may witness a power of attorney document?

The Committee recommends the Powers of Attorney Act require all power of attorney documents to be witnessed by two witnesses, one of whom is authorised to witness affidavits or is a medical practitioner.

### Who should not be able to witness power of attorney documents?

The current legislation makes it clear that the principal and representative cannot themselves witness an enduring power of attorney document.<sup>378</sup>

One of the witnesses to an enduring power of attorney (financial) is permitted to be a family member of either the principal or the representative, whereas relatives are prohibited from witnessing enduring powers of attorney (guardianship). Relatives are not defined in the Guardianship and Administration Act, but the Instruments Act defines relative to mean:

- spouse or domestic partner
- son or daughter
- mother or father
- brother, sister, half-brother, half-sister, adoptive brother, adoptive sister, step-brother or step-sister

Advance Directives Review Committee, above n 289, 32-33.

<sup>&</sup>lt;sup>378</sup> Instruments Act 1958 (Vic) s 125(1); Guardianship and Administration Act 1986 (Vic) s 35A(2).

Instruments Act 1958 (Vic) s 125(2); Guardianship and Administration Act 1986 (Vic) s 35A(2).

- grandfather or grandmother
- grandson or granddaughter
- uncle or aunt
- niece or nephew. 380

Seniors Rights Victoria's submission observed that the current wording of the Instruments Act does not prevent the one witness who is a relative of the principal or representative from being the same person who is authorised to witness statutory declarations.<sup>381</sup> This significantly reduces the protection offered by the requirement to have an authorised witness.

There was support among participants for prohibiting the relatives of parties from acting as witnesses to all types of power of attorney documents. Mr Ronald Jones, a Justice of the Peace, suggested there should be even wider restrictions, prohibiting witnessing by anyone 'who could benefit in some way from the document, such as a beneficiary in a will'. 383

However, not all participants supported banning family members as witnesses. The Ministerial Advisory Council of Senior Victorians stated that a relative may be in a good position to assess a principal's understanding of the document as 'sometimes a relative is very aware of how the donor/appointer has been coping in recent times'. 384

The Committee agrees it is important that witnesses are independent. It appears that many power of attorney documents are currently witnessed by people who may benefit from the document, such as family members, leaving principals susceptible to undue influence or the perception of undue influence. Therefore, the Committee recommends that the powers of attorney legislation prohibit persons who could benefit from the document from witnessing all power of attorney documents. The new legislation should provide a wide definition of persons who could benefit from the document, including relatives of a party to the document. Relative should be defined as broadly as possible, extending beyond the current definition in the Instruments Act. The existing prohibition on parties witnessing the document should also remain.

To help ensure that ineligible witnesses do not witness these documents, the Committee recommends that the standard forms for creating powers of attorney require each witness to declare that he or she is acting independently and is not a party or the relative of a party to the document.<sup>385</sup>

\_

Instruments Act 1958 (Vic) s 114(1). Note the Guardianship and Administration Act 1986 (Vic) does define 'nearest relative': see s 3(1). It is unclear whether this definition also applies to 'relative', see Moores Legal Pty Ltd, Submission 31, 3.

Seniors Rights Victoria, Submission 38, 26.

Office of the Public Advocate, Submission 9, 21.

Ronald T Jones, JP, Submission 2, 2.

Ministerial Advisory Council of Senior Victorians, Submission 48, 3.

See, for example, Office of the Public Guardian, United Kingdom, Lasting power of attorney - property and financial affairs (2009), 9 and Department of Justice and Attorney-General, Queensland Government, Short form, above n 266, 11.

## Recommendation 16: Excluding parties and their relatives from witnessing power of attorney documents

The Committee recommends:

- a) the Powers of Attorney Act prohibit the following persons from witnessing a power of attorney document:
  - a party to the document
  - any person who could benefit from the document
- b) the Powers of Attorney Act define a person who could benefit from the power of attorney document. This definition should be broad and include a relative of a party to the document
- c) the forms for creating all powers of attorney require each witness to declare that he or she is not a party to the document and is not related to any party to the document.

## Can a representative's employees witness power of attorney documents?

State Trustees brought to the Committee's attention that there is some debate about whether an employee of a trustee company is eligible to witness a document that appoints the trustee company as a representative. State Trustees' submission observed that some people argue this creates a conflict of interest. However, State Trustees disagrees with this view and suggested that the matter be clarified in statute. For example, the Queensland legislation states that a representative's employee is not disqualified from acting as a witness, while acting in the ordinary course of their employment. 387

The Committee did not receive any evidence to suggest that the employees of trustee companies or other organisations that may be appointed as representatives are not appropriate witnesses. The Committee therefore agrees that the new legislation should clarify that an employee of a representative can witness power of attorney documents in the ordinary course of their employment.

### Recommendation 17: Witnessing by a representative's employees

The Committee recommends the Powers of Attorney Act clarify that a person is not excluded from witnessing a power of attorney document merely because he or she is an employee of the representative who witnesses the document while acting in the ordinary course of employment.

State Trustees Limited, Submission 58, 9-10.

<sup>&</sup>lt;sup>387</sup> *Powers of Attorney Act 1998* (Qld) s 31(2).

### Proving the witness's authority

While one witness to an enduring power of attorney (financial) must be a person authorised to take a statutory declaration, the approved form for creating this document does not require witnesses to state their qualification or occupation. In contrast, the statutory form for creating an enduring power of attorney (guardianship) requires both witnesses to state their occupation.

Several participants in the Inquiry argued that the approved form should require each witness, or at least the person signing as an authorised witness, to state his or her qualification to sign the document. The Australian Bankers' Association commented, 'The absence of stating in what capacity the witness is acting means that it is unclear whether the witness is authorised to take statutory declarations.' Lawyer and settlement clerk, Mr Grant Sturgeon, stated that in his experience, property settlements have sometimes had to be cancelled, with the Titles Office demanding that the authorised witness sign a separate statutory declaration confirming his or her authority to sign the power of attorney document. <sup>389</sup>

The Committee believes the forms for creating powers of attorney under the new powers of attorney legislation should require the authorised witness to state the basis of his or her qualification to witness the document, for example that the witness is a legal practitioner. This will provide business certainty by enabling third parties such as banks to quickly and easily establish whether the document has been witnessed by a person able to take affidavits or a medical practitioner, as will be required by the new legislation.

## Recommendation 18: Authorised witness to state the basis of his or her qualification to witness the document

The Committee recommends the forms for creating all powers of attorney require the authorised witness to state the basis of his or her qualification to witness the document.

## 4.3.4 Supporting witnesses

As noted earlier in this chapter, the role of witnesses extends far beyond merely witnessing a principal's signature. Many Inquiry participants called for witnesses to be supported and educated to make sure they can effectively fulfil this challenging role.

### Information and education for witnesses

The Trustee Corporations Association of Australia informed the Committee that prospective witnesses often enquire whether there are sample questions they can ask to determine whether a principal has capacity to create a power of attorney

Australian Bankers' Association Inc, Submission 55, 4-5.

<sup>389</sup> Grant Sturgeon, Submission 35, 1. See also Law Institute of Victoria, Submission 41, 11; State Trustees Limited, Submission 58, 9.

document.<sup>390</sup> A number of participants supported the development of checklists of questions that witnesses may, or even must, ask a principal before they can witness the document.<sup>391</sup>

In Queensland the Office of the Adult Guardian has produced guidelines to assist witnesses to assess a principal's capacity to make an enduring power of attorney. The guidelines outline a suggested process to determine capacity, highlight a list of behaviours that may indicate impaired decision-making capacity and assist the witness to determine if the principal should be referred for a more specialised capacity assessment. One Inquiry participant commented that the specific questions and checklists provided by the Queensland guidelines are extremely useful for witnesses.

Some participants also supported training for witnesses.<sup>394</sup> OPA called for a targeted education campaign for authorised witnesses to reinforce the important nature of their role.<sup>395</sup>

OPA's submission emphasised that the information provided for witnesses should make it clear that a witness should not sign the document if he or she has any doubts about the principal's capacity or voluntariness. <sup>396</sup>

The Committee's view is that witnesses play an essential role in ensuring that power of attorney documents are correctly and appropriately executed. The Committee recommends that the witnesses' obligations in performing their role should be simply summarised on the forms for creating powers of attorney.

In the next chapter the Committee recommends the development of a practical resource to assist all people involved in assessing capacity, both at the time powers of attorney are created and activated. The Committee believes that this resource should include specific information to assist witnesses to perform their role, such as a checklist of questions to identify that a principal understands the nature and effect of the document, and information to assist witnesses to identify undue influence and impaired decision-making capacity. It should also include information about when and how to seek an expert assessment of a principal's capacity.

The Committee notes that the capacity resource is likely to be substantial, and there would be benefit in having a short guide specifically targeted at witnesses.

The Committee believes training for authorised witnesses will also assist in ensuring that witnesses are aware of their obligations. The training and information resources for witnesses should be developed in conjunction with relevant professional bodies,

\_

Trustee Corporations Association of Australia, Submission 27, att, 2.

Office of the Public Advocate, Submission 9, 21; Robert C Bolch, JP, Submission 3, 5.

Office of the Adult Guardian, Queensland, Capacity guidelines for witnesses of enduring powers of attorney (2005), 1.

<sup>393</sup> Debra Parnell, Policy Officer, Council on the Ageing Victoria, Transcript of evidence, Melbourne, 22 October 2009, 2-3.

<sup>&</sup>lt;sup>394</sup> Alfred Health, Submission 66, 4; Robert C Bolch, JP, Submission 3, 5; Sue Field, Submission 61, above n 262, 2.

Office of the Public Advocate, Submission 9, 25; John Chesterman, Transcript of evidence, above n 258, 5, 13. See also Ministerial Advisory Council of Senior Victorians, Submission 48, 4.

Office of the Public Advocate, *Submission 9*, 21.

such as the Law Institute of Victoria. Narrowing the classes of people who can witness powers of attorney, as recommended in the previous section, will allow for the more effective targeting and distribution of information and training to persons likely to witness these documents.

## Recommendation 19: Power of attorney forms to emphasise the role and obligations of witnesses

The Committee recommends that the forms for creating powers of attorney include information about the role and obligations of witnesses.

#### Recommendation 20: Education, training and resources for witnesses

The Committee recommends the Victorian Government, in conjunction with relevant professional organisations, develop and provide:

- a) practical resources to help witnesses fulfil their role, including checklists to assess a principal's understanding of the nature and effect of the document and information about how to identify evidence of duress and evidence that may displace the presumption of capacity
- b) education and training for authorised witnesses about their role.

### Access to interpreters

The Ministerial Advisory Council of Senior Victorians told the Committee that many people who are asked to witness power of attorney documents do not have access to interpreters and may ask family members to interpret, leaving the principal vulnerable to abuse. The Council suggested that interpreters should be widely available to ensure that the witness can independently assess the principal's understanding of the document and make sure that a principal is not being pressured to sign the document.<sup>397</sup>

The Committee agrees it is important that witnesses have access to qualified and independent interpreters when required. Given the complexity of powers of attorney, as far as is practicable, interpreters should be accredited by the National Accreditation Authority for Translators and Interpreters (NAATI) at the professional level. The resources for witnesses that the Committee has recommended above should emphasise the importance of using qualified interpreters and promote the availability of appropriate interpreting services.

## **Protecting witnesses**

The Trustee Corporations Association of Australia suggested that the powers of attorney legislation should provide a 'good faith' defence for witnesses 'to help overcome witness reluctance'. <sup>398</sup> No other participants commented on this issue.

<sup>&</sup>lt;sup>397</sup> Ministerial Advisory Council of Senior Victorians, Submission 48, 3.

Trustee Corporations Association of Australia, *Submission 27*, att, 2.

The Committee did not receive any evidence that principals have difficulty finding witnesses who are willing to witness power of attorney documents. The Committee believes that the education, training and resources for witnesses recommended above will help witnesses understand their role and complete it with confidence. On the basis of the evidence received, the Committee does not believe it is necessary to specifically include a good faith defence for witnesses in the proposed new powers of attorney legislation.

## 4.4 Informing and supporting principals

Evidence presented to the Committee suggested that principals who create powers of attorney sometimes do not fully understand the implications of the document they are signing. This may not necessarily be due to impaired decision-making capacity: principals may simply lack information about powers of attorney. <sup>399</sup>

Seniors Rights Victoria's submission provided a case study demonstrating that competent people may sometimes sign documents they do not fully understand:

"I thought at the time that the POA [power of attorney] was a good idea but did not realise the extent of the power I had handed over to my children. I was not aware of placing conditions in the document to protect me – but these are my children!" Stated by older woman who had major surgery and gave EPOA [enduring power of attorney] to adult children for the time she was in hospital. Her bank balance dropped \$20,000.00 and they threatened to put her away (in a nursing home) if she did not stop causing trouble by asking about her money. 400

Some participants observed that it is not uncommon for the creation of a power of attorney document to be instigated by someone other than the principal, often a family member. <sup>401</sup> In these cases it is particularly important that the principal fully appreciates the effect of the document he or she is signing. National Seniors Australia stated:

In order to further ensure the integrity and validity of EPAs [enduring powers of attorney] to be valid, older Victorians should have access to information on their rights before entering an EPA. While some good information is currently available ... [a]necdotal evidence suggests many older Australians enter an EPA without being fully informed of their rights, such as their right to limit the powers of their attorney, or their ability to revoke an EPA in writing provided they still have capacity. 402

The comments of Inquiry participants are consistent with research in this area. One Queensland study found that many older people who had made enduring powers of attorney could not recall the details of the forms they had signed. 403

Jeni Lee, *Transcript of evidence*, above n 280, 11.

Seniors Rights Victoria, Submission 38, 18.

<sup>&</sup>lt;sup>401</sup> L Richmond, *Transcript of evidence*, Melbourne, 30 March 2010, 18; Hunt & Hunt, *Submission 24*, 3; Letter from Peter Maloney, Chief Executive Officer, Public Trustee, Tasmania, to Chair, Victorian Parliament Law Reform Committee, 3 November 2009, 2-3. See also Owen Ames and John Harley, 'Elder abuse: Being part of the solution, not part of the problem' (2001) 23(3) *Law Society Bulletin* 34, 35.

National Seniors Australia, Submission 54, 3.

<sup>403</sup> Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people Trends & issues in crime and criminal justice, no. 139, Australian Institute of Criminology (1999), 2-3.

### 4.4.1 Information and educational resources for principals

Participants in the Inquiry stressed the importance of principals being provided with information about powers of attorney, both when they are contemplating making these, and at the time they sign the documents.<sup>404</sup>

Inquiry participants suggested principals need to be informed about a number of specific areas. These include:

- the importance of obtaining legal advice (see section 4.4.2)
- when the powers come into effect (see section 4.7)
- how to revoke the document (see section 4.8)
- the implications of creating multiple power of attorney documents (see section 4.8.2)
- the need to periodically review the document (see section 4.9)
- the need to let others, such as family members and doctors, know about the power of attorney (see section 4.11)
- the type of skills representatives require (see section 6.2.2)
- appointing multiple representatives (see section 6.3.1)
- the nature of representatives' powers (see section 6.4)
- the ability to limit representatives' powers and impose conditions on the exercise of powers (see section 6.4.4)
- the duties of representatives (see section 6.5)
- how representatives will make decisions (see section 6.5.3)
- where to report abuse (see section 7.1.5)
- appointing a personal monitor to oversee the operation of the power of attorney (see section 7.2.4)
- the need to register the power of attorney document (see section 8.2).

These matters are all explored in detail either in this chapter or in subsequent chapters, with the Committee agreeing that it would be valuable for principals to have access to easy-to-understand information about all these topics.

State Trustees Limited, Submission 58, 6-8; Australian Bankers' Association Inc, Submission 55, 5.

In the UK a principal is required to state when signing a power of attorney that he or she has read prescribed information and the witness is also required to certify this fact. The prescribed information is set out at the start of each power of attorney form and includes information about the purpose of the power of attorney, when the representative can act, how the representative must act and how the power can be revoked. Another approach, as discussed earlier in this chapter, is to require a witness to explain the document to the principal.

The Committee agrees it is important that principals are provided with all relevant information about powers of attorney at the time they make or are contemplating making powers of attorney. The *Take control* kit currently covers many of the areas that the Committee has identified as important for principals, but it is not specifically targeted at principals.

The Committee recommends that information and educational materials should be developed specifically for principals, setting out in simple language the key features of power of attorney documents and important information that principals need to know. Locating this material in separate guides for principals will help ensure that the power of attorney forms are kept as simple as possible, as recommended earlier in this chapter. Based on the evidence received in this Inquiry, the Committee does not recommend making it mandatory for principals to read this information. However, these materials should be distributed widely to ensure that all principals have access to them.

The materials for principals should be developed in conjunction with the new power of attorney forms, taking into account the same considerations about format and language. This information should be available in a variety of community languages, as recommended earlier in this chapter.

#### Recommendation 21: Resources for principals

The Committee recommends the Victorian Government provide simple, easy-tounderstand information and educational materials for people making and contemplating making powers of attorney.

## 4.4.2 Access to legal advice

The *Take control* kit currently advises people wishing to make a power of attorney that it may be wise to seek legal advice in a range of circumstances, such as if they are struggling to fill out the forms, have a complicated family situation or have complex financial arrangements. 406

Several participants in the Inquiry stressed the importance of principals obtaining legal advice at the time they create powers of attorney. The Law Institute of Victoria

Mental Capacity Act 2005 (UK) c 9, sch 1 cl 2(1); Office of the Public Guardian, United Kingdom, above n 385, 2; Office of the Public Guardian, United Kingdom, Lasting power of attorney for health and welfare (2009), 2.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 27.

expounded that a lawyer will 'explain the legal and practical implications of making the instrument'. Southport Community Legal Service's submission stated that very few people read the forms, relying on the person preparing the document, both in terms of completing it correctly and understanding what it does. Other participants highlighted that legal advice helps ensure that documents are correctly executed. 409

Some participants even suggested that a principal should be required to obtain legal advice before executing a power of attorney. However, others expressed concern that requiring principals to seek legal advice may discourage some people from creating powers of attorney. Ms Dorothy Trezise, a participant in the Committee's Seniors' Forum, told the Committee, 'I think a lot of them [seniors] would be frightened of lawyers. They have not had dealings with them and perhaps mistrust them. I think finances are another consideration.'

The House of Representatives Standing Committee on Legal and Constitutional Affairs' 2007 report on *Older people and the law* recommended a national scheme to enable all powers of attorney to be prepared with the advice of a legal practitioner. 412

Many participants in this Inquiry supported the implementation of such a scheme. The Federation of Community Legal Centres' submission stated many community legal centres, while providing general advice about powers of attorney, do not have the resources to offer detailed assistance to those wanting to make a power of attorney. The Hume Riverina Community Legal Centre told the Committee that many of its clients are not able to afford a private solicitor:

We have instituted a subsidised scheme in our region in association with the Albury & District Law Society and North East Law Association so that a power of attorney can be drafted for one of our clients by a participating law firm for \$50. However, most clients struggle to afford this amount, especially if they require all the powers and a will for \$50 each. 414

Ms Dahni Houseman of Seniors Rights Victoria expressed the view that involving a lawyer in all cases would not be cost prohibitive and could be rolled out through community legal centres:

Unless there are a lot of concerns around capacity, it is not necessarily going to take a substantial amount of time to actually draft and execute these documents. For example, at the clinics that we run at the moment we have 45 minute appointments, and that is often enough time to explain everything that needs to be explained and execute two or three different types of documents. 415

\_

Law Institute of Victoria, Submission 41, 13. See also Jeni Lee, Submission 57, 3.

Southport Community Legal Service Inc, Submission 43, 1.

Phillip Hamilton, Submission 7, 2-3. See also Paula Chatfield, Transcript of evidence, above n 273, 4.

Jeni Lee, Submission 57, 3; Dahni Houseman, Policy Officer, Seniors Rights Victoria, Transcript of evidence, Melbourne, 22 October 2009, 7; Australian Bankers' Association Inc, Submission 55, 5.

Dorothy Trezise, Director, United Way Geelong, *Transcript of evidence*, Melbourne, 30 March 2010, 17.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 317, 86-87.

Federation of Community Legal Centres (Victoria) Inc, Submission 47, 5.

Hume Riverina Community Legal Service, Submission 36, 3.

Dahni Houseman, *Transcript of evidence*, above n 410, 7.

The Committee recognises the benefits of legal advice and assistance in creating powers of attorney. Legal advice helps ensure that forms are properly executed and that principals are informed about their choices and rights.

However, the Committee does not believe it should be compulsory for those wishing to create a power of attorney to obtain legal advice. As noted previously in this chapter, 'doing it yourself' is appealing to many people and should remain an option. The Committee's view is that the information for principals should emphasise the legal significance of the documents and encourage people to seek legal advice. Where principals do wish to obtain legal advice, the Committee agrees that this should be available at no or low cost to those without the financial means to obtain private legal assistance.

## Recommendation 22: Legal support and advice for people wishing to make powers of attorney

The Committee recommends the Victorian Government implement a scheme to support and enable members of the community to access legal advice and assistance to create powers of attorney.

## 4.5 Acceptance of appointment by representatives

All representatives appointed under enduring powers of attorney (financial) and enduring powers of attorney (guardianship) are currently required to sign a statement accepting their appointment. If a representative fails to accept his or her appointment, the enduring power of attorney is ineffective in relation to that representative.

The Instruments Act requires that a representative's statement of acceptance for an enduring power of attorney (financial) document include undertakings to:

- exercise the powers with reasonable diligence to protect the principal's interests
- avoid acting where there is any conflict between the interests of the principal and the representative
- exercise the powers in accordance with the Act. 416

The representative's acceptance does not need to be witnessed.

Representatives appointed under an enduring power of attorney (guardianship) must also accept their appointment. When accepting their appointment they must undertake to exercise their powers honestly and in accordance with the Act. <sup>417</sup> The representative's acceptance must be witnessed by two witnesses, who must certify that the representative has signed freely and voluntarily and appears to understand

-

<sup>&</sup>lt;sup>416</sup> Instruments Act 1958 (Vic) s 125B(5).

Guardianship and Administration Act 1986 (Vic) s 35A(2)(b), sch 4.

the effect of the document. $^{418}$  One of the witnesses must be a person authorised to witness statutory declarations.

Representatives appointed by a general (non-enduring) power of attorney document are not required to formally accept their appointment.

Participants in the Inquiry were generally supportive of the requirement for representatives to accept their appointment and suggested there should be consistent acceptance requirements for all types of powers of attorney, including general (non-enduring) powers. Requiring representatives to sign an acceptance has a number of advantages such as ensuring that ineligible representatives are not appointed, making sure that representatives understand their powers, duties and the seriousness of the document and providing a sample of the representative's signature.

Several participants also emphasised that requiring the representative's acceptance to be witnessed may help highlight the gravity of the role the representative is agreeing to undertake. The submission of law firm Moores Legal stated that witnessing 'may assist in creating a higher level of understanding of the responsibilities of the attorneys under the documents and reduce the abuse of the powers'. 423

The Committee believes that requiring representatives to accept their appointment serves a number of important functions. Most significantly it provides an opportunity to educate representatives about the nature of their roles and responsibilities. This is discussed in detail in chapter six.

The Committee's view is that consistent acceptance requirements should be implemented for representatives appointed under all types of power of attorney. This will reduce confusion among principals appointing representatives for different roles at the same time. The forms for creating powers of attorney recommended earlier in this chapter should contain a section requiring the representative to accept the appointment.

The Committee also considers that the representative's signing of the acceptance of appointment should be witnessed. Witnessing serves to highlight the importance of the role the representative is agreeing to undertake. However, the Committee did not receive any specific evidence about the class or classes of people who are appropriate to witness a representative's acceptance of appointment. The Committee believes that requiring one witness who is authorised to take an affidavit would be sufficient. Witnesses of this class will be in a good position to assess whether a representative understands the role he or she is accepting. In addition, this will create synergies with the witnessing requirements for principals. The legislation should clarify that the

report - Enduring powers of attorney Report no 47 (1988), 13.

Guardianship and Administration Act 1986 (Vic) s 35A(2)(c), sch 4.

Seniors Rights Victoria, Submission 38, 21, 23; Moores Legal Pty Ltd, Submission 31, 2-3; Trustee Corporations Association of Australia, Submission 27, 3.

<sup>420</sup> Department of Justice and Attorney-General, Queensland Government, Long form, above n 266, 22.

Ministerial Advisory Council of Senior Victorians, Submission 48, 3; Moores Legal Pty Ltd, Submission 31, 2-3.
 Australian Law Reform Commission, Community law reform for the Australian Capital Territory: Third

<sup>&</sup>lt;sup>423</sup> Moores Legal Pty Ltd, *Submission 31*, 3. See also Ministerial Advisory Council of Senior Victorians, *Submission 48*, 3.

person witnessing the representative's acceptance of appointment does not need to be the same person who witnesses the principal's creation of the power of attorney.

### Recommendation 23: Acceptance of appointment by representatives

The Committee recommends the Powers of Attorney Act:

- a) require all representatives to formally accept their appointment
- b) require representatives' acceptance of appointment to be witnessed by a person authorised to witness affidavits. This witness does not need to be the same person who witnessed the creation of the power of attorney document by the principal.

# 4.6 What happens if a power of attorney document is not correctly executed?

The Committee heard that errors in the execution of power of attorney documents are relatively common. The Victorian Bar observed:

The anecdotal experience of donors and donees under the current legislative regime is that they have found that compliance with the formality requirements in respect of an Enduring Power of Attorney is time consuming and not a simple process and at least as complex as execution of a Will. The risk of non compliance is high. 424

Mr Sturgeon, a lawyer and settlement clerk, estimated that about a third of the enduring power of attorney (financial) documents he checks at property settlements are defective.<sup>425</sup>

Several participants argued the fact that a power of attorney document is not correctly executed should not necessarily mean that the power is invalid. State Trustees and the Law Institute of Victoria stated that the Supreme Court should be able to dispense with the formal requirements for execution of a power of attorney. The Victorian Bar suggested that the Victorian Civil and Administrative Tribunal (VCAT) should also have that power. The Bar's submission stated:

At present a failure to comply with formality requirements of creation or revocation of a power of attorney results in the invalidity of a Power of Attorney in the first instance and unexpected continuation of the Power of Attorney in the case where revocation is ineffective. In the case of a will a Court can dispense with the requirements of execution or revocation in limited circumstances which can prevent injustice and there is no reason why similar provisions should not apply to Powers of Attorney. 427

Legislation in the UK permits the Court of Protection to declare that a document that is not in the prescribed form should be treated as if it were in the correct form if

The Victorian Bar, Submission 40, 5.

Grant Sturgeon, Submission 35, 1.

<sup>426</sup> State Trustees Limited, Submission 58, 23; Law Institute of Victoria, Submission 41, 19.

The Victorian Bar, Submission 40, 5. See also Wills Act 1997 (Vic) s 9.

satisfied that the principal 'intended it to create a lasting [enduring] power of attorney'. 428

The Committee hopes that the introduction of simplified forms and increased support for principals and witnesses recommended previously in this chapter will help ensure that most power of attorney documents are correctly executed. In addition, in chapter eight the Committee proposes the introduction of a compulsory registration system for enduring powers of attorney, which will include a basic check that all documents meet formality requirements prior to registration.

However, the Committee recognises that there will always be a risk of error in creating these documents and agrees it is important that powers of attorney are not unjustly invalidated if they do not meet formal requirements, such as the requirement to be in the statutory form. This is particularly an issue in relation to enduring powers because when principals have impaired decision-making capacity they are not able to create a new document if the enduring powers they initially created are found to be invalid for technical reasons.

The Committee believes VCAT should be able to declare an enduring power of attorney document valid despite the fact that it does not meet formal requirements, if it finds that the principal intended the document to be a power of attorney document.

### Recommendation 24: Dispensing with formal execution requirements

The Committee recommends the Powers of Attorney Act provide that VCAT may declare valid an enduring power of attorney (financial) or enduring power of attorney (guardianship) document that does not meet the formal requirements, if satisfied that the principal intended the document to be such a power of attorney, that the principal had the necessary capacity to make a power of attorney and that the principal signed the document freely and voluntarily.

# 4.7 When should an enduring power of attorney commence?

Currently a principal may specify a time, circumstance or occasion when the powers under an enduring power of attorney (financial) commence. If the principal does not stipulate a commencement time, the power is exercisable from when the document is made. In contrast, an enduring power of attorney (guardianship) only commences when a principal is unable to make a particular decision himself or herself because of impaired decision-making capacity. 430

Several participants in the Inquiry highlighted that some members of the community rely on their representatives to make decisions or undertake tasks for them not because they have impaired decision-making capacity, but rather because they are

<sup>428</sup> *Mental Capacity Act* 2005 (UK) c 9 sch 1 cl 3(2).

<sup>429</sup> Instruments Act 1958 (Vic) s 117(2).

Guardianship and Administration Act 1986 (Vic) s 35B(1).

immobile or frail. Mr Gardner told the Committee 'my mother is 95, she goes to work every day, but her mobility is limited. My brother does all of her banking but does it under her direction and control. She signed one of these, but she has not given up her autonomy'. 431

Mr John Billings, Deputy President of VCAT's Guardianship List, provided another example:

I had a hearing only this morning for a quiet young man who had a very serious car accident at the beginning of the year. His parents were appointed as his administrators. He has been fortunate, I suppose, to have now recovered sufficient cognitive capacity where he can manage his affairs himself.

At the end of the hearing, after I revoked his administration order, we had a discussion about an enduring power of attorney. It is likely that he will now make one for his parents. He was physically injured as well as suffering a brain injury, and there will be times when he will want his parents to physically go to the bank or talk to Centrelink — or the TAC or whoever it is — to do some of those things that he cannot do, more for physical reasons than cognitive ones. 432

Mr Gardner pointed out that activating an enduring power of attorney while a principal still has capacity has the advantage of enabling the principal to regulate the representative's behaviour. In addition, it allows the principal and representative to work together, building a strong foundation on which the representative can continue to operate when a principal is no longer able to make decisions for himself or herself. Mr Tony Fitzgerald, the Managing Director of State Trustees, told the Committee:

Sometimes if we start off a relationship under an EPA [enduring power of attorney] where they have capacity, you build up an experience with that client and you know what their preferences are, so if they do lose capacity and we continue on with that relationship, you would just continue on with implementing their preferences because you have built up that experience and you have got records on the file and that kind of thing about what their preferences are. 434

Mr Fitzgerald estimated that approximately half of the 700 clients whose affairs are managed by State Trustees under an enduring power of attorney (financial) still have the capacity to make decisions themselves but have elected to pass this responsibility on to State Trustees. 435

OPA's submission stated that the current arrangements, whereby the different types of enduring powers can potentially come into effect at different times, are confusing for members of the community. It argued that all enduring powers should be able to be exercised prior to a principal having impaired decision-making capacity, with a principal able to specify that the powers are only operative at certain times. This would eliminate the need to make complex assessments about capacity and instil a

Julian Gardner, *Transcript of evidence*, above n 268, 9. See also Ministerial Advisory Council of Senior Victorians, *Submission 48*, 3; Office of the Public Advocate, *Submission 9*, 13; Tony Fitzgerald, Managing Director, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 4.

John Billings, *Transcript of evidence*, above n 349, 6.

Julian Gardner, Transcript of evidence, above n 268, 9.

Tony Fitzgerald, *Transcript of evidence*, above n 431, 5.

<sup>435</sup> Ibid, 3.

supported decision-making approach. <sup>436</sup> Dr John Chesterman, the Manager of Policy and Education at OPA, explained the advantages of his office's proposal:

If you say enduring powers of attorney can only be activated when someone lacks capacity and a person has fluctuating capacity, there is a question about whether the enduring power of attorney is being appropriately used. It is even more complex if you think about capacity as being decision-specific ... A person might have capacity to make some decisions but not others, so the appropriate use of an enduring power of attorney here, requiring the person to be lacking capacity in regard to the particular decision, is very difficult to be certain about. 437

Several other participants also supported the approach suggested by OPA. 438

However, the Committee heard that some members of the community may be reluctant to grant powers that operate immediately. Ms Janet Wood, President of the Council on the Ageing Victoria, expressed the view that people might delay making powers of attorney if these gave powers to others before the principal was ready. Mr Billings emphasised the need for people to have choice about when to activate enduring powers:

I can well understand somebody who is fit and well, has no cognitive impairment and is able to manage their affairs, saying, 'I want to appoint my son or daughter to be my attorney, but I do not want them in control until I cannot do it myself'. There will be others somewhere in the middle who will say, 'I do not want to lose control until I am incapacitated, but I know I am going to need help from time to time, so I want my attorney to be able to start doing things for me straightaway, provided it is in consultation with me and it is carrying out my wishes'... <sup>440</sup>

The Committee recognises that some principals want to activate an enduring power of attorney while they still have capacity to make decisions themselves. This can provide support for a frail or immobile principal and allows the principal and representative to work together, establishing a strong rapport and understanding. The Committee believes that immediate activation of the power should be the default position for both enduring powers of attorney (financial) and enduring powers of attorney (guardianship).

However, the Committee acknowledges that not all principals wish enduring powers to come into effect while they still have capacity. Therefore principals should be able to elect a future date, event or occasion on which the power comes into force. This means that some powers will continue to be activated when a principal has impaired decision-making capacity. The Committee discusses the need to establish a clear process for assessing capacity in the next chapter.

\_

Office of the Public Advocate, Submission 9, 13-14.

John Chesterman, *Transcript of evidence*, above n 258, 3. See also Office of the Public Advocate, *Supported decision-making: Options for legislative recognition*, report prepared by John Chesterman (2010), 6-7.

<sup>&</sup>lt;sup>438</sup> Julian Gardner, Submission 21, 5; The Victorian Bar, Submission 40, 8; Terry Carney, Transcript of evidence, above n 257, 5; Financial Planning Association of Australia Ltd, Submission 49, 2.

Janet Wood, President, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 6. See also Debra Parnell, *Transcript of evidence*, above n 393, 6; Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, *Transcript of evidence*, Melbourne, 1 October 2009, 5.

John Billings, *Transcript of evidence*, above n 349, 6.

The Committee believes that the information and educational materials for principals recommended earlier in this chapter should provide clear information about the options for activating enduring powers of attorney.

### Recommendation 25: Time of activation of an enduring power of attorney

The Committee recommends the Powers of Attorney Act provide:

- a) a principal may elect to make an enduring power of attorney (financial) or enduring power of attorney (guardianship) effective immediately or upon another date, event or occasion
- b) if a principal does not specify when an enduring power commences, it commences immediately.

## 4.8 Revoking powers of attorney

Powers of attorney can be revoked in a number of ways.

The Instruments Act states that enduring powers of attorney (financial) may be revoked:

- by the principal in writing<sup>441</sup>
- automatically by the principal's death 442
- according to its terms, for example, if the document states it is only operable for a specified period<sup>443</sup>
- to the extent that a power of attorney confers power on a representative, automatically by the resignation, death, legal incapacity or insolvency of that representative 444
- automatically by the creation of a later inconsistent power of attorney (discussed in detail below). 445

In addition, VCAT may revoke a representative's appointment under an enduring power of attorney (financial) after a principal has lost capacity, where the Tribunal considers that it is in the principal's best interests. 446 VCAT's powers, including its powers to revoke representatives' appointments, are discussed in detail in chapter seven.

<sup>441</sup> Instruments Act 1958 (Vic) s 125I.

Instruments Act 1958 (Vic) s 125K.

<sup>443</sup> Instruments Act 1958 (Vic) s 125L.

Instruments Act 1958 (Vic) ss 125M-125P. Note a representative can only resign while the principal has capacity. If the principal has impaired decision-making capacity, VCAT's approval is required.

Instruments Act 1958 (Vic) s 125J.

Instruments Act 1958 (Vic) s 125X.

The circumstances in which general (non-enduring) powers of attorney can be revoked are not codified, but are similar to those in which enduring powers of attorney (financial) are revoked. Importantly, a general (non-enduring) power of attorney is revoked automatically when a principal has impaired decision-making capacity. When the circumstance is revoked automatically when a principal has impaired decision-making capacity.

The Guardianship and Administration Act stipulates that an enduring power of attorney (guardianship) may be revoked:

- by the principal in writing 449
- automatically by the creation of a later enduring power of attorney (guardianship)<sup>450</sup>
- automatically if the representative becomes involved in a professional or administrative capacity in the principal's care or treatment or provides accommodation to the principal<sup>451</sup>
- by VCAT if it is satisfied that the representative is not willing or able to act, or has not acted in the principal's best interests (discussed further in chapter seven). 452

Participants in the Inquiry raised a number of issues about how powers of attorney may be revoked which are explored in this section.

# 4.8.1 Should a principal be able to revoke a power of attorney orally?

If a principal has capacity, he or she may revoke a power of attorney document at any time. The revocation of general (non-enduring) and enduring powers of attorney (financial) may be oral or in writing. The Secretary of the Department of Justice has approved a form for revoking an enduring power of attorney (financial), but the use of this form is optional. The approved form for revoking an enduring power of attorney (financial) requires two witnesses and does not require either of the witnesses to be qualified. However, the version of the form in the *Take control* kit states that one of the witnesses should be authorised to take statutory declarations.

A principal with capacity can revoke an enduring power of attorney (guardianship) at any time, but this must be in the statutory form and be witnessed by two witnesses.

<sup>447</sup> See generally Berna Collier and Shannon Lindsay, Powers of attorney in Australia and New Zealand (1992) The Federation Press, ch 11.

<sup>&</sup>lt;sup>448</sup> Ibid, 222. Cf *Instruments Act 1958* (Vic) s 115(2) which makes it clear that an enduring power of attorney (financial) is not revoked if the principal subsequently has impaired decision-making capacity.

See Guardianship and Administration Act 1986 (Vic) s 35C(2), sch 4 form 2.

Guardianship and Administration Act 1986 (Vic) s 35C(1).

Guardianship and Administration Act 1986 (Vic) s 35A(4)-(5).

Guardianship and Administration Act 1986 (Vic) s 35D.

<sup>453</sup> Instruments Act 1958 (Vic) s 125H; Berna Collier and Shannon Lindsay, above n 447, 219-220.

Instruments Act 1958 (Vic) s 125I; Victorian Government, above n 310, 442-443.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 56.

One witness must be authorised to take statutory declarations and neither witness is permitted to be a party to the document or a relative of a party. 456

Some participants noted the different requirements for a principal to revoke a power of attorney and called for consistency and certainty. For instance, Seniors Rights Victoria's submission stated, 'It is fair to say that there is confusion in the community as to how to revoke an enduring instrument ... It is important that members of the public be provided with a clear and certain process for the revocation of enduring instruments.' 457

In particular, there was concern among Inquiry participants about permitting oral revocation. St Vincent's Hospital's submission stated that oral revocation creates confusion for staff because it is not clear when a power of attorney has been revoked verbally. According to Mr Lachlan Wraith, who gave evidence on behalf of the Trustee Corporations Association, requiring revocations to be in writing ensures that there are the same 'checks and balances' in place for a document's revocation as there are for its creation.

Ms Angela Burton, General Manager of Personal Financial Solutions at State Trustees, told the Committee that her organisation's policy 'is to make sure we get a document signed to revoke, because under current legislation it can be verbal. You can say, 'I revoke you', and we do not think that is strong enough. It is more for the protection of the client than for us.' Palliative Care Victoria argued that oral revocation should not be allowed unless there is a registration system for powers of attorney that enables these to be 'filed'. Registration is dealt with in chapter eight.

The Committee agrees that oral revocations potentially create uncertainty for all users of power of attorney documents. Permitting oral revocation is inconsistent with the extensive formal requirements for creating a power of attorney document. The Committee therefore recommends that a principal can only revoke both enduring or non-enduring powers of attorney in writing, using a form set out in the powers of attorney legislation. However, the Committee recognises that, in some instances, a principal may not be in a position to formally revoke the power of attorney document, for example, in a situation where a decision needs to be made quickly. In chapter seven the Committee recommends a number of offences for representatives, including an offence of not acting honestly and with reasonable diligence in regard to the principal's expressed wishes. The Committee believes that this offence will provide some protection where a principal who has capacity has tried to revoke a power of attorney orally or has instructed a representative to act in a particular way, but that representative has continued to act under that power, or has acted in a way that is contrary to the principal's instructions.

See Guardianship and Administration Act 1986 (Vic) s 35C, sch 4 form 2.

Seniors Rights Victoria, Submission 38, 30. See also Lionel H Parrott, Submission 67, 1.

<sup>458</sup> St Vincent's Hospital, Submission 19, 1. See also Paula Chatfield, Transcript of evidence, above n 273, 5-6.

Lachlan Wraith, *Transcript of evidence*, above n 343, 3.

Angela Burton, General Manager, Personal Financial Solutions, State Trustees Limited, Transcript of evidence, Melbourne, 22 October 2009, 7.

Palliative Care Victoria, Submission 70, 1.

The standard forms for revoking powers of attorney should be developed in conjunction with the forms for creating a power of attorney discussed earlier in this chapter. The revocation of a power of attorney should have the same witnessing requirements as the creation of a power of attorney. In addition, other recommendations made in this chapter in relation to power of attorney documents, should apply equally to revocations, such as requiring these forms to be published in community languages and allowing VCAT to dispense with formal requirements where appropriate.

### Recommendation 26: Revoking a power of attorney document

The Committee recommends the Powers of Attorney Act:

- a) provide that the revocation of all powers of attorney made under the Act must be in writing in the approved form
- b) set out an approved form for revoking a power of attorney
- c) provide that the revocation of a power of attorney must be witnessed in the same way as a document creating a power of attorney.

# 4.8.2 What is the effect of creating multiple enduring powers of attorney?

The Instruments Act provides that if a principal creates an enduring power of attorney (financial), an earlier document is revoked to the extent of any inconsistency. <sup>462</sup> The approved form has a standard clause declaring that all previous enduring powers of attorney (financial) are revoked. A principal is required to cross out this option if he or she does not wish it to apply. <sup>463</sup>

The creation of an enduring power of attorney (guardianship) automatically revokes any previous appointments. 464

Evidence to the Committee suggests that it is not uncommon for people to execute multiple enduring power of attorney documents over a period of time. This can give rise to confusion, particularly at a time of crisis when a power of attorney document may need to be used. In the case of a general (non-enduring) power of attorney, the principal will be able to clarify which power prevails, but a principal with impaired decision-making capacity may not be able to provide this direction.

Ms Debra Parnell, Policy Officer with the Council on the Ageing Victoria, told the Committee:

people can in fact have numerous people with powers of attorney and not realise that that is what they have done, that in one circumstance someone suggested they

<sup>462</sup> Instruments Act 1958 (Vic) s 125J.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 40; Victorian Government, above n 310, 439.

Guardianship and Administration Act 1986 (Vic) s 35C(1).

sign this form and that person could then do various things for them; and that at another time they can sign another one, and they do not necessarily understand the implications of what they have done ... I know Seniors Rights Victoria has had cases where someone had five different powers of attorney, and it was a nightmare trying to work out which ones needed to be revoked, which ones were valid, and therefore which ones were questionable. 465

State Trustees suggested the default position should be that a new enduring power of attorney document should automatically revoke a previous one 'unless the donor has specified that the prior document is not intended to be revoked, and specifies in the new document (a) the name of the previously appointed attorneys and the (b) type of enduring power granted to the attorney(s)'. 466

The Victorian Bar preferred the current provision in the Instruments Act, whereby a power of attorney is only revoked by a later document to the extent of any inconsistency, suggesting this should be extended to apply to all enduring powers of attorney. The New South Wales Trustee and Guardian has also highlighted that sometimes it might be useful for two powers of attorney to co-exist, as 'a principal may appoint 2 or more attorneys to do different tasks in respect of different assets with each power of attorney form stating the limitation of each attorney'. 468

The Australian Bankers' Association expressed the view that a registration system would simplify this area, with the registration of a new power of attorney document automatically revoking a previously registered document. Registration is discussed further in chapter eight.

The Committee agrees there needs to be more clarity about the effect of the creation of a later enduring power of attorney document. The Committee recommends legislation should provide that, unless a principal specifies otherwise, a later enduring power of attorney will automatically revoke an earlier one that relates to the same powers. The Committee recognises that sometimes principals may want multiple powers of attorney to co-exist. The new enduring power of attorney form or forms should be designed in a way that allows principals to specify any previous power that they do not wish to revoke.

The information and educational materials for principals recommended previously in this chapter should highlight the need to specify any existing powers of attorney that the principal does not want to revoke when creating a new power of attorney document.

Debra Parnell, *Transcript of evidence*, above n 393, 4-5.

State Trustees Limited, Submission 58, 7. See also Land and Property Management Authority, New South Wales, above n 282, 14.

The Victorian Bar, Submission 40, 14. See, for example, Powers of Attorney Act 1998 (Qld) s 50(1).

NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales,
 Review of the *Powers of Attorney Act 2003* (2009), 21 as attachment to letter from Imelda Dodds, above n 374.
 Australian Bankers' Association Inc, *Submission 55*, 8-9.

### Recommendation 27: A later enduring power of attorney document revokes an earlier one

The Committee recommends:

- a) the Powers of Attorney Act provide that, unless a principal specifies otherwise, a later enduring power of attorney (financial) will revoke an earlier enduring power of attorney (financial) and a later enduring power of attorney (guardianship) will revoke an earlier enduring power of attorney (guardianship)
- b) the forms for creating enduring powers of attorney allow a principal to specify any previous enduring powers of attorney that the principal does not wish to revoke.

# 4.8.3 Should a principal have to inform representatives and third parties that a power of attorney has been revoked?

The Instruments Act currently provides protection for representatives and third parties who rely in good faith on a general (non-enduring) power of attorney or an enduring power of attorney (financial) that has been revoked. However, the representatives or third parties are not protected if they had notice of the revocation.

The *Take control* kit currently advises principals to keep a record of how he or she informed each representative of the revocation and to retrieve any copies of the document that the representative might have. The kit also advises principals to inform relevant third parties such as banks about the revocation.<sup>471</sup>

While no participants in this Inquiry raised this issue, the Committee notes that legislation in the ACT and Queensland places a positive obligation on the principal to inform a representative that his or her power has been cancelled. For example, the Queensland legislation requires a principal to take reasonable steps to advise all representatives that a power has been revoked. The legislation in that state also sets out a specific offence for representatives who knowingly rely on a power of attorney that has been revoked. The Committee deals with offences in chapter seven of this report.

The Committee believes the Victorian legislation should require principals to take reasonable steps to inform a representative that a power has been revoked. The revocation forms and the information and educational materials for principals should promote this obligation, as well as encouraging principals to inform third parties, such as banks.

<sup>470</sup> Instruments Act 1958 (Vic) ss 110, 125U.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 53, 55, 59.

Powers of Attorney Act 1998 (Qld) s 46. See also Powers of Attorney Act 2006 (ACT) s 55.

Powers of Attorney Act 1998 (Qld) s 71.

In chapter eight of this report the Committee recommends a compulsory registration system for all documents creating and revoking enduring powers of attorney. This will provide more certainty for third parties who wish to rely on a power of attorney.

### Recommendation 28: Informing representatives that a power of attorney has been revoked

The Committee recommends the Powers of Attorney Act provide that if a principal revokes a power of attorney, the principal must take reasonable steps to inform all representatives that the power has been revoked.

### 4.8.4 What happens if VCAT appoints a guardian or administrator?

The Guardianship and Administration Act states that the appointment of a representative under an enduring power of attorney (guardianship) is not revoked if VCAT appoints a guardian or administrator for the principal. He pointed out that the corresponding provision of the Instruments Act stipulates that if VCAT makes an administration order, a representative may exercise powers under an enduring power of attorney (financial) only to the extent authorised by VCAT.

The Committee agrees that the new legislation should clarify the effect of VCAT appointing a guardian or administrator on an appointment under an enduring power of attorney. While the Committee received limited evidence about this, it suggests that the current provision in the Instruments Act should apply to both enduring powers of attorney (financial) and enduring powers of attorney (guardianship). This should only apply where the VCAT appointee has the same kind of powers as the representative, for example, the appointment of an administrator should not affect the powers of a representative appointed under an enduring power of attorney (guardianship), as the former has financial powers and the latter has guardianship powers.

#### Recommendation 29: Effect of guardianship or administration order

The Committee recommends the Powers of Attorney Act provide:

- a) if VCAT makes an administration order in relation to a principal, the representative may exercise power under an enduring power of attorney (financial) only to the extent authorised by the Tribunal
- b) if VCAT makes a guardianship order in relation to a principal, the representative may exercise power under an enduring power of attorney (guardianship) only to the extent authorised by the Tribunal.

Guardianship and Administration Act 1986 (Vic) s 35D(3).

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Submission 37*, 5; *Instruments Act 1958* (Vic) s 125G.

### 4.9 Keeping powers of attorney up to date

The submission of Aged and Community Care Victoria (ACCV) argued that powers of attorney could automatically expire after a certain period of time to protect against abuse. ACCV suggested that automatic expiry after three years would ensure that representatives appointed long ago but who have become disenfranchised from a principal's life, cannot come back and assume responsibility for the principal's affairs. 476

Mr Paul Zanatta, Manager of Community Care, Policy and Small Rural Health with ACCV, told the Committee that if a shelf life for powers of attorney was introduced 'one would have a greater surety that the person who has made the power of attorney has contemplated it reasonably recently in their lives; it is not something from the dim past'. <sup>477</sup>

However, not all participants supported this approach. Ms Birch from Victoria Legal Aid stated:

you make a power of attorney to guarantee that you will be looked after when you no longer have capacity. If you have to redo that every three years, then you have the whole question of diminished capacity happening on a regular basis, whereas if you do it properly the first time, then it can continue as long as you want it to continue, whether that be when you lose capacity or not. To me, the requirement that that should happen is really building a lot of blocks to what people might want to do and to being able to be quite satisfied and have confidence in what the decision is that they have already made. 478

An alternative approach is to encourage principals to regularly review their powers of attorney to make sure these documents reflect their current needs. For example, Gippsland Community Legal Service urges principals to review their powers of attorney every two to three years.<sup>479</sup>

The Committee recognises that powers of attorney do sometimes become outdated and may not reflect the people presently in a principal's life, or the principal's current wishes and needs. However, the Committee does not support the automatic expiry of these documents. This would increase the costs associated with making powers of attorney and create a risk that a person does not have a power of attorney in place when he or she needs it.

The Committee considers that the best way of ensuring that powers of attorney are up to date is to encourage principals to review their arrangements regularly, much in the same way they are encouraged to periodically review their will. The information and educational materials for principals that the Committee has recommended should include information about the need to regularly review powers of attorney.

<sup>&</sup>lt;sup>476</sup> Aged & Community Care Victoria, *Submission 53*, 6.

Paul Zanatta, *Transcript of evidence*, above n 268, 4.

Sarnia Birch, *Transcript of evidence*, above n 370, 7. See also Jean Thomas, *Transcript of evidence*, Melbourne, 30 March 2010, 4.

<sup>479</sup> Letter from Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, to Executive Officer, Victorian Parliament Law Reform Committee, 1 October 2009. See also Robert C Bolch, JP, Submission 3, 12.

# 4.10 Certifying copies of power of attorney documents

Principals and representatives are advised to keep original power of attorney documents in a safe place, so a representative needs certified copies to produce to third parties such as banks and aged care facilities. A certified copy of a document is authenticated so that the copy is treated as a true and accurate copy of the original.

The Instruments Act sets out the requirements for certifying copies of general (non-enduring) power of attorney documents as well as enduring powers of attorney (financial). These may be certified by a range of professionals including legal practitioners, financial services licensees, public notaries and justices of the peace. Each page of the copy is required to be certified. A certified copy may also be used to create further certified copies. There are no requirements for certification set out in the Guardianship and Administration Act for enduring powers of attorney (guardianship).

Two submissions called for a consistent certification process for all power of attorney documents. 483

Southport Community Legal Service also observed that the length of the current enduring power of attorney (financial) form makes certification difficult, as the certifier is required to sign each page.<sup>484</sup>

A further issue was raised by the Respecting Patient Choices – Advance Care Planning Program at Northern Health which stated that sometimes legal practitioners keep the original power of attorney document, leaving the representative with an uncertified copy. It suggested that educating lawyers was a way of addressing this issue. The education of legal professionals is considered in chapter nine of this report.

The Committee believes it is important that a representative appointed under a power of attorney is able to produce verified copies of that document to third parties. This will be the case even if a registration system is introduced as recommended in chapter eight, as to protect the principal's privacy, the Committee's proposed system will only verify key information in a power of attorney document in most cases, rather than providing unlimited access to information in the document. The Committee believes the certification provisions set out in the Instruments Act should apply to all powers of attorney. The simplified forms the Committee has recommended earlier in this chapter should be of a length that means that certification is not overly onerous.

<sup>&</sup>lt;sup>480</sup> Instruments Act 1958 (Vic) ss 111(1), 125ZH.

<sup>&</sup>lt;sup>481</sup> Instruments Act 1958 (Vic) ss 111(1), 125ZG(2).

<sup>&</sup>lt;sup>482</sup> Instruments Act 1958 (Vic) ss 111(2), 125ZI.

The Victorian Bar, Submission 40, 7; State Trustees Limited, Submission 58, 22-23.

<sup>484</sup> Southport Community Legal Service Inc, Submission 43, 2. See also Glenn Dickson, above n 303, 9.

Respecting Patient Choices – Advance Care Planning, Submission 13, 1.

#### Recommendation 30: Certified copies of power of attorney documents

The Committee recommends the Powers of Attorney Act provide that a power of attorney document can be proved by a copy of the document that is certified as a true and complete copy of the original by a legal practitioner, financial services licensee, regulated principal under the Corporations Act, justice of the peace, public notary or any other officer authorised by law to administer an oath or by a person of a prescribed class.

# 4.11 Making sure power of attorney documents can be located when needed

Sometimes it may be difficult to locate a power of attorney document when it is needed. For instance, the Trustee Corporations Association of Australia informed the Committee that there have been situations where an administrator has been appointed because it was not known that the principal had created an enduring power of attorney (financial). 486

The Royal District Nursing Service reported that difficulties locating powers of attorney are not uncommon:

The admission process becomes problematic for our staff in that those clients who suffer some form of dementia or cognitive impairment are often confused and unable to reliably advise if they have a POA [power of attorney] and/or the type of POA and/or its whereabouts. And there may be no other person present at the admission (or available later) to assist with this information.

Alternatively, the client/representative may confirm that a particular type of POA exists but our staff are unable to obtain evidence of the original or a copy at the relevant time or at a later stage  $\dots^{487}$ 

Registration, which is discussed in chapter eight, is one way of ensuring that information about these documents is able to be accessed when needed. However, that will only be the case if documents are registered at the time they are created rather than when they are activated.

Mr Dale Reddick, the Advocacy and Support Worker with the Gippsland Community Legal Service's Rights Advocacy and Support Program, told the Committee that the Service has produced a wallet card that is given to all clients who make enduring powers of attorney. The card records the name of the appointed representative and the principal is advised to carry the card in his or her wallet at all times. The South Australian review recommended the production of fridge magnets and wallet cards to help highlight the fact that a principal has an enduring power of attorney in place. 489

Trustee Corporations Association of Australia, *Submission* 27, att, 3.

<sup>&</sup>lt;sup>487</sup> Royal District Nursing Service, *Submission 25*, 2. See also Joyce Jeffs, *Transcript of evidence*, Melbourne, 30 March 2010, 6-7; Southern Health, *Submission 30*, 2.

Letter from Dale Reddick, above n 479; Dale Reddick, Transcript of evidence, above n 439, 8.

Advance Directives Review Committee, above n 289, 35.

Professor Carney from the Faculty of Law at the University of Sydney suggested there could be a 'smartcard-type solution, in the same way that we bring to attention that someone has dementia or serious medical condition ... Perhaps it could be part of a wearable USB stick or something!' 490

The *Take control* kit urges principals to give a certified copy of the power of attorney document to their representative and other relevant people such as doctors, solicitors, accountants and stockbrokers. <sup>491</sup> Encouraging principals to let appropriate people know about the existence of a power of attorney document was also a key strategy recommended by the review in South Australia. <sup>492</sup>

A further strategy identified as part of the South Australian review is for accreditation standards for health, medical and aged care facilities to require that powers of attorney be checked on admission and filed on a patient's record. <sup>493</sup> A number of such institutions participating in this Inquiry reported that it is their standard practice to record information about powers of attorney upon admission. <sup>494</sup>

The Committee agrees it is important that relevant people know about the existence of powers of attorney and for copies of the documents to be able to be located when required. In chapter eight of this report the Committee recommends a registration system for power of attorney documents, which will make it easier for these documents to be located when needed. However, the Committee proposes that access to the register and the information in it be restricted to ensure that principals' privacy is protected. Therefore, there is still a need for additional mechanisms to notify others, particularly the principal's family and friends, about the existence of a power of attorney.

The Committee believes that the information and educational materials for principals recommended earlier in this chapter should encourage principals to let key people and organisations know about the existence of powers of attorney. In addition, the Committee recommends the Victorian Government develop and widely distribute promotional materials including a wallet card for principals that provides the name and contact details of the principal's representative. This will help ensure that a representative can be located quickly in an emergency.

### Recommendation 31: Making sure powers of attorney can be located when needed

The Committee recommends the Victorian Government produce and distribute promotional materials such as wallet cards that can be used to inform people about the existence of a power of attorney and provide basic information such as representatives' names and contact details.

Terry Carney, *Transcript of evidence*, above n 257, 5.

Victoria Legal Aid and Office of the Public Advocate, above n 311, 31.

Advance Directives Review Committee, above n 289, 34-35.

<sup>&</sup>lt;sup>493</sup> Ibid, 37.

Royal District Nursing Service, *Submission 25*, 1; Lucy Cordone, General Counsel, St Vincent's Hospital, *Transcript of evidence*, Melbourne, 14 December 2009, 3; Latrobe Regional Hospital, *Submission 44*, 2.

# 4.12 The portability of powers of attorney – Mutual recognition

The Instruments Act currently provides for the recognition in Victoria of powers of attorney created in other Australian jurisdictions. The Act provides that if an enduring power of attorney (financial) complies with the requirements of the state or territory in which it was made then, to the extent that the powers it gives could validly be granted by a Victorian enduring power of attorney (financial), it is to be treated as a power made under the Instruments Act. This provision is based on a model endorsed by the Standing Committee of Attorneys-General in 2000.

There are no mutual recognition provisions in relation to enduring powers of attorney (guardianship) or general (non-enduring) powers of attorney. 497

The Commonwealth *Older people and the law* report described the current mutual recognition arrangements as 'at best limited'.<sup>498</sup> The current arrangements require each individual document to be interpreted, which is a significant impost on business such as banks.

Mr Reddick of the Gippsland Community Legal Service told the Committee he suggests to retirees intending to move interstate that they make another power of attorney in their new state of residence. He observed, 'They have to start from scratch.'

Mr Reddick emphasised that the current mutual recognitions are even more problematic for people travelling:

the original power is still in existence but, depending on where they will be, if and when anything happens, will determine how the power is directed or restricted.

Normally attendees at the community information sessions will say, 'What does that mean? I cannot tell from that'. We will just say that the powers are recognised across jurisdictions, but it may be subject to the limitations of that state. In practical terms we cannot tell people what that means so it creates that level of uncertainty. 'What is the point of donating power here in Victoria before I go off on a 12-month trip, because I do not know what is going to happen if I get crook in Queensland or the Northern Territory' is the number one response from attendees. <sup>500</sup>

As noted in the previous chapter, many participants supported national harmonisation of power of attorney laws, but recognised that this was a longer-term goal. There was strong support for improving mutual recognition provisions in the short term. For

\_

<sup>495</sup> Instruments Act 1958 (Vic) s 116. Note this applies to powers of attorney created prior to 2004 when this amendment came into effect: see s 125ZO.

Victoria, Parliamentary debates, Legislative Assembly, 28 August 2003, 215 (The Hon. Rob Hulls, Attorney-General), 217.

John Billings, Submission 37, above n 475, 4; Seniors Rights Victoria, Submission 38, 17.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 317, 77. See also Australian Bankers' Association Inc, *Submission 55*, 5; Alfred Health, *Submission 66*, 4; Seniors Rights Victoria, *Submission 38*, 17; State Trustees Limited, *Submission 58*, 17.

Dale Reddick, *Transcript of evidence*, above n 439, 3.

<sup>500</sup> Ibid. See also David Fara, Deputy President, Association of Independent Retirees (AIR) Ltd (Victorian Division), *Transcript of evidence*, Melbourne, 22 October 2009, 2.

example, the Federation of Community Legal Centres stated that the Victorian Government 'should do everything in its power to promote cross-jurisdictional recognition of powers of attorney'. <sup>501</sup>

The Committee acknowledges that the national harmonisation of power of attorney laws that the Committee recommended in chapter three is likely to be a protracted process. In the meantime, the Committee believes it is important to provide certainty for the many Australians who travel interstate each year. Therefore the Committee recommends that the Victorian Government work through the Standing Committee of Attorneys-General to implement effective mutual recognition provisions Australia-wide, including in relation to enduring powers of attorney (guardianship).

The primary purpose of mutual recognition is to promote 'the goal of freedom of movement ... in a national market in Australia'. In order for this goal to be achieved in relation to powers of attorney in Victoria, the Committee believes that the proposed new Powers of Attorney Act should provide for the maximum recognition in Victoria of all power of attorney documents validly executed in another Australian state or territory. This will ensure that members of the community who travel interstate are not inconvenienced if they need to rely on their power of attorney document outside their home state or territory. However, the Committee acknowledges that there may be some circumstances in which powers of attorney created interstate cannot be recognised in Victoria, for example, if a power of attorney document purports to give a power that cannot be given by such a document in Victoria.

It may also be possible to improve the mutual recognition of power of attorney documents by linking this to a registration system. This is discussed in chapter eight.

#### Recommendation 32: Mutual recognition of powers of attorney

The Committee recommends:

- a) the Victorian Government, through the Standing Committee of Attorneys-General, actively promote and support the implementation of effective mutual recognition provisions for all enduring powers of attorney
- b) the Powers of Attorney Act provide, to the maximum extent possible, for recognition in Victoria of all power of attorney documents validly executed in another Australian state or territory.

Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 3. See also Trustee Corporations Association of Australia, *Submission 27*, 4; State Trustees Limited, *Submission 58*, 17.

Mutual Recognition Act 1992 (Cth) s 3.

### Case study 8: The power of attorney was 'worthless' 504

In 1969 Mrs Bonnie Malherbe purchased a unit using her own savings and a loan of \$4700. Mrs Malherbe met Mr Samuel Malherbe in 1971 and they married in October 1972. In March 1974 Mrs Malherbe completed repaying the loan on the unit, and the mortgage was discharged. The couple had separate bank accounts and Mr Malherbe made no contribution towards the loan repayments.

Mr Malherbe suffered a stroke in 1983 from which he never fully recovered. In May 1994, Mr Malherbe signed an enduring power of attorney (financial), appointing Mrs Malherbe as his representative. The document was witnessed by the couple's GP who was of the view that Mr Malherbe 'fully understood the significance of his actions' in signing the document. By 1999, Mr Malherbe's physical and mental health had deteriorated to the extent that Mrs Malherbe was no longer able to care for him at home and he was placed in a nursing home.

Mr Malherbe had three sons from a previous marriage. In December 1998 one son, Paul, obtained his father's signature on another enduring power of attorney (financial) document, which appointed him as Mr Malherbe's representative. Shortly after, Mr Malherbe's solicitor, Mr Harry Silver, lodged a caveat in Mr Malherbe's name on Mrs Malherbe's unit, claiming an interest in the property.

Justice Beach of the Supreme Court felt there was a 'clear inference' that Paul had obtained his father's signature on the enduring power of attorney (financial) in order to lodge the caveat. The Court was provided with a medical report dated April 1998 in which Mr Malherbe's GP concluded that Mr Malherbe was not 'of sound mind with respect to signing any legal documents within the last six months at least'. The Court found that, considering Mr Malherbe's lack of capacity, 'the power of attorney given to Paul was worthless'.

Mrs Malherbe sought an order from the Supreme Court of Victoria that the caveat on her property be removed and her costs for the legal proceedings be paid by Mr Silver. The Court recognised that Mr Silver may have initially believed that the enduring power of attorney granted Paul Malherbe valid authority. However he had been informed by Mrs Malherbe's solicitors that Mr Malherbe was 'without capacity and in care' and that Mrs Malherbe had purchased and paid for the property herself. Justice Beach found Mr Silver's actions after this point 'indefensible' and ordered the caveat be removed and that Mr Silver pay Mrs Malherbe's costs.

-

<sup>&</sup>lt;sup>504</sup> *Malherbe v Silver* [1999] VSC 176.

### **Chapter 5: Capacity**

The concept of capacity is fundamental to powers of attorney: a principal must have capacity to create a power of attorney and in some cases these powers may be activated when a principal has impaired decision-making capacity. At present the Victorian legislation and associated supports provide little guidance about what capacity is and how it should be assessed. In this chapter the Committee considers mechanisms for providing more certainty about capacity and capacity assessments.

### 5.1 Why is capacity important to this Inquiry?

The terms of reference for this Inquiry ask the Committee to consider the issue of capacity in the context of when an enduring power of attorney is created and activated.

A power of attorney is only valid if the principal has the necessary capacity at the time the document is created. The *Instruments Act 1958* (Vic) and the *Guardianship and Administration Act 1986* (Vic) do not define capacity, although as noted in the previous chapter, the Instruments Act provides that a person may make an enduring power of attorney (financial) if he or she 'understands the nature and effect' of the document. The Act sets out the matters witnesses should consider in determining whether a person has such an understanding. Chapter four considered how capacity to create a power of attorney should be established.

A non-enduring power of attorney is automatically revoked when a principal has impaired decision-making capacity. In addition, a principal may specify that an enduring power of attorney (financial) comes into effect when he or she has impaired decision-making capacity. However, the Instruments Act does not provide any guidance about determining when a principal has impaired decision-making capacity.

Enduring powers of attorney (guardianship) are only activated when a principal has impaired decision-making capacity. The Guardianship and Administration Act states that a representative is only able to exercise powers under an enduring power of attorney (guardianship) if, and only to the extent that the principal 'subsequently becomes unable by reason of a disability to make reasonable judgments in respect of any of the matters'. Disability is defined as an 'intellectual impairment, mental disorder, brain injury, physical disability or dementia'. Again the legislation does not provide any guidance about how to establish whether a person is unable to make reasonable judgments.

The activation of some enduring powers of attorney (financial) and all enduring powers of attorney (guardianship) when a principal has impaired decision-making capacity means that it is important to be able to identify when a principal has

<sup>&</sup>lt;sup>505</sup> Instruments Act 1958 (Vic) ss 118, 125A(1).

Berna Collier and Shannon Lindsay, Powers of attorney in Australia and New Zealand (1992) The Federation Press, 222.

<sup>&</sup>lt;sup>507</sup> Instruments Act 1958 (Vic) ss 117(1), 115(2).

Guardianship and Administration Act 1986 (Vic) ss 35B(1), 3.

impaired decision-making capacity and is unable to make a decision for himself or herself.

### 5.2 A consistent approach to capacity

Capacity is an extremely complex concept and the Committee heard that the different approaches contained in the various statutes cause widespread confusion and uncertainty. Many participants in this Inquiry called for a consistent approach to capacity in relation to all powers of attorney.<sup>509</sup>

The concept of capacity fundamentally underpins many other laws as well. Several Inquiry participants supported a uniform approach to capacity in all relevant laws, including the *Mental Health Act 1986* (Vic), the *Medical Treatment Act 1988* (Vic) and the *Victorian Civil and Administrative Tribunal Act 1998* (Vic). A standard approach to capacity was recently recommended in New South Wales. As noted in chapter three, some Inquiry participants advocated for an even broader approach, with a single capacity-based legislative regime for all substitute decision making in Victoria.

The House of Representatives Standing Committee on Legal and Constitutional Affairs *Older people and the law* report in 2007 found that the current ad hoc approach to assessing capacity throughout the nation 'does not provide an adequate level of transparency and protection of the interests of people making enduring powers of attorney'. The report recommended a nationally consistent approach to assessing capacity.

The Committee agrees that a uniform approach to capacity should be adopted in the proposed new Victorian Powers of Attorney Act, covering general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). Such an approach was recommended in chapter four in relation to capacity to create a power of attorney. Other elements of a consistent approach in terms of defining and assessing capacity are considered in the remainder of this chapter.

The Committee agrees there would be merit in adopting a uniform approach to capacity in all relevant laws in Victoria, as well as nationally. This is outside the scope of the Committee's present Inquiry, but the Committee draws this matter to the attention of the Victorian Government. In addition, the Committee notes that the Victorian Law Reform Commission is considering capacity extensively as part of its

Australian Medical Association Victoria, Submission 69, 2; Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 2; Seniors Rights Victoria, Submission 38, 18-19; Council on the Ageing Victoria, Submission 39, 3; David Goldberg, Solicitor and Senior Adviser, Australian Medical Association Victoria, Transcript of evidence, Melbourne, 17 December 2009, 3-4; Paula Chatfield, Legal Counsel, St Vincent's Hospital, Transcript of evidence, Melbourne, 14 December 2009, 3-4.

<sup>&</sup>lt;sup>510</sup> Carers Victoria, Submission 65, 4, 9.

Standing Committee on Social Issues, *Substitute decision-making for people lacking capacity* Report 43, New South Wales Legislative Council (2010), xx.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 91.

concurrent review of the Guardianship and Administration Act. Given the considerable overlap between the work of the Committee and the Commission, the Committee urges the Commission to take the Committee's findings and recommendations about capacity into account in its review.

#### Starting from a presumption of capacity 5.3

There is a common law presumption that all adults have capacity to make their own decisions unless there is evidence to the contrary. 513 Some jurisdictions have specifically incorporated a statutory presumption of capacity into their powers of attorney legislation. For example, the Mental Capacity Act 2005 (UK) states 'a person must be assumed to have capacity unless it is established that he lacks capacity'. 514

Participants in this Inquiry called for the Victorian powers of attorney legislation to be founded on a presumption of capacity. Professor Peteris Darzins, Professor of Geriatric Medicine at Monash University and Director of Geriatric Medicine at Eastern Health stated, 'In the whole process the emphasis has to be on allowing people to act as though they are capable, and should there be sufficient evidence of lack of capacity then something else needs to be done, rather than the other way around. '515

Inquiry participants emphasised that a presumption of capacity enhances a principal's human rights, in particular, the right of 'persons with disabilities [to] enjoy legal capacity on an equal basis with others in all aspects of life' enshrined in the International Convention on the Rights of Persons with Disabilities. 516 Ms Janet Wood, President of the Council on the Ageing Victoria, warned that 'unless we start from the notion of the continuing right to make decisions, we can end up with legislation that is too protective and takes away a right too early'. 517

The Committee agrees that the powers of attorney legislation should be founded on a presumption of capacity. This would promote principals' rights by making it clear that all decisions about a person's capacity should start from the assumption that a person has capacity to make decisions for himself or herself. The presumption will

Borthwick v Carruthers (1787) 99 ER 1300.

Mental Capacity Act 2005 (UK) c 9 s 1(2). See also Powers of Attorney Act 1998 (Qld) sch 1 principle 1; Powers of Attorney Act 2006 (ACT) s 18; Standing Committee on Social Issues, above n 511, 44, 62 which recommended legislating for a presumption of capacity in NSW.

Peteris Darzins, Professor of Geriatric Medicine, Monash University, and Director of Geriatric Medicine, Eastern Health, Transcript of evidence, Melbourne, 30 March 2010, 3.

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12(2) (entered into force 3 May 2008).

Janet Wood, President, Council on the Ageing Victoria, Transcript of evidence, Melbourne, 22 October 2009, 3. See also Law Institute of Victoria, Submission 41, 9-10; Mental Health Legal Centre Inc, Submission 59, 4; Carers Victoria, Submission 65, 12; Robyn Mills, Transcript of evidence, above n 509, 6; Kristen Pearson, President, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 2; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2; Stephen Taffe, Legal Counsel, Alfred Health, Transcript of evidence, Melbourne, 14 December 2009, 4; Seniors Rights Victoria, Submission 38, 27; Royal College of Nursing Australia, Submission 26, 3.

be displaced if there is evidence that a person has impaired decision-making capacity. What that evidence would look like is explored later in this chapter.

#### **Recommendation 33: Presumption of capacity**

The Committee recommends the Powers of Attorney Act provide that a person is presumed to have capacity to make his or her own decisions.

### 5.4 Defining capacity

Many participants in this Inquiry called for the powers of attorney legislation to provide a clear definition of capacity. For instance, the Australian Medical Association submitted that the current definitions of capacity are 'varied and erratic' and 'unhelpful to a donor, witness, doctor and the operation of the law'. 518

Participants referred the Committee to the definitions of capacity set out in legislation in some other jurisdictions.

For example, the powers of attorney legislation in Queensland defines capacity as meaning that, in relation to a particular matter, the person is capable of:

- understanding the nature and effect of decisions about the matter
- freely and voluntarily making decisions about the matter
- communicating the decisions in some way. <sup>519</sup>

The Australian Capital Territory (ACT) legislation states that a person has decision-making capacity 'if the person can make decisions in relation to the person's affairs and understands the nature and effect of the decisions'. 520

The legislation in both those jurisdictions also defines impaired decision-making capacity, which is basically the absence of capacity for a matter or decision. <sup>521</sup>

The Mental Capacity Act in the United Kingdom (UK) defines lack of capacity rather than capacity. It states that 'a person lacks capacity in relation to a matter if at the material time he is unable to make a decision for himself in relation to the matter because of an impairment of, or a disturbance in the functioning of, the mind or

Australian Medical Association Victoria, Submission 69, 2. See also Robyn Mills, Transcript of evidence, above n 509, 2; Seniors Rights Victoria, Submission 38, 19; Council on the Ageing Victoria, Submission 39, 3; David Goldberg, Transcript of evidence, above n 509, 3-4; Carers Victoria, Submission 65, 10-12; Paula Chatfield, Transcript of evidence, above n 509, 3-4.

Powers of Attorney Act 1998 (Qld) s 3, sch 3; Guardianship and Administration Act 2000 (Qld) s 3, sch 4.

Powers of Attorney Act 2006 (ACT) s 9.

Powers of Attorney Act 1998 (Qld) s 3, sch 3; Guardianship and Administration Act 2000 (Qld) s 3, sch 4;
 Powers of Attorney Act 2006 (ACT) ss 9, 91.

brain'. 522 The Act specifies that a person who is unable to make a decision for himself or herself is someone who is unable to:

- understand information relevant to the decision
- retain that information
- use or weigh that information as part of the process of making the decision
- communicate the decision. 523

Professor Darzins advocated a similar approach, although he suggested different terminology:

capable people know the issues they face, they also know the possible approaches for dealing with those issues, they also appreciate the reasonably foreseeable consequences of choices and their decisions are not based on a delusional construct.

If all of those things are true, then the people are capable even if they have cognitive impairment. This latter bit is important because doctors will often confuse the presence of cognitive impairment, such as Alzheimer's dementia, a stroke or acquired brain injury; they will say that because there is cognitive impairment present the person lacks capacity, or that if the cognitive impairment has been present for a long time and is profound then that equals lack of capacity. That may be true, but the logic is not right. It is possible for people to have cognitive impairment without lacking capacity. <sup>524</sup>

Most Inquiry participants who called for capacity to be defined in the Victorian legislation did not specify what that definition should be. Of those who did, Seniors Rights Victoria advocated the adoption of the UK approach. The Mental Health Legal Centre suggested a combined approach incorporating the ability to:

- understand the information relevant to making the decision
- retain the relevant information
- weigh up the relevant information
- communicate a decision. 526

Ms Elizabeth Mullaly, Manager of Psychology Services at Caulfield Hospital, who gave evidence on behalf of the Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, also supported this definition, stating 'I think the layperson could probably get a bit of a grip on that by maybe having those four ... As neuropsychologists we would agree that they are the

<sup>&</sup>lt;sup>522</sup> Mental Capacity Act 2005 (UK) c 9 s 2(1).

<sup>&</sup>lt;sup>523</sup> Mental Capacity Act 2005 (UK) c 9 s 3(1).

Peteris Darzins, *Transcript of evidence*, above n 515, 3. See also Peteris Darzins, *Assessment of decision-making capacity: Key concept summary*, supplementary evidence received 30 March 2010, 1; Kristen Pearson, *Transcript of evidence*, above n 517, 4-5.

Elizabeth Samra, Lawyer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 7.

Mental Health Legal Centre Inc, Submission 59, 5.

vital elements that need to be in place.' The Victorian Coalition of Acquired Brain Injury Service Providers also advocated a similar approach, but preferred 'ability to appreciate the consequences of a decision' to 'ability to retain the information relevant to the decision'. 528

The Committee agrees the Victorian powers of attorney legislation should contain a clear definition of capacity. The Committee believes it is important that the definition should be as simple as possible to provide maximum clarity for those assessing capacity and affected by capacity assessments. Thus the Committee's preferred definition incorporates the four elements that are common to most of the definitions of capacity examined: the ability to understand the relevant information, the ability to retain the relevant information, the ability to weigh up the relevant information and the ability to communicate a decision in some way.

The Committee believes the legislation should also define impaired decision-making capacity, which is the absence of capacity for a particular decision. Those making capacity assessments should be provided with guidance materials to help them apply these definitions consistently and transparently. This will be discussed later in this chapter.

\_

Elizabeth Mullaly, Member of College of Clinical Neuropsychologists Course Approval Sub-Committee, and Manager of Psychology Services, Caulfield Hospital, Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, Transcript of evidence, Melbourne, 14 December 2009, 7. See also Standing Committee on Social Issues, above n 511, 35 which recommended a similar approach; Kelly Purser, Ellis S Magner and Jeanne Madison, 'Competency and capacity: The legal and medical interface' (2009) 16(5) Journal of Law and Medicine 789, 796; The Law Society of New South Wales, When a client's capacity is in doubt: A practical guide for solicitors (2009), 2 citing Jenna MacNab, 'Capacity: A practical guide for lawyers' (2008) 46(5) Law Society Journal 68, 71. See also Attorney General's Department, New South Wales, Capacity toolkit: Information for government and community workers, professionals, families and carers in New South Wales (2008), 18; H Bennett and P Hallen, 'Guardianship and financial management legislation: What doctors in aged care need to know' (2005) 35(8) Internal Medicine Journal 482, 486; Malcolm Parker, 'Judging capacity: Paternalism and the risk-related standard' (2004) 11(4) Journal of Law and Medicine 482, 485-486; John Dawson and Annegret Kämpf, 'Incapacity principles in mental health laws in Europe' (2006) 12(3) Psychology, Public Policy, and Law 310, 310; Deborah O'Connor and Barbara Purves (eds), Decision-making, personhood and dementia: Exploring the interface (2009) Jessica Kingsley Publishers, 12; Clare Delany, 'Decision-making and consent' in Rosemary Kennedy (ed), Allied health professionals and the law (2008) The Federation Press, 99, 101 citing Thomas Grisso and Paul S Appelbaum, Assessing competence to consent to treatment: A guide for physicians and other health professionals (1998) Oxford University Press, and John Devereux and Malcolm Parker, 'Competency issues for young persons and older persons' in Ian Freckelton and Kerry Peterson (eds), Disputes and dilemmas in health law (2006) The Federation Press, 58. See also Peteris Darzins, D William Molloy and David Strang (eds), Who can decide? The six step capacity assessment process (2000) Memory Australia Press, 9; Mental Capacity Act 2005 (UK) c 9 ss 2, 3; Department of Human Services, Victoria, With respect to age - 2009: Victorian Government practice guidelines for health services and community agencies for the prevention of elder abuse (2009), 98-99.

### Recommendation 34: Defining capacity and impaired decision-making capacity

The Committee recommends the Powers of Attorney Act:

- a) state that a person has capacity to make a decision if he or she has:
  - the ability to understand the information relevant to making the decision
  - the ability to retain the relevant information
  - the ability to weigh up the relevant information

**AND** 

- the ability to communicate the decision in some way.
- b) state that a person has impaired decision-making capacity if, in relation to a decision, he or she does not have:
  - the ability to understand the information relevant to making the decision
  - the ability to retain the relevant information
  - the ability to weigh up the relevant information

OR

- the ability to communicate the decision in some way.

# 5.5 Protecting rights when assessing capacity – Principles to guide capacity assessments

Decisions about a person's capacity potentially have significant human rights implications. For example, a decision that a person does not have capacity to choose where to live may deprive a person of his or her freedom.

Participants in this Inquiry emphasised the importance of ensuring that principals' rights are protected when capacity assessments are made. In particular, participants highlighted the rights of recognition and equality before the law and freedom of movement under the *Charter of Human Rights and Responsibilities Act* 

above n 512, 91; Peteris Darzins, D William Molloy and David Strang (eds), above n 527, 3, 10.

Seniors Rights Victoria, Submission 38, 24 citing Law Reform Commission, Ireland, Consultation paper on law and the elderly LRC CP 23 - 2003 (2003), 11. See also Attorney General's Department, New South Wales, Are the rights of people whose capacity is in question being adequately promoted and protected? A discussion paper (2006), 3; House of Representatives Standing Committee on Legal and Constitutional Affairs,

2006 (Vic)<sup>530</sup> and the rights of equal legal capacity and access to support to exercise legal capacity set out in the *International Convention on the Rights of Persons with Disabilities*.<sup>531</sup>

Inquiry participants suggested a number of principles to promote rights that could be incorporated into power of attorney laws to underpin decisions about a person's capacity.

### 5.5.1 Capacity is decision-specific

Many Inquiry participants emphasised that capacity is decision-specific: a person may have capacity to make decisions in relation to some matters but not others. For instance, a person may be able to make day-to-day decisions about financial matters or his or her lifestyle, but may not be able to make larger, more complex decisions, for example, whether to sell a house or to enter an aged care facility. <sup>532</sup>

Inquiry participants expressed concern that the current Victorian legislation encourages an all or nothing approach to capacity. For example, Carers Victoria stated:

Individuals rarely lose all capacity at a particular moment. Most people with lifelong intellectual disabilities and people with acquired brain injuries including degenerative disorders – that is people with capacity disabilities – have the capacity to make some decisions and express their preferences in relation to others. Notwithstanding that they are often reliant on others to operationalise their decisions.

We believe that the current laws that suggest a person either has capacity or they don't are too simplistic and fail to capture peoples lived experience. 533

Similarly, former Public Advocate Julian Gardner told the Committee that 'it is very important that we see capacity as not being a black and white concept. People's capacity can vary from day to day, and it can vary from decision to decision'. 534

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12 (entered into force 3 May 2008).

<sup>&</sup>lt;sup>530</sup> Charter of Human Rights and Responsibilities Act 2006 (Vic) ss 8, 12, 13.

Law Institute of Victoria, Submission 41, 9. See also Sue Connelly, Committee member, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 6.

Carers Victoria, Submission 65, 10. See also David Goldberg, Transcript of evidence, above n 509, 3; Royal District Nursing Service, Submission 25, 3; Stephen Taffe, Transcript of evidence, above n 517, 4; Jeni Lee, Submission 57, 2; Seniors Rights Victoria, Submission 38, 24; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2; Robyn Mills, Transcript of evidence, above n 509, 3; Attorney General's Department, New South Wales, above n 527, 19; Queensland Law Reform Commission, Shaping Queensland's guardianship legislation: Principles and capacity Discussion paper (2008), 10, 108; Advance Directives Review Committee, Planning ahead: Your health, your money, your life. First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy (2008), 3, 10; Terry Carney and Patrick Keyzer, 'Planning for the future: Arrangements for the assistance of people planning for the future of people with impaired capacity' (2007) 7(2) Queensland University of Technology Law and Justice Journal 255, 257.

Julian Gardner, Transcript of evidence, Melbourne, 1 October 2009, 3. See also Advance Directives Review Committee, above n 533, 46.

Inquiry participants stated that approaches that provide findings of global incapacity may compromise the rights of a principal who may be able to make some decisions. <sup>535</sup>

The definitions of capacity in the UK and Queensland discussed earlier in this chapter make it clear that capacity is decision-specific, that is, the principal's decision-making capacity must be considered in relation to each decision that needs to be made. The Committee's proposed definitions of capacity and impaired decision-making capacity also recognise that capacity is decision-specific.

### 5.5.2 Capacity fluctuates

Participants in this Inquiry also highlighted that capacity is not static. A person's capacity to make decisions may fluctuate depending on factors such as his or her mental and physical health or the time of day. <sup>537</sup> Mr David Goldberg, Solicitor and Senior Adviser with the Australian Medical Association, emphasised that capacity is not 'on-off' and 'can vary from day to day, hour to hour'. <sup>538</sup>

The Committee was provided with many examples of fluctuating capacity. The Victorian Coalition of Acquired Brain Injury Service Providers submitted:

In the case of acquired brain injury, capacity may change considerably from immediate post-injury phase to later stages in the rehabilitation process, which for many people is life-long ... there is a significant risk that assessments (particularly those done in acute settings) indicating a lack of capacity can quickly become outdated. This leaves the person with the injury without appropriate legal recognition of their right to determine their own future. 539

Similarly, Ms Wood from the Council on the Ageing Victoria told the Committee:

some of the case studies from SRV [Seniors Rights Victoria] have involved people who have gone into hospital for three weeks, six weeks, say, and in the interim their worldly goods have been taken away from them. They come out having recovered from whatever it was, and it is all gone. We must not have legislation that allows an anticipation of incapacity. Increasingly as one can recover from a stroke or all sorts of things like that, there needs to be more of an emphasis on permanent incapacity; we need to be very careful about temporary incapacity. <sup>540</sup>

\_

Alzheimer's Australia Vic, Submission 32, 4; Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 4.

Mental Capacity Act 2005 (UK) c 9 s 2(2), 3(3); Powers of Attorney Act 1998 (Qld) s 3, sch 3. See also Attorney General's Department, New South Wales, above n 527, 19-23; Rachel Zombor, Clinical neuropsychologist, 'The role of neuropsychology in the assessment of decision making capacity' (Paper presented at the Australian Guardianship and Administration Council: Social inclusion: The future of ageing, disability and substituted decision-making conference, Brisbane, 19-20 March 2009), 4; Mental Health Legal Centre Inc, Submission 59, 4-5.

Alzheimer's Australia Vic, *Submission 32*, 4; Attorney General's Department, New South Wales, above n 527, 19-23; Attorney General's Department, New South Wales, above n 529, 3.

David Goldberg, *Transcript of evidence*, above n 509, 3.

Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2. See also Tom Worsnop, Chair, Victorian Coalition of Acquired Brain Injury Service Providers, Transcript of evidence, Melbourne, 30 March 2010, 2, 4.

Janet Wood, Transcript of evidence, above n 517, 3.

These examples highlight the importance of capacity assessments, particularly decisions about impaired decision-making capacity, not being a one-off. The UK legislation specifically recognises that impaired decision-making capacity may be 'permanent or temporary'. The code of practice which supports that legislation suggests that decisions about a person's capacity should be regularly reviewed, and if a person is unable to make a decision for himself or herself, consideration should be given to delaying the decision to see whether the person might be able to make the decision at some other time. This approach was supported by some Inquiry participants. This approach was supported by some Inquiry participants.

### Case study 9: He was unable to return home as his house had been emptied out 544

'A man appointed his two sons as his joint attorneys and enduring guardians. During a hospital admission the sons decided their father would be unable to return to his own home and sold his household belongings and his car and put his house up for sale. When the man was ready to be discharged from hospital he was effectively unable to return home as the house had been emptied out and the sons refused to return his belongings.'

# 5.5.3 Capacity should not be assumed based on a person's appearance

The UK's Mental Capacity Act states that 'a lack of capacity cannot be established merely by reference to a person's age or appearance'. <sup>545</sup> The code of practice which supports the Act gives some further guidance about what this means:

The Act deliberately uses the word 'appearance', because it covers all aspects of the way people look. So for example, it includes the physical characteristics of certain conditions (for example, scars, features linked to Down's syndrome or muscle spasms caused by cerebral palsy) as well as aspects of appearance like skin colour, tattoos and body piercings, or the way people dress (including religious dress). 546

Mr Robert Bolch, a Justice of the Peace who made a submission to the Inquiry, observed that some witnesses judge that a person with motor neurone disease who is unable to physically sign a power of attorney document does not have capacity to make the document and commented, 'Because a donor is unable to control a pen to sign a document does not relate to their reasoning processes ...' 547

<sup>&</sup>lt;sup>541</sup> Mental Capacity Act 2005 (UK) c 9 s 2(2).

Department for Constitutional Affairs, United Kingdom, Mental Capacity Act 2005 Code of practice, The Stationery Office (2007), 49-50. See also Attorney General's Department, New South Wales, above n 527, 19-23.

<sup>&</sup>lt;sup>543</sup> Carers Victoria, Submission 65, 11; Mental Health Legal Centre Inc, Submission 59, 4-5.

Office of the Public Advocate, *Submission* 9, 9.

Mental Capacity Act 2005 (UK) c 9 s 2(3). See also Attorney General's Department, New South Wales, above n 527, 33-35.

Department for Constitutional Affairs, United Kingdom, above n 542, 43.

Robert C Bolch, JP, Submission 3, 8. See also Elizabeth Mullaly, Transcript of evidence, above n 527, 4.

# 5.5.4 Capacity assessments should focus on a person's ability to make a decision not the outcome of a decision

The Committee heard that capacity assessments should focus on a person's ability to make a decision, rather than judging the actual decisions he or she makes. Ms Catherine Leslie, Legal and Policy Officer at the Mental Health Legal Centre, told the Committee that impaired decision-making capacity should not be assumed merely because a person makes a decision that someone else might perceive to be unwise:

I think it is probably fair to say that we often make decisions which people find uncomfortable and difficult to deal with, but we should be supporting the person to be able to understand the relevant information, if that is necessary, and to be able to allow that person to be informed to make a wise or unwise decision. We should not be judging simply on the basis of what kind of decision they have made, whether that decision itself is reasonable but looking at the way in which the person has made the decision. <sup>548</sup>

Professor Darzins agreed, observing that 'the vast majority of irrational choices are made by competent people'. 549

The UK legislation specifically states that 'a person is not to be treated as unable to make a decision merely because he makes an unwise decision'. A recent review of substitute decision-making in New South Wales recommended that a similar provision should be inserted into all legislation which deals with capacity in that state. The Committee recognises that in some circumstances certain behaviour may be evidence of impaired decision-making capacity.

## 5.5.5 People should be supported to make their own decisions where possible

In chapter three of this report the Committee recommended that the proposed new Victorian Powers of Attorney Act should be underpinned by the notion of supported decision making. This is in line with the *International Convention on the Rights of Persons with Disabilities* which stipulates that people with disabilities must be given the necessary support to exercise their legal capacity. <sup>552</sup>

Participants in this Inquiry emphasised that there are a range of factors that influence a person's ability to make a decision for himself or herself. Alzheimer's Australia Victoria wrote: 'A person's overall capacity to make decisions can be enhanced by

Catherine Leslie, Transcript of evidence, above n 535, 3-4. See also Seniors Rights Victoria, Submission 38, 28; Mental Health Legal Centre Inc, Submission 59, 5.

Peteris Darzins, *Transcript of evidence*, above n 515, 4.

<sup>&</sup>lt;sup>550</sup> *Mental Capacity Act* 2005 (UK) c 9 s 1(4).

<sup>551</sup> Standing Committee on Social Issues, above n 511, 63.

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12(2) (entered into force 3 May 2008).

personal strengths, good service provision, information and support. Personal limitations, poor service provision and lack of support can limit it'. 553

Similarly, Carers Victoria submitted that the Committee 'should consider whether an individual and or caring family's lack of access to appropriate resources may be the source of, or be contributing to an individual's capacity disability'. State Carers Victoria suggested that legislation should enshrine the right of people with impaired decision-making capacity to support to make their own decisions, including access to advocacy, support to build their capacity, appropriate communication aids and materials in their first language. The Victorian Coalition of Acquired Brain Injury Service Providers also emphasised the importance of access to communication aids, observing that even experienced health professionals sometimes mistake a lack of ability to communicate for a lack of capacity to make a decision.

Ms Kerry Stringer, Community Partnerships Manager at the Summer Foundation, who gave evidence on behalf of the Victorian Coalition of Acquired Brain Injury Service Providers, told the Committee that people with brain injuries are often not supported to make decisions, even when they are capable of making decisions themselves:

We had an example recently of a young man who is in a locked unit in a specialist residential aged care facility whose father has power of attorney and who has retired interstate. Our view is that that young man could live somewhere else. Again, I do not think there is any malice or intent around the decision making. The parent likes it where he is because he is safe. He does not have to worry about him. He is not getting those phone calls at midnight. He kind of retired. Our view is that if the young man were given the opportunity to look at alternative accommodation and spend some time out in the community, he would be able to make a decision about an alternative place to live. <sup>557</sup>

Many other jurisdictions have introduced a principle that a person should not be treated as unable to make a decision unless all practicable steps to help him or her to do so have been taken without success. For example, legislative principles underpinning the powers of attorney legislation in Queensland include a principle that a person 'must be given any necessary support, and access to information, to enable the adult to participate in decisions affecting the adult's life'. <sup>558</sup>

Alzheimer's Australia Vic, Submission 32, 4. See also Attorney General's Department, New South Wales, above n 529, 3.

<sup>&</sup>lt;sup>554</sup> Carers Victoria, Submission 65, 11.

<sup>555</sup> Ibid.

Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2. See also Tom Worsnop, Transcript of evidence, above n 539, 2.

Kerry Stringer, Community Partnerships Manager, Summer Foundation, Victorian Coalition of Acquired Brain Injury Service Providers, *Transcript of evidence*, Melbourne, 30 March 2010, 5.

Powers of Attorney Act 1998 (Qld) sch 1 cl 7(3). See also Powers of Attorney Act 2006 (ACT) sch 1 cl 1.6; Mental Capacity Act 2005 (UK) c 9 s 1(3); Department for Constitutional Affairs, United Kingdom, above n 542, 22-24, ch 3; Attorney General's Department, New South Wales, above n 527, 42-44.

### 5.5.6 Principles to guide capacity assessments – The Committee's view

In chapter three the Committee recommended that the proposed new Powers of Attorney Act be supported by a statement of general principles that must be applied by all those exercising powers or functions under the Act in relation to a principal with impaired decision-making capacity. The statement of principles suggested by the Committee has three elements: an articulation of relevant human rights contained in international conventions, an emphasis on ensuring that all decisions about a person's capacity uphold his or her rights and a focus on ensuring that all decisions and actions under enduring powers of attorney promote principals' interests and wellbeing. The first component was discussed in chapter three and the third component will be discussed in the next chapter. The Committee considers the second element in this section.

The Committee recognises that decisions about a person's capacity have the potential to significantly impact that person's human rights. The Committee believes that providing principles to guide capacity assessments as part of the proposed general principles underpinning the Powers of Attorney Act will help ensure that principals' rights are at the forefront when capacity is assessed.

The Committee therefore recommends that the statement of general principles to be contained in the new Powers of Attorney Act should include fundamental principles to guide the conduct of capacity assessments. These principles should be that capacity is decision-specific and fluctuates over time, capacity should not be assumed based on a person's appearance, a person must not be presumed to have impaired decision-making capacity simply because he or she makes a decision that others may view as unwise and a person should not be treated as unable to make a decision if it is possible for him or her to make that decision with appropriate support.

The definition of capacity that the Committee recommended earlier in this chapter also recognises that capacity fluctuates and is decision-specific. However, the Committee believes that incorporating these concepts as separate principles will highlight their importance to people making decisions about others' capacity. The last suggested principle provides more specific guidance about how the principle of supported decision making, which the Committee has proposed as a fundamental pillar of the new Act, should be implemented.

Later in this chapter the Committee recommends the development of guidance materials to assist people making capacity assessments. This material should highlight these principles and provide practical information about how each principle can be implemented.

#### Recommendation 35: Principles to guide capacity assessments

The Committee recommends the statement of principles underpinning the Powers of Attorney Act recommended in recommendation 3 include principles to guide capacity assessments. These principles should include:

- capacity is specific to each decision to be made
- impaired decision-making capacity may be temporary or permanent
- capacity should not be assumed based on a person's appearance
- a person must not be presumed to have impaired decision-making capacity merely because he or she makes a decision that is, in the opinion of others, unwise
- a person should not be treated as unable to make a decision if it is possible for him or her to make that decision with appropriate support.

### 5.6 Who should assess capacity?

As noted in chapter four, the two witnesses to both enduring powers of attorney (financial) and enduring powers of attorney (guardianship) documents are required to certify that the principal appeared to have the capacity to create the document. The creation of a general (non-enduring) power of attorney does not need to be witnessed and consequently there is no check of a principal's capacity to create that document. In that chapter, the Committee recommended the introduction of consistent witnessing requirements for all powers of attorney, including general (non-enduring) powers of attorney.

Neither the Instruments Act nor the Guardianship and Administration Act provide any guidance about how impaired decision-making capacity should be assessed in terms of the activation of powers of attorney or who should perform the capacity assessment. In practice, it is usually the representative who assesses that a principal does not have capacity to make a particular decision, thus activating an enduring power.

### 5.6.1 When is a medical assessment of capacity required?

Most participants in this Inquiry agreed that assessment of capacity is a legal not a medical issue.<sup>560</sup> However, as the Australian Medical Association's submission observed, 'The threshold for reaching the legal conclusion regarding capacity will

\_\_\_

Instruments Act 1958 (Vic) s 125A; Guardianship and Administration Act 1986 (Vic) s35A, sch 4 form 1.
 David Goldberg, Transcript of evidence, above n 509, 3; Kristen Pearson, Transcript of evidence, above n 517, 56. See also Sue Field, 'Assessing mental capacity' (2008) 86 Precedent 45, 47; Peteris Darzins, D William Molloy and David Strang (eds), above n 527. Cf John Myers, Submission 56, 16; The Royal Women's Hospital, Submission 8, 1.

often require medical input.'561 The Committee received a variety of evidence about when a medical assessment of capacity to create and activate a power of attorney is necessary.

### Medical assessment of capacity to create a power of attorney

The Committee heard that requiring medical evidence of capacity at the time a power of attorney is created would provide more certainty about the validity of a document. The Australian Association of Gerontology told the Committee that having an expert assessment of capacity to create a power of attorney, for example by a general practitioner:

could provide certainty not only to the person who is giving EPoA [enduring power of attorney] to someone, but also to the financial or health-care provider who has to act on that authority at some time in the future. This would also help to reduce vexatious claims from family/others that the person was not competent when they completed the document. <sup>562</sup>

Some participants felt that a medical assessment of capacity should be obtained in all cases to avoid any doubt about a principal's capacity at the time he or she creates the document. Seniors Rights Victoria proposed that, if a registration system is introduced, the capacity assessment could be lodged with the registration body.

However, other participants expressed concern that requiring a medical assessment of capacity in all cases would discourage members of the community from making powers of attorney. <sup>565</sup> In particular, there was concern about the additional costs that would be imposed on principals. The Ministerial Advisory Council of Senior Victorians stated 'we do not want to suggest anything that will mean additional costs for seniors'. <sup>566</sup> State Trustees observed that powers of attorney are often created in crisis situations and there may be no time to obtain a formal capacity assessment:

Not infrequently, enduring powers (and Wills) are required to be prepared as a matter of urgency because the donor has left it till the "last minute" (for example, they are in hospital about to have an operation); avoidable delays that arise in such circumstances may result in the donor's wishes ultimately being thwarted because they lose capacity before the document can be executed ... <sup>567</sup>

A professional assessment of capacity can take between six and sixteen hours. 568

Australian Medical Association Victoria, Submission 69, 2. See also David Goldberg, Transcript of evidence, above n 509, 3.

Australian Association of Gerontology Inc, Submission 15, 1. See also The Royal Women's Hospital, Submission 8, 1; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, att, 7.

Association of Independent Retirees (AIR) Ltd (Victorian Division), *Submission 12*, 4; Audrey Cooke, *Submission 5*, 1; Murray Mallee Community Legal Service, *Submission 22*, 2.

Seniors Rights Victoria, Submission 38, 25.

Office of the Public Advocate, *Submission 9*, 12; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Transcript of evidence*, Melbourne, 1 October 2009, 4; Mental Health Legal Centre Inc, *Submission 59*, 5-6.

Ministerial Advisory Council of Senior Victorians, Submission 48, 2. See also The Institute of Legal Executives (Victoria), Submission 33, 2.

<sup>567</sup> State Trustees Limited, Submission 58, 8.

Attorney General's Department, New South Wales, above n 529, 13.

Most participants who commented on this issue supported obtaining a medical assessment only when there is doubt about a principal's capacity to create a power of attorney. Dr Kristen Pearson, President of the Australian & New Zealand Society for Geriatric Medicine, told the Committee, 'I do not think doctors want to get involved with every person who signs an enduring power of attorney ... But clearly there will be a role for medical practitioners where there is doubt ...' Former Public Advocate, Mr Gardner, expressed the view that the cases where a specialised medical assessment is needed are quite rare. <sup>570</sup>

The Victorian Bar suggested that the legislation could include an evidentiary provision stating that if a registered medical practitioner signs a capacity certificate then 'unless the contrary is proved, a person has capacity to create a Power of Attorney'. <sup>571</sup>

Guidance materials for capacity assessors in New South Wales, Queensland and the UK suggest obtaining an expert capacity assessment where there is doubt about a person's capacity to create a power of attorney document. The Law Institute of Victoria commended the 'warning signs' that may warrant a further investigation of capacity set out in the Law Society of New South Wales' guidelines When a client's capacity is in doubt: A practical guide for solicitors. These include:

- the principal has difficulty with memory or recall
- the principal has difficulty communicating
- the principal is disorientated
- the principal is in hospital or an aged care facility. 573

Case studies 10 and 11 provide two examples of when an expert assessment of capacity, in those instances performed by neuropsychologists, was helpful in determining a principal's capacity to create a power of attorney.

Kristen Pearson, Transcript of evidence, above n 517, 5. See also John Billings, Submission 37, above n 562, att, 7; Catherine Leslie, Transcript of evidence, above n 535, 5; John Billings, Transcript of evidence, above n 565, 4; Moreland Community Legal Centre Inc, Submission 51, 3; Law Institute of Victoria, Submission 41, 12; Robert C Bolch, JP, Submission 3, 8; Mental Health Legal Centre Inc, Submission 59, 5; Matthew Hughes, Submission 62, 1; Lillian Jeter, Executive Director, Elder Abuse Prevention Association, Transcript of evidence, Melbourne, 30 March 2010, 3.

Julian Gardner, *Transcript of evidence*, above n 534, 9.

The Victorian Bar, Submission 40, 3. See also Australian Medical Association Victoria, Submission 69, 2.

Department for Constitutional Affairs, United Kingdom, above n 542, 59-61; Attorney General's Department, New South Wales, above n 527, 56-57; Office of the Adult Guardian, Queensland, *Capacity guidelines for witnesses of enduring powers of attorney* (2005), 3.

Law Institute of Victoria, *Submission 41*, 12 citing The Law Society of New South Wales, above n 527, 4. See also Seniors Rights Victoria, *Submission 38*, 25.

### Case study 10: 'Her cognitive impairment was not readily apparent', 574

'Mrs S was assessed in a memory clinic. She had a three year history of gradual cognitive decline despite good health. She had four adult children who had come into conflict over the best approach to her care needs. One daughter had recently taken her to a solicitor to have her sign an Enduring Power of Guardianship (EPG) but the lawyer and other siblings requested that her capacity be assessed. Mrs S was pleasant, co-operative and aware of her surroundings. She did not appear confused on the surface and she carried out superficial social conversation without difficulty. Her neuropsychological results showed preservation of long term (remote) memories but severe impairment of recent memory and new learning. Other mild cognitive difficulties were present, including problems with reasoning, and the results suggested established Alzheimer's disease. There was no evidence of fluctuation in mental state. Close questioning revealed that she had a very general understanding of the meaning of an EPA [enduring power of attorney] but she had no appreciation that her children were in conflict about her care and she had no appreciation of the basis of that conflict (i.e. her increasing care needs and how they could best be met). Explanation did not help because she was unable to retain the information. She tended to operate on the basis of a long term tendency to be agreeable. The neuropsychologist considered that her formal cognitive assessment results were consistent with her performance when interviewed about the matter under consideration. The neuropsychologist felt that Mrs S was unable, on the basis of her memory impairment and her problems with reasoning, to appreciate the likely effect of signing an EPG to one daughter over and above her other children.

It is possible that if Mrs S had been subjected to a brief structured discussion using leading questions in the presence of her daughter she may have appeared to have had the requisite understanding to sign. Her cognitive impairment was not readily apparent at a superficial conversational level but the neuropsychological assessment raised concerns that were subsequently confirmed by a more detailed discussion of her circumstances and wishes.'

### Case study 11: Twelve months after her stroke she had regained capacity to appoint a representative 575

'Mrs K had a large stroke affecting the areas at the front of her brain on the left side. In the early months, her husband was appointed administrator as she had never signed an EPA [enduring power of attorney] and lacked the capacity to do so. Twelve months later she was attending rehabilitation. She continued to have considerable problems with expression of ideas but she had reasonable comprehension of language and she could read. A specialised neuropsychological assessment using yes/no responses to a series of questions (following education) suggested that she had regained the capacity to appoint a financial EPA and she indicated that she would choose to appoint her husband. Her administration order was subsequently revoked.'

-

Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, Submission 68, 6.

<sup>&</sup>lt;sup>575</sup> Ibid, 7.

### A medical assessment of capacity at time of activation?

There were also divided views about whether a medical assessment of capacity should be required in order to activate an enduring power of attorney.

Several participants supported requiring all representatives to obtain a medical certificate confirming that the principal has impaired decision-making capacity, with some suggesting the certificate should be provided to the Office of the Public Advocate or a registration body before an enduring power can take effect. Some participants suggested that the medical certificate should include an indication of whether the impaired decision-making capacity is likely to be permanent. The state of the provided to the Office of the Public Advocate or a registration body before an enduring power can take effect. The participants suggested that the medical certificate should include an indication of whether the impaired decision-making capacity is likely to be permanent.

Others suggested that a medical assessment is only necessary when there is doubt about a principal's capacity.<sup>578</sup> Again resources in other jurisdictions provide guidance about when an expert assessment is needed at the activation stage. For example, the UK code of practice states that a formal capacity assessment should be sought in a range of circumstances including where:

- the decision is complicated or has serious consequences
- the principal challenges a finding of impaired decision-making capacity
- there is disagreement between family members, carers and/or professionals about the principal's capacity
- there is a conflict of interest between the assessor and the principal.<sup>579</sup>

### Medical assessment of capacity – The Committee's view

The Committee believes that requiring medical certification of a principal's capacity to create or activate a power of attorney would be time-consuming, costly and may discourage the use of powers of attorney. While the Committee notes there was support among Inquiry participants, particularly for medical certification in order to activate a document, it does not believe that this fits well into a framework based on the notion that capacity fluctuates and is decision-specific, as discussed earlier in this chapter. The Committee considers alternative ways of evidencing the activation of an enduring power of attorney below.

The Committee's view is that a medical assessment of capacity should be sought whenever there is doubt about a principal's capacity to create a power of attorney or

\_

St Kilda Legal Service Co-Op Ltd, Submission 50, 2-3; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 3; Royal Victorian Association of Honorary Justices, Submission 28, 1; Jeni Lee, Submission 57, 4; Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 8; Royal College of Nursing Australia, Submission 26, 3; Murray Mallee Community Legal Service, Submission 22, 2; Hunt & Hunt, Submission 24, 3; Lucie O'Brien, Policy Officer, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 4; Seniors Rights Victoria, Submission 38, 29; D Constantine and L Richmond, Submission 75, 1.

Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 3; St Vincent's Hospital, *Submission 19*, 6; Paula Chatfield, *Transcript of evidence*, above n 509, 6.

Australian Association of Gerontology Inc, Submission 15, 1.

Department for Constitutional Affairs, United Kingdom, above n 542, 59-60.

to make a particular decision. The capacity resource that the Committee proposes later in this chapter and the information for witnesses recommended in the previous chapter should make it clear in what circumstances referral for a medical assessment of capacity is required.

# 5.6.2 Who should conduct medical assessments of capacity?

The Committee heard there are a range of professionals who may be well placed to provide medical assessments of capacity. Many participants supported general practitioners playing a role in assessing capacity, as they may have an ongoing relationship with the principal. Other potentially suitable assessors identified by participants include neuropsychologists, geriatricians, psychiatrists, psychologists, psychogeriatricians, gerontologists, neurologists and nurse practitioners. 581

Some jurisdictions have specially trained capacity assessors. Professor Darzins told the Committee that in Ontario, Canada, assessors complete a short course and work as private practitioners on a fee-for-service basis. The assessors are not required to have a health care background. Professor Darzins did not recommend the Canadian approach, proposing instead more training for existing professionals, including those in the legal and health care sectors. Other participants in this Inquiry also suggested the use of specialised memory clinics to assess capacity. See

Resources developed to support powers of attorney legislation in some other jurisdictions provide advice about referrals to appropriate professionals for expert medical capacity assessments. 585

The Committee notes that there are a range of professionals who may be well placed to provide assessments of capacity. The most appropriate person to conduct a capacity assessment will depend on a range of factors such as any medical conditions the principal has, the type of decision and when it needs to be made. Some

David Goldberg, *Transcript of evidence*, above n 509, 4; St Kilda Legal Service Co-Op Ltd, *Submission 50*, 2; John Myers, *Submission 56*, 16; Jenny Chapman, *Transcript of evidence*, above n 576, 8; Sue Field, above n 560, 47; Mental Health Legal Centre Inc, *Submission 59*, 5; Luke Delaney, Chair, 'Guidelines for Preparation of Neuropsychological Reports for the Guardianship List of the Victorian Civil and Administrative Tribunal' Working Party and Senior Clinical Neuropsychologist, Alcohol Related Brain Injury Australian Services, Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Transcript of evidence*, Melbourne, 14 December 2009, 7.

Nikki Isaks, Senior Social Worker, Royal District Nursing Service, *Transcript of evidence*, Melbourne, 17 December 2009, 6; Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Submission 68*, 2, 3; Law Institute of Victoria, *Submission 41*, 12; Tom Worsnop, *Transcript of evidence*, above n 539, 3. See also Alzheimer's Association Queensland, *Common questions*, <a href="http://www.alzheimersonline.org/facts/faq.php">http://www.alzheimersonline.org/facts/faq.php</a>, viewed 29 June 2010; The Law Society of New South Wales, above n 527, 7; American Bar Association Commission on Law and Ageing and American Psychological Association, *Assessment of older adults with diminished capacity: A handbook for lawyers* (2005), 32-33; Attorney General's Department, New South Wales, above n 527, 12, 54-55; Kristen Pearson, *Transcript of evidence*, above n 517, 5; Aged & Community Care Victoria, *Submission 53*, 3; Stelvio Vido, Executive General Manager, Strategic and Support Services, Royal District Nursing Service, *Transcript of evidence*, Melbourne, 17 December 2009, 7.

Peteris Darzins, *Transcript of evidence*, above n 515, 6.

<sup>&</sup>lt;sup>583</sup> Ibid, 8

<sup>&</sup>lt;sup>584</sup> Aged & Community Care Victoria, Submission 53, 3.

The Law Society of New South Wales, above n 527, 7.

assessments may require a multidisciplinary approach combining the skills and expertise of different professionals. The capacity resources the Committee suggests later in this chapter should provide information about the types of professionals to whom referrals for assessments of capacity can be made.

# 5.7 What evidence is required to activate an enduring power of attorney?

As noted above, there is currently no process for formally activating an enduring power of attorney that is operable when the principal has impaired decision-making capacity.

Inquiry participants raised concerns that the totally unregulated activation process is susceptible to abuse. For example, the Federation of Community Legal Centres was 'concerned that in some instances, attorneys exercise their powers under an enduring power of attorney despite the fact that the donor has capacity. This is particularly concerning with respect to mentally ill donors who sometimes lose capacity for short periods and then regain it'. 586

Some organisations have implemented processes that require representatives to provide proof that a principal has impaired decision-making capacity. Credit Union Australia, for instance, informed the Committee that it requires a letter from the principal's doctor stating that the principal has impaired decision-making capacity before it will accept any enduring power of attorney document. Earlier in this chapter, the Committee considered and ultimately rejected a requirement for medical evidence of impaired decision-making capacity in all cases.

The Committee received two other suggestions for evidencing the activation of an enduring power of attorney. Firstly, two submissions argued that where an enduring power of attorney is stated to come into effect when a principal has impaired decision-making capacity, the evidence needed to prove this should be set out in the document creating the power. 588

Secondly, Mr John Billings, Deputy President of the Guardianship List at the Victorian Civil and Administrative Tribunal, suggested that representatives could be required to record:

- the date the enduring power was activated
- the date from which the principal had impaired decision-making capacity
- the medical or other evidence on which the representative based his or her belief that the donor had impaired decision-making capacity. 589

-

Federation of Community Legal Centres (Victoria) Inc, Submission 47, 3. See also Paula Chatfield, Transcript of evidence, above n 509, 3.

Credit Union Australia Ltd, Submission 18, 3. See also Hunt & Hunt, Submission 24, 2.

Seniors Rights Victoria, Submission 38, 29. See also Financial Planning Association of Australia Ltd, Submission 49, 2.

John Billings, Submission 37, above n 562, 6.

The Committee believes it would be too onerous to require all principals to specify in the enduring power of attorney document the evidence needed to activate an enduring power upon impaired decision-making capacity. However, it notes that principals can currently choose to do this and believes this option should remain.

The Committee agrees that representatives should be encouraged to record when they activate an enduring power of attorney and on what evidence they based that assessment. This will provide more certainty in relation to these decisions. In chapter seven the Committee recommends a legislative requirement that representatives formally notify the activation of an enduring power, including to any personal monitor appointed by the principal and the body responsible for registering powers of attorney. The Committee does not believe this notice needs to contain any medical certification or other formal evidence. However, encouraging representatives to make a note of the evidence on which their assessment is based will help ensure that this is documented and provide valuable evidence should the decision to activate the document be challenged.

In the next chapter the Committee recommends resources for representatives to support them in their role. These resources should encourage representatives to record information about the activation of enduring powers of attorney.

# 5.8 Providing guidance and support to those assessing capacity

Many participants in this Inquiry highlighted the complexity of capacity assessments and called for more guidance and support for those making these assessments.

### 5.8.1 Resources to assist with capacity assessments

There was strong support among Inquiry participants for the development of guidelines or other resources to assist people to make capacity assessments both at the time of creation and activation of power of attorney documents. For example, Mr Luke Delaney, Senior Clinical Neuropsychologist with Alcohol Related Brain Injury Australian Services, who gave evidence on behalf of the Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychology Society, told the Committee, 'I think the idea of guidelines is a good way of dealing with this issue, because this is legislation that is going to affect everyone, and there are going to be people of varying levels of knowledge and understanding actually acting under the legislation ...' Southern Health suggested that guidelines, in conjunction with training would allow more people to make capacity assessments without referral or delay.

Southern Health, Submission 30, 2.

Luke Delaney, *Transcript of evidence*, above n 580, 7. See also Victoria Legal Aid, *Submission 42*, 3-4; Law Institute of Victoria, *Submission 41*, 8; Peteris Darzins, *Transcript of evidence*, above n 515, 8; Royal District Nursing Service, *Submission 25*, 3; Southern Health, *Submission 30*, 2; Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Submission 68*, 5; Hume Riverina Community Legal Service, *Submission 36*, 3; Elizabeth Mullaly, *Transcript of evidence*, above n 527, 7; Lillian Jeter, *Transcript of evidence*, above n 569, 3.

In particular, participants drew the Committee's attention to resources that have been developed in other jurisdictions to guide those involved in assessing capacity.

### New South Wales Capacity toolkit

The Attorney General's Department of New South Wales published a *Capacity toolkit* in 2008 following an extensive consultation process. The toolkit provides a wide range of information and guidance about capacity and capacity assessments. It is aimed at a broad audience, including those concerned about the ability of a person to make a decision for himself or herself and those who may need to assess a person's capacity. 593

The toolkit is written in a simple style and contains a breadth of information, including:

- What is capacity?
- Principles for assessing capacity.
- When should capacity be assessed?
- Who can assess capacity?
- Assessing capacity in relation to personal life, health and money and property.
- Supported decision making.

Alzheimer's Australia Victoria highly commended the New South Wales toolkit and suggested that a similar resource could be developed in Victoria. A recent South Australian review also praised the toolkit and recommended that it be adapted for use throughout Australia. Septimental S

### Law Society of New South Wales – Guide for solicitors

The Law Society of New South Wales has produced *When a client's capacity is in doubt – A practical guide for solicitors*. This guide outlines a range of information including what is capacity, indicators of lack of capacity and when to seek the appointment of a substitute decision maker. It provides specific guidance on the lawyer's role in assessing capacity and includes information about communicating with clients, keeping records of capacity assessments and when and how to make referrals for medical assessments of capacity.

Attorney General's Department, New South Wales, above n 527, 6-7; Attorney General's Department, New South Wales, above n 529.

Attorney General's Department, New South Wales, above n 527, 11-12.

Margaret Brown, Adjunct Research Fellow, Hawke Research Institute, University of South Australia, Alzheimer's Australia Vic, *Transcript of evidence*, Melbourne, 22 October 2009, 3-4; Alzheimer's Australia Vic, *Submission 32*, 5.

Advance Directives Review Committee, *Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies* (2008), 42-43.

While the *Capacity toolkit* developed by the NSW Attorney General's Department has a broader audience, the Law Society's guide specifically focuses on issues relevant to legal practitioners. <sup>596</sup> The Law Society advised that it developed the guide because many of its members sought guidance about assessing capacity. The Law Society informed the Committee that the guide has been well received by lawyers and is widely used.

Several Inquiry participants from the legal sector praised this publication.<sup>597</sup>

### United Kingdom Mental Capacity Act 2005 Code of practice

The UK *Mental Capacity Act 2005 Code of practice* was issued in 2007 and aims to provide guidance about how to apply the Act. The Office of the Public Guardian, which developed the code, told the Committee, 'The Code was developed as a single document to keep all appropriate guidance and information in one place to avoid any confusion.' <sup>598</sup>

The code explains how the Act operates on a day-to-day basis and offers examples of best practice to carers and practitioners. It includes information about what is capacity, how to assess capacity, principles to guide capacity assessments and supported decision making.

The Office of the Public Guardian advised that over 45 400 copies of the code have been sold and that feedback received about the code had been 'broadly positive'. <sup>599</sup> The Office also indicated that the code may be reviewed in the future.

Carers Victoria was strongly supportive of the UK code and advocated for such a code to be introduced in Victoria. <sup>600</sup>

#### Other resources

The Committee heard that a number of other professional bodies have developed resources on capacity assessment for their members.

The Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society (APS) provided the Committee with the *Guidelines* for preparation of neuropsychological reports for the Guardianship List of the Victorian Civil and Administrative Tribunal (VCAT) which were developed by the APS. These guidelines provide information on the preparation of neuropsychological reports for VCAT, but do not address how to conduct an assessment of capacity. 601

Letter from Mary Macken, President, The Law Society of New South Wales, to Chair, Victorian Parliament Law Reform Committee, 25 May 2010.

Law Institute of Victoria, *Submission 41*, 12; Hume Riverina Community Legal Service, *Submission 36*, 3; Seniors Rights Victoria, *Submission 38*, 25.

Letter from Martin John, Public Guardian and Chief Executive, Office of the Public Guardian, United Kingdom, to Chair, Victorian Parliament Law Reform Committee, 27 May 2010, att, 1.

Ibid, att, 2.
 Carers Victoria, Submission 65, 13; Penny Paul, Coordinator, Carer Consultations, Carers Victoria,
 Transcript of evidence, Melbourne, 17 December 2009, 5-6.

The Australian Psychological Society Ltd, Guidelines for the preparation of neuropsychological reports for the Guardianship List of the Victorian Civil and Administrative Tribunal (VCAT) (2009), 4, 11. See also

Dr Pearson of the Australian & New Zealand Society for Geriatric Medicine told the Committee there were a variety of relevant resources for general practitioners. She drew the Committee's attention to the Royal Australian College of General Practitioners' *Medical care of older persons in residential aged care facilities* which sets out information about assessing capacity. 603

### Resources to assist with capacity assessments – The Committee's view

The Committee acknowledges that assessing capacity can be extremely challenging. The Committee agrees there would be value in a national approach to this difficult issue, including the development of guidance materials that can be used nationwide. However, the development of resources at a national level is likely to be a protracted process, and such resources will only be of limited assistance with the application of state-based legislation.

The Committee therefore recommends that a resource based on the Attorney General's Department of New South Wales's *Capacity toolkit* should be developed to support the Victorian legislation. Such a resource would be extremely valuable to all people involved in capacity assessments. The Committee notes that this resource will have utility beyond the specific powers of attorney under review in this Inquiry, being equally relevant to other capacity-based legislation.

The new resource should contain a range of information relevant to those making capacity assessments, as well as principals and their families, such as what is capacity, indicators of lack of capacity, the process for assessing capacity, key principles for assessing capacity, when an expert assessment of capacity is required and supported decision making. The resource should be developed in consultation with relevant professionals, including lawyers and health care professionals.

The proposed capacity resource would supplement the resources to assist witnesses recommended in the previous chapter, providing more detailed information and guidance about how witnesses and others should approach capacity assessments.

The Committee notes that different professions involved in capacity assessments such as lawyers and medical practitioners may also develop their own guidelines and resources. These will supplement the general resource and provide specific support and assistance to members of particular professions.

### **Recommendation 36: Capacity resource**

The Committee recommends the Victorian Government, in consultation with a wide range of stakeholders, including representatives from the legal and health care sectors, develop a comprehensive resource about capacity and capacity assessments, based on the New South Wales *Capacity toolkit*.

Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, Submission 68, 4.

Kristen Pearson, Transcript of evidence, above n 517, 8.

<sup>603</sup> The Royal Australian College of General Practitioners, Medical care of older persons in residential aged care facilities (4th edition, 2006), 12.

### 5.8.2 Educating capacity assessors

Some participants in this Inquiry suggested that people who assess capacity should receive specialised training. Professor Darzins stated:

We need to educate legal professionals, financial professionals and health-care professionals, and each of those needs different types of input. They need to know that capacity is presumed present, that there is rebuttal if there is sufficient evidence, and what that evidence could be. They need to know how to find the evidence using a capacity assessment process, and they need to know that there is a need for proper documentation. I think that is all doable for those different audiences. <sup>604</sup>

Mr Matthew Hughes, Principal Neuropsychologist at the Alfred Hospital and Acting Chair of the Victorian Section of the College of Clinical Neuropsychologists of the APS, saw a role for his profession in educating capacity assessors:

It is useful to probably consider a neuropsychologist to be involved in educating legal and health-care professionals, because this is certainly an area of expertise for us. We are often making these kinds of assessments, so we can certainly help streamline and improve the process for others. <sup>605</sup>

However, not all participants agreed there was a need for training about capacity. The Mental Health Legal Centre submitted that members of the health and legal professions are constantly required to assess capacity and do not require any additional education in this area.

Given the complexity of capacity assessments, the Committee agrees that there may be benefit in providing education to those conducting capacity assessments in relation to powers of attorney. This would help ensure a more consistent approach to assessing capacity. This education and training could be rolled out in conjunction with the new capacity resource recommended in the previous section and the training for authorised witnesses the Committee recommended in the previous chapter. This education and training could also form part of broader education about powers of attorney for the legal, health and community sectors which the Committee discusses in chapter nine.

### 5.8.3 Tools for assessing capacity

The Committee heard that assessing capacity can be extremely difficult, even for experienced professionals.

Some participants in this Inquiry called for the development of new tools to help assess capacity to ensure that professionals apply a consistent approach to capacity assessments. <sup>607</sup>

Peteris Darzins, *Transcript of evidence*, above n 515, 8. See also Southern Health, *Submission 30*, 2; Jeni Lee, *Submission 57*, 2, 4; Margaret Brown, *Transcript of evidence*, above n 594, 3-4.

Matthew Hughes, Principal Neuropsychologist, The Alfred Hospital, and Acting Chair, Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Transcript of evidence*, Melbourne, 14 December 2009, 3. See also Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Submission 68*, 5.

Mental Health Legal Centre Inc, Submission 59, 5.

Jeni Lee, Submission 57, 2, 4; Aged & Community Care Victoria, Submission 53, 3; Alzheimer's Australia Vic, Submission 32, 4-5.

Currently there are a variety of different tools used for assessing capacity, with different professions preferring different tools. The two tools most commonly referred to by participants in this Inquiry were the Six step capacity assessment process and the Mini mental state examination.

### Six step capacity assessment process

Several participants in this Inquiry strongly recommended the six step process to the Committee. <sup>608</sup> Figure 2 outlines the six step capacity assessment process.

Step 1 Step 2 Step 3 Step 4 Step 5 Step 6 Pre-assessment Capacity assessment Post tasks assessment tasks Educate Gather Assess about the information capacity Ensure Engage Act on the issues, about issues, with valid person in outcome choices and available respect to trigger the capacity of the reasonably choices and the is assessment assessment foreseeable consequences specific present process consequences of choices issues of choices

Figure 2: The six step capacity assessment process<sup>609</sup>

One of the creators of the six step process, Professor Darzins, told the Committee:

I suggest that the six-step capacity assessment process be followed. You could make a power-of-attorney specific tool, but the trouble is that when people are dealing with capacity issues around the time of the power of attorney, they are often dealing with other things as well — health care, finances and personal care. Why would you make one approach for powers of attorney which does not link to all the others? That is why I would recommend that you made a more global approach that is part of a suite of assessments rather than being very specific around power of attorneys. <sup>610</sup>

Some of the resources to assist with capacity assessments mentioned earlier in this chapter draw on the six step capacity assessment process.<sup>611</sup>

#### Mini mental state examination

Several participants in this Inquiry indicated that they used the Mini mental state examination to assist with capacity assessments. 612 The Mini mental is a cognitive

Kristen Pearson, Transcript of evidence, above n 517, 4; Southern Health, Submission 30, 2; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2; Seniors Rights Victoria, Submission 38, 25.

Peteris Darzins, Assessment of decision-making capacity: Key concept summary, above n 524, 2-3.

Peteris Darzins, *Transcript of evidence*, above n 515, 8.

For example, The Royal Australian College of General Practitioners, above n 603, 12.

screening tool commonly used to quickly assess a person's level of overall cognitive functioning using a brief standard questionnaire. This includes an assessment of a person's orientation, attention, registration and recall, language and the ability to follow simple oral and written instructions on a 30 point scale. <sup>613</sup>

The Mini mental has been criticised as a capacity assessment tool for focusing on cognitive function and not being able to measure a person's capacity to make a specific decision. In addition, the assessment is language based and may be influenced by a person's age, education, culture and language. However, the Mini mental may provide an initial overall assessment of cognitive functioning and help the assessor to determine if more testing is required.

Mr David Stokes, Neuropsychologist and Senior Manager Professional Practice at the APS, supported the limited use of the Mini mental:

As long as they accept that it is a screen only, and there are lots of false positives that come out of those sorts of processes and that people who look okay there in actual fact do have difficulties; and that people who do badly on those can sometimes be not too bad. It is a blunt instrument ... You never rely on one particular task, so I would be happy that the mini mental state had been done because it is not a bad starting point, but you would also want to make sure that the set of guideline questions that have been set down, as we do in other settings, had also been administered ... 617

A modified Mini mental has also been developed to assess the overall cognitive functioning of older people, in particular, for dementia. It can be administered in 12 minutes by a qualified health professional using a standard set of questions. <sup>618</sup>

#### Other tools

There are a number of other tools used by professionals to assess capacity including the MacArthur Competence Assessment Tool and the Aid to Capacity Evaluation. These tools are used to evaluate a person's capacity to make a specific medical

Nikki Isaks, *Transcript of evidence*, above n 581, 6-7; Lucy Cordone, General Counsel, St Vincent's Hospital, *Transcript of evidence*, Melbourne, 14 December 2009, 7.

MF Folstein, SE Folstein and PR McHugh, "'Mini-mental state": A practical method for grading the cognitive state of patients for the clinician' (1975) 12(3) *Journal of Psychiatric Research* 189; American Bar Association Commission on Law and Ageing and American Psychological Association, above n 581, 21-22, 66.

American Bar Association Commission on Law and Ageing and American Psychological Association, above n 581, 21-22, 66; Peteris Darzins, D William Molloy and David Strang (eds), above n 527, 8; Terry Carney, 'Judging the competence of older people: An alternative?' (1995) 15 Ageing & Society 515, 520; Attorney General's Department, New South Wales, above n 529, 15.

American Bar Association Commission on Law and Ageing and American Psychological Association, above n 581, 21-22, 66; Peteris Darzins, D William Molloy and David Strang (eds), above n 527, 8; Australian Association of Gerontology Inc, Submission 15, 1; Jaklina Michael, Cultural Liaison Coordinator, Royal District Nursing Service, Transcript of evidence, Melbourne, 30 March 2010, 15-16.

American Bar Association Commission on Law and Ageing and American Psychological Association, above n 581, 21-22, 66.

David Stokes, Senior Manager Professional Practice, Australian Psychological Society and Neuropsychologist, Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society, *Transcript of evidence*, Melbourne, 14 December 2009, 5.

Jaklina Michael, *Transcript of evidence*, above n 615, 15-16; Dementia Collaborative Research Centres, *Cognition assessment measures*, <a href="http://www.dementia-assessment.com.au/cognitive/index.html">http://www.dementia-assessment.com.au/cognitive/index.html</a>>, viewed 29 June 2010.

treatment decision.<sup>619</sup> Participants in this Inquiry did not comment on the usefulness of these tools.

## Tools for assessing capacity – The Committee's view

The Committee notes there are a range of tools available to assist professionals to make capacity assessments. It notes that a standard capacity assessment tool may be difficult to apply in a variety of circumstances and may discourage the development of new approaches to capacity assessment. The Committee suggests that the resources recommended earlier in this chapter should provide information about possible tools available to assist with capacity assessments.

# 5.9 A gap in the law: people who cannot make enduring powers of attorney and cannot obtain a guardianship order

All submissions to the Inquiry, with two exceptions, were based on the fundamental premise that a principal requires capacity to make a valid power of attorney document. The submissions of both Palliative Care Victoria and Carers Victoria suggested that, in certain circumstances, people with impaired decision-making capacity should be able to make enduring powers of attorney. 621

Ms Penny Paul, Carers Victoria's Coordinator, Carer Consultations, stated:

Carers Victoria believes that individuals with a capacity disability are deemed to lack the capacity to appoint an enduring power of attorney and are conceptualised in law as perpetual children. The failure of current laws to provide a mechanism whereby individuals with capacity disabilities can appoint substitute decision-makers has left substitute decision making to family members, service providers and friends who are undertaking this task in an unregulated, underdeveloped and poorly conceptualised manner.

The problem with the Victorian Guardianship and Administration Act 1986 is that it appoints a substitute decision-maker, an inherently restrictive appointment, and so guardianship orders are rarely made. 622

Ms Marianne Dalton, the parent carer of an adult with an intellectual disability who gave evidence on behalf of Carers Victoria, told the Committee that VCAT initially appointed her as her daughter's guardian, but the guardianship order was subsequently removed. She stated:

you have to go back after three years, and in that time something happened. There was some change, and the next time we went to the three-year review the presiding officer said, and this is a quote, 'It is now important for people to be independent',

Thomas Grisso and Paul S Appelbaum, above n 527; Edward Etchells et al, 'Assessment of patient capacity to treatment' (1999) 14(1) *Journal of General Internal Medicine* 27.

Attorney General's Department, New South Wales, above n 529, 15.

Palliative Care Victoria, Submission 70, 1.

Penny Paul, *Transcript of evidence*, above n 600, 4-5. See also Carers Victoria, *Submission 65*, 9, 12.

and he said that a guardianship order was unnecessary, 'because in a crisis you can get temporary guardianship in 24 hours'. 623

Ms Dalton commented 'for the last 11 years of my father's life I acted with his enduring power of attorney ... and I can tell you no-one sent him to aged care or anywhere else without contacting me. Why cannot my daughter have those same rights as my father did?' 624

Carers Victoria argued that individuals with an intellectual disability who might be otherwise viewed as lacking capacity should be able to make enduring powers of attorney. It suggested that such appointments could be made at a specially convened meeting attended by the individual, their family, primary carer, service providers, proposed representatives, an advocate and representative of the Guardianship List of VCAT. Carers Victoria drew the Committee's attention to the system in Alberta, Canada, where a court can make an order appointing a co-decision maker where an adult's ability to make decisions is significantly impaired but he or she can make decisions with good support.

The Office of the Public Advocate did not agree with the proposal put forward by Carers Victoria. Dr John Chesterman, Manager of Policy and Education at the Office of the Public Advocate, wrote:

Our concern relates to the potential problems that might arise if a person with an intellectual disability is able to assign rights through an enduring power of attorney without necessarily understanding the full extent or nature of the power being assigned. At present, the general rule is that a person can sign an enduring power of attorney if they understand the document they are signing ... Expanding this out so that one could enter an enduring power of attorney without showing that level of understanding could lead to problems of abuse of the power. The suggestion by Carers Victoria is that a number of safeguards could be put in place, including having an organisation like ours play a role in a special meeting where the enduring power of attorney would be executed. Our view is that when safeguards like this are in place (which would be necessary) the meeting itself would not differ dramatically from a guardianship/administration hearing.

The Committee recognises there is a gap in the law where VCAT determines that a person is not in need of a guardian but a person does not have capacity to appoint a representative. The Committee's view is that the Victorian Government should give further consideration to the suggestion by Carers Victoria that the ability to make powers of attorney should be expanded to include people with impaired decision-making capacity, who would not meet a traditional test of capacity. The Committee notes that the Victorian Law Reform Commission's review of the Guardianship and Administration Act will also consider this issue.

Marianne Dalton, Parent carer of an adult with a capacity disability, Carers Victoria, *Transcript of evidence*, Melbourne, 17 December 2009, 3-4.

<sup>&</sup>lt;sup>624</sup> Ibid, 4.

<sup>625</sup> Carers Victoria, Submission 65, 9, 12.

<sup>626</sup> Adult Guardianship and Trusteeship Act, SA 2008, c A-4.2; Penny Paul, Transcript of evidence, above n 600, 5-6; Letter from Penny Paul, Coordinator, Carer Consultations, Carers Victoria, to Executive Officer, Victorian Parliament Law Reform Committee, 18 January 2010, 1 and attachment: Government of Alberta, Co-decision-making: Adult Guardianship and Trusteeship Act.

<sup>627</sup> Email from John Chesterman, Manager, Policy and Education, Office of the Public Advocate, to Executive Officer, Victorian Parliament Law Reform Committee, 31 December 2009.

# **Chapter 6: The role of representatives**

This chapter considers the role, powers and duties of people acting as representatives for others under powers of attorney. It considers the skills and attributes that representatives require and identifies strategies for ensuring that appropriate representatives are appointed. This chapter also highlights the need for more clarity about the powers and duties that representatives have. Finally, the Committee explores ways for providing information, education and support to representatives to assist them to carry out their often highly challenging role.

# 6.1 Recognising the contribution of representatives

When a person agrees to undertake the role of representative, he or she assumes a large and sometimes uncertain commitment. The vast majority of representatives are volunteers who accept the role either out of friendship or a sense of duty. It can be a demanding and time-consuming role. Mr Adrian Cohen, an individual who is a representative for his elderly parents, told the Committee, 'I am totally responsible for both my parents' wellbeing. It is a 24/7 operation, and I have been doing it for about six years.' 628

Representatives offer principals peace of mind: principals know there will be someone they trust acting for them when they are no longer able to make decisions for themselves. Over and above this, the role that representatives take on also provides significant benefits to the State in terms of reduced reliance on formal mechanisms — such as courts, tribunals and public guardians — for providing decision-making support to people with impaired decision-making capacity.

In considering the role, powers and duties of representatives in this chapter and the next, the Committee acknowledges there is a significant tension between providing checks on representatives' appointment and the performance of their role, and ensuring that representatives are not discouraged from undertaking what must sometimes seem an onerous and thankless task. In this report the Committee endeavours to strike a balance between valuing the contribution representatives make, and providing mechanisms to ensure that appropriate people are appointed as representatives and that they are accountable for their actions.

# 6.2 Appointing appropriate representatives

Evidence to the Committee emphasised that the selection of a suitable person to act as a representative is a key factor in ensuring that a power of attorney is used appropriately. In particular, participants in the Inquiry highlighted the need for principals to appoint representatives they trust and who they feel know them well. 630

Adrian Cohen, *Transcript of evidence*, Melbourne, 30 March 2010, 2.

Letter from Peter Maloney, Chief Executive Officer, Public Trustee, Tasmania, to Chair, Victorian Parliament Law Reform Committee, 3 November 2009, 2.

Office of the Public Advocate, Submission 9, 16.

Very little is known about the types of people who are currently acting as representatives. Research conducted in Queensland suggests that principals are most likely to appoint their children as their representatives. Other common appointments include parents, siblings, grandchildren, friends, legal practitioners, accountants and trustee companies.

The *Instruments Act 1958* (Vic) allows principals to appoint the head of a religious chapter or order as a representative under an enduring power of attorney (financial). The Committee did not receive any evidence about the extent to which this occurs.

### 6.2.1 What skills do representatives need?

The representative's role may be highly complex and demanding. Representatives may have to make complicated legal and financial arrangements on the principal's behalf, such as selling property or organising government entitlements. Mr Cohen told the Committee he had used his parents' enduring power of attorney (financial) for a diverse range of functions including:

banking; employing professionals, doctors, lawyers, case managers; dealing with share registries for dividends, takeovers; paying bills like gas, electricity, rates, insurance policies, owners corporations meetings – doing levies for them and proxies; and sorting the nursing home accounts, the health insurance, the chemist bills, lease statements, disclosure statements on leases and club memberships. 633

A Queensland study found that managing assets on behalf of older people under an enduring power of attorney was extremely complex, with many representatives finding the role 'onerous and beyond their capacities'. While research in this area has predominantly focused on the pressures faced by representatives with financial and legal decision-making responsibilities, many lifestyle decisions are equally complex and may need to be made in a crisis situation, such as a decision to move the principal into an aged care facility.

Given the demands inherent in the representative's role, participants stressed the importance of representatives having the right skills. Ms Sue Field, NSW Trustee and Guardian Fellow in Elder Law at the University of Western Sydney, identified three essential traits for representatives: 'financial acumen, integrity and availability'. 635

Ms Angela Burton, General Manager of Personal Financial Solutions at State Trustees Limited, gave an example of an inappropriate appointment: 'There was a father who had nominated his daughter. She had three young children. She was busy.

-

<sup>631</sup> Lindy Willmott and Liza Windle, 'Witnessing EPAs: Empirical research' (2007) 27 Queensland Lawyer 238, 241.

<sup>632</sup> Instruments Act 1958 (Vic) s 122.

Adrian Cohen, *Transcript of evidence*, above n 628, 2.

Cheryl Tilse, Deborah Setterlund et al, 'Minding the money: A growing responsibility for informal carers' (2005) 25 *Ageing & Society* 215, 225.

<sup>&</sup>lt;sup>635</sup> Sue Field, NSW Trustee and Guardian Fellow in Elder Law, University of Western Sydney, *Submission 61*, 3.

She just did not have the time to look after his financial affairs.' In that case the father revoked the daughter's appointment and appointed State Trustees instead.

Mr David Davis, a solicitor who gave evidence on behalf of the Law Institute of Victoria, told the Committee that different functions might require representatives with different skills: 'I see that analogous to trying to get horses for courses: who is the best person for the job in relation to this particular set of assets, or who might be best for that particular category of tasks required to look after my personal affairs if I lose capacity?' 637

Participants in the Inquiry identified a variety of individuals who may, because of their profession, be particularly suited to the role of representative. These include solicitors, accountants, statutory trustee companies or the public trustee. The Trustee Corporations Association of Australia's submission suggested that appointing a professional trustee is more critical when there are large sums of money involved or there are family tensions. The submission of the public trustee is more critical when there are large sums of money involved or there are family tensions.

### 6.2.2 Helping principals appoint appropriate representatives

The Committee heard evidence that principals are generally not well informed about who would be suitable to appoint as a representative. In particular, there was concern that some principals feel obliged to appoint family members as their representatives, and do not give appropriate consideration to other alternatives. <sup>640</sup> Dr John Chesterman, Manager of Policy and Education at the Office of the Public Advocate, told the Committee that the selection of the right representative is critical:

When people come and ask us in public presentations ... and they say, 'What is your view? Should we sign an enduring power of attorney?' Our response always is, 'Yes, but only if you can really trust the person you are appointing. You should not appoint people who you think you should appoint, like your eldest child just because they are your eldest child. If you have someone you can absolutely trust, then you should sign an enduring power of attorney. If you have not, then you ought not to'. 641

The *Take control* kit produced by the Office of the Public Advocate and Victoria Legal Aid provides a checklist of factors to consider when appointing a representative. The kit encourages principals to:

- appoint someone you can trust
- appoint someone who can act according to your interests and not their own

Angela Burton, General Manager, Personal Financial Solutions, State Trustees Limited, Transcript of evidence, Melbourne, 22 October 2009, 7.

David Davis, Elder Law Committee, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 8.

Seniors Rights Victoria, *Submission 38*, 20-21.

Trustee Corporations Association of Australia, *Submission* 27, 3.

Sue Field, Submission 61, above n 635, 3; Lionel H Parrott, Submission 67, 1.

John Chesterman, Manager, Policy and Education, Office of the Public Advocate, *Transcript of evidence*, Melbourne, 22 October 2009, 7-8.

- appoint someone who is likely to be able to take on the role when it is needed
- appoint someone who is happy to take on the role
- appoint someone who will listen to what you want and respect your preferences even after you have lost legal capacity
- appoint someone who is able to take on the role and can make appropriate and competent decisions. 642

Seniors Rights Victoria suggested that legislation should provide guidelines about who can be appointed as a representative, including that a representative

- may be a relation or close friend
- must be capable of considering the consequences of decisions and have the ability to understand and reason
- may be a solicitor, accountant, statutory trustee company or the public trustee.<sup>643</sup>

The decision about who to appoint as a representative is an important one which can have significant consequences for the principal. The Committee believes that principals and those advising them, such as lawyers, should be provided with a range of information about the sorts of skills and personal qualities that representatives should have. This will help ensure that suitable people are appointed to the position.

The Committee's view is that the appropriate place for information about the desirable skills and attributes of representatives is in educational material for principals and their advisers rather than in legislation. The matters set out in *Take control* are a good starting point for educating people contemplating making a power of attorney about suitable appointments.

#### Recommendation 37: Providing guidance about appropriate representatives

The Committee recommends the Victorian Government provide simple, easy-tounderstand information and educational materials for principals, and persons likely to advise principals, which includes information about the sorts of skills representatives require and guidance about appropriate appointments.

Victoria Legal Aid and Office of the Public Advocate, *Take control: A kit for making powers of attorney and guardianship* 10th edition (2007), 16-17.

Seniors Rights Victoria, Submission 38, 20-21.

# 6.2.3 Should there be restrictions on who can be appointed as a representative?

While there is no regulation of who may be appointed under a general (non-enduring) power of attorney, legislation currently imposes a number of restrictions on who can be appointed as a representative under an enduring power of attorney (financial) and an enduring power of attorney (guardianship). Representatives under these documents are required to be at least 18 years of age. People involved in a professional or administrative capacity in the principal's care and treatment or in the provision of accommodation to the principal are excluded from appointment under an enduring power of attorney (guardianship). In addition, people who are insolvent cannot be appointed under an enduring power of attorney (financial).

A number of participants in the Inquiry suggested that persons not of good character should be excluded from being appointed as a representative. Ms Sally Costar, Manager of the Patient and Family Services at Caulfield Hospital, who gave evidence on behalf of Alfred Health, provided two examples of appointments that placed principals at risk:

At Alfred Health we have been in the position where we have had an attorney who had a prior conviction for fraud take over the estate of a person who lacked competency, and indeed we believe the attorney was involved in some fraud with the estate. It became extremely difficult for the elderly person involved, who really had no way through. It was really only through the influence of her friends contacting the hospital that we even realised that things were amiss. I think there was no way of this elderly person — the donor — being aware of the attorney's background ...

I have also been in the position of taking a case to VCAT for a guardianship application where it became evident that the enduring power of attorney was in existence and it was being held by a person who was in jail for a very serious offence and was going to be in jail for a prolonged period. That person was administering the estate over the internet. It seemed to me that there were no checks whatsoever in what was occurring to that estate ... 647

Several other participants also supported excluding people with a criminal conviction involving dishonesty or fraud, or serving current prison sentences, from being appointed as representatives. Seniors Rights Victoria recommended that a person previously removed as a representative by the Victorian Civil and Administrative Tribunal (VCAT) or another tribunal in Australia for misusing his or her power should also be excluded from acting as a representative. The Royal District Nursing Service suggested probity or police checks prior to appointment as a

<sup>644</sup> Instruments Act 1958 (Vic) s 119(4); Guardianship and Administration Act 1986 (Vic) s 35A(3).

<sup>&</sup>lt;sup>645</sup> Guardianship and Administration Act 1986 (Vic) s 35A(4)-(5).

<sup>646</sup> Instruments Act 1958 (Vic) s 121.

Sally Costar, Manager, Patient and Family Services, Caulfield Hospital, Alfred Health, *Transcript of evidence*, Melbourne, 14 December 2009, 2. See also Stephen Taffe, Legal Counsel, Alfred Health, *Transcript of evidence*, Melbourne, 14 December 2009, 2; Alfred Health, *Submission 66*, 3.

<sup>&</sup>lt;sup>648</sup> Lionel H Parrott, Submission 67, 1; Seniors Rights Victoria, Submission 38, 21; The Victorian Bar, Submission 40, 4.

Seniors Rights Victoria, Submission 38, 21.

representative, while the Victorian Bar proposed placing the onus on representatives by making it an offence for an excluded representative to accept an appointment. <sup>650</sup>

Some Inquiry participants acknowledged that excluding certain classes of people from being appointed as representatives may preclude people who would perform the role effectively. Ms Costar shared an example with the Committee:

I was faced with exactly this kind of dilemma with an application to VCAT for a person to become an administrator. At the VCAT hearing the proposed attorney identified himself as an undischarged bankrupt, and the VCAT member said, 'I cannot proceed and appoint you'. Even though everyone in the room felt that this person would have in fact done a very good job, the VCAT member could not proceed. My feeling is that if you are really going to do something about the exploitation that we know is occurring, unfortunately we probably do have to exclude some people who could in fact do a very good job. There are some instances where barriers should be put up, and my view is that fraud, perhaps serious crime and bankruptcy should be put up as hurdles. <sup>651</sup>

Her colleague, Alfred Health's Legal Counsel, Mr Stephan Taffe, suggested there could be a process for people deemed to be ineligible to apply to a court or VCAT for permission to act as a representative. 652

Seniors Rights Victoria submitted that the eligibility requirements for representatives should be consistent for all types of powers of attorney, both enduring and non-enduring. While acknowledging that the current arrangements recognise the greater powers that representatives have under enduring documents, Seniors Rights Victoria argued that consistency would provide clarity and help protect against elder abuse.

The Committee believes that excluding inappropriate persons from acting as representatives is an important protection against abuse. The Committee did not receive any evidence to suggest that the current restrictions on representatives are not effective and therefore is of the view that these should be retained.

In addition, the Committee considers there should be some further restrictions on who can be appointed as a representative under an enduring power of attorney (financial). While recognising the benefits of consistency, the Committee is of the view that the increased risk of abuse of financial powers when a principal has impaired decision-making capacity means that these powers should be subject to extra protections. It recommends that people with a criminal conviction of a type involving dishonesty should be prohibited from acting as a representative under an enduring power of attorney (financial). This restriction is warranted because the nature of the offence directly relates to the type of powers with which a representative is entrusted.

-

Royal District Nursing Service, Submission 25, 5; The Victorian Bar, Submission 40, 4; Robert Shepherd, Barrister, The Victorian Bar, Transcript of evidence, Melbourne, 22 October 2009, 4.

Sally Costar, Transcript of evidence, above n 647, 5.

<sup>652</sup> Stephen Taffe, *Transcript of evidence*, above n 647, 6.

<sup>653</sup> Seniors Rights Victoria, Submission 38, 20.

The Committee believes that legislation should not be overly restrictive and should, as far as possible, respect principals' rights to appoint a representative of his or her choice. Therefore, while noting the concerns raised by some participants, it does not recommend any further restrictions on who can be appointed as a representative. In addition, in order to ensure that appropriate representatives are not automatically excluded, the legislation should set out a process by which a principal can apply to VCAT for permission to appoint someone who is otherwise excluded. This might be appropriate, for example, when a conviction was relatively minor and occurred a long time in the past. 654

In the Committee's opinion, requiring potential representatives to undergo a police check prior to their appointment would introduce an unnecessary bureaucratic hurdle to the process of creating a power of attorney, potentially discouraging use of these documents. Instead the Committee recommends that a person should be required to declare that he or she is eligible to act as a representative at the time he or she accepts the appointment, as is currently required in Queensland. This builds on the Committee's recommendation in chapter four that all representatives should be required to accept their appointment.

The legislation should provide that it is an offence for a person who is ineligible to accept an appointment as a representative.

#### Recommendation 38: Excluding unsuitable representatives

The Committee recommends the Powers of Attorney Act provide:

- a) a person is not eligible to be appointed as a representative under an enduring power of attorney (financial) if he or she has previously been convicted of an offence involving dishonesty
- b) a principal is entitled to apply to VCAT for approval to appoint as a representative under an enduring power of attorney (financial) a person who has previously been convicted of an offence involving dishonesty
- c) when accepting an appointment as a representative, a person must declare that he or she is eligible to be appointed as a representative
- d) a person who accepts an appointment as a representative when he or she is not eligible, is guilty of an offence.

Department of Justice and Attorney-General, Queensland Government, Enduring power of attorney: Long form Form 2, Queensland Powers of Attorney Act 1998 (Section 44(1)) (2002), 22-23. See also Seniors Rights Victoria, Submission 38, 21.

143

Victoria does not currently have spent conviction legislation. Such legislation allows some old criminal convictions to be disregarded. The Committee notes that the Standing Committee of Attorneys-General has developed a draft model spent conviction law, see Standing Committee of Attorneys-General, *Current projects and achievements: Spent convictions*, <a href="http://www.scag.gov.au/lawlink/SCAG/Il\_scag.nsf/pages/scag\_achievements#Spent%20convictions">http://www.scag.gov.au/lawlink/SCAG/Il\_scag.nsf/pages/scag\_achievements#Spent%20convictions</a>>, viewed 29 April 2010.

# 6.2.4 Should there be restrictions on the number of principals for whom an individual representative can act?

Alfred Health also raised concerns that individual representatives can act for an unlimited number of principals, which may leave some principals susceptible to abuse. Ms Costar told the Committee:

We could have a circumstance where a person has four maiden aunts, all of whom require an attorney. So the person steps in and does that. That is a different situation, I think, from a person who befriends people and takes on the task of being their attorney. We have certainly come across that situation. I have come across that situation in the hospital where I have come across the same attorney a couple of times, several years apart. It was just one of those things that I came across this person. It alarmed me, but I have had no evidence at all that there was anything amiss. His view was that he was helping elderly people who lived in his street. 656

Alfred Health suggested there could be restrictions on the number of enduring powers of attorney that an individual representative could hold. No other participants raised this point.

While noting concerns that some people may serially seek out vulnerable people and try to gain appointment as their representative, the Committee has not received sufficient evidence to make recommendations about this matter. The Committee believes that the recommendations in the next chapter which aim to increase accountability and transparency around the exercise of powers by representatives are a more appropriate way of addressing this issue.

# Case study 12: The representative was a stranger to the principal 657

'One of the most alarming incidents of confirmed misuse/abuse of this type of matter that came to RDNS' [Royal District Nursing Service's] attention in recent years, involved a private individual who acted as an attorney under an Enduring Power of Guardianship for an elderly RDNS client. The attorney was a "stranger" to the client and in fact carried out this function as his commercial "business" (the name of which was unregistered). We initiated applications for both guardianship and revocation of the Enduring Power of Attorney, and were successful in both. Some of the alarming features of this case included this client being invoiced for numerous expenses of the attorney to which the attorney was not legally entitled nor was there evidence of such expenses incurred. And the attorney had been negotiating the sale of the client's house (without the client's knowledge) with a real estate agent who had assumed that the attorney was the owner of the house and had no knowledge that the attorney was acting outside his powers. Of ongoing concern to us was that whilst our client's interests were protected by the outcome of the VCAT hearing, this attorney appeared to be at liberty to continue his "business" and act as attorney for other vulnerable elderly persons, as we were advised that it would be difficult to establish an offence in the circumstances.'

Royal District Nursing Service, Submission 25, 4.

<sup>656</sup> Sally Costar, Transcript of evidence, above n 647, 7. See also Alfred Health, Submission 66, 3.

6.3 Appointing multiple and alternative

# 6.3 Appointing multiple and alternative representatives

# 6.3.1 How many representatives should a principal be able to appoint?

Currently in Victoria a principal may appoint a single representative, two or more joint representatives (who act together) or two or more joint and several representatives (who may act together or alone) under a general (non-enduring) power of attorney and an enduring power of attorney (financial). The *Guardianship and Administration Act 1986* (Vic) permits only a single representative to be appointed. <sup>659</sup>

Some participants in the Inquiry argued that the number and types of appointments that can be made should be consistent across all powers. Commercial law firm Hunt & Hunt's submission stated, 'As most donors want continuity they insist that the same representatives be appointed, which is not possible under the current legislation.' However, Mr John Billings, Deputy President of VCAT's Guardianship List, expressed the view that 'it would be unnecessarily complicated if not unworkable for there to be joint and several enduring guardians with authority to make a decision about, say a person's accommodation ...' While the Committee notes Mr Billings' misgivings, it observes that multiple representatives can be appointed for personal and lifestyle matters in some other Australian jurisdictions.

Several participants said that appointing multiple representatives may protect principals from abuse by providing a check on the action of individual representatives. Aged & Community Care Victoria suggested that the protective effect of joint appointments is so strong that these should be encouraged as the norm, with single representatives only able to be appointed with VCAT's approval. 664

Another advantage of appointing multiple representatives is that, where representatives are authorised to act separately, there is an increased chance that a decision-maker will be available to make a decision when needed. 665

However, not all participants agreed that appointing multiple representatives safeguards against abuse. Mr D Constantine, a participant in the Committee's Seniors' Forum, highlighted the danger that one representative will exert influence over the other, thus negating any benefits associated with joint appointments. 666

<sup>658</sup> Instruments Act 1958 (Vic) ss 107(1), 119(1).

<sup>&</sup>lt;sup>659</sup> Guardianship and Administration Act 1986 (Vic) s 35A(1), sch 4 form 1.

<sup>660</sup> Hunt & Hunt, Submission 24, 4. See also Alzheimer's Australia Vic, Submission 32, 3; The Institute of Legal Executives (Victoria), Submission 33, 1; Seniors Rights Victoria, Submission 38, 23; Ronald T Jones, JP, Submission 2, 1-2; Jeni Lee, Submission 57, 2, 4.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, 4-5.

<sup>662</sup> See, for example, Powers of Attorney Act 1998 (Qld) s 32(1); Powers of Attorney Act 2006 (ACT) s 13(2).

<sup>&</sup>lt;sup>663</sup> Royal College of Nursing Australia, Submission 26, 4; Seniors Rights Victoria, Submission 38, 23.

<sup>&</sup>lt;sup>664</sup> Aged & Community Care Victoria, Submission 53, 3. See also Ronald T Jones, JP, Submission 2, 2.

Royal College of Nursing Australia, *Submission 26*, 4; L Richmond, *Transcript of evidence*, Melbourne, 30 March 2010, 7; Janice Reynolds, *Transcript of evidence*, Melbourne, 30 March 2010, 14.

D Constantine, *Transcript of evidence*, Melbourne, 30 March 2010, 13.

The Committee agrees that principals should be able to appoint more than one representative for all types of powers of attorney. This will provide consistency, potentially provide a safeguard against abuse and, in the case of several appointments, ensure that a decision-maker is available when required. To ensure that principals understand the implications of appointing more than one representative, the educational materials for principals recommended in chapter four should clearly outline the benefits and issues to consider when appointing multiple representatives.

#### Recommendation 39: Appointing multiple representatives

The Committee recommends the Powers of Attorney Act provide that a principal may appoint one or more representatives.

#### 6.3.2 How can multiple representatives make decisions?

At present multiple representatives may only be appointed jointly (to act together) or jointly or severally (to act together or alone). Some participants in the Inquiry called for more flexibility in the appointment of representatives. In particular, there was support for allowing the appointment of a number of representatives and permitting some of them to act together. Representatives appointed in this way are often called composite representatives.

The Law Institute of Victoria informed the Committee that clients regularly request lawyers to draft a document appointing their children as composite representatives:

A parent may, for example, wish to appoint three children to act as his or her attorneys. Rather than making the appointment joint and several, in which case any of the children can act individually, the parent may feel more comfortable requiring two of the children to agree before any action is taken. 668

Composite representatives are currently able to be appointed in some other Australian jurisdictions. For example, legislation in the Australian Capital Territory (ACT) permits a principal to authorise representatives 'to act together or separately, or in any combination'. This allows arrangements such as multiple representatives acting as a simple majority or as a two-thirds majority.

The Law Institute suggested that legislation allowing composite representatives to be appointed should be retrospective. Ms Laura Helm, a Policy Adviser with the Institute, told the Committee that 'many people have actually tried to encompass this type of arrangement already'. <sup>670</sup>

146

-

Hume Riverina Community Legal Service, Submission 36, 2; St Kilda Legal Service Co-Op Ltd, Submission 50, 4; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2; Moores Legal Pty Ltd, Submission 31, 4.

Law Institute of Victoria, Submission 41, 7.

<sup>669</sup> Powers of Attorney Act 2006 (ACT) ss 13, 25.

Laura Helm, Policy Adviser, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 4. See also Law Institute of Victoria, *Submission 41*, 7.

The Committee recognises that members of the community may wish to appoint different numbers of representatives to act in different ways. It agrees that the powers of attorney legislation should permit the appointment of composite representatives. This change should apply to power of attorney documents made prior to the commencement of this provision in order to validate any appointments already made in this way.

#### Recommendation 40: Decision making by multiple representatives

The Committee recommends the Powers of Attorney Act provide that a principal can appoint multiple representatives to act jointly, jointly and severally, or in any combination, for example as a majority. This provision should apply to powers of attorney made prior to the commencement of this provision.

# 6.3.3 What happens if multiple representatives cannot agree?

St Kilda Legal Service's submission highlighted the need for a dispute resolution mechanism if multiple representatives cannot agree on a decision in relation to the exercise of their powers under a power of attorney.<sup>671</sup> Legislation in the ACT and Queensland allows representatives to apply to a court or tribunal for direction in such instances.<sup>672</sup>

The Committee acknowledges that representatives who cannot agree can currently apply to VCAT under its general powers to give directions and advisory opinions in relation to enduring powers of attorney (financial). However, it feels there would be benefit in explicitly stating in the legislation that multiple representatives who cannot agree should be able to apply to VCAT. This option would only be available for representatives appointed under enduring powers, as VCAT does not have jurisdiction in relation to general (non-enduring) powers of attorney.

The Committee believes that alternative dispute resolution mechanisms could be helpful to representatives who disagree about a particular decision to be made. There are a plethora of alternative dispute resolution mechanisms currently available in Victoria, including mediation.<sup>674</sup> Alternative dispute resolution is informal, cost-effective, time-effective and often less stressful for parties.

The Committee suggests that educational materials and information for representatives should provide information about alternative dispute resolution options and services and encourage representatives to access these services when they have a dispute. Educational materials for representatives is discussed in more detail later in this chapter.

<sup>&</sup>lt;sup>671</sup> St Kilda Legal Service Co-Op Ltd, Submission 50, 4.

Powers of Attorney Act 2006 (ACT) s 27; Powers of Attorney Act 1998 (Qld) s 80(2).

<sup>673</sup> Instruments Act 1958 (Vic) ss 125V, 125Z, 125ZA.

Victorian Parliament Law Reform Committee, Inquiry into alternative dispute resolution and restorative justice (2009), 26-33.

#### Recommendation 41: VCAT directions when representatives cannot agree

The Committee recommends the Powers of Attorney Act provide that representatives can apply to VCAT for directions when they cannot agree about a decision to be made in relation to the exercise of their power under an enduring power of attorney (financial) or an enduring power of attorney (guardianship).

# 6.3.4 Avoiding uncertainty where there are multiple representatives

# What happens if the principal does not specify how multiple representatives are appointed?

The Committee notes that legislation in the ACT and Queensland provides that if a principal does not specify how two or more attorneys are appointed, it is presumed they are appointed jointly.<sup>675</sup> While no participants in this Inquiry raised this issue, the Committee believes that the Victorian legislation should contain such a presumption in order to avoid any doubt.

#### Recommendation 42: Presumption that representatives are appointed jointly

The Committee recommends the Powers of Attorney Act provide that if a principal does not specify how two or more representatives are appointed, they are presumed to be appointed jointly.

### What happens if the power of one joint representative ends?

Another matter the Committee considers needs to be clarified is what happens when representatives are appointed to act together (jointly) and the power of one of them ends, for instance, the representative dies or resigns. Currently the Instruments Act stipulates that the power of all joint representatives automatically ends if the power of one joint representative is revoked. 676

This matter was not raised by Inquiry participants, however, a recent New South Wales review found widespread concern in that state about the automatic revocation of joint powers, in particular that principals were left without representation when there were still other representatives willing and able to act. The review recommended that where one joint representative's power ends, the remaining representative or representatives should be allowed to continue to act unless the principal has expressly stated otherwise in the power of attorney document. <sup>677</sup> Legislation in Queensland currently contains such a provision and the Committee

Powers of Attorney Act 1998 (Qld) s 78; Powers of Attorney Act 2006 (ACT) s 26.

<sup>676</sup> *Instruments Act 1958* (Vic) s 125R(2).

Land and Property Management Authority, New South Wales, *Review of the Powers of Attorney Act 2003*, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009), 15.

believes that, to avoid uncertainty and injustice, Victorian legislation should include a similar provision. <sup>678</sup>

### Recommendation 43: Effect of one joint representative's power ending

The Committee recommends the Powers of Attorney Act provide that, unless the power of attorney document states otherwise, when a joint representative's power ends, any remaining representative or representatives may continue to exercise power under the power of attorney document.

# 6.3.5 How many alternative representatives should a principal be able to appoint?

An alternative representative is a person appointed by the principal to stand in place of the representative if the representative is unable to act. Current Victorian legislation allows the appointment of one alternative representative under both an enduring power of attorney (financial) and enduring power of attorney (guardianship).

Several participants in the Inquiry called for a principal to be able to appoint more than one alternative representative. The Law Institute of Victoria's submission noted that under previous Victorian legislation, it was not uncommon for principals to appoint multiple alternative representatives. The Institute told the Committee that a couple often appointed each other as representatives and appointed two or more children as alternatives:

This practice had a number of benefits. Children appointed jointly and severally as attorneys could share the workload. In the event of the death or incapacity of one child, the donor parent was not left unrepresented. The appointment of two attorneys also reduced the likelihood of abuse by representatives. Finally, allowing the children to be appointed as alternative attorneys promoted harmony in the family by treating all children equally. 680

Mr Lachlan Wraith, who gave evidence on behalf of the Trustee Corporations Association of Australia, told the Committee that there appeared to be 'no good reason' why the number of alternative representatives is currently restricted. Again the Law Institute of Victoria suggested that this change should be retrospective to validate any existing arrangements made by people who have not understood the current limitation.

Alternative representatives have an important function, ensuring that a principal is not left without representation when the representative he or she initially appointed is

Instruments Act 1958 (Vic) s 120(1); Guardianship and Administration Act 1986 (Vic) s 35A(1A).

Powers of Attorney Act 1998 (Qld) s 59A.

Law Institute of Victoria, Submission 41, 7. See also Laura Helm, Transcript of evidence, above n 670, 4; St Kilda Legal Service Co-Op Ltd, Submission 50, 4; State Trustees Limited, Submission 58, 11; Trustee Corporations Association of Australia, Submission 27, att, 3; Hume Riverina Community Legal Service, Submission 36, 2; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 2; MF Spottiswood, Submission 45, 3; Jeni Lee, Submission 57, 4.

Lachlan Wraith, Senior Manager, Trusts & Estates, Equity Trustees Ltd, Trustee Corporations Association of Australia, *Transcript of evidence*, Melbourne, 22 October 2009, 3.

unable to act. The Committee agrees that principals should be free to appoint multiple alternative representatives for all types of powers of attorney and recommends that this apply to powers of attorney made prior to the commencement of this legislative change in order to validate appointments already made in this way.

#### Recommendation 44: Appointing multiple alternative representatives

The Committee recommends the Powers of Attorney Act provide that a principal may appoint one or more alternative representatives. This provision should apply to powers of attorney made prior to the commencement of this provision.

# 6.3.6 When should alternative representatives be able to act?

Currently an alternative representative is able to act if the representative dies, is absent or is legally incapable. Several submissions argued that an alternative representative should be able to act when a representative is unwilling to act. The Law Institute's submission highlighted that a principal 'who appoints one attorney and an alternative is left unrepresented where that first attorney is unwilling to act'. 683

State Trustees' submission argued that legislation should set out a clear process to allow a representative to relinquish power to an alternative without recourse to VCAT. It suggested that an alternative representative should be able to commence acting upon receipt of a signed notice stating that the representative is not willing to act. Where the representative is not willing to provide a signed notice, the matter could be brought to VCAT under the Tribunal's existing powers.

At present when a principal has impaired decision-making capacity, a representative can only resign with the leave of a court or VCAT. This process is designed to ensure that a representative cannot resign unilaterally, leaving a principal without a representative. State Trustees suggested that its proposed process should be taken as a relinquishing of power rather than a resignation by the representative.

The Committee believes that it is important to ensure that a principal who has lost capacity has a representative who is willing to make necessary decisions on his or her behalf. In addition to the current arrangements that allow an alternative to act where a representative dies, is absent or has impaired decision-making capacity, the Committee is of the view that there should be a mechanism allowing a representative to relinquish power to an alternative representative without application to VCAT.

The Committee recommends that a representative who wishes to relinquish his or her powers provide a signed notice to the alternative representative nominated in the

150

<sup>682</sup> Instruments Act 1958 (Vic) s 120(2); Guardianship and Administration Act 1986 (Vic) s 35A(1B).

<sup>683</sup> Law Institute of Victoria, Submission 41, 7. See also Trustee Corporations Association of Australia, Submission 27, att, 3.

State Trustees Limited, Submission 58, 10-11.

<sup>685</sup> Instruments Act 1958 (Vic) ss 125V, 125Z, 125ZA.

Instruments Act 1958 (Vic) s 125M(2); Guardianship and Administration Act 1986 (Vic) s 35D.

enduring power of attorney document stating that he or she is not willing to act in that role. A copy of that notice should also be provided to the body responsible for registering enduring power of attorney documents discussed in chapter eight. The representative's relinquishment would be non-revocable.

#### Recommendation 45: Relinquishing power to an alternative representative

The Committee recommends the Powers of Attorney Act provide that a representative can relinquish all powers under an enduring power of attorney to an alternative representative nominated in the enduring power of attorney document by providing a signed notice stating that the representative is not willing to act in that role to the alternative representative and to the body responsible for registering enduring power of attorney documents.

# 6.4 The powers of representatives

### 6.4.1 What powers do representatives have?

The Instruments Act provides that under an enduring power of attorney (financial) a principal may authorise a representative 'to do anything on behalf of the donor that the donor can lawfully authorise an attorney to do'. <sup>687</sup> A representative appointed under a general (non-enduring) power of attorney has similarly broad powers. <sup>688</sup> The legislation does make it clear that a representative under an enduring power of attorney (financial) is not authorised to make a decision about the principal's medical treatment. <sup>689</sup>

Under the Guardianship and Administration Act, an enduring power of attorney (guardianship) confers on the representative specified powers or 'all the powers and duties which the guardian would have if he or she were a parent and the appointer his or her child ...' 690

Evidence to the Committee suggests there is widespread uncertainty about the powers representatives have. Inquiry participants described the current legislation as 'vague' and 'circular'. 692

The extent of representatives' powers under an enduring power of attorney (financial) was particularly contentious. The Committee heard two distinct views: firstly that this power gives the representative all powers (including those that might be considered to fall within guardianship) and, secondly, that it confers only financial and legal powers.

<sup>&</sup>lt;sup>687</sup> Instruments Act 1958 (Vic) s 115(1).

<sup>688</sup> *Instruments Act 1958* (Vic) s 107(1).

<sup>689</sup> Instruments Act 1958 (Vic) s 125F(1).

Guardianship and Administration Act 1986 (Vic) s 35B.

Belinda Evans, Senior Advocate, Elder Rights Advocacy, Transcript of evidence, Melbourne, 22 October 2009, 4.

Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 3.

Ms Leonie Schween, Manager, Legal and Privacy at the Royal District Nursing Service, told the Committee that the Service treats the enduring powers of attorney (financial) as giving the representative very broad powers:

With respect to a financial power of attorney, some time ago I actually sought external legal advice to confirm what is the extent of those powers when it came to a request for information ... The advice was that a financial power of attorney, unless it specifically is limited in some respect, grants powers of decision making as wide as the decision-making powers that the person about whom the power of attorney is made would have ... the advice was that it is wide ranging and there is no limitation, really, to the decision making ... <sup>693</sup>

Mr Robert Shepherd, a barrister who gave evidence on behalf of The Victorian Bar, took a similar view. He advised:

the Common Law scope of a General Power permits an attorney to do "all things" which the principal can do and which are "capable of delegation". These things extend beyond purely financial matters and the main limitation on the Power is whether the thing is capable of delegation. <sup>694</sup>

Mr Shepherd concluded that the use of the word 'financial' in the heading of the proforma enduring power of attorney (financial) form in the *Take control* kit limits the power to financial matters. However, he suggested that if 'financial' was removed from the heading, then the representative would have the power to do all things capable of delegation.

Matters that are not capable of delegation are those requiring 'personal skill or discretion' such as making a will or swearing an affidavit. 695

The view that enduring powers of attorney (financial) are limited to financial and legal matters was more prevalent among Inquiry participants. Many participants started from the assumption that representatives appointed under enduring powers of attorney (financial) cannot make personal or lifestyle decisions. For example, Mr Paul Zanatta, Manager of Community Care, Policy and Small Rural Health with Aged & Community Care Victoria, told the Committee:

what is sometimes happening is that the person with enduring power of attorney (financial) is actually de facto assuming a whole lot of other things that are really in the space of an enduring guardian ... sometimes people only have one single authority because they have neglected to make any other type of authority, and hence what is happening is that the power of attorney (financial) is receiving reports about the person's health. They are making other decisions about the person and about their physical needs. 696

Leonie Schween, Manager, Legal and Privacy, Royal District Nursing Service, Transcript of evidence, Melbourne, 17 December 2009, 3.

Robert Shepherd, The Victorian Bar, Further information on the legal authority in terms of the common law scope of a power of attorney, supplementary evidence received 12 November 2009.

<sup>&</sup>lt;sup>695</sup> Berna Collier and Shannon Lindsay, Powers of attorney in Australia and New Zealand (1992) The Federation Press, 41.

Paul Zanatta, Manager, Community Care, Policy and Small Rural Health, Aged & Community Care Victoria,
 Transcript of evidence, Melbourne, 14 December 2009, 5. See also Lynne Barratt, Elder Law Committee,
 Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 9-10; Seniors Rights Victoria,
 Submission 38, 21-23, 35; Aged & Community Care Victoria, Submission 53, 4-5; Hume Riverina

The *Take control* kit, which is the main source of information about how to make a power of attorney in Victoria, states that a representative's powers under an enduring power of attorney (financial) are limited to legal and financial matters. <sup>697</sup>

It is not surprising that there was considerable confusion among participants in the Inquiry about the extent of representatives' powers under powers of attorney (financial). In their authoritative book on powers of attorney in Australia and New Zealand, Berna Collier and Shannon Lindsay state 'there is a complete dearth of authority as to the extent to which an attorney can be authorised to make decisions about the personal welfare of the donor, as distinct from his or her property matters, unless authorised by statute to make such decisions'. For this reason many jurisdictions, including Victoria, have introduced legislation specifically allowing representatives to make appointments that relate to personal and lifestyle matters, including medical treatment. 699

## 6.4.2 Defining representatives' powers

Many participants in the Inquiry called for more clarity about the powers that representatives have. Currently neither the legislation nor the power of attorney forms provide any guidance about representatives' powers. As noted above, the information in the *Take control* kit is based on the assumption that representatives under general (non-enduring) powers of attorney and enduring powers of attorney (financial) have the ability to make financial and legal decisions only. However, Ms Belinda Evans, Elder Rights Advocacy's Senior Advocate, told the Committee it is not clear what that means: 'Some might say certain types of lifestyle decisions, such as choosing where to live, could be seen as a type of legal decision.' She suggested that representatives need to be given a clear statement of what they can and cannot do."

Several participants referred the Committee to the general (non-enduring) power of attorney form which was used in Victoria prior to 1980. This form contained a comprehensive guide to representatives' powers, most of which related to financial matters. The Davis of the Law Institute of Victoria told the Committee that the old form 'enumerated the kinds of roles that the attorney could do, so it was not exclusive but it certainly gave very good guidelines as to, 'I can do this, and I can do that'.'

In particular, many participants argued that the powers of attorney legislation should provide more clarity about the powers that representatives have under each type of

Community Legal Service, Submission 36, 4; Elder Rights Advocacy, Submission 63, 3-4; Belinda Evans, Transcript of evidence, above n 691, 5.

Victoria Legal Aid and Office of the Public Advocate, above n 642, 28.

<sup>&</sup>lt;sup>698</sup> Berna Collier and Shannon Lindsay, above n 695, 154.

Robin Creyke, 'Enduring powers of attorney: Cinderella story of the 80s' (1991) 21 Western Australian Law Review 122, 142-143; Australian Law Reform Commission, Community law reform for the Australian Capital Territory: Third report - Enduring powers of attorney Report no 47 (1988), 23-24.

Belinda Evans, *Transcript of evidence*, above n 691, 3.

Tibid, 4. See also Elder Rights Advocacy, Submission 63, 4; Lynne Barratt, Transcript of evidence, above n 696, 10.

Law Institute of Victoria, Submission 41, 13, att.

David Davis, *Transcript of evidence*, above n 637, 8.

power of attorney document.<sup>704</sup> There were two suggested approaches: first that legislation should fully list the powers granted to representatives and, second, that legislation should describe the types of powers granted and provide examples.

The first approach was advocated by the Financial Planning Association of Australia. The Association's submission stated that representatives under enduring powers of attorney (financial) should be granted powers in relation to:

- real property
- tangible personal property
- stocks and bonds
- commodities and options
- banks and other financial institutions
- operation of an entity or business
- insurance and annuities
- estates, trust and other beneficial interests
- claims and litigation
- personal and family maintenance
- benefits from government programs and civil/military service
- retirement plans
- taxes
- gifts within certain limits. 705

The submission argues that the powers in each of these areas should be described in detail in legislation 'so that donors thoroughly understand the powers they are granting and so that attorneys are not left unable to complete necessary tasks on behalf of their respective donors'. The Association suggests that there should be a further range of powers that principals should only be able to grant individually and explicitly, such as the power to make gifts outside statutory limits and the power to change a beneficiary designation.

\_

Kristen Pearson, President, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 2; Elder Rights Advocacy, Submission 63, 4; Laura Helm, Transcript of evidence, above n 670, 3; John Chesterman, Transcript of evidence, above n 641, 9-10; Robert Shepherd, Transcript of evidence, above n 650, 5.

Financial Planning Association of Australia Ltd, Submission 49, 2.

<sup>&</sup>lt;sup>706</sup> Ibid, 3.

However, Mr Alistair Craig, Senior Corporate Lawyer with State Trustees, told the Committee there was a danger in specifying representatives' powers in detail in legislation:

the risk there is we cannot necessarily foresee all the types of circumstances that will arise in a person's life.

Just by way of example, we did not have binding superannuation nominations until relatively recently. The risk is that you will miss something if you try to specify exhaustively what they can do. There might be points you can make about what they cannot do.  $^{707}$ 

A different approach has been taken in Queensland and the ACT. Legislation in those jurisdictions sets out the types of powers that representatives have and provides examples (see figure 3). This approach was widely endorsed by Inquiry participants. Ms Diane Tate, Director Financial Services, Corporations, Community Policy, at the Australian Bankers' Association, told the Committee that this model 'is attractive for businesses because there is certainty as to what is provided statutorily and legally'.

However, State Trustees' submission noted that there may be overlap between different powers. For example, it suggests it may be difficult to identify the types of legal matters that are considered to be 'financial matters' as some legal matters are of a personal nature, such as those relating to protection of a principal's privacy. The submission concludes 'such statutory delineations would therefore require careful consideration'. The ACT legislation addresses this particular concern by providing that legal matters that relate to the principal's finances and property fall within 'property matter' and legal matters that relate to the principal's personal care fall within 'personal care matter'.

While expressing support for the model in the ACT and Queensland legislation, Dr Chesterman from the Office of the Public Advocate told the Committee his preference was to retain the current terminology of financial, guardianship and medical treatment as 'they have a history'. He also indicated that it was important to retain synergies between representatives appointed under enduring powers of attorney (guardianship) and guardians appointed by VCAT, as their powers are similar.

Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, Transcript of evidence, Melbourne, 22 October 2009, 9.

Powers of Attorney Act 2006 (ACT) ss 10-12; Powers of Attorney Act 1998 (Qld) sch 2. See also Mental Capacity Act 2005 (UK) c 9 ss 17, 18.

John Billings, Submission 37, above n 661, 6-7; Law Institute of Victoria, Submission 41, 6, 13; Laura Helm, Transcript of evidence, above n 670, 6.

Diane Tate, Director, Financial Services, Corporations, Community Policy, Australian Bankers' Association Inc, *Transcript of evidence*, Melbourne, 17 December 2009, 7.

State Trustees Limited, Submission 58, 10.

Powers of Attorney Act 2006 (ACT) ss 10 example 16, 11 example 9.

John Chesterman, *Transcript of evidence*, above n 641, 9.

# Figure 3: Defining representatives' powers: ACT Powers of Attorney Act<sup>714</sup>

#### 10 Meaning of property matter

In this Act:

property matter, for a principal, means a matter relating to the principal's property.

#### Examples of property matters a power of attorney may deal with

- 1 paying maintenance and accommodation expenses for the principal and the principal's dependants
- 2 paying the principal's debts and expenses
- 3 receiving and recovering amounts payable to the principal
- 4 carrying on the principal's trade or business
- 5 performing contracts entered into by the principal
- 6 discharging a mortgage over the principal's property
- 7 paying rates, taxes and other outgoings for the principal's property
- 8 insuring the principal or the principal's property
- 9 preserving or improving the principal's estate
- 10 investing in authorised investments for the principal
- 11 continuing investments of the principal, including taking up rights to share issues, or options for new shares, to which the principal becomes entitled because of the principal's shareholding
- 12 undertaking transactions for the principal involving the use of the principal's property as security for the benefit of the principal
- 13 undertaking a real estate transaction for the principal
- 14 dealing with land under the Land Titles Act 1925 for the principal
- 15 withdrawing amounts from, or depositing amounts into, an account of the principal held with an authorised deposit-taking institution
- 16 legal matters in relation to the principal's finances and property

#### 11 Meaning of personal care matter

In this Act:

*personal care matter*, for a principal, means a matter, other than a health care matter, special personal matter or special health care matter relating to the principal's personal care, including the principal's welfare.

#### Examples of personal care matters a power of attorney may deal with

- 1 where the principal lives
- 2 who the principal lives with
- 3 whether the principal works and, if the principal works, where and how the principal works
- 4 what education or training the principal gets
- 5 whether the principal applies for a licence or permit
- 6 the principal's daily dress and diet
- 7 whether to consent to a forensic examination of the principal
- 8 whether the principal will go on holiday and where
- 9 legal matters relating to the principal's personal care

Powers of Attorney Act 2006 (ACT) ss 10-11.

The Committee agrees that the current legislative descriptions of representatives' powers are too vague. This lack of clarity may contribute to both intentional and inadvertent misuse of powers. While the Committee does not think it is wise, or even possible, to fully enunciate the powers of representatives in legislation, it considers that the legislation should describe the powers representatives have under each type of power of attorney. These new provisions should be modelled on the Queensland and ACT legislation, but should retain the terminology with which Victorians are familiar. The Committee identifies a number of specific areas of uncertainty about representatives' powers in the next section, and the Committee's findings on these issues should be incorporated into the legislated definitions of 'financial and legal matters' and 'guardianship matters'.

The Committee believes it is important that representatives and third parties understand what representatives' powers are, and in particular the limits of the powers. These should be highlighted on the power of attorney forms as well as in information and educational materials for representatives and third parties. The Committee makes specific recommendations about educational materials for representatives later in this chapter and considers the educational needs of third parties such as banks and health care providers in chapter nine.

#### Recommendation 46: Legislated clarity about representatives' powers

The Committee recommends the Powers of Attorney Act:

- a) provide that representatives appointed under an enduring power of attorney (financial) have powers in relation to financial and legal matters
- b) provide that representatives appointed under an enduring power of attorney (guardianship) have powers in relation to guardianship matters
- c) define 'financial and legal matters' and 'guardianship matters' and give examples of the type of matters that a representative may deal with under each power.

# 6.4.3 Clarifying and extending representatives' powers

The Committee invited participants to comment on powers that representatives should and should not have. Participants identified a range of specific areas where there is currently uncertainty about representatives' powers. In addition, there were several suggestions for extending representatives' powers.

### Power to restrict a principal's visitors

The use of enduring powers of attorney (financial) to restrict principals' visitors caused a great deal of concern among participants. Seniors Rights Victoria's submission included this example:

Lynne Barratt, *Transcript of evidence*, above n 696, 9-10; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 4; Elder Rights Advocacy, *Submission 63*, 3.

Maria was refused access to visit her friend who resided in an aged care facility, based on the directions of the friend's attorney. Based on the powers donated in the POA [power of attorney] instrument the attorney did not have the power to refuse access and neither did the aged care facility. 716

The Royal District Nursing Service informed the Committee that representatives often try to restrict a principal's visitors without the principal's knowledge and for reasons not related to the principal's welfare.<sup>717</sup>

The Committee is concerned about the apparent widespread use of power of attorney documents to restrict visitors in a way that may compromise principals' rights. The Committee believes that defining the various powers in legislation as recommended earlier in this chapter will make it clear that only those with powers relating to guardianship are in a position to restrict principals' visitors.

Later in this chapter the Committee also recommends providing more guidance for representatives about how they make decisions on a principal's behalf and adopts an approach that focuses on the principal's autonomy and rights. The adoption of such a model will help ensure that a principal's visitors are not excluded on inappropriate grounds.

It is important that both representatives and third parties such as aged care service providers, are aware of the limits of a representative's power and the basis on which power should be exercised. The Committee considers education for these groups in chapters seven and nine.

## Case study 13: The representative did not have the power to restrict visitors <sup>718</sup>

'An aged care facility was restricting a resident's son from visiting his mother. He contacted the facility and asked to speak to his mother and he was told by the DON [Director of Nursing] that he could not speak with her or see her unless he had permission from his sister who held an enduring power of attorney (financial) from the resident. The son was estranged from his sister and other siblings and had had little contact with his mother for some time. The son contacted our office [Elder Rights Advocacy] for advice. An advocate visited the resident and she indicated that she wanted to see her son. The advocate explained to the DON that the sister did not have the power to restrict her brother from visiting his mother. The DON said that she did not get involved in family disputes and stated again that the son would have to obtain permission to visit from his sister. The advocate suggested that the son contact the Office of the Public Advocate and VCAT for advice. Unfortunately, his mother passed away before he was able to obtain further advice.'

<sup>&</sup>lt;sup>716</sup> Seniors Rights Victoria, *Submission 38*, 21.

<sup>&</sup>lt;sup>717</sup> Royal District Nursing Service, *Submission 25*, 4.

Elder Rights Advocacy, Submission 63, 3-4.

#### Power to access a principal's will

At common law a representative is not entitled to access a principal's will unless access is specifically authorised in the power of attorney document or is required for the management of the principal's affairs under the power of attorney.<sup>719</sup>

The Trustee Corporations Association of Australia argued that any bequests in a will may affect a representative's dealings with assets. Therefore the Association suggested that representatives with financial powers should have the same right to access a principal's will as administrators appointed by VCAT. Administrators are permitted to read a represented person's will either before or after that person's death.

The Law Institute of Victoria and State Trustees recommended legislation clarify that representatives are not entitled to obtain either the original or a copy of the principal's will unless it is specifically stated in the power of attorney document. They suggested that the holder of a will could be authorised to inform the representative of any specific gift in the will if the representative produces a signed medical certificate indicating that the principal has impaired decision-making capacity. 722

The Committee recognises that bequests in a will may impact on a representative's management of a principal's financial and legal matters. However, the Committee does not consider that it has received sufficient evidence on this issue to justify overriding the long-standing common law rule that a representative does not have a right to access a principal's will. In the short term, the Committee suggests that information and education materials for both principals and representatives should make it clear that a representative does not have the authority to access a principal's will, unless it is specifically stated in the power of attorney document. In the longer term, the Committee suggests that the Victorian Government consults further on this issue, particularly in light of arguments canvassed in the next chapter that representatives should be under an obligation to protect the entitlements of beneficiaries under a will.

### Power to access personal and health information

Several participants in the Inquiry also raised the issue of the power of a representative under an enduring power of attorney (financial) to access the principal's personal and health information. Solicitor Ms Jeni Lee informed the Committee, 'There are often breaches of privacy by the financial attorney assuming the power extends to accessing the donor's medical records without the donor's consent.'

Ken Aitken, 'Enduring powers and wills' (1999) 73(12) Law Institute Journal 94, 94.

Trustee Corporations Association of Australia, *Submission* 27, 3.

<sup>&</sup>lt;sup>721</sup> Guardianship and Administration Act 1986 (Vic) s 58G.

Law Institute of Victoria, Submission 41, 18; State Trustees Limited, Submission 58, 16. See also Hunt & Hunt, Submission 24, 4.

Jeni Lee, Submission 57, 3. See also Paul Zanatta, Transcript of evidence, above n 696, 4; Janice Hadgraft, Manager, Residential Services, Aged & Community Care Victoria, Transcript of evidence, Melbourne, 14 December 2009, 4.

The Committee recognises that people with impaired decision-making capacity are particularly vulnerable to breaches of their privacy. Mr Julian Gardner, the former Public Advocate, told the Committee, 'In my experience privacy is one of the first rights that seems to go out of the window when people lose capacity.' 724

The Committee believes that clarifying in legislation the powers that representatives have under different types of powers of attorney will provide more certainty about who is entitled to access personal and health information. The Committee's view is that the Victorian powers of attorney legislation should specify that the principal's health care and legal matters relating to the principal's personal care fall within the definition of a 'guardianship matter'. Educational materials for representatives and third parties, which will be discussed later in this report, also play an important role in alerting people about the extent of representatives' powers.

#### Power to act as a trustee

Under common law, trustee powers cannot be delegated, as they involve the exercise of 'personal skill or discretion'. The Instruments Act restates this rule in relation to representatives under general (non-enduring) power of attorney documents. <sup>726</sup>

The Committee received conflicting evidence about this restriction. The Trustee Corporations Association of Australia argued that representatives should be entitled to act as trustees. The Wictorian Bar argued that the existing prohibition should remain.

The Committee does not feel that it received sufficient evidence to make a recommendation about this issue.

#### Power to lodge a Queen's caveat

State Trustees submitted that it would be useful for a representative under an enduring power of attorney (financial) to be able to lodge a caveat (notice) to protect a principal's interest in land, for example, where a principal has been persuaded to sell property at below market cost. 729

The Registrar of Titles can enter a Queen's caveat on the title of a property on behalf of 'a person of unsound mind' to prevent any fraudulent or improper dealing with that property. <sup>730</sup>

State Trustees informed the Committee that the Registrar of Titles accepts VCAT guardianship and administration orders as evidence of 'unsoundness of mind' and has entered caveats lodged on behalf of represented persons. However, the Registrar has refused to exercise his discretion to allow a caveat to be lodged by a representative

Julian Gardner, *Transcript of evidence*, Melbourne, 1 October 2009, 3.

Berna Collier and Shannon Lindsay, above n 695, 41.

<sup>&</sup>lt;sup>726</sup> Instruments Act 1958 (Vic) s 107(2).

Trustee Corporations Association of Australia, *Submission* 27, 3.

The Victorian Bar, Submission 40, 3. See also Robert Shepherd, Transcript of evidence, above n 650, 3.

<sup>729</sup> State Trustees Limited, *Submission* 58, 17.

<sup>&</sup>lt;sup>730</sup> See *Transfer of Land Act 1958* (Vic) s 106(1)(a).

under a power of attorney. Mr Craig from State Trustees told the Committee, 'In effect, it is my understanding a policy decision at the Titles Office'. 731

State Trustees suggested that VCAT be given the power to make a declaration that an enduring power of attorney is valid, the representative's powers are currently exercisable, the powers extend to protection of the principal's interest in real estate and that the principal has impaired decision-making capacity. State Trustees informed the Committee that allowing such a caveat to be lodged 'would be a valuable protection for the donor against third parties who could otherwise unscrupulously take advantage of the donor'. 732

VCAT currently has wide powers to make declarations about enduring powers of attorney (financial). State Trustees advised that it is not aware of any applications to obtain an order of the type it has suggested under VCAT's current powers. However, Mr Craig expressed the view that if VCAT made such an order under the existing legislation 'it is not clear that the Registrar of Titles would be bound to act on the declaration/direction'. 734

No other participants in the Inquiry raised this issue.

The Committee agrees that an important role of a representative under an enduring power of attorney (financial) is to protect the property of a principal. The Committee's view is that VCAT's existing powers are broad enough for a representative to apply for a declaration of the sort suggested by State Trustees. The Committee believes that the Titles Office should accept such a declaration as sufficient evidence to allow a caveat to be lodged on the principal's behalf.

#### Power to refuse medical treatment

Several submissions called for the powers of representatives appointed under enduring powers of attorney (guardianship) to be extended to include the power to refuse medical treatment on behalf of a principal.

While representatives under enduring powers of attorney (guardianship) can withhold consent to treatment, they cannot refuse medical treatment. Mr Gardner explained this distinction to the Committee:

under an enduring power of guardianship I can give somebody all of the powers that can be given to somebody under an enduring power of attorney for medical treatment, except one — that is, the power to refuse treatment. So what we have is a situation that occurs from time to time in hospitals where a decision can be made to withhold consent for treatment. But that is not the same as refusing treatment. If that subtlety is difficult to grasp, I share the difficulty, and certainly doctors find it very difficult. <sup>735</sup>

Alistair Craig, *Transcript of evidence*, above n 707, 4.

State Trustees Limited, *Submission* 58, 17.

<sup>733</sup> *Instruments Act 1958* (Vic) s 125Z.

Email from Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, to Executive Officer, Victorian Parliament Law Reform Committee, 30 June 2010.

Julian Gardner, *Transcript of evidence*, above n 724, 3. See also Julian Gardner, *Submission 21*, 4-5; John Chesterman, *Transcript of evidence*, above n 641, 4.

The distinction between withholding consent and refusing treatment is significant, because a medical practitioner can overrule a decision to withhold consent, but a decision to refuse treatment can only be challenged by applying to VCAT. 736

A guardian appointed by VCAT has both the power to withhold consent and to refuse medical treatment. 737

At present a principal must separately appoint an agent under the *Medical Treatment Act 1988* (Vic) if he or she wishes to authorise a person to refuse treatment when he or she has impaired decision-making capacity.<sup>738</sup>

Mr Gardner and a number of other participants suggested that the powers of representatives under enduring powers of attorney (guardianship) should be extended to allow them to refuse treatment. Mr Gardner stated that this would affect only a very small number of cases. However, in his view this change may enhance the takeup rate of enduring powers of attorney (guardianship) 'because why give somebody powers in relation to making decisions about medical treatment when you cannot give them all of the powers that you want to give them under another document'. <sup>740</sup>

The Committee notes concerns about the overlapping roles of representatives under enduring powers of attorney (guardianship) and agents appointed under the Medical Treatment Act. However, the Medical Treatment Act is outside the scope of the Committee's terms of reference, and the Committee is not in a position to make a recommendation about this issue.

The Committee recognises it is important that principals fully understand the need to appoint an agent under the Medical Treatment Act if the principal wishes to have another person refuse medical treatment when he or she has impaired decision-making capacity. The proforma form for creating an enduring power of attorney (guardianship) in the *Take control* kit currently highlights this. The Committee believes that the new forms to be developed as a result of the recommendations in chapter four of this report should also highlight this distinction, as should information and educational materials for principals.

# 6.4.4 Principals' ability to limit representatives' powers

A principal is able to impose conditions or limitations on a representative's powers under both enduring and non-enduring powers of attorney. Some participants in the Inquiry told the Committee that limiting a representative's powers is one way a

162

Medical Treatment Act 1988 (Vic) ss 5C-5D; Guardianship and Administration Act 1986 (Vic) ss 42L-42M.
See also Office of the Public Advocate, Submission 9, 19.

<sup>&</sup>lt;sup>737</sup> Medical Treatment Act 1988 (Vic) s 5B; Guardianship and Administration Act 1986 (Vic) ss 37, 39.

<sup>738</sup> Medical Treatment Act 1988 (Vic) s 5A.

Julian Gardner, Transcript of evidence, above n 724, 3-4. See also Julian Gardner, Submission 21, 4-5; Elder Rights Advocacy, Submission 63, 6; Office of the Public Advocate, Submission 9, 19.

Julian Gardner, *Transcript of evidence*, above n 724, 4.

Victoria Legal Aid and Office of the Public Advocate, above n 642, 51.

<sup>&</sup>lt;sup>742</sup> Instruments Act 1958 (Vic) s 115(1)(b); Guardianship and Administration Act 1986 (Vic) s 35B(1); Berna Collier and Shannon Lindsay, above n 695, 7-10.

principal can algrify the powers a representative has potentially reducing abuse, or at

principal can clarify the powers a representative has, potentially reducing abuse, or at least making abuse more obvious. <sup>743</sup>

A participant in the Committee's Seniors' Forum, Ms Margaret Scott Simmons, told the Committee that she had acted as a representative under a power of attorney that was 'exceedingly detailed on exactly what the people who had the power could do and could not do'. The stated that the certainty provided by the power made her feel 'very secure' in her role as a representative.

Ms Lee, a solicitor who gave evidence on behalf of the Law Institute of Victoria, told the Committee that a representative could include conditions:

such as 'The home is not to be sold unless', so that there are conditions included in the financial power, and there is space on the document for those. There might even be a suggestion as to how that is managed. Then everybody knows exactly what is to happen, and the family is aware of when the house is to be sold, so it does not surprise the family that mum is to be kept at home as long as she wants to be, that the money, instead of going to school fees, is now to be spent on her care to make sure she stays at home ... <sup>745</sup>

However, evidence suggested that principals are not always aware of their ability to restrict representatives' powers. The Even principals who are aware that this option is available may not feel comfortable limiting powers. Ms Jenny Chapman, Chief Social Worker at Werribee Mercy Hospital, who gave evidence on behalf of Palliative Care Victoria, stated, 'We always explain that they can limit powers and what that means but some people are just not prepared to consider that and say, 'No. I would rather just leave it and see how it happens, trust my doctors, and my family can fight it out at the bedside'.'

The Australian Bankers' Association expressed concern that limitations on representatives' powers can give rise to confusion unless they are very clear. However, Professor Terry Carney from the Faculty of Law at the University of Sydney told the Committee that unpublished data:

demonstrates that citizens are pretty responsible about the kinds of ways that they write down the kinds of instructions they would like and the conditions and so on attached to the way in which substitute decision making might be exercised by somebody. 749

Margaret Brown, Adjunct Research Fellow, Hawke Research Institute, University of South Australia, Alzheimer's Australia Vic, *Transcript of evidence*, Melbourne, 22 October 2009, 5; L Richmond, *Transcript of evidence*, above n 665, 15.

Margaret Scott Simmons, *Transcript of evidence*, Melbourne, 30 March 2010, 10.

Jeni Lee, Elder Law Committee and Disability Law Committee, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 7. See also David Davis, Transcript of evidence, above n 637, 8; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Transcript of evidence, Melbourne, 1 October 2009, 6.

National Seniors Australia, Submission 54, 3; Jeni Lee, Submission 57, 3.

Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 8.

Australian Bankers' Association Inc, *Submission 55*, 7-8. See also Ministry of Justice, United Kingdom, *Reviewing the Mental Capacity Act 2005: Forms, supervision and fees* Response to consultation (2009), 35.

<sup>&</sup>lt;sup>49</sup> Terry Carney, Professor of Law, The University of Sydney, *Transcript of evidence*, Melbourne, 1 October 2009, 4.

The *Take control* kit advises principals to seek legal advice if they wish to impose conditions or limitations on a representative's powers. <sup>750</sup>

The current form for creating enduring powers of attorney (financial) has a space for principals to write conditions, limitations and instructions. Principals are directed to cross this section out if they do not wish to impose any restrictions. Several participants suggested that the form be redesigned to make it clear that principals have considered this option and exercised a choice, for example, by providing a box for principals to tick if they do not wish to impose any conditions on a representative's powers. <sup>751</sup>

The Committee believes that allowing the principal to place restrictions on a representative's power is an important protection against abuse. The information and educational materials for principals that the Committee has recommended in chapter four should clearly outline the principal's ability to limit a representative's powers and provide practical examples of the sorts of conditions and limitations that can be imposed. In addition, the forms for creating powers of attorney should be designed in a way that encourages principals to seriously consider the option of restricting the powers they delegate by requiring them to exercise a choice.

## 6.4.5 Which representative's decisions take precedence?

The Instruments Act currently provides that if the decision of a VCAT-appointed guardian or a representative under an enduring power of attorney (guardianship) conflicts with the decision of a representative under an enduring power of attorney (financial), then the decision of the former prevails. Evidence to the Committee suggests that this hierarchy of decision making is not well understood, and there was concern that, in practice, it is the decision of the financial representative that takes precedence. The provided representative that takes precedence.

Participants in the Inquiry acknowledged the overlapping powers of different representatives. Dr Chesterman told the Committee 'choosing where someone lives has financial implications. The enduring guardian and the attorney have to work that out between themselves ...'<sup>754</sup>

The Committee believes that providing clearer definitions of the powers that each type of representative has will help clarify the boundaries of each representative's decision making. However, it acknowledges that at times there may be overlap between the roles of various representatives and, in particular, many decisions have financial implications. The Committee believes the current provision, which highlights that the decision of those with powers in relation to guardianship matters prevails over those with powers over financial and legal matters, should be restated in the powers of attorney legislation.

Victoria Legal Aid and Office of the Public Advocate, above n 642, 9.

State Trustees Limited, Submission 58, 16-17; Law Institute of Victoria, Submission 41, 17; Moores Legal Pty Ltd, Submission 31, 4. See, for example, Department of Justice and Attorney-General, Queensland Government, above n 655, 9-10.

<sup>&</sup>lt;sup>752</sup> *Instruments Act 1958* (Vic) s 125F.

<sup>&</sup>lt;sup>753</sup> St Vincent's Hospital, Submission 19, 2; Carers Victoria, Submission 65, 13.

John Chesterman, *Transcript of evidence*, above n 641, 13.

The Committee considers that the legislation needs to provide greater clarity about what it means for one representative's decision to prevail over another's decision. Evidence to the Committee suggests that this issue most often arises where a representative with powers in relation to guardianship matters determines that a principal needs costly care that requires the representative with powers in relation to financial matters to make sufficient funds available. This may necessitate, for example, selling assets such as shares or the family home. The Committee considers that the wording of the legislation should make it clear that the representative with financial powers must take such available steps as may be necessary to allow the decision of the representative with guardianship powers to be implemented.

The evidence presented to the Committee suggests that the hierarchy of decision making is currently not well understood by either principals or representatives. This should be highlighted in information and educational materials for representatives. In particular, the Committee recognises that the fact that the decisions of a person with guardianship powers will prevail means that principals need to give careful consideration to who they appoint as a representative, appointing a person that they trust and in whose judgement they have full confidence. The Committee recommended the development of guidance materials for principals and persons likely to advise principals about appropriate appointments in the previous chapter.

# Recommendation 47: Decision of representative with guardianship powers to prevail

The Committee recommends the Powers of Attorney Act provide that if there is conflict between a representative with powers in relation to guardianship matters (whether appointed by VCAT or a principal) and a representative with powers in relation to financial and legal matters, then the decision of the representative with powers in relation to guardianship matters prevails, and the representative with powers in relation to financial and legal matters must take such available steps as may be necessary to allow the decision of the representative with powers in relation to guardianship matters to be implemented.

# 6.5 The duties of representatives

# 6.5.1 What are the duties of representatives?

Representatives occupy a position of 'trust and confidence', called a fiduciary relationship, which gives rise to particular obligations and duties at common law. These duties include acting in the principal's interests, not using his or her position for profit, avoiding conflicts of interests, acting in accordance with instructions, and acting with due care, skill and diligence. Representatives who do not act in accordance with their fiduciary duties may be personally liable for any loss or damage.

Berna Collier and Shannon Lindsay, above n 695, 142, 163-169.

165

<sup>&</sup>lt;sup>755</sup> Hospital Products Ltd v United States Surgical Corporation (1984) 156 CLR 41, para 68 (Mason J).

# Figure 4: Specific responsibilities of representatives<sup>757</sup>

• **Duty to keep records.** You must keep accurate records of dealings and transactions made under the power as VCAT or the Supreme Court or the Public Advocate may require you to produce them.

You must keep these records separate from your own affairs where possible.

For example, if you dispose of an asset you should keep records about the disposal.

Where there are joint attorneys, then it is sufficient that by agreement one of the attorneys will retain a record or account of transactions or dealings.

• **Duty to keep property separate.** You must keep your property separate from the donor's property unless you and the donor own the property jointly.

If the donor's capacity to make decisions is impaired, you must also get approval from VCAT or the Supreme Court for any transactions that have not been authorised in this document.

• **Duty to avoid transactions that involve conflict of interest.** You must not enter into transactions that could or do bring your interests (or those of your relation, business associate or close friend) into conflict with those of the donor.

However, you may enter into such a transaction if it has been authorised in this document or by VCAT or the Supreme Court.

The common law duties imposed on representatives are supplemented by legislative duties. Under the Instruments Act representatives have a duty to keep accurate records and accounts about dealings made using a power of attorney (financial). <sup>758</sup> In addition, as noted earlier in chapter four, when accepting an appointment under an enduring power of attorney (financial), a representative is required to state that he or she undertakes to exercise the powers with reasonable diligence and to protect the principal's interests, to avoid any conflict of interest between the representative's interests and the principal's interests, and to exercise the powers conferred in accordance with the Act. <sup>759</sup>

The Guardianship and Administration Act requires representatives under an enduring power of attorney (guardianship) to act 'in the best interests' of the principal. Acting in the principal's best interests is defined as:

- acting as an advocate
- encouraging the person to participate as much as possible in community life

Victoria Legal Aid and Office of the Public Advocate, above n 642, 37.

Instruments Act 1958 (Vic) s 125D.

<sup>&</sup>lt;sup>759</sup> *Instruments Act 1958* (Vic) s 125B(5).

<sup>&</sup>lt;sup>760</sup> Guardianship and Administration Act 1986 (Vic) ss 28(1), 35B(5).

- encouraging and assisting the person to make their own decisions
- protecting the person from neglect, abuse or exploitation
- consulting the person and taking into account his or her wishes.<sup>761</sup>

The *Take control* kit and the Department of Justice's website also provide information for representatives about their duties under enduring powers of attorney (financial). This information is summarised in figure 4.

### 6.5.2 Providing clarity about representatives' duties

Some participants in the Inquiry argued that representatives' duties should be set out clearly in legislation. This approach is taken in some other jurisdictions, for example, legislation in Queensland requires representatives to:

- exercise power honestly and with reasonable diligence to protect the principal's interests 763
- not knowingly exercise or purport to exercise a power that has been revoked 764
- not use confidential information gained through being a representative. 765

The Queensland legislation also sets out penalties for representatives who fail to abide by these duties.

Mr Shepherd of the Victorian Bar told the Committee that spelling out such specific duties helps guide representatives in their actions. In his view the Queensland requirements are sufficient, and a more detailed list of obligations would be too onerous on representatives. <sup>766</sup>

Seniors Rights Victoria's submission suggested that the obligations of representatives specified in legislation should include acting honestly and with a reasonable degree of care, as well as acting in accordance with the terms of the power. <sup>767</sup>

The Financial Planning Association of Australia submitted that there should be a high standard of non-waivable duties that are owed to principals, including duties to act in good faith and within the scope of the power. However, the Association suggested that some other duties, such as the duty to preserve the principal's estate

Guardianship and Administration Act 1986 (Vic) s 28(2).

Laura Helm, Transcript of evidence, above n 670, 3; Kristen Pearson, Transcript of evidence, above n 704, 2; Council on the Ageing Victoria, Submission 39, 5; Seniors Rights Victoria, Submission 38, 22-23, 35; Victoria Legal Aid, Submission 42, 4; Law Institute of Victoria, Submission 41, 13-14; State Trustees Limited, Submission 58, 14; Jeni Lee, Submission 57, 4.

<sup>&</sup>lt;sup>763</sup> *Powers of Attorney Act 1998* (Qld) s 66(1).

Powers of Attorney Act 1998 (Qld) s 71.

Powers of Attorney Act 1998 (Old) s 74A.

Robert Shepherd, *Transcript of evidence*, above n 650, 6-7.

Seniors Rights Victoria, Submission 38, 22-23.

plan, could be waived by an individual principal who appoints a non-professional representative. 768

The Trustee Corporations Association of Australia noted that administrators and trustees are required to invest in accordance with the prudent person principle and suggested this requirement could be extended to representatives. The prudent person principle requires the trustee to exercise the 'care, diligence and skill that a prudent person would exercise in managing the affairs of other persons'. There is a higher level of prudence expected of professional trustees and administrators.

It is important for representatives to have a comprehensive understanding of their responsibilities. The Committee believes that a clear legislative statement of representatives' key duties will provide a firm foundation for educating representatives about the requirements of their role. The Committee's view is that the fundamental duties of representatives that should be highlighted by legislation are:

- the duty to act honestly and in good faith
- the duty to put the principal's interests ahead of the representative's interests
- the duty to act within the powers granted
- the duty to keep accurate records.

These duties will also be supplemented by principles that guide decision making by representatives, which are discussed in the next section.

In chapter four the Committee recommended that all representatives be required to accept their appointment. The Committee believes that representatives' duties should be highlighted at this stage and that representatives should be required to specifically acknowledge the duties and undertake to exercise their powers in accordance with them.

Not acting in accordance with these legislative duties potentially gives rise to civil or criminal liability, as well as potentially providing a basis for the revocation of the representative's appointment. The Committee discusses these issues further in chapter seven.

The Committee also believes that representatives need to be well-informed about their duties and responsibilities. Third parties also need to be aware of the duties in order to be able to monitor the actions of representatives and report any suspected abuse. The Committee makes specific recommendations for the education of representatives and third parties about representatives' duties later in this chapter, as well as in chapter nine.

770 Trustee Act 1958 (Vic) s 6(1).

Financial Planning Association of Australia Ltd, Submission 49, 3.

<sup>&</sup>lt;sup>769</sup> Trustee Corporations Association of Australia, *Submission 27*, att, 4.

#### Recommendation 48: The duties of representatives

The Committee recommends the Powers of Attorney Act:

- a) provide that representatives have duties to act honestly and in good faith, to put the principal's interests ahead of the representative's interests, to keep accurate records and to act within their powers
- b) require representatives to undertake to act in accordance with the legislated duties when accepting an appointment as a representative.

### 6.5.3 How should representatives make decisions?

At present there is very little guidance for representatives about how they should make decisions on a principal's behalf. As noted earlier, the Guardianship and Administration Act requires representatives appointed under an enduring power of attorney (guardianship) to act in the principal's 'best interests', and failure to do so is a ground for revoking the representative's appointment.<sup>771</sup> There is no equivalent requirement in the Instruments Act.

Information in the *Take control* kit and on the Department of Justice's website sets out general principles to guide representatives in their decision making. These are set out in figure 5.

## Figure 5: General principles to guide decision making by representatives 772

- presuming that the donor has the capacity to make a particular decision until there is conclusive evidence that this is not the case
- recognising their right to participate in decisions affecting their life to the maximum extent for which they have capacity
- respecting the donor's human worth and dignity and equal claim to basic human rights, regardless of their capacity
- recognising the donor's role as a valued member of society and encouraging their self-reliance and participation in community life
- taking into account the importance of the donor's existing supportive relationships, values and cultural and linguistic environment
- ensuring that your decisions are appropriate to the donor's characteristics and needs
- recognising the donor's right to confidentiality of information.

Guardianship and Administration Act 1986 (Vic) ss 28, 35B(5), 35D(1)(b).

Victoria Legal Aid and Office of the Public Advocate, above n 642, 37. See also Department of Justice, Victoria, Office of the Public Advocate, *Information for attorneys*, <a href="http://www.publicadvocate.vic.gov.au/file/file/Powerofattorney/Information\_for\_Attorneys\_from\_the\_Department\_of\_Justice.pdf">http://www.publicadvocate.vic.gov.au/file/file/Powerofattorney/Information\_for\_Attorneys\_from\_the\_Department\_of\_Justice.pdf</a>, viewed 29 April 2010.

Mr Adrian Cohen, an individual who has been acting as a representative for his elderly parents for a number of years, told the Committee about how he approaches making decisions for his parents:

I think about what my parents would do in the particular situation, what they would have done and what they would have said they would have done. Then I think about my parents' best interests, about how they would like the situation solved, which is probably different from what they might say about how they would like it done. Then I normally go into a conservative situation. If it is a multiple choice-type situation where you could go high risk or low risk, I would go for low risk. Normally, I would take a while to make the decision, because my idea about a decision-making process is you decide about a week out from the event and then in the last couple of days you actually make the decision so that you have time to change it or rethink it. 773

# The role of principals' wishes and 'best interests' in decision making

Several participants in the Inquiry supported the introduction of a 'best interests' requirement for representatives under all types of power of attorney documents, with some stakeholders suggesting there should be a clear legislative explanation of what acting in the principal's best interests entails.<sup>774</sup> However, other participants viewed this as a paternalistic approach which may potentially facilitate abuse. Professor Carney told the Committee:

It looks good when I say, 'I have decided this in your best interests'. It is a bit like the father giving the kid a belting in the old days: 'This hurts me more than it hurts you'. Why am I doing it? 'It is in your best interests, son'. 'Why is it in my best interests?'. 'Just believe me; it is in your best interests'. It is not a very good guide.<sup>775</sup>

## Case study 14: While the principal was ill his son arranged for him to sign a power of attorney 776

'The client attended the Legal Service seeking to have a power of attorney revoked. The client stated that while he was ill his son attended with a lawyer and arranged for the client to sign a will and power of attorney. The client stated that subsequently his son provided him with an inadequate weekly allowance. The client was also concerned that his personal financial documents had been taken from his home. The client held the view that his son was not managing his affairs in his best interests.'

Adrian Cohen, *Transcript of evidence*, above n 628, 3.

Australian Bankers' Association Inc, Submission 55, 9; Alfred Health, Submission 66, 3; David Fara, Deputy President, Association of Independent Retirees (AIR) Ltd (Victorian Division), Transcript of evidence, Melbourne, 22 October 2009, 4-5; Kristen Pearson, Transcript of evidence, above n 704, 2; Julian Gardner, Submission 21, 4; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 4; Julian Gardner, Transcript of evidence, above n 724, 10; Seniors Rights Victoria, Submission 38, 38-39; Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 9-10; John Chesterman, Transcript of evidence, above n 641, 3.

Terry Carney, *Transcript of evidence*, above n 749, 7. See also Mental Health Legal Centre Inc, *Submission* 59, 6; Lynne Barratt, *Transcript of evidence*, above n 696, 9.

St Kilda Legal Service Co-Op Ltd, Submission 50, 3.

The Convention on the Rights of Persons with Disabilities promotes the rights of people to determine their own affairs, even when they have impaired decision-making capacity. The Victorian Equal Opportunity and Human Rights Commission argued that when a representative needs to make a decision 'the appropriate approach in a given scenario will fall on a continuum between supported and substitute decision making, with the precise point being dependent on the circumstances'. Supported decision-making approaches were discussed in detail in chapter five.

The Committee heard that principals' human rights are better protected by a decision-making approach that focuses on their wishes and preferences, rather than their perceived 'best interests'. The Mental Health Legal Centre's submission argued:

the wishes of the donor must be the paramount consideration guiding the donee ... Aspiring to a supported decision-making model would require that obligations upon the donee are strengthened to ensure the wishes of the donor are respected unless clear evidence can be produced as to why this should not be the case. <sup>780</sup>

Ms Debra Parnell, the Council on the Ageing's Policy Officer, told the Committee that powers of attorney are most commonly used to make decisions on behalf of older people who have lived their lives independently and managed their own affairs successfully for many years:

the focus should be on the attorney carrying out the wishes of the individual. It really should be a responsibility of the attorney that they endeavour to understand what the wishes of the person are, what their preferences are, what their lifestyle has been, and therefore what sorts of choices that person would conceivably make. The focus is on carrying out the person's wishes rather than acting in their best interests.

As we have said, people have the right to make bad choices, and it is not about taking over someone's life; it is about enabling their choices to continue to be carried out whether there is lack of capacity or whether there are illness and mobility issues that prevent them from carrying out those tasks themselves.<sup>781</sup>

The Office of the Public Advocate (OPA) acknowledged the difficulty of requiring principals' wishes to be given precedence in all situations. OPA's submission states:

Principals may lose the ability to make reasonable judgements and yet retain the ability to voice preferences. Any requirement that such wishes be observed by an attorney or enduring guardian would in effect render EPAs [enduring powers of attorney] superfluous. At the same time, OPA believes it is important to place in the legislation the principle that representatives and enduring guardians, in deciding

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3, art 12 (entered into force 3 May 2008). See also Charter of Human Rights and Responsibilities Act 2006 (Vic) s 32(2).

Victorian Equal Opportunity & Human Rights Commission, Submission 52, 7.

Julian Gardner, *Transcript of evidence*, above n 724, 3, 10; Mental Health Legal Centre Inc, *Submission 59*, 6.

Mental Health Legal Centre Inc, *Submission 59*, 6. See also Dahni Houseman, Policy Officer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 4.

Debra Parnell, Policy Officer, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 3.

how to act, must consult with, and where reasonable act upon, the expressed wishes of principals, and take into account any pre-incapacity views the principal had. <sup>782</sup>

Ms Lee, a solicitor who gave evidence on behalf of the Law Institute of Victoria, emphasised the importance of principals clearly stating their wishes in the power of attorney document. Other participants stressed the need for a principal to appoint a representative that they know well. For example, Mr Tony Fitzgerald, the Managing Director of State Trustees, told the Committee that his organisation often works with principals while they have capacity and this can be helpful in guiding decisions when a principal's decision-making capacity is impaired:

Sometimes if we start off a relationship under an EPA [enduring power of attorney] where they have capacity, you build up an experience with that client and you know what their preferences are, so if they do lose capacity and we continue on with that relationship, you would just continue on with implementing their preferences because you have built up that experience and you have got records on the file and that kind of thing about what their preferences are.<sup>784</sup>

#### Principles to guide decision making

As noted in chapter three, principles underpinning powers of attorney legislation in the UK, Queensland and the ACT reinforce the importance of maximising principals' rights to make their own decisions and to participate in decision making. <sup>785</sup>

A number of participants in the Inquiry supported the introduction of such legislative principles in Victoria, with some suggesting that a requirement to consult with the principal about their wishes should be at their core. <sup>786</sup>

OPA's submission pointed out that sometimes there may be tension between various principles, for example, a principal's wishes may conflict with the duty of the representative to protect the principal. OPA concludes that this risk is inherent in granting powers of attorney and emphasises the need for principals to appoint an appropriate representative in the first place. Mr Gardner gave one example where there is a conflict between these various factors:

If you have somebody who has never really had any money in their life and has been very careful and finishes up with \$10 000 in their older age and says, 'My wish is that you not spend that; I want it to go to my grandchildren', and the attorney says, 'Hang on, you need a motorised wheelchair for your quality of life', what are the best interests in those circumstances?<sup>788</sup>

172

Office of the Public Advocate, *Submission 9*, 16. See also Health Services Commissioner, *Submission 46*, 2; John Chesterman, *Transcript of evidence*, above n 641, 3, 10-11; Victorian Coalition of Acquired Brain Injury Service Providers, *Submission 71*, 2; Jeni Lee, *Submission 57*, 4.

Jeni Lee, *Transcript of evidence*, above n 745, 7.

Tony Fitzgerald, Managing Director, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 5. See also Office of the Public Advocate, *Submission 9*, 20.

Powers of Attorney Act 2006 (ACT) s 44, sch 1 cl 1.6; Powers of Attorney Act 1998 (Qld) s 76, sch 1 cl 7.
See also Mental Capacity Act 2005 (UK) c 9 ss 1, 4, 9(4).

Office of the Public Advocate, *Submission 9*, 16; Seniors Rights Victoria, *Submission 38*, 22-23, 39; Dahni Houseman, *Transcript of evidence*, above n 779, 4; John Chesterman, *Transcript of evidence*, above n 641, 3.

Office of the Public Advocate, *Submission 9*, 16.

Julian Gardner, *Transcript of evidence*, above n 724, 10. See also Alistair Craig, *Transcript of evidence*, above n 707, 5; John Chesterman, *Transcript of evidence*, above n 641, 10-11.

However, not all participants agreed that there should be legislated principles to guide representatives' decision making. Mr Shepherd, who gave evidence on behalf of the Victorian Bar, told the Committee that these would be 'too general' and instead the focus should be on specific duties such as the duty to act honestly.<sup>789</sup>

Some submissions suggested there should be sanctions for failing to act in accordance with the new proposed principles, in particular, the duty to consult. It was suggested that this could include criminal sanctions, <sup>790</sup> liability to compensate the principal or both. <sup>791</sup> Seniors Rights Victoria suggested that not acting in accordance with the legislative principles should be a ground for revoking the power of attorney document. <sup>792</sup>

#### Dr Chesterman of OPA told the Committee that such principles:

would only ever be guidelines; you could not make them enforceable. Ultimately it is the attorney who has to make the judgement call, and our belief is that that is why you should nominate someone you trust as your attorney and let them make the judgement call. <sup>793</sup>

In chapter three the Committee recommended that the new Powers of Attorney Act contain a statement of general principles to be applied by all those exercising powers or functions under the Act. The Committee has already dealt with the first two components of the proposed statement of principles: an articulation of relevant human rights contained in international conventions (see chapter three) and an emphasis on ensuring that all decisions about a person's capacity uphold his or her rights (see chapter five). The final component of the proposed principles is a focus on ensuring that all decisions and actions under powers of attorney promote principals' interests and wellbeing.

The Committee agrees that a decision-making approach that promotes principals' rights should be enshrined in the principles underpinning the powers of attorney legislation. The principles to guide representatives' decision making should focus on the principal's wishes. This highlights the need for ongoing communication between the principal and representative, which continues when the principal has impaired decision-making capacity.

The principles should also enshrine the principal's right to participation in decision making. The Committee believes that there is still a role for the principal's 'best interests' to be considered but prefers the more modern terminology 'the personal and social wellbeing of the principal'. This should be defined in legislation, with the definition including the elements currently listed in the definition of best interests in the Guardianship and Administration Act, as well as the general principles set out in figure 5 above.

Robert Shepherd, *Transcript of evidence*, above n 650, 6-7.

Stephen Taffe, Transcript of evidence, above n 647, 2-3; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 2; Seniors Rights Victoria, Submission 38, 22-23; Mental Health Legal Centre Inc, Submission 59, 6.

Julian Gardner, *Transcript of evidence*, above n 724, 10.

<sup>&</sup>lt;sup>792</sup> Seniors Rights Victoria, Submission 38, 22, 39.

John Chesterman, *Transcript of evidence*, above n 641, 3-4.

<sup>&</sup>lt;sup>794</sup> Ibid, 11.

The Committee recognises that such general principles would be guidelines only, setting out an approach to decision making but not providing specific assistance to help a representative make a particular decision. The Committee believes that more specific guidance, including examples of how to behave in accordance with the principles, should be set out in information and educational materials for representatives. Tailored educational materials for representatives are discussed in more detail later in this chapter.

The Committee recommends that the stage at which a representative accepts his or her appointment should be used to highlight the approach that must be taken to making decisions under a power of attorney. Thus, as with the duties discussed earlier in this chapter, when representatives accept their appointment they should be required to undertake to make decisions in accordance with the guiding principles.

It is also important that third parties are aware of the decision-making approach that should be used by representatives. This will enable them to offer support where appropriate, or to report behaviour that might constitute abuse. Thus the information and education materials for third parties must also highlight the principles by which representatives should make decisions. The Committee makes specific recommendations about educational materials for third parties in chapter nine.

The Committee considers the potential consequences of not acting in accordance with the legislated principles further in chapter seven, when it explores the civil and criminal liability of representatives and VCAT's powers to revoke the appointment of a representative.

#### Recommendation 49: Principles to guide decision making by representatives

The Committee recommends the statement of principles underpinning the Powers of Attorney Act recommended in recommendation 3 include principles to guide decision making by representatives. These principles should include:

- the starting point for any decision making should be the principal's wishes
- principals should be encouraged to participate in decision making, even when they have impaired decision-making capacity
- representatives must act in a way that promotes the personal and social wellbeing of the principal. This should be defined as including matters such as recognising the principal's role as a valued member of society, taking into account the principal's existing supportive relationships, values and cultural and linguistic environment and recognising the principal's right to confidentiality of information.

## Recommendation 50: Representatives' undertaking to act in accordance with the statement of principles

The Committee recommends the Powers of Attorney Act require representatives when accepting an appointment to undertake to act in accordance with the statement of principles.

# 6.6 Increasing representatives' understanding of their role, powers and duties

# 6.6.1 How well do representatives understand their role, powers and duties?

Evidence suggests that many representatives have only a rudimentary understanding of their role, powers and duties. One Australian study of representatives under enduring powers of attorney (financial) found, 'Limited knowledge of their responsibilities was fairly common'. Anecdotal evidence from New South Wales suggests that in most cases where principals' funds were misused, the representatives were 'simply misguided as to the nature and extent of their duties, or believed that their actions somehow benefited the principal'.

Participants in this Inquiry also highlighted that many representatives do not have a good understanding of the nature and requirements of their role. St Vincent's Hospital's submission stated, 'Donees do not understand their duties and or obligations often. Genuine mistakes are made ...' In particular, many participants were concerned that representatives did not understand the scope of their powers. This issue was predominantly raised in relation to persons appointed under enduring powers of attorney (financial) purporting to make lifestyle decisions, which was discussed earlier in this chapter.

Mr John Hogan, a participant in the Committee's Seniors' Forum, highlighted the confusion that some people acting as representatives have about the scope of their powers:

I have used my mother's enduring power of attorney which was made prior to 2004. The heading was enduring power of attorney and I thought that covered everything but I found out subsequently that it did not handle medical issues at all; it was only suitable for financial matters.<sup>799</sup>

State Trustees suggested that many non-professional representatives do not understand the concept of conflict of interest and their duty to avoid it, even though this is set out in the current form for accepting their appointment. 800

Ms Robyn Mills, Acting Director of Civil Law Services at Victoria Legal Aid, told the Committee that Legal Aid's telephone advice line receives calls on a daily basis from:

Cheryl Tilse, Jill Wilson et al, 'Older people's assets: A contested site' (2005) 24(s1) Australasian Journal on Ageing s51, s55.

Department of Lands, New South Wales, *Review of the Powers of Attorney Act 2003* Issue paper (2009), 4-5. See also Land and Property Management Authority, New South Wales, above n 677, 6.

<sup>&</sup>lt;sup>797</sup> St Vincent's Hospital, *Submission 19*, 2. See also Australian Association of Gerontology Inc, *Submission 15*, 2; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 4.

Seniors Rights Victoria, Submission 38, 35; Jeni Lee, Submission 57, 2-3; Sue Connelly, Committee member, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Transcript of evidence, Melbourne, 22 October 2009, 6.

John Hogan, *Transcript of evidence*, Melbourne, 30 March 2010, 4.

State Trustees Limited, Submission 58, 13-14.

people who hold powers of attorney who are not actually certain as to what they involve and what they can or cannot do, or they have held a power of attorney and find that they are at VCAT where VCAT is in the process of looking at whether they are going to revoke the power of attorney because they have misused the power and they have not actually had a full understanding of what was required through the use of the power in the first place. <sup>801</sup>

## 6.6.2 Providing information and support for representatives

Research has identified that educating representatives about their responsibilities is a key strategy in addressing the misuse of power of attorney documents. 802 Many participants in the Inquiry also identified the education of representatives as a priority and identified a number of strategies for achieving this.

#### **Training for representatives**

A number of participants suggested that representatives could receive training about their role, powers and duties, with some expressing the view that such training should be compulsory. Several participants suggested there should be testing or certification of representatives' ability to undertake the role.

## Information about role, powers and duties on powers of attorney forms

It was also put to the Committee that the forms for creating powers of attorney should contain information about representatives' role, powers and duties. For example, the New South Wales power of attorney form provides important information for principals and representatives and sets out how the power can be used, as well as the obligations of representatives. A recent review in that state has suggested that this information could be expanded further. However, the provision of detailed information on forms in Queensland has been criticised on the basis that it makes the forms unnecessarily long and 'does not appear to have any effect on the behaviour of some attorneys'.

176

Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 3.

Cheryl Tilse, Jill Wilson and Deborah Setterlund, 'The mismanagement of the assets of older people: The concerns and actions of aged care practitioners in Queensland' (2003) 22(1) Australasian Journal on Ageing 9, 13. See also Anne-Louise McCawley et al, 'Access to assets: Older people with impaired capacity and financial abuse' (2006) 8(1) The Journal of Adult Protection 20, 30.

Wictoria Legal Aid, Submission 42, 4; L Richmond, Transcript of evidence, above n 665, 8; Robyn Mills, Transcript of evidence, above n 801, 5.

Lionel H Parrott, Submission 67, 1; Royal District Nursing Service, Submission 25, 4; Leonie Schween, Transcript of evidence, above n 693, 4. See also David Davis, 'Protecting elderly victims of economic crime' (2007) 81(9) Law Institute Journal 46, 49.

Financial Planning Association of Australia Ltd, Submission 49, 1; Hunt & Hunt, Submission 24, 4; Phillip Hamilton, Submission 7, 3; Grant Sturgeon, Submission 35; Trustee Corporations Association of Australia, Submission 27, 3; Margaret Brown, Transcript of evidence, above n 743, 7; Jeni Lee, Submission 57, 4.

Department of Lands, New South Wales, *Powers of attorney in New South Wales* (2005); Land and Property Management Authority, New South Wales, above n 677, 6.

Glenn Dickson, 'The enduring power of attorney form: Should it be changed?' (Paper presented at the Queensland Law Society Symposium 2007, Brisbane, 2-3 March 2007), 1. See also Department of Justice and Attorney-General, Queensland Government, above n 655, 17-21; Department of Justice and Attorney-

#### **Educational resources for representatives**

There was also support for developing a range of educational materials specifically for representatives, such as information booklets and kits, explaining their roles, powers and duties. <sup>808</sup> Mr Cohen, an individual who has experience acting as a representative, informed the Committee that he found the *Take control* kit 'very helpful' and added, 'I think that is an excellent starting point for somebody who has absolutely no idea of what they are doing.' <sup>809</sup>

While many participants praised the *Take control* kit, it is a general resource, predominantly targeting principals and people contemplating making a power of attorney. The Office of the Public Advocate has published two detailed guides for people appointed as guardians and administrators. It was suggested that these publications could provide a good template for resources targeting representatives. <sup>810</sup> VCAT also runs training sessions for newly-appointed guardians and administrators which provide the opportunity to ask questions about their roles. <sup>811</sup>

Research suggests that any information provided for representatives must be practical and easy to understand in order not to discourage people from taking on the role of representative. Some participants in the Inquiry also emphasised the need for materials for representatives to be simple and functional, providing examples of situations the representative might encounter and step-by-step guides about how to respond to particular situations or dilemmas. States

#### Educating representatives when they accept their appointment

In chapter four the Committee recommended that all representatives should be required to accept their appointments as representatives. Many Inquiry participants suggested that the acceptance of appointment should be used to educate representatives about their role, powers and duties. It appears that this opportunity is not optimised at present. Ms Audrey Cooke, a member of the community who made a submission to the Inquiry, stated that when she accepted an appointment as a representative for a close friend 'the solicitor was more interested in getting the

General, Queensland Government, *Enduring power of attorney: Short form* Form 3, Queensland *Powers of Attorney Act 1998* (Section 44(1)) (2003), 12-16.

Peter MacCallum Cancer Centre, Submission 20, 1; Victorian Equal Opportunity & Human Rights Commission, Submission 52, 7-8.

Adrian Cohen, *Transcript of evidence*, above n 628, 3.

Leonie Schween, Transcript of evidence, above n 693, 4. See also Office of the Public Advocate, Administration guide: A guide for people appointed as administrators under the Guardianship and Administration Act 1986 (2009); Office of the Public Advocate, Good guardianship: A guide for guardians appointed under the Guardianship and Administration Act (2008).

Victorian Civil and Administrative Tribunal (VCAT), VCAT information sessions for newly appointed guardians and administrators: Dates for 2010, <a href="http://www.vcat.vic.gov.au/CA256902000FE154/Lookup/guardianship/\$file/information\_sessions-for\_guardians\_and\_administrators\_2010.pdf">http://www.vcat.vic.gov.au/CA256902000FE154/Lookup/guardianship/\$file/information\_sessions-for\_guardians\_and\_administrators\_2010.pdf</a>, viewed 29 April 2010.
 Cheryl Tilse, Deborah Setterlund et al, above n 634, 225.

Australian Bankers' Association Inc, *Submission 55*, 9; NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales, Review of the *Powers of Attorney Act 2003* (2009), 16-18 as attachment to letter from Imelda Dodds, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010.

signature from me than to ensure my understanding of my duties in that particular document'. 814

Commercial law firm Hunt & Hunt informed the Committee that the firm provides the principal with an information booklet about powers of attorney and asks the principal to provide a copy of the booklet to the representative before the representative signs the statement of acceptance. The firm's submission stated, 'We believe that the information booklet assists in the education of both the donor and attorney on exactly what powers he/she/they have under each enduring power of attorney.' 815

Some participants suggested that representatives should have to sign a statement that they have read prescribed information and understand the duties imposed on them, as required by legislation in the United Kingdom. Ms Mills from Victoria Legal Aid told the Committee that this would make it easier to prove fraud, 'You are saying, 'Yes, here is this knowledge' and that you have this knowledge which would make a criminal proceeding easier to prove because you are saying that you have received this knowledge and you have signed that you have understood this.' An alternative approach, suggested by the Royal College of Nursing Australia, is that a lawyer should be required to sign a statement stating that the power of attorney documents have been explained to and understood by the potential representative.

Witnesses giving evidence on behalf of the Law Institute of Victoria suggested that representatives could be referred to a lawyer for independent legal advice to make sure they fully understand the role they are taking on. 819

## Educating and supporting representatives – The Committee's view

While the Committee acknowledges that the education of representatives cannot eliminate abuse of powers of attorney, it believes that education is an important step in reducing inadvertent misuse by well-meaning representatives.

The Committee believes that the powers and duties of representatives should be clearly summarised on the forms for creating powers of attorney, as well as the forms for accepting appointment as a representative. This should be supplemented by more specific information and educational materials specifically tailored for representatives which could be modelled on material currently provided to newly-appointed guardians and administrators. This information should be available in a range of community languages, as recommended in chapter four.

178

Audrey Cooke, Submission 5, 1. See also Janice Reynolds, Transcript of evidence, above n 665, 16-17.

Hunt & Hunt, Submission 24, 3.

Mental Capacity Act 2005 (UK) c 9 sch 1 cl 2(1)(d).

Robyn Mills, *Transcript of evidence*, above n 801, 5. See also Lynne Barratt, *Transcript of evidence*, above n 696, 9.

<sup>&</sup>lt;sup>818</sup> Royal College of Nursing Australia, Submission 26, 4.

Lynne Barratt, Transcript of evidence, above n 696, 10; David Davis, Transcript of evidence, above n 637, 7-8.

Throughout this report the Committee identifies a range of specific material that should be covered in educational resources for representatives, including information about:

- enhancing the principal's capacity to make decisions for himself or herself (see section 5.5)
- what to do if multiple representatives do not agree (see section 6.3.3)
- how to resign or relinquish power to an alternative representative (see section 6.3.6)
- representatives' powers and duties (see sections 6.4 and 6.5)
- which representatives' decisions have precedence (see section 6.4.5)
- how to approach decision making on the principal's behalf (see section 6.5.3)
- the representative's right to be paid (see section 6.7)
- the role of a personal monitor appointed by the principal (see section 7.2.4)
- making gifts using the principal's funds (see section 7.2.5)
- notifying activation of an enduring power of attorney document (see section 7.2.8)
- criminal offences for the abuse of powers of attorney (see section 7.3.2)
- the need to register the power of attorney document (see section 8.2).

The Committee does not support mandatory education for representatives on the basis that it may be overly onerous and deter people from agreeing to act as representatives. However, voluntary information sessions, similar to those currently conducted for VCAT-appointed guardians and administrators, would provide an important opportunity for representatives to clarify their role and obligations.

There are currently no specific advice and support services available for representatives who may wish to ask questions or receive clarification about their role, powers and duties. The Committee believes that the Victorian Government should provide advice and ongoing support to representatives, in particular a telephone advice service. The Committee believes that OPA may be well placed to provide this service. Such a role would be consistent with OPA's current provision of education and support in relation to powers of attorney, as well as its broader role of promoting the rights of people with disabilities. The Committee acknowledges that the provision of support for representatives would need to be separated from OPA's advocacy role in VCAT applications to avoid potential conflicts of interests.

In the Committee's view representatives' acceptance of their appointment provides an important opportunity to educate them about their role, powers and duties. The Committee does not wish to overly formalise the representative's acceptance, as this may discourage people from taking on the role. However, it highlights that this is an ideal time for representatives to be provided with the educational materials and information about how to access support services for representatives.

#### Recommendation 51: Educating and supporting representatives

The Committee recommends:

- a) the forms for creating powers of attorney and accepting an appointment as a representative set out in the Powers of Attorney Act provide a summary of representatives' powers and duties
- b) the Victorian Government produce simple, easy-to-understand information and educational materials for representatives
- c) the Victorian Government provide advice and ongoing support to representatives, including through a telephone advice service and information sessions for new representatives.

## 6.7 Should representatives be entitled to be paid?

The current Victorian legislation is silent about whether representatives are entitled to be paid for their work. There is no common law prohibition on the reasonable remuneration of representatives. 820

Several participants in the Inquiry highlighted that allowing non-professional representatives to pay themselves using the principal's money potentially facilitates abuse and called for regulation in this area. Mr Billings of VCAT submitted that legislation should clarify that representatives are not entitled to be remunerated unless it is specifically authorised in the power of attorney document. The Trustee Corporations Association of Australia suggested that representatives could be required to have minimum professional qualifications if they are to be paid. Under the Guardianship and Administration Act, non-professional administrators are not entitled to be paid unless specifically authorised by VCAT.

In New South Wales representatives are prohibited from conferring a benefit on themselves unless it is explicitly set out in the power of attorney document. <sup>824</sup> The form for creating powers of attorney and accompanying explanatory material in that state highlight the need to specify in the power of attorney document if a principal wishes a representative to be paid. <sup>825</sup> Such a clause is currently standard in documents appointing a professional trustee such as State Trustees.

Berna Collier and Shannon Lindsay, above n 695, 63.

John Billings, Submission 37, above n 661, 9.

<sup>822</sup> Trustee Corporations Association of Australia, Submission 27, att, 4.

<sup>623</sup> Guardianship and Administration Act 1986 (Vic) s 47A(1).

<sup>&</sup>lt;sup>824</sup> Powers of Attorney Act 2003 (NSW) s 12, sch 3 cl 1-3. See also Powers of Attorney Act 2006 (ACT) s 34.

Department of Lands, New South Wales, above n 806, 4.

The Committee believes that setting out clear guidance on the remuneration of representatives will provide an important protection against abuse. The legislation should clarify that a representative is not entitled to be paid unless payment is specifically authorised in the power of attorney document. This restriction should be highlighted in the forms for creating powers of attorney as well as in the educational materials for principals and representatives recommended in chapter four and in the previous section respectively. Representatives would still be able to recover out-of-pocket expenses that they incur in executing their duties as representatives, as long as appropriate records are maintained.

#### Recommendation 52: Remuneration of representatives

The Committee recommends the Powers of Attorney Act provide that a representative is not entitled to be paid unless payment is specifically authorised by the power of attorney document.

## Case study 15: The son used his mother's money for his household expenses 826

'An elderly woman with limited English had been living with her son who was also her attorney. The son had used his powers under the EPA [enduring power of attorney] to pay for the household expenses from his mother's finances. After she moved to an aged care facility, the son continued to finance his household expenses with funds from her finances. The attorney's siblings became concerned and sought to have his power revoked.'

-

Office of the Public Advocate, Submission 9, 9.

## **Chapter 7: Protecting principals from abuse**

This chapter considers how principals can be protected from abuse of a validly created power of attorney document. While there is no single, quick-fix strategy to prevent abuse, in this chapter the Committee explores a plethora of approaches including increased supervision of representatives' actions and specific criminal offences for abuse of powers of attorney. The Committee also considers strategies for increasing the detection and reporting of abuse and empowering the Victorian Civil and Administrative Tribunal (VCAT) to provide even better outcomes where there is evidence of abuse.

# 7.1 Detecting, reporting and investigating abuse of powers of attorney

As noted in chapter two, there is very little data available about the extent of abuse of powers of attorney. However, evidence to this Inquiry suggests that abuse is not uncommon, particularly in relation to financial powers.

The Committee heard that abuse of power of attorney documents is difficult to detect, and even when it is detected, it is very rarely reported and investigated. This section explores the barriers to detecting, reporting and investigating abuse, and identifies strategies for increasing responses to abuse of powers of attorney.

### 7.1.1 Why is abuse difficult to detect?

Powers of attorney are private documents, with no accountability requirements. 827 This makes it extremely difficult to detect abuse. Mr Lachlan Wraith, who gave evidence on behalf of the Trustee Corporations Association of Australia, told the Committee:

One of the difficulties here is that there is perhaps a much lower reporting of incidents of financial abuse of the elderly than of other forms of crime because where the abuse is successfully orchestrated there is no way of detection. Presumably, or typically, someone who gets a power of attorney will also get themselves appointed as executor of the will and there is no-one who is able to determine what has happened. 828

Even the victims themselves may not be aware that they are being abused. Ms Jill Linklater, Alzheimer's Australia Victoria's Policy and Advocacy Officer, stated:

if a person with dementia has diminished capacity to make a decision, often they do not know and they are not aware of these issues happening; that they are being defrauded and there are fraudulent things happening around their finances and that whoever is the attorney is actually taking advantage of them. 829

Office of the Public Advocate, Submission 9, 7.

Lachlan Wraith, Senior Manager, Trusts & Estates, Equity Trustees Ltd, Trustee Corporations Association of Australia, *Transcript of evidence*, Melbourne, 22 October 2009, 4.

Jill Linklater, Policy and Advocacy Officer, Alzheimer's Australia Vic, *Transcript of evidence*, Melbourne, 22 October 2009, 6. See also Sue Field, 'Assessing mental capacity' (2008) 86 *Precedent* 45, 46.

Staff of service providers, such as banks and health and aged care services, may be in a position to detect abuse; however, identifying abuse of powers of attorney poses a number of challenges. Unlike physical abuse, there is often little tangible evidence of financial abuse. <sup>830</sup> It may also be hard to distinguish between the erratic behaviour of a person who has fluctuating capacity and his or her abuse by a representative. <sup>831</sup> In addition, it can be difficult to determine whether abuse has occurred or whether the representative has made an unwise or negligent financial decision. <sup>832</sup>

The Committee heard that members of the community may also be well placed to detect and report abuse, but may lack an understanding of what abuse is and may consider some abusive transactions using 'family money' acceptable behaviour within a family environment. 833

#### 7.1.2 Why is abuse not reported?

Even when there is suspicion that a power of attorney is being abused, the Committee heard that these concerns are not always reported. 834

Research suggests that where abuse is reported, it is usually family members, friends or professionals who make the reports. 835

There are a number of reasons why abuse may not be reported even if it is detected.

Victims who are aware of the abuse may be reluctant to report it. Abuse of a power of attorney is usually perpetrated by a close family member and a principal may be unwilling to report that person because of social and emotional ties, or because he or she depends on the abuser for care. 836

Again service providers may be in a good position to report abuse. Credit Union Australia submitted that staff in the financial services sector 'must be diligent and act on concern of any transaction/behaviour that may raise a "red flag". 837 A

Anne-Louise McCawley et al, 'Access to assets: Older people with impaired capacity and financial abuse' (2006) 8(1) *The Journal of Adult Protection* 20, 23.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 82.

Bisid 14 citing State Government of Victoria, submission 121 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2007), 17.

Alzheimer's Australia Vic, Submission 32, 6 citing Anne-Louise McCawley et al, above n 830, 30.

John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 4.

Age Concern New Zealand, Elder abuse and enduring power of attorney: A special report from the Age Concern New Zealand Elder Abuse and Neglect Database covering the period 1 July 2002 to 31 December 2003 (2004), 19-20.

Belinda Evans, Senior Advocate, Elder Rights Advocacy, *Transcript of evidence*, Melbourne, 22 October 2009, 5; Russell G Smith, 'Fraud and financial abuse of older persons' (2000) 11(3) *Current Issues in Criminal Justice* 273, 276; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 82; Cheryl Tilse, Jill Wilson and Deborah Setterlund, 'The mismanagement of the assets of older people: The concerns and actions of aged care practitioners in Queensland' (2003) 22(1) *Australasian Journal on Ageing* 9, 12; Adam Graycar and Marianne James, 'Crime and older Australians: Understanding and responding to crime and older people' (Paper presented at the Family Futures: Issues in Research and Policy - 7th Australian Institute of Family Studies Conference, Sydney, 24-26 July 2000), 9.

Credit Union Australia Ltd, Submission 18, 3.

Queensland study found that most aged care workers who had a concern about the management of an older person's assets had made a referral or sought advice. However, that research identified a number of barriers to reporting abuse including clients' hesitance to take action and staff members' lack of time and information. Sale Concerns about potentially breaching privacy laws may also prevent some staff members from reporting abuse. Sale

# 7.1.3 Increasing the capacity of service providers to detect and report abuse

Participants in this Inquiry identified that service providers such as health and aged care professionals and financial institution staff have the potential to play an increased role in detecting and reporting abuse.

Alzheimer's Australia Victoria's submission argued that clear and consistent protocols should be developed for those working in the aged care sector, setting out how to report and respond to suspected abuse. For example, the *With respect to age* booklet has been developed in relation to elder abuse as part of the Victorian Government's Elder Abuse Prevention Strategy. This resource, targeted at those working in the health and community sectors, aims to provide information about elder abuse and how to respond. 841

Participants also identified education and training for service providers as a key strategy to increase reporting of abuse. Ms Dahni Houseman, Seniors Rights Victoria's Policy Officer, told the Committee educating service providers would make them more confident to challenge a representative who is acting outside his or her power:

There is a general perception that 'If you are an attorney, we just have to listen to what you want', and that is where this is occurring. If education was undertaken for the staff to understand that there are different types of instruments and that they give people different powers, then that might also give them more confidence in being able to say, 'Well, actually, no, we can't assist you with that kind of decision; you don't have the power to do that'. 842

Educating professionals is also a key component of the Victorian Government's Elder Abuse Prevention Strategy. The 2005 report of the Elder Abuse Prevention Project which laid the foundation for the Government's current strategy found that professionals, particularly health professionals, may be in a good position to identify and report elder abuse, but that identifying and responding to abuse requires

<sup>&</sup>lt;sup>838</sup> Cheryl Tilse, Jill Wilson and Deborah Setterlund, above n 836, 12-13.

State Government of Victoria, above n 832, 19.

Alzheimer's Australia Vic, Submission 32, 6 citing Anne-Louise McCawley et al, above n 830, 23, 30.

Department of Human Services, Victoria, With respect to age - 2009: Victorian Government practice guidelines for health services and community agencies for the prevention of elder abuse (2009). Note the code of practice that supports the Mental Capacity Act 2005 (UK) c 9 also contains information about identifying and responding to abuse, see Department for Constitutional Affairs, United Kingdom, Mental Capacity Act 2005 Code of practice, The Stationery Office (2007), 134-135.

Dahni Houseman, Policy Officer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 5.

'awareness and good clinical judgement'. 843 The report found varying skill levels throughout the service sector and recommended resources and training for health care workers, as well as those in the financial and legal sectors.<sup>844</sup> The roll out of With respect to age has been accompanied by training for health and community sector workers.845

Inquiry participants also suggested that bank and other financial institution staff should to be trained to detect the potential abuse of powers of attorney. 846 The 2005 report Older people and the law recommended that the Australian Government work in cooperation with the banking and financial sector to develop national, industrywide protocols and a training program to assist banking staff to detect and report alleged financial abuse.<sup>847</sup>

An example of a training program for the financial services sector is the course for bank staff in monitoring older people's finances run by the Loddon Campaspe Community Legal Centre. This course aims to teach staff how to recognise and deal with vulnerability in customers, understand the characteristics of vulnerable customers and potential abusers and provide strategies for detecting impaired decision-making capacity in customers.<sup>848</sup>

While participants in this Inquiry generally supported the need to enhance the capacity of service providers to respond to abuse, a 2009 Victorian review found that education and empowerment of service providers may not significantly increase the detection and reporting of abuse. In particular, it was observed that 'even with additional knowledge practitioners and financial institution staff may be reluctant to act because of confidentiality issues or because of lack of conviction that their actions would improve outcomes'. 849

The Committee believes that professionals working in the financial, health, aged care and community sectors may be well placed to detect and report instances of abuse of powers of attorney. It believes that protocols should be developed to assist professionals working in these areas and that training should be provided to support these protocols. These should build on and support existing initiatives, such as the Victorian Government's response to elder abuse.

The Committee recognises that the financial services sector operates at a national level and that developing Victoria-specific resources and training would be counterproductive. Therefore it supports the calls in the Older people and the law report for national responses to educating and training bank staff.

Office of Senior Victorians, Strengthening Victoria's response to elder abuse Report of the Elder Abuse Prevention Project (2005), 35.

Ibid, 36, 32.

Department of Human Services, Victoria, With respect to age 2009 - New guide and information sessions, <a href="http://www.dhs.vic.gov.au/rrhacs/ph\_bulletin/with-respect-to-age/\_nocache">http://www.dhs.vic.gov.au/rrhacs/ph\_bulletin/with-respect-to-age/\_nocache</a>, viewed 5 July 2010.

Australian Bankers' Association Inc, Submission 55, 10. See also Anne-Louise McCawley et al,

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 42-43.

Frances Gibson, Financial & consumer credit issues for older consumers in central Victoria, report for Loddon Campaspe Community Legal Centre (2008), 84-85.

Georgia Lowndes et al, Financial abuse of elders: A review of the evidence, Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2009), 6.

The Committee discusses providing more general information and education about powers of attorney to service providers in chapter nine.

#### Recommendation 53: Increasing detection and reporting of abuse

The Committee recommends the Victorian Government:

- a) develop and implement protocols and training for professionals in the health, aged care and community sectors about detecting and reporting abuse of powers of attorney
- b) work in conjunction with the Australian Government, other jurisdictions and the banking and finance sector to develop and implement a national system for detecting and reporting suspected financial abuse and a training program for banking staff.

#### **Mandatory reporting**

One option raised in the literature is mandatory reporting of abuse of powers of attorney. This would mean that a professional who suspects abuse would be required to report it. However, several recent studies have concluded that there is insufficient evidence that mandatory reporting is effective. No participants in this Inquiry suggested mandatory reporting of abuse of powers of attorney.

#### **Protecting whistleblowers**

Several participants in this Inquiry called for legislative protection for professionals who report abuse of powers of attorney or participate in an investigation about abuse ('whistleblowers'). Mr Robert Shepherd, a barrister who gave evidence on behalf of the Victorian Bar, told the Committee:

Members of the community need to feel that they are able to bring their concerns to the attention of relevant authorities. In this regard consideration of whistleblower and privacy laws may be made with a view to protecting those who make complaints about what they perceive to be ... elder abuse. 852

Current Victorian legislation protects whistleblowers from actions such as reprisal and civil and criminal liability. However, that protection only applies to a person who discloses improper conduct by a public body or official and therefore has limited application to the reporting of abuse of powers of attorney.

A recent South Australian review found that providing whistleblower protection to professionals such as nurses, aged care workers and bank staff would encourage

David Davis, 'Protecting elderly victims of economic crime' (2007) 81(9) Law Institute Journal 46, 48.

Office of Senior Victorians, above n 843, 28-30; Georgia Lowndes et al, above n 849, 7.

Robert Shepherd, Barrister, The Victorian Bar, *Transcript of evidence*, Melbourne, 22 October 2009, 3. See also The Victorian Bar, *Submission 40*, 3; Alzheimer's Australia Vic, *Submission 32*, 6 citing Anne-Louise McCawley et al, above n 830, 30.

Whistleblowers Protection Act 2001 (Vic) pt 3, ss 1, 3-4.

these professionals to report and follow up concerns about abuse of powers of attorney. 854

The Committee believes that the proposed new Powers of Attorney Act should provide clear protection for whistleblowers who report abuse of powers of attorney. This may encourage reporting of suspected abuse. Whistleblower protection is particularly important for those working in the banking, health, community and aged care sectors.

#### **Recommendation 54: Protecting whistleblowers**

The Committee recommends the Powers of Attorney Act provide protection from civil and criminal liability for people who, in good faith, report abuse of powers of attorney.

## 7.1.4 Educating the general community about abuse

The Committee heard that many members of the community do not have a good understanding of powers of attorney and their abuse.

Alzheimer's Australia Victoria drew the Committee's attention to inconsistent community attitudes. Its submission highlighted that:

stealing and exploiting another person's finances is called theft and is a criminal offence. However, when misappropriation of finances occurs with an enduring power of attorney or a family member, it is currently referred to as financial abuse and frequently there are no consequences for the abuser. 855

To address this, Alzheimer's Australia Victoria suggested more education and discussion in the community about the meaning of 'family money' and financial abuse. 856

General community education is a key component of the Victorian Government's strategy to address abuse of older members of the community. The 2005 report of the Victorian Elder Abuse Prevention Project stated:

Families, and the community more generally, also need to be aware that some older people suffer abuse and neglect from those close to them. Where the abuse is a result of ignorance or negligence, information and education would be instrumental in alerting family members and other carers to what behaviours are inappropriate. Furthermore, the range of behaviours that may constitute criminal behaviour, also need to be better understood. 857

Advance Directives Review Committee, *Planning ahead: Your health, your money, your life. First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy* (2008), 19-20, 63-64. See also Anne-Louise McCawley et al, above n 830, 30; Western Canada Law Reform Agencies, *Enduring powers of attorney: Areas for reform* (2008), xiv.

Alzheimer's Australia Vic, Submission 32, 5.

lbid, 5-6 citing Anne-Louise McCawley et al, above n 830, 30.

Office of Senior Victorians, above n 843, 31, 6-7.

In response to this report, the Victorian Government has launched a communication strategy to promote awareness of elder abuse generally. This includes a range of promotional materials such as question and answer sheets, information brochures and pocket guides. A key element of the communication strategy is the promotion of Seniors Rights Victoria as a source of further information, support, referral and legal advice. 858

However, there is a dearth of evidence about the effectiveness of general community education campaigns on abuse rates. A recent review found that, while community awareness campaigns have increased the number of calls to help lines, there is no evidence that they lead to an overall reduction in elder abuse. 859

The Committee notes that members of the community, including victims and their close family and friends, may not always recognise abuse of powers of attorney. While acknowledging that community education may not have any significant impact on reducing abuse, the Committee believes it may encourage the increased detection and reporting of abuse. Therefore the Committee recommends the Victorian Government implement a general community education campaign aimed at increasing community awareness of abuse of powers of attorney. This should include information about the types of actions that constitute abuse and how to report abuse.

The Committee acknowledges the synergies that its recommended community education program has with other education activities, such as the Victorian Government's current campaign to raise awareness about elder abuse. The education campaign proposed by the Committee should be developed to complement these existing campaigns.

#### Recommendation 55: Educating the general community about abuse

The Committee recommends the Victorian Government develop and implement an education campaign to increase general community awareness about abuse of powers of attorney.

## 7.1.5 Clarifying where to report abuse

Several participants in this Inquiry called for there to be a single organisation with a clear responsibility for receiving reports about abuse of powers of attorney. At present, many people who may detect abuse may simply not know where to report it. For example, Ms Shirley Glover, a social worker at Wimmera Care Group, asked 'who do we report to if we know the Power of Attorney is being abused?' 861

Barbara Mountjouris, Director, Office of Senior Victorians, 'Empowerment, partnership and collaboration to prevent elder abuse' (Paper presented at the 10th International Federation on Ageing Conference, Melbourne, 3-6 May 2010).

Georgia Lowndes et al, above n 849, 6.

The Victorian Bar, Submission 40, 3; Julian Gardner, Transcript of evidence, Melbourne, 1 October 2009, 4; Lionel H Parrott, Submission 67, 2.

Shirley Glover, Submission 6, 2.

Suggestions for who could receive reports included the Office of the Public Advocate (OPA), <sup>862</sup> an ombudsman and a specially-established review board. <sup>864</sup>

In the United Kingdom (UK), the Office of the Public Guardian plays a role in receiving reports and investigating abuse of power of attorney documents. The Office of the Public Guardian may direct a Court of Protection Visitor to investigate and, in serious cases will refer the matter to the Court of Protection which may cancel the enduring power of attorney or prevent it from being registered. Recent review in South Australia recommended that the Public Advocate in that state should have the primary responsibility for receiving complaints.

The Committee believes that providing clarity about how to report abuse of powers of attorney may encourage reporting of suspected abuse. It believes that OPA, with its mandate to protect the rights of people with cognitive disabilities is well placed to receive complaints about abuse of powers of attorney. This role should be promoted widely in educational materials for principals, professionals and the general community recommended throughout this report.

#### Recommendation 56: Office of the Public Advocate to receive reports of abuse

The Committee recommends the Victorian Government empower the Office of the Public Advocate to receive reports of suspected abuse of powers of attorney.

### 7.1.6 Investigating abuse

Evidence to this Inquiry suggests that even when suspected abuse of powers of attorney is reported, cases are often not investigated and prosecuted. In particular, participants expressed concern that the police are currently reluctant to investigate cases of suspected abuse. 867

Dr John Chesterman, Manager of Policy and Education at OPA, told the Committee:

anecdotal evidence suggests that criminal levels of abuse of enduring powers of attorney are rarely reported to police and that in any case police are reluctant to follow through reported abuses of enduring powers of attorney simply because it is so difficult to prove. <sup>868</sup>

Similarly, notary and specialist lawyer, Professor Phillip Hamilton, submitted, 'In my experience, even the most blatant fraud by an attorney stimulates no activity by

Jeni Lee, Submission 57, 4; Julian Gardner, Transcript of evidence, above n 860, 4.

Lionel H Parrott, Submission 67, 2.

Royal College of Nursing Australia, Submission 26, 4.

Department for Constitutional Affairs, United Kingdom, above n 841, 134-135. Note the Office of the Public Guardian also registers power of attorney documents: see chapter eight of this report.

Advance Directives Review Committee, above n 854, 60-61. See also Standing Committee on Social Issues, Substitute decision-making for people lacking capacity Report 43, New South Wales Legislative Council (2010), 122; Western Canada Law Reform Agencies, above n 854, xiv.

Lillian Jeter, Executive Director, Elder Abuse Prevention Association, *Transcript of evidence*, Melbourne, 30 March 2010, 4. See also Office of the Public Advocate, *Submission 9*, 17; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 26-27.

John Chesterman, *Transcript of evidence*, above n 834, 4.

the police force towards enforcement. Usually of comparatively small amounts, such fraud never gets past first base with the Fraud Squad ... '869

Ms Lillian Jeter, Executive Director of the Elder Abuse Prevention Association, told the Committee that there is a need for further education and training of police that abuse of powers of attorney is a criminal rather than a civil matter.<sup>870</sup>

One of the outcomes of the Victorian Government's Elder Abuse Prevention Strategy discussed earlier in this chapter is the amendment of the *Victoria Police code of practice for the investigation of family violence* to include a section on responding to incidents of elder abuse, including financial abuse. <sup>871</sup> However, the Committee did not receive any evidence about whether this has resulted in increased police willingness to investigate abuses including powers of attorney.

Several participants in this Inquiry suggested that OPA could have an enhanced role in investigating abuse of powers of attorney. Mr Wraith from the Trustee Corporations Association of Australia told the Committee:

If there was the ability for, perhaps, the Office of the Public Advocate or some other statutory body to have a power of investigation which would enable a determination of a threshold level of capacity before there is the intervention of a court-like procedure such as VCAT, it is submitted that it would vastly increase the detection of incidents of loss of capacity in the community and substantially reduce the opportunity for financial or elder abuse. That would obviously be a very costly exercise to set up, but we do not believe the cost would be disproportionate to the benefit to the community.<sup>872</sup>

However, not all participants agreed that it was appropriate for a body such as OPA to investigate criminal matters. Ms Toni Higgins, Specialist Mental Health, Human Rights and Civil Law with Victoria Legal Aid, expressed the view:

in the end the police will have to investigate it. It becomes very murky if you get someone like the Office of the Public Advocate investigating criminal matters; it would actually be diluting some of their very important functions in relation to supporting people with disabilities. <sup>873</sup>

The role of the UK Office of the Public Guardian in receiving and investigating reports of abuse was noted earlier in this chapter. Similarly, Queensland's Adult Guardian is empowered to investigate complaints and allegations against representatives acting under powers of attorney. The Adult Guardian's powers include suspending a representative's power under an enduring power of attorney for up to three months if the representative has not adequately protected the principal's

191

Phillip Hamilton, Submission 7, 3.

Lillian Jeter, *Transcript of evidence*, above n 867, 6-7.

<sup>871</sup> Victoria Police, Code of practice for the investigation of family violence (2004), 19. See also Office of Senior Victorians, Victorian Government elder abuse prevention strategic implementation plan 2006-09 (2007), 11.

Lachlan Wraith, *Transcript of evidence*, above n 828, 5. See also Jeni Lee, *Submission 57*, 4; Terry Carney, Professor of Law, The University of Sydney, *Transcript of evidence*, Melbourne, 1 October 2009, 7.

<sup>873</sup> Toni Higgins, Specialist Mental Health, Human Rights and Civil Law, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 5.

<sup>&</sup>lt;sup>874</sup> Guardianship and Administration Act 2000 (Qld) s 174.

interests, has neglected his or her duties or has contravened the powers of attorney legislation. 875

Ms Houseman of Seniors Rights Victoria suggested that if a registration system is established, the body responsible for registering powers of attorney could play a role in investigating abuse. <sup>876</sup> Registration is discussed in chapter eight of this report.

While the Committee has recommended that OPA be empowered to receive reports of suspected abuse of powers of attorney, it does not believe that OPA should have detailed investigatory functions. Instead the Committee recommends that where OPA suspects abuse, it should make referrals to the police where appropriate, bring action in VCAT or provide guidance and support to others who may be in a position to make an application to VCAT, such as a personal monitor.

Later in this chapter the Committee also considers the role of VCAT in identifying abuse of powers of attorney and referring matters for investigation and potential prosecution.

# 7.2 Measures to prevent abuse of validly executed powers of attorney

In chapter four of this report the Committee considered a number of strategies aimed at reducing abuse of powers of attorney at the time the documents are created. In this section the Committee explores a range of measures for reducing the abuse of powers of attorney once these powers have been validly created. Evidence to the Committee suggested that abuse of enduring powers of attorney at this stage is much more common than pressure or duress to create a document. A further measure that may have an impact on abuse of valid power of attorney documents is registration, which is discussed in detail in the next chapter.

## 7.2.1 Balancing protection and utility

The Committee heard that the limited accountability requirements imposed on representatives makes the abuse of enduring powers of attorney, especially financial powers, relatively easy. For example, solicitor Ms Jeni Lee noted that at present attorneys are relied on 'to exercise diligently their responsibilities with no form of structured or regular accountability'. Similarly, Mr Stephen Taffe, Alfred Health's Legal Counsel, told the Committee:

The current legislation does not require independent scrutiny of the exercise of powers of attorney, and we think that may not be necessary in the case of medical and lifestyle powers of attorney, because in those cases the decisions usually

<sup>&</sup>lt;sup>875</sup> Guardianship and Administration Act 2000 (Qld) s 195.

Dahni Houseman, *Transcript of evidence*, above n 842, 4.

John Chesterman, *Transcript of evidence*, above n 834, 4, 8.

<sup>&</sup>lt;sup>878</sup> Jeni Lee, Submission 57, 2. See also Moreland Community Legal Centre Inc, Submission 51, 4; Stephen Taffe, Legal Counsel, Alfred Health, Transcript of evidence, Melbourne, 14 December 2009, 2-3; Lionel H Parrott, Submission 67, 2; Ministerial Advisory Council of Senior Victorians, Submission 48, 4; Debra Parnell, Policy Officer, Council on the Ageing Victoria, Transcript of evidence, Melbourne, 22 October 2009, 8; Seniors Rights Victoria, Submission 38, 34-35.

involve a range of health-care providers and other professionals. However, in the case of attorneys with power to make legal and financial decisions there is significant scope for abuse of vulnerable donors.<sup>879</sup>

However, while many participants in this Inquiry called for increased supervision of representatives' use of enduring powers, the Committee was warned that increasing accountability may have some negative impacts. Earlier in this report the Committee highlighted that powers of attorney are flexible, low cost and empowering for the principal. The more regulated the use of powers of attorney becomes, the greater the risk that these benefits are lost. 880

In particular, participants expressed concern that increasing accountability requirements would deter both principals and representatives from these arrangements. Ms Lucie O'Brien, the Federation of Community Legal Centres' Policy Officer, told the Committee that 'all of the measures to improve accountability need to be weighed against the risk that you will make powers of attorney so onerous that people will not enter into them any more which is obviously of concern'. 881 Dr Chesterman of OPA stated:

I am bearing that in mind whenever I am thinking about increasing the regulation of attorneys and their requirement to submit periodic reports and so on, which I am generally not in favour of; it would be a disincentive for a busy person to be occupying that role. 882

In this section the Committee explores a range of options for preventing abuse of validly created enduring power of attorney documents. In considering these options, the Committee endeavours to strike a balance between ensuring that principals are protected and ensuring that the benefits these arrangements offer are not unduly compromised.

## 7.2.2 Reducing abuse – The evidence gaps

Participants in this Inquiry put forward a variety of suggestions for preventing the abuse of enduring powers of attorney after they have been lawfully created. However, there were divergent views about which strategies would be most effective.

The Committee's consideration of protections against abuse have been hampered by a lack of evidence about the efficacy of the various measures proposed. A 2009 review of measures to address the financial abuse of elders found:

NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales, Review of the *Powers of Attorney Act 2003* (2009), 15 as attachment to letter from Imelda Dodds, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010.

Stephen Taffe, *Transcript of evidence*, above n 878, 2.

Lucie O'Brien, Policy Officer, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 9. See also Georgia Lowndes et al, above n 849, 7.

John Chesterman, Transcript of evidence, above n 834, 12. See also Terry Carney, Transcript of evidence, above n 872, 6-7; Lucie O'Brien, Transcript of evidence, above n 881, 7, 9; Cheryl Tilse, Jill Wilson, Deborah Setterlund and Linda Rosenman, 'Older people's assets: A contested site' (2005) 24(s1) Australasian Journal on Ageing s51, s52; Anne-Louise McCawley et al, above n 830, 30; NSW Trustee & Guardian, above n 880, 13; Australian Law Reform Commission, Community law reform for the Australian Capital Territory: Third report - Enduring powers of attorney Report no 47 (1988), 8; Department of Lands, New South Wales, Review of the Powers of Attorney Act 2003 Issue paper (2009), 6-7.

Many approaches have been advocated for preventing or detecting and ameliorating financial elder abuse. There is a dearth of evidence about the effectiveness of these interventions. This may be due to a lack of evidence of effectiveness rather than a lack of effectiveness. Interventions may be effective but this is yet to be demonstrated ... Optimal protection of elders' assets is likely to require a number of approaches. 883

The Committee acknowledges that there is unlikely to be a single solution for addressing abuse of powers of attorney at the post-creation stage. Therefore, the Committee recommends a multifaceted approach, with the solutions in this section focusing on increased supervision of representatives' activities. The Committee believes that increased accountability for representatives in addition to clearer criminal sanctions for perpetrators of abuse discussed later in this chapter, strengthened requirements for creating a document considered in chapter four and increased education of representatives recommended in chapter six, will provide a solid framework for protecting principals from abuse.

### 7.2.3 Reporting and auditing requirements

Participants in this Inquiry called for increased formal monitoring of representatives' activities. In particular, there was concern that in the absence of any ongoing monitoring, currently third parties are relied on to bring possible abuse to VCAT's attention. <sup>884</sup>

VCAT currently has a limited oversight role in relation to the management of affairs by a representative under an enduring power of attorney (financial). Under the *Instruments Act 1958* (Vic) VCAT can order a representative to lodge accounts or other documents with VCAT or that the accounts be examined or audited by a person appointed by VCAT.<sup>885</sup>

Mr John Billings, Deputy President of the Guardianship List at VCAT, told the Committee that VCAT does use these powers, including requiring some representatives to lodge accounts annually. However, he acknowledged that 'we are passive, so if no-one had ever made an application in respect of those donors, we would not know about them'. 886

In contrast, there are much more stringent obligations in place for administrators appointed by VCAT under the *Guardianship and Administration Act 1986* (Vic). Administrators are required to lodge annual accounts which must provide 'a full and true account of the assets and liabilities of that estate'. These accounts are either lodged with VCAT or with another body that examines and audits the accounts and submits a report to VCAT.

<sup>883</sup> Georgia Lowndes et al, above n 849, 5. See also NSW Trustee & Guardian, above n 880, 7-8, 12-15; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 25-26.

Seniors Rights Victoria, Submission 38, 34.

Instruments Act 1958 (Vic) s 125ZB.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Transcript of evidence*, Melbourne, 1 October 2009, 7-8.

Guardianship and Administration Act 1986 (Vic) s 58.

Mr Billings told the Committee that VCAT currently receives annual accounts from about 7000 private administrators which it refers for professional examination. 888 Only a small number of these accounts result in a reassessment of the administration

order (about 150 per year) and very few of these reassessments lead to the revocation of an administrator's appointment (up to 10 each year). 889

#### **Annual statements**

Some participants in the Inquiry highlighted the discrepancy in reporting requirements for administrators and representatives appointed under enduring powers of attorney (financial). Mr Wraith told the Committee:

Presently under the Guardianship and Administration Act — appointments by VCAT — there is a requirement for annual reporting of accounts. It is difficult to see why there should be a distinction between someone acting under a power of attorney for someone who has lost capacity and an appointment by VCAT. In both instances it is an individual acting in a fiduciary capacity for someone who is unable to scrutinise the activities of the person who is acting on their behalf. There does not appear to be any reason why the reporting requirements would be any different. <sup>890</sup>

A number of other participants also supported a requirement that representatives under enduring powers of attorney (financial) complete annual accounts or statements. <sup>891</sup> It was suggested that these should be submitted to VCAT, <sup>892</sup> the body responsible for registering power of attorney documents, <sup>893</sup> guardians or representatives under enduring powers of attorney (guardianship) <sup>894</sup> or personal monitors. <sup>895</sup>

However, some participants expressed concern that requiring the submission of annual accounts would place an unreasonable burden on representatives. Mr John Hogan, a participant in the Committee's Seniors' Forum, told the Committee:

I think one of the problems we will face if you add annual returns and an enormously onerous responsibility on people who accept powers of attorney is that you will finish up with things essentially being handled by solicitors or trustee companies ... You do not want a system that is too onerous for people at the bottom end of the market. If somebody is going to ask me to handle an annual return of how I spent every dollar for mum or dad or whatever and why I have done this and why I have done that, and I am doing it for nothing, it is a big ask.

John Billings, *Transcript of evidence*, above n 886, 7.

Letter from John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), to Executive Officer, Victorian Parliament Law Reform Committee, 29 October 2009, 3.

Lachlan Wraith, *Transcript of evidence*, above n 828, 6. See also Jeni Lee, Elder Law Committee and Disability Law Committee, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 5; Jeni Lee, *Submission 57*, 3.

Shirley Glover, Submission 6, 2; Jeni Lee, Submission 57, 3; Seniors Rights Victoria, Submission 38, 34; David Davis, Elder Law Committee, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 10. See also Laura Helm, Policy Adviser, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 3; Moreland Community Legal Centre Inc, Submission 51, 4.

Jeni Lee, Submission 57, 3.

<sup>893</sup> Seniors Rights Victoria, Submission 38, 34-35; David Davis, Transcript of evidence, above n 891, 10; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, 7.

<sup>&</sup>lt;sup>894</sup> Carers Victoria, Submission 65, 15.

<sup>&</sup>lt;sup>895</sup> Ibid.

John Hogan, Transcript of evidence, Melbourne, 30 March 2010, 10.

Mr Billings also questioned whether representatives would be willing to take on the role if such demanding obligations were introduced. 897

Ms O'Brien of the Federation of Community Legal Centres suggested that, to reduce the administrative burden on representatives, they could make these annual reports by telephone, rather than in writing:

One alternative might be to make an annual telephone appointment to ring in to OPA, or to whichever other body is responsible for the regulation of this area, to report orally on how you are exercising your powers. If for any reason OPA has any concerns, they can then ask you to supply documentary evidence and possibly even refer it for investigation. That occurred to us as one possible means of making it less onerous for people to comply with reporting the obligations that we are talking about. 898

She suggested that this might be particularly useful for representatives who have low levels of literacy or limited English language skills.

Some other participants suggested that, rather than requiring annual accounts, representatives could be required to submit an annual plan about the management of the principal's affairs or an annual statutory declaration stating that they are complying with all duties. 900

#### **Auditing**

Some participants in the Inquiry supported the introduction of regular audit requirements. However, Mr Wraith, who gave evidence on behalf of the Trustee Corporations Association of Australia, warned that auditing 'would impose a very significant expense'. 902

Several Inquiry participants suggested that audits could only be required in some cases, such as for large estates,  $^{903}$  or where the principal specifically requires it in the power of attorney document.  $^{904}$ 

Another option, proposed by several participants, including Mr Wraith, is a system of random audits. The Committee was told that random auditing would increase accountability, while being less onerous than requiring regular auditing or annual submission of accounts. The 2007 *Older people and the law* report found that

196

John Billings, *Transcript of evidence*, above n 886, 7. See also Mental Health Legal Centre Inc, *Submission 59*, 11; NSW Trustee & Guardian, above n 880, 19-20.

Lucie O'Brien, *Transcript of evidence*, above n 881, 9.

Stephen Taffe, *Transcript of evidence*, above n 878, 2-3. See also Georgia Lowndes et al, above n 849, 6.

<sup>&</sup>lt;sup>900</sup> Seniors Rights Victoria, Submission 38, 35; Council on the Ageing Victoria, Submission 39, 5.

<sup>901</sup> Seniors Rights Victoria, Submission 38, 34; Stephen Taffe, Transcript of evidence, above n 878, 2-3.

Lachlan Wraith, *Transcript of evidence*, above n 828, 6.

Stephen Taffe, Transcript of evidence, above n 878, 2-3; David Fara, Deputy President, Association of Independent Retirees (AIR) Ltd (Victorian Division), Transcript of evidence, Melbourne, 22 October 2009, 5.

State Trustees Limited, *Submission 58*, 13. See also Land and Property Management Authority, New South Wales, *Review of the Powers of Attorney Act 2003*, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009), 11.

David Fara, *Transcript of evidence*, above n 903, 5; Debra Parnell, *Transcript of evidence*, above n 878, 8; Ronald T Jones, JP, *Submission* 2, 3; Australian Association of Gerontology Inc, *Submission* 15, 2; Lachlan Wraith, *Transcript of evidence*, above n 828, 6.

periodic random audits may offer some benefits, but noted that such a system should be carefully designed to ensure it is not too onerous for representatives. 906

A recent study identified that auditing accounts could deter 'flagrant financial abuse' but was unlikely to prevent more subtle abuse. 907

#### Reporting and auditing requirements – The Committee's view

The Committee notes current concerns about the lack of accountability for representatives with financial responsibilities. However, the Committee fears that introducing more stringent formal reporting requirements will mean that people may be discouraged from taking on the role of representative. The Committee believes that a more appropriate option is to encourage principals to make arrangements for oversight of the operation of their power of attorney documents. The Committee envisages that this will primarily be in the form of oversight by a personal monitor, as discussed in the next section.

#### 7.2.4 Personal monitors

There was considerable support among participants in this Inquiry for enabling the principal to appoint a personal monitor to oversee the operation of enduring powers of attorney. This approach was seen as being personally empowering for the principal, as well as being less formal than other approaches such as increased auditing and reporting requirements. 909

Several other jurisdictions have introduced a mechanism for oversight by personal monitors. In the UK a principal may specify up to five 'named persons' in an enduring power of attorney document. These people must be notified when the document is to be registered and are able to object to the registration. <sup>910</sup> In New York in the United States, a principal may appoint a monitor who has authority to request information about financial transactions made by the representative. <sup>911</sup>

OPA is a particularly strong proponent of personal monitors. Dr Chesterman explained the role that OPA envisages a personal monitor could play:

The interested persons would not have any role to play in the exercise or execution of an enduring power of attorney, but they would be a person whom a principal can nominate as having an interest in the wellbeing and financial affairs of the principal. <sup>912</sup>

<sup>&</sup>lt;sup>906</sup> House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 95-96.

Georgia Lowndes et al, above n 849, 6.

Tony Fitzgerald, Managing Director, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 6; Lachlan Wraith, *Transcript of evidence*, above n 828, 8; Lionel H Parrott, *Submission 67*, 1; Margaret Brown, Adjunct Research Fellow, Hawke Research Institute, University of South Australia, Alzheimer's Australia Vic, *Transcript of evidence*, Melbourne, 22 October 2009, 4; Office of the Public Advocate, *Submission 9*, 17-18; Seniors Rights Victoria, *Submission 38*, 35.

Ocatherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 5-6.

The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian Regulations 2007 (UK) reg 2, 6.

<sup>911</sup> NY General Obligations Law § 5-1509 (2009). See also Representation Agreement Act, RSBC 1996, c 405 s 12.

John Chesterman, *Transcript of evidence*, above n 834, 4. See also Dahni Houseman, *Transcript of evidence*, above n 842, 6.

Participants in this Inquiry suggested a number of potential functions a personal monitor could perform:

- receiving accounts from the representative 913
- being notified if the representative makes any gifts to himself or herself or any large gifts to others<sup>914</sup>
- being notified if the representative plans to sell the principal's family home or other major asset <sup>915</sup>
- being notified of an intention to make a major change to the principal's lifestyle, for example, a plan to move the principal into residential care <sup>916</sup>
- being notified of the activation or an application to register an enduring power of attorney document. 917

Personal monitors could potentially challenge a representative's decisions or actions in VCAT. 918

Ms Houseman of Senior Rights Victoria supported the concept of a personal monitor but suggested:

You might need to limit, though, the kinds of decisions that they would need to be informed of, because if they were to be informed of every single action that was undertaken by the attorney, that would be quite prohibitive. <sup>919</sup>

The Committee was presented with two possible approaches for empowering personal monitors. Firstly, the principal could be able to specify the monitor's role in the enduring power of attorney document. This approach was recently recommended in South Australia. A second view is that the legislation should require personal monitors to be notified of certain decisions or actions. For example, OPA suggested that representatives should be required to notify a personal monitor within 30 days if the representative financially benefits from the exercise of power under an enduring power of attorney. Dr Chesterman explained:

Of course if there is abuse of the enduring power of attorney, it is unlikely the person would send the letter and tell the interested person, so you have to make that

\_

Carers Victoria, Submission 65, 15.

John Chesterman, Transcript of evidence, above n 834, 4, 10, 13. See also Catherine Leslie, Transcript of evidence, above n 909, 5-6; Office of the Public Advocate, Submission 9, 17-18.

<sup>&</sup>lt;sup>915</sup> Carers Victoria, Submission 65, 13; Dahni Houseman, Transcript of evidence, above n 842, 6.

Carers Victoria, *Submission 65*, 13; John Chesterman, *Transcript of evidence*, above n 834, 10; Dahni Houseman, *Transcript of evidence*, above n 842, 6.

<sup>917</sup> Moreland Community Legal Centre Inc, Submission 51, 4; Seniors Rights Victoria, Submission 38, 35.

<sup>918</sup> Seniors Rights Victoria, Submission 38, 39; John Chesterman, Transcript of evidence, above n 834, 4; Office of the Public Advocate, Submission 9, 18.

Dahni Houseman, *Transcript of evidence*, above n 842, 6.

Advance Directives Review Committee, Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008), 46-47.

Office of the Public Advocate, Submission 9, 18.

a sub-penalty that would attach to an attorney receiving the benefit and not notifying interested persons.  $^{922}$ 

The Victorian Privacy Commissioner, Ms Helen Versey, warned that appointing a personal monitor potentially has privacy implications for both principals and representatives:

The issue of interested persons obviously raises other issues, because you then have another person having access to an individual's personal information. I think in that sort of case you would need to consider the sort of model you would want to set up. Firstly, who is appropriate to be an interested person and ensuring that the person who has the power of attorney is given notice that there has been an interested person appointed. I think people should be allowed to have a choice as to whether they want to take on the obligations of holding a power of attorney, and they need to know they are going to be overseen and there will be a third party having access to their information and how they are handling the power of attorney. 923

Some participants also observed that not all principals will have a family member or friend who they can appoint as a personal monitor. To address this it was suggested that representatives of a company such as a law firm or financial institution could be eligible to be appointed on a fee for service basis. 925

The Committee believes that allowing principals to nominate one or more personal monitors is an efficient, cost-effective and empowering way of providing additional protection for principals. The functions that monitors could play include receiving accounts, being notified of gifts to the representative or large gifts to others, being notified of plans to sell or deal with the principal's family home or other major assets and being notified of an intention to make a significant change to the principal's lifestyle. The Committee also believes that personal monitors could play a role in overseeing the appropriate activation and registration of an enduring power of attorney document. The personal monitor's potential functions in relation to activation are discussed later in this chapter, while registration is discussed in chapter eight.

The personal monitor should have clear standing to bring an application in VCAT regarding an enduring power of attorney. This is discussed later in this chapter.

The Committee does not believe that the role of personal monitors should be mandated by legislation. The Committee considers that the principal is best placed to determine the role that a personal monitor should play. To ensure that principals are informed about their ability to appoint a monitor, information about personal monitors should be included on enduring power of attorney forms and the information and educational materials for principals recommended in chapter four. Information that should be provided to principals should include the benefits of appointing a monitor, the types of people or organisations that may be suitable to appoint and the functions a monitor could perform.

John Chesterman, *Transcript of evidence*, above n 834, 10, 4, 13. See also Catherine Leslie, *Transcript of evidence*, above n 909, 5-6.

<sup>&</sup>lt;sup>923</sup> Helen Versey, Privacy Commissioner, Office of the Victorian Privacy Commissioner, *Transcript of evidence*, Melbourne, 14 December 2009, 3.

Debra Parnell, *Transcript of evidence*, above n 878, 7.

Tony Fitzgerald, *Transcript of evidence*, above n 908, 6-7. See also Advance Directives Review Committee, above n 920, 46-47.

The Committee also considers that representatives need to be well informed about the role of the monitor and the obligations that representatives have to the monitor. This information should be included in the resources for representatives that the Committee recommended in the previous chapter.

Finally, the Committee recognises that personal monitors will perform a very important role, overseeing the operation of a document which is otherwise not subject to independent scrutiny. To assist them in this role, the Committee believes that monitors should be provided with information and resources about their functions and, in particular, what to do if they consider that an enduring power of attorney has been misused.

#### **Recommendation 57: Personal monitors**

The Committee recommends:

- a) the Powers of Attorney Act provide that a principal may appoint one or more personal monitors to oversee the operation of an enduring power of attorney (financial) or an enduring power of attorney (guardianship)
- b) the Victorian Government produce simple, easy-to-understand information and educational materials for personal monitors.

### 7.2.5 Regulating gifts and other benefits

The Committee heard that it is not uncommon for representatives to make gifts to themselves or others. For example, solicitor Ms Lee commented:

Many family members of others appointed attorneys who are alleged to have abused their power often claim that the donor has contributed to the welfare of the family in the past and claim that therefore they can access funds to pay personal accounts such as school fees. 926

The Instruments Act is silent about the ability of representatives to confer gifts or other benefits on themselves and others. In contrast, the Guardianship and Administration Act makes it clear that administrators are able to make donations or gifts of a seasonal nature, or for a special event, to relatives or close friends of the person subject to the administration order. The value of the gift must be reasonable in the circumstances, particularly in regard to the represented person's financial situation. The administrator, or a charity with which the administrator is connected, may receive a gift or donation, however the administrator must notify VCAT in writing if the gift is \$100 or over in value.

<sup>&</sup>lt;sup>926</sup> Jeni Lee, Submission 57, 3.

<sup>&</sup>lt;sup>927</sup> Guardianship and Administration Act 1986 (Vic) s 50A.

## Case study 16: 'A significant proportion of her assets had been transferred by the attorney to himself' 928

'An elderly lady had appointed one of her late husband's friends as her enduring attorney. During a hospital admission she was diagnosed with dementia. The hospital suggested to the attorney that he needed to commence exercising his powers as attorney. Friends were alarmed when they discovered a significant proportion of her assets had been transferred by the attorney to himself. They also reported that the attorney had cameras installed in the woman's house to track her movements and her visitors. Initially, the woman was supportive of the attorney but when an application for revocation of the power of attorney was heard at VCAT it became apparent she had not been fully aware of the actions of the attorney. The attorney did not dispute that he had transferred money/shares to himself but said he did so on the basis of a long-standing gifting agreement. At the VCAT hearing the attorney agreed to refund some of the money and resigned as attorney. An independent administrator was appointed.'

Legislation in some other jurisdictions provides greater clarity about the rights of representatives to make gifts using a principal's funds. For example, in New South Wales a representative under an enduring power of attorney (financial) may only make a gift or confer a benefit if the power of attorney document specifically authorises that gift or benefit. Another approach is taken in Queensland and the United Kingdom, where reasonable gifts are permitted, similar to the regime currently in place for administrators in Victoria.

State Trustees supported the strict approach to gifts in the New South Wales legislation. <sup>931</sup> However, OPA did not agree, stating that 'many attorneys are close family members to whom the making of gifts may be appropriate'. <sup>932</sup>

OPA also felt that subjecting representatives to the same requirements as are currently in place for administrators was not a workable alternative. OPA indicated that requiring representatives to notify VCAT of gifts to themselves of \$100 or more 'would be administratively burdensome and would have limited value unless resources were given to VCAT to investigate reports of gifts'. <sup>933</sup> Instead, as noted in the previous section, OPA suggested a requirement that personal monitors be notified whenever the representative financially benefits from the exercise of power under an enduring power of attorney.

The Committee acknowledges that the current arrangements for providing gifts and other benefits are susceptible to abuse. It believes that the powers of attorney legislation should provide more guidance about when and to whom a representative may make a gift, and the appropriate value of gifts.

<sup>928</sup> Office of the Public Advocate, Submission 9, 8.

Powers of Attorney Act 2003 (NSW) ss 11-13.

Powers of Attorney Act 1998 (Qld) s 88; Mental Capacity Act 2005 (UK) c 9 s 12. See also Powers of Attorney Act 2006 (ACT) ss 38, 39.

<sup>931</sup> State Trustees Limited, Submission 58, 14.

Office of the Public Advocate, *Submission 9*, 17.

<sup>933</sup> Ibid.

The Committee recognises that many principals may wish to continue to make gifts to friends and family members after they have impaired decision-making capacity. It believes requiring all such gifts to be specifically stipulated in the enduring power of attorney document is too onerous. Instead the Committee recommends that a provision similar to that which currently regulates gifts by administrators should be introduced, allowing reasonable gifts to family and friends. Gifts and donations to representatives and charities with which representatives are involved should be permitted as long as they meet this criteria.

The Committee agrees that it would be administratively burdensome to require VCAT to approve larger gifts to representatives. Instead, it recommends that principals be encouraged to give personal monitors a role in monitoring gifts to representatives.

#### Recommendation 58: Regulating gifts

The Committee recommends the Powers of Attorney Act provide that a representative can make a gift of the principal's property, including to the representative, only if:

- the gift is reasonable in the circumstances, particularly in view of the principal's financial situation AND
- the gift:
  - is to a relative or close friend of the principal and is of a seasonal nature or for a special event OR
  - the gift is a type of donation that the principal made when he or she had capacity or might reasonably be expected to make.

## 7.2.6 Regulating dealings with major assets

Several participants in the Inquiry suggested there could be special restrictions introduced in relation to the representative's dealings with the principal's major assets, in particular property.

Mr Billings proposed that representatives who are appointed jointly and severally could be prohibited from acting severally (separately) when disposing of property or other significant assets. <sup>934</sup> However, Mr D Constantine, a participant in the Committee's Seniors' Forum, was sceptical about whether this would prevent abuse as, in his experience, one joint representative often exerts pressure over the other. <sup>935</sup>

Another suggestion was that VCAT ratification should be required for certain decisions such as selling the family home or disposing of assets worth over

D Constantine, *Transcript of evidence*, Melbourne, 30 March 2010, 13.

John Billings, Submission 37, above n 893, 8.

\$100 000. 936 Mr Tony Fitzgerald, Managing Director of State Trustees, told the Committee that when acting as an administrator State Trustees frequently seeks VCAT's guidance when dealing with major assets:

What we are able to do is go to VCAT and say that we want to dispose of the property and use those funds to pay the accommodation bond to enable a client to go into a certain level of care because of the fact that they cannot look after themselves ... We would go to VCAT and ask, 'Can you authorise us to go ahead and take this action?' We are protected then by VCAT.

We think that is a very important part of the current process that we use to protect ourselves in making that decision ... The other thing it does is also give the opportunity for a family member who might want to take issue with that decision to go and have themselves heard at VCAT under that same process. <sup>937</sup>

The Committee notes that many concerns in relation to powers of attorney, arise out of the use or disposal of a major asset, particularly the principal's home. The Committee is not convinced by the evidence received that requiring representatives appointed jointly and severally to act together to dispose of property or other major assets will significantly reduce abuse. The Committee notes that principals are currently able to specify that joint and several representatives must act together in particular transactions.

The Committee also believes that it would be too burdensome on both the representative and VCAT to require VCAT's approval for all dealings with major assets. It notes, however, that representatives currently have discretion to seek such approval if they wish.

The Committee considers that the personal monitor recommended earlier in this chapter can potentially provide valuable oversight in relation to dealings with significant assets. It suggests that the information and educational materials for principals should emphasise the benefit of requiring a personal monitor to be notified of such transactions.

## 7.2.7 Protecting gifts made under a will

The Law Institute of Victoria, State Trustees and the Trustee Corporations Association of Australia all called for legislative protection for gifts designated in the principal's will. 938

Currently the Guardianship and Administration Act requires an administrator who sells or disposes of an interest in property which is specifically gifted in a will, to keep a separate account and record of the money or other property received. <sup>939</sup> The proceeds will then flow to the beneficiary of the will. The New South Wales powers

Ocarers Victoria, Submission 65, 14; John Billings, Submission 37, above n 893, 8; State Trustees Limited, Submission 58, 16; Royal College of Nursing Australia, Submission 26, 2, 4.

Tony Fitzgerald, *Transcript of evidence*, above n 908, 6.

Trustee Corporations Association of Australia, *Submission 27*, att, 4; State Trustees Limited, *Submission 58*, 15-16; Law Institute of Victoria, *Submission 41*, 17; Lachlan Wraith, *Transcript of evidence*, above n 828, 3. See also Matthew Groves, 'Adeptly avoiding ademption' (2010) 84(08) *Law Institute Journal 36*, 40.

Guardianship and Administration Act 1986 (Vic) s 53.

of attorney legislation contains a similar provision in relation to dealings by representatives. <sup>940</sup>

The Committee believes that a representative has an obligation to uphold the wishes that a principal has set out in his or her will about the distribution of assets after his or her death. However, it notes that it will not be workable to introduce an obligation for representatives to protect the entitlements of beneficiaries without overturning the long-standing common law rule that representatives do not generally have a right of access to the principal's will, as was discussed in the previous chapter. In view of the limited evidence the Committee received on this issue, the Committee recommends that the Victorian Government consult further about whether legislation should protect the interests of beneficiaries under the principal's will and how this could effectively be achieved.

#### Recommendation 59: Protecting gifts made in the principal's will

The Committee recommends the Victorian Government conduct further consultation about whether the Powers of Attorney Act should protect the interests of beneficiaries under a principal's will.

# 7.2.8 Notifying activation of an enduring power of attorney document

As was noted in chapter five, it can be difficult to establish when an enduring power of attorney that is expressed to commence when a principal has impaired decision-making capacity is activated. At present it is usually the representative who decides that a principal has impaired decision-making capacity, thus activating such a document. The fact that capacity may fluctuate and applies to each decision to be made means that an enduring power of attorney may come in and out of operation.

Three Inquiry participants suggested that representatives should be required to give notice of intention to activate an enduring power of attorney to family members or other persons designated in the power of attorney document, for example, personal monitors.<sup>941</sup>

Another suggestion put forward by the Royal College of Nursing Australia is that a hearing should be held where all interested parties can attend and put forward any concerns before an enduring power of attorney can be activated in an aged care setting. <sup>942</sup>

As discussed earlier in this chapter, the personal monitors appointed by the principal in the UK must be notified when the document is lodged for registration, which is required before the power can be activated. The monitor then has the opportunity

204

Powers of Attorney Act 2003 (NSW) s 22.

Moreland Community Legal Centre Inc, *Submission 51*, 4; Seniors Rights Victoria, *Submission 38*, 39; Lachlan Wraith, *Transcript of evidence*, above n 828, 8.

<sup>&</sup>lt;sup>942</sup> Royal College of Nursing Australia, Submission 26, 4.

The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian Regulations 2007 (UK) reg 2, 6.

to object to the registration. Registration is considered in more detail in the next chapter.

Similarly, a recent Canadian report recommended that before a representative can commence acting under an enduring power of attorney, he or she must issue a formal notice to the principal and to any person designated in the enduring power of attorney document, or if nobody is designated, to the principal's immediate family. 944

The Committee agrees that requiring a representative to notify others when an enduring power of attorney document that states that it commences when a principal has impaired decision-making capacity is first utilised provides a safeguard against abuse. The Committee believes it is appropriate that the representative notify the personal monitor, the body responsible for registering power of attorney documents and any other persons nominated by the principal in the enduring power of attorney document.

Earlier in this chapter the Committee suggested that the principal should have to specify the role of the personal monitor in the enduring power of attorney document. However, the Committee believes that ensuring the appropriate activation of an enduring power of attorney that commences upon impaired decision-making capacity is such an important function that, if a principal has nominated a personal monitor, the legislation should require that person to be notified.

### Recommendation 60: Notification of activation of an enduring power of attorney

The Committee recommends the Powers of Attorney Act provide that prior to commencing to act under an enduring power of attorney (financial) or enduring power of attorney (guardianship) that states that it commences when a principal has impaired decision-making capacity, the representative must give notice of his or her intention to activate that enduring power of attorney to the personal monitor or monitors, any other persons designated in the power of attorney document and the body responsible for registering enduring power of attorney documents in Victoria.

#### 7.2.9 Performance bonds

It was suggested to the Committee that representatives could be required to take out a performance bond prior to commencing to act. Performance bonds are bonds issued by a financial institution following payment of a fee as security against any loss as a result of behaviour in carrying out a particular role. Mr Fitzgerald of State Trustees described these bonds as 'a bit like an insurance policy'. Administrators appointed in the UK can currently be ordered by a court to pay such bonds to the Public Guardian.

Western Canada Law Reform Agencies, above n 854, xviii.

Tony Fitzgerald, *Transcript of evidence*, above n 908, 8. See also David Davis, above n 850, 49.

Tony Fitzgerald, *Transcript of evidence*, above n 908, 8.

<sup>947</sup> Mental Capacity Act 2005 (UK) c 9 s19(9)(a); Department for Constitutional Affairs, United Kingdom, above n 841, para 8.45.

However, Professor Peteris Darzins, Professor of Geriatric Medicine, Monash University and Director of Geriatric Medicine, Eastern Health, noted that performance bonds may discourage the use of power of attorney documents. 948

The evidence examined by the Committee does not suggest that performance bonds are a workable option for reducing abuse of power of attorney documents. In particular, the Committee is concerned that such measures may discourage representatives from taking on this important role.

### 7.3 Criminal offences

Many participants in the Inquiry observed that the abuse of powers of attorney is often not perceived as or dealt with as a criminal offence.

Currently the Instruments Act does not set out any specific offences in relation to abuse of enduring powers of attorney (financial) or general (non-enduring) powers of attorney. The Guardianship and Administration Act provides general sanctions for a contravention of that Act which can be applied to breaches of enduring power of attorney (guardianship) provisions. <sup>949</sup>

There was strong support among Inquiry participants for creating specific offences for abuse of powers of attorney. <sup>950</sup> Alzheimer's Australia Victoria submitted that people who intentionally strip older people of their assets should be held accountable. <sup>951</sup> Barrister Mr Shepherd, who gave evidence on behalf of the Victorian Bar, told the Committee:

what we are looking at here in one sense is abuse of the worst kind. People are losing capacity. People are importuning them. People are taking advantage of them. These are matters which have always been regulated by the criminal law and treated seriously. 952

The Committee heard that creating criminal offences that specifically apply to abuse of powers of attorney would highlight the criminal nature of abuse to both representatives and the general community, as well as increasing awareness of representatives' roles and obligations. For example, former Victorian Public Advocate, Mr Julian Gardner stated:

A further approach to the abuse of financial powers could be to make it more explicit that the abuse of a power for the benefit of the attorney is a criminal offence. In some cases, attorneys benefit themselves in part because they believe

Guardianship and Administration Act 1986 (Vic) s 80.

Alzheimer's Australia Vic, Submission 32, 6 citing Anne-Louise McCawley et al, above n 830, 30.

206

Peteris Darzins, Professor of Geriatric Medicine, Monash University, and Director of Geriatric Medicine, Eastern Health, *Transcript of evidence*, Melbourne, 30 March 2010, 8. See also Alberta Law Reform Institute, *Enduring powers of attorney: Safeguards against abuse* Final report no. 88 (2003), 23.

<sup>950</sup> Hunt & Hunt, Submission 24, 4; Australian Bankers' Association Inc, Submission 55, 9; Stephen Taffe, Transcript of evidence, above n 878, 3; Lionel H Parrott, Submission 67, 2.

Alzheimer's Australia Vic, Submission 32, 6 citing Anne-Louise McCawley et al, above n 830, 30.

Robert Shepherd, *Transcript of evidence*, above n 852, 3.

<sup>953</sup> Ibid; Victoria Legal Aid, Submission 42, 4; Hunt & Hunt, Submission 24, 4; Stephen Taffe, Transcript of evidence, above n 878, 3; Lionel H Parrott, Submission 67, 2; John Billings, Submission 37, above n 893, 6-7; Seniors Rights Victoria, Submission 38, 41-42; Australian Bankers' Association Inc, Submission 55, 9;

that they are entitled or will become entitled to certain assets. This lack of clarity in their minds about the duty to act solely in the principal's best interest could be countered by providing for a criminal offence and by a warning about criminal liability in the documentation. <sup>954</sup>

Ms Jeter of the Elder Abuse Prevention Association also told the Committee that criminal sanctions would help prevent abuse of powers of attorney:

To me it would be one more step in prevention. It would be one more signal to those who want to go down that route or want to not properly use their authority under the law, meaning the power of attorney. If they knew that not only would they be punished or possibly be sanctioned under the civil but also now the criminal, then I think that would give a little bit of a wake-up call to the numerous abusers out there who are using that very legal document that I am sure you all hold very dear. I do, too. It would be a wake-up call that they could be not only civilly but criminally punished. 955

However, there are a number of issues associated with criminal offences in relation to powers of attorney. Firstly, such offences are extremely difficult to prove. Ms Robyn Mills, Acting Director of Civil Law Services at Victoria Legal Aid, told the Committee that there are few cases prosecuted because the criminal burden of proof is extremely high and it is very hard to prove crimes such as fraud. 956

Another issue is that victims of abuse of powers of attorney may also be reluctant to make complaints of a criminal nature against representatives, who are often family members. 957

Thirdly, some participants expressed concern that criminal sanctions may result in fewer people being willing to act as representatives. While recognising the need for 'more sanctions than currently exist', Mr Billings acknowledged that 'it would put a lot of people off. They think they are being asked to do something to help their aged parent, and then they start reading that they might go to jail or something if they slip up in some way.'958

Seniors Rights Victoria provided the Committee with helpful information about the use of offence provisions in the Queensland powers of attorney legislation. Based on this research, Seniors Rights Victoria concluded:

It would appear that the penalty provisions of the POA Act [Powers of Attorney Act 1998 (Qld)] are rarely utilised, and where they are considered, no breach has been found or if a breach has been found, that breach has been relied upon as a basis to revoke the donee and/or order compensation rather than to apply a penalty.

It could arguably be inferred from these findings that the penalty provisions have served their purpose in that the increase in awareness of the obligations and

Lillian Jeter, *Transcript of evidence*, above n 867, 6.

<sup>&</sup>lt;sup>954</sup> Julian Gardner, Submission 21, 4.

Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, *Transcript of evidence*, Melbourne, 17 December 2009, 5.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 30. See also Janet Wood, President, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 7-8; Financial Planning Association of Australia Ltd, *Submission 49*, 3.

John Billings, Transcript of evidence, above n 886, 6-7. See also Land and Property Management Authority, New South Wales, above n 904, 12-13.

penalties for breach of those obligations under the POA Act has resulted in very few findings of misuse and abuse of powers under the Act. 959

Participants in this Inquiry emphasised the importance of education of representatives about appropriate behaviour. The Australian Bankers' Association suggested that sanctions should be introduced in conjunction with educational materials for representatives which cover issues such as exercising powers in an honest way and managing conflicts of interests. 960

Those participants in favour of specific offences for abuse of powers of attorney suggested that these could either be included in the *Crimes Act 1958* (Vic) or in the powers of attorney legislation.

### 7.3.1 Specific offences in the Crimes Act

The Crimes Act currently provides a number of offences that can be used to prosecute abuse of powers of attorney, including theft and fraud. 961 Case study 17 provides an example of the use of existing provisions under the Crimes Act to prosecute a representative who stole money from her parents.

OPA proposed that the Crimes Act should be amended to make it clear that any fraudulent use of powers under an enduring power of attorney is theft if it materially benefits a representative. Dr Chesterman told the Committee:

Another way of guarding against the abuse of enduring powers of attorney is to clarify that such abuse will often constitute theft. We recognise that a specific legislative provision to this extent is technically superfluous — it is theft anyway — but the Office of the Public Advocate is of the view that it would help clarify that abuse of enduring powers of attorney often does amount to theft. On that basis we are recommending that the Crimes Act should be amended to have a specific provision saying that the fraudulent exercise of power under an enduring power of attorney constitutes theft. <sup>962</sup>

The Canadian Criminal Code provides a specific offence of theft by a person holding a power of attorney. 963

### 7.3.2 Specific offences in the powers of attorney legislation

A number of participants in the Inquiry supported introducing specific offences into the powers of attorney legislation.

208

<sup>959</sup> Seniors Rights Victoria, Submission 38S, 3.

Australian Bankers' Association Inc, Submission 55, 9. See also Seniors Rights Victoria, Submission 38, 41-42; Stephen Taffe, Transcript of evidence, above n 878, 3; Jeni Lee, Submission 57, 3-4; Victoria Legal Aid, Submission 42, 4; Hunt & Hunt, Submission 24, 4; Lionel H Parrott, Submission 67, 2; Alzheimer's Australia Vic, Submission 32, 6; John Billings, Transcript of evidence, above n 886, 6-7; Robert Shepherd, Transcript of evidence, above n 852, 3; Julian Gardner, Submission 21, 4.

<sup>&</sup>lt;sup>961</sup> Crimes Act 1958 (Vic) ss 74, 81, 82, 83, 83A.

John Chesterman, Transcript of evidence, above n 834, 4. See also Office of the Public Advocate, Submission 9, 18.

<sup>&</sup>lt;sup>963</sup> *Criminal Code*, RSC 1985, c C-46 s 331. See also s 332.

Several participants drew the Committee's attention to offences contained in the Queensland powers of attorney legislation. That legislation creates six offences

• dishonestly inducing the making or revoking of a general (non-enduring) power of attorney 964

- dishonestly inducing the making or revoking of an enduring power of attorney 965
- not acting honestly and with reasonable diligence 966
- exercising a revoked power<sup>967</sup>

specific to powers of attorney:

- prohibited use of confidential information <sup>968</sup>
- failing to keep property separate.<sup>969</sup>

There was support among Inquiry participants for the introduction of a number of offences based on the Queensland model, in particular, an offence of procuring a power of attorney by threat or deception, and an offence of failing to exercise power honestly and with reasonable diligence to protect the principal's interests.

In addition, Seniors Rights Victoria suggested there should be an offence of failing to keep accurate records. The Instruments Act currently provides that a representative under an enduring power of attorney (financial) must keep and preserve accurate records and accounts of all dealings and transactions made under the power. <sup>973</sup>

However, legislation in Western Australia and South Australia specifically makes it an offence to fail to keep accurate records. 974

The Committee believes that, given the degree of trust placed in representatives, and the lack of oversight of their role, it is appropriate for criminal sanctions to be imposed on representatives who do not behave appropriately. While it is currently a criminal offence to abuse a power of attorney, the creation of specific offences would highlight the criminal nature of abuse to both representatives and the general community.

Powers of Attorney Act 1998 (Qld) s 26.

Powers of Attorney Act 1998 (Qld) s 61.

Powers of Attorney Act 1998 (Qld) s 66.

Powers of Attorney Act 1998 (Qld) s 71.

Powers of Attorney Act 1998 (Qld) s 74A.

<sup>969</sup> Powers of Attorney Act 1998 (Qld) s 86.

The Victorian Bar, Submission 40, 3; Robert Shepherd, Transcript of evidence, above n 852, 3; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 4.

Robert Shepherd, *Transcript of evidence*, above n 852, 3; Seniors Rights Victoria, *Submission 38*, 41-42; Australian Bankers' Association Inc, *Submission 55*, 9.

<sup>&</sup>lt;sup>972</sup> Seniors Rights Victoria, Submission 38, 41-42.

<sup>973</sup> *Instruments Act 1958* (Vic) s 125D.

Guardianship and Administration Act 1990 (WA) s 107(1)(b); Powers of Attorney and Agency Act 1984 (SA) s 8.

### Case study 17: Internet scam victim stole from parents 975

At the start of 2008, Ms T began to assist her parents with cleaning, shopping and financial transactions. Her parents were 86 and had moved into an aged care facility in Wodonga. Ms T travelled from Melbourne each week to help her parents.

Ms T 'had a difficult and strained relationship with her parents'.

In April 2008 her parents appointed Ms T as their representative under an enduring power of attorney (financial). She used the power of attorney to set up internet banking to help pay her parents' bills. However, over the next six months she also used the internet banking to transfer approximately \$11 000 to her own account.

Ms T also used her father's credit cards to withdraw over \$31 000 from ATMs over three months. Ms T then made an internet application for another credit card in her father's name. Over the next two months, she used this credit card to purchase goods worth almost \$9 500.

Ms T was the victim of an e-mail scam. Around the time she began caring for her parents, she received an e-mail which said that if she paid a transfer fee she would receive a cheque for \$80 000. She paid the fee, but then more 'fees' and 'taxes' arose. In total, she sent about \$30 000 to the scammers, using her parents' money when her own savings ran out.

Ms T had intended to pay back her parents when she received the \$80 000 and 'didn't mean it to get this bad'.

In total, Ms T took \$51 914.81 from her parents.

When Ms T was interviewed by the police in February 2009, she made full admissions, and she pleaded guilty to theft and obtaining financial advantage from her parents when the case went to court. Her lawyer admitted that it was 'serious offending ... given the age and vulnerability of the victims. There was an abuse of trust.'

However, Ms T had since repaid approximately \$1500 to a credit card company, and set up a payment plan of \$50 a fortnight with them. She sent her parents two cheques for \$250, and was trying to obtain a loan so she could repay the rest of the money to her parents.

Magistrate Andrew Capell said, 'the offending, I accept, can be directly related to you becoming involved in a scam where you lost money ... through naivety - and also, I suppose, the belief that it might make your life a bit more pleasant ... All it did was, you dug yourself a much bigger hole.'

The magistrate convicted Ms T and gave her a suspended sentence of three months' imprisonment. He also placed her on a community based order for 12 months. He said that if she had not pleaded guilty, he would have sentenced her to five months' imprisonment.

<sup>&</sup>lt;sup>975</sup> R v Carmel Tracy (Unreported, VMC, Magistrate Capell, 18 January 2010, 30 March 2010).

The Committee believes that offences of abusing powers of attorney should be included in the proposed new Victorian Powers of Attorney Act. It believes that housing the offences within this legislation is consistent with the approach recommended in chapter three for stand-alone legislation covering all aspects of powers of attorney.

Based on the evidence received in this Inquiry the Committee recommends four offences that should apply to both enduring and non-enduring powers of attorney:

- procuring a power of attorney by threat or deception
- not acting honestly and with reasonable diligence to protect the principal's interests, having regard to the principal's expressed wishes
- knowingly exercising powers under a revoked power of attorney
- failing to keep accurate records.

In recommending the creation of these specific criminal offences, the Committee is cognisant of the risk of discouraging representatives from taking on this important role. The Committee believes that the offences will highlight the significance of the role that representatives are taking on and the obligations it carries. Information about the offences should be summarised on the power of attorney forms and also in the information materials for representatives recommended in chapter six.

The Committee acknowledges it is difficult to prove criminal offences in relation to powers of attorney. While there is unlikely to be an increase in prosecutions, the Committee believes that creating criminal offences will increase awareness of the responsibilities and have a deterrent effect.

#### Recommendation 61: Criminal offences for abuse of powers of attorney

The Committee recommends the Powers of Attorney Act provide the following offences:

- procuring a power of attorney by threat or deception
- not acting honestly and with reasonable diligence to protect the principal's interests, having regard to the principal's expressed wishes
- knowingly exercising powers under a revoked power of attorney
- failing to keep accurate records.

### 7.4 Providing redress for victims of abuse

The Committee heard that it is currently very difficult for victims of abuse of powers of attorney to recover assets or seek compensation.

Mr Billings informed the Committee that most power of attorney cases that come to VCAT involve the abuse of an enduring power of attorney (financial). Currently, VCAT's only recourse is to revoke the power. If the principal wishes to obtain other outcomes, he or she will need to instigate separate civil litigation. Ms Laura Helm, the Law Institute of Victoria's Policy Adviser, observed, 'At present people suffering financial loss must pursue complex and costly litigation if they are to recover any money lost.'

However, very few victims of abuse of powers of attorney launch civil litigation. There are a number of barriers to civil action, including the difficulty of proving abuse, the fact that a principal may not wish to compromise the relationship with a representative and that the principal may have impaired decision-making capacity. <sup>978</sup>

The cost of civil litigation was seen as a particularly significant obstacle. Notary and specialist lawyer Professor Phillip Hamilton submitted, 'It is no comfort to a defrauded family member that a right of recovery exists ... if they have no funds to pursue it ...'<sup>979</sup> In addition, the sums involved in the abuse of powers of attorney are usually insufficient to justify the cost of civil litigation.<sup>980</sup> Mr Fitzgerald of State Trustees told the Committee:

In our experience, by the time VCAT is involved it is too late. In cases of abuse, the attorney has already misappropriated the money and transferred it out of the name of the donor into their own name or family's name or whatever ... Sometimes we are appointed by VCAT as administrator to try to recover the situation, but in the majority of cases we would find that it is too late — the money has been misappropriated already and it is gone. Of course then the person has no money with which to pursue the person who misappropriated the funds; they have got no money to fund any sort of legal action ... <sup>981</sup>

Many participants in the Inquiry supported empowering VCAT to provide redress for people who are victims of abuse of power of attorney documents. In particular, there

John Billings, *Transcript of evidence*, above n 886, 3; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 4.

Laura Helm, *Transcript of evidence*, above n 891, 4. See also Terry Carney, *Transcript of evidence*, above n 872, 7; Rosslyn Monro, 'Elder abuse and legal remedies: Practical difficulties?' (2002) 81 *Reform* 42, 44-45.

Seniors Rights Victoria, Submission 38, 37; Belinda Evans, Transcript of evidence, above n 836, 5; Janet Wood, Transcript of evidence, above n 957, 7-8; Rosslyn Monro, above n 977, 44-45; Office of Senior Victorians, The mismanagement & misappropriation of older people's assets: How can the financial service sector help? A 'roundtable discussion' Elder abuse prevention project (2005), 9; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 29 citing Council on the Ageing SA, submission 77 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006), 8.

<sup>979</sup> Phillip Hamilton, Submission 7, 3. See also House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 29-30.

Debra Parnell, *Transcript of evidence*, above n 878, 8; Phillip Hamilton, *Submission 7*, 3; Terry Carney, *Transcript of evidence*, above n 872, 7; Laura Helm, *Transcript of evidence*, above n 891, 4; Rosslyn Monro, above n 977, 44-45.

Tony Fitzgerald, Transcript of evidence, above n 908, 4.

was support for allowing VCAT to award compensation, restitution or set aside transactions. 982 Alternatively, lawyer Ms Lee suggested that VCAT could make an order that could be lodged with a superior court. 983

Seniors Rights Victoria drew the Committee's attention to the *Sentencing Act 1991* (Vic). Under that Act if a court finds a person guilty of an offence, it can order the person to pay compensation for any loss or damage to property. Seniors Rights Victoria noted that 'VCAT does not however, have any power to make orders for the claw-back of the donor's funds where they have been misappropriated as a result of misuse of power conferred upon an attorney. 1985

In contrast, legislation in South Australia and Queensland make it clear that a representative may be liable to pay compensation to the principal. Several participants supported the introduction of a similar requirement in Victoria. For example, Ms Elizabeth Samra, a lawyer with Seniors Rights Victoria, told the Committee that in Victoria:

the donor or the person acting on behalf of the donor needs to go to another forum in order to seek compensation, whereas in Queensland there are provisions for breaches of the duties of the attorney, including fines ... and also remedies available – for example compensation where there has been misuse of power which is clearly enunciated in the legislation. <sup>987</sup>

Not all participants agreed that it was appropriate for VCAT to have power to award compensation. Mr Alistair Craig, Senior Corporate Lawyer with State Trustees Limited, informed the Committee that this:

would be perhaps putting in VCAT's hands powers that it is not used to wielding at the moment. You would obviously have questions of procedural fairness and so forth, so that might be a better process through the courts. In theory that might be something that could be dealt with perhaps, up to a certain quantum, because it is very difficult to actually recoup moneys. <sup>988</sup>

However, Mr Billings did not agree with those reservations. He stated that it 'would be relatively inexpensive and certainly convenient for VCAT, after inquiring into alleged abuse, to make relevant findings and appropriate orders about restitution'. He informed the Committee that VCAT's Civil Claims and Residential Tenancies Lists have the power and their members, the expertise, to award compensation. There

Seniors Rights Victoria, Submission 38, 40-41; Jeni Lee, Submission 57, 3; Law Institute of Victoria, Submission 41, 17; Toni Higgins, Transcript of evidence, above n 873, 5; John Billings, Submission 37, above n 893, 8; Murray Mallee Community Legal Service, Submission 22, 3; Lionel H Parrott, Submission 67, 2; Ministerial Advisory Council of Senior Victorians, Submission 48, 4; John Billings, Transcript of evidence, above n 886, 2.

Jeni Lee, Submission 57, 3.

<sup>984</sup> Sentencing Act 1991 (Vic) s 86.

Seniors Rights Victoria, Submission 38, 40.

Powers of Attorney and Agency Act 1984 (SA) s 7; Powers of Attorney Act 1998 (Qld) s 66.

Elizabeth Samra, Lawyer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 3. See also John Billings, *Submission 37*, above n 893, 7; Office of the Public Advocate, *Submission 9*, 18-19.

Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, *Transcript of evidence*, Melbourne,

<sup>22</sup> October 2009, 4.

John Billings, Submission 37, above n 893, 8. See also John Billings, Transcript of evidence, above n 886, 2.

is no limit on the amount of compensation that can be awarded by the Civil Claims List in relation to claims about goods or services. <sup>990</sup>

Mr Billings told the Committee that many members of the Guardianship List are also lawyers who sit in the Civil Claims and Residential Tenancies Lists. While Guardianship List members come from a wide range of backgrounds, Mr Billings advised that only those who are lawyers are able to make orders in relation to an enduring power of attorney (financial).

The Committee agrees that VCAT has the potential to provide a simple and cost effective mechanism for providing redress for victims of abuse of enduring powers of attorney. It therefore recommends that where VCAT finds that a power of attorney has been abused, it should be able to order the representative to compensate the principal for any loss. This will ensure that deserving victims are compensated, without having to pursue time-consuming and expensive separate legal action.

#### Recommendation 62: Compensation for victims of abuse

The Committee recommends the Powers of Attorney Act provide that where VCAT finds that an enduring power of attorney has been abused, VCAT may order a representative to compensate a principal for any loss.

### 7.5 VCAT's powers

As outlined in chapter two, VCAT's Guardianship List has a range of powers in relation to both enduring powers of attorney (financial) and enduring powers of attorney (guardianship). These include:

- revoking the appointment of a representative 993
- giving advisory opinions<sup>994</sup>
- varying or suspending the document. 995

Participants in this Inquiry highlighted that VCAT is an efficient and cost-effective jurisdiction and that it has the potential to play an increased role in relation to powers of attorney. In the previous section the Committee recommended empowering VCAT to award compensation to victims of abuse. Some participants suggested some additional powers for VCAT which are discussed in this section.

<sup>&</sup>lt;sup>990</sup> Fair Trading Act 1999 (Vic) ss 107, 108.

<sup>&</sup>lt;sup>991</sup> John Billings, Submission 37, above n 893, 8.

<sup>&</sup>lt;sup>992</sup> Ibid, att, 5.

<sup>993</sup> Instruments Act 1958 (Vic) s 125X; Guardianship and Administration Act 1986 (Vic) s 35D(1).

<sup>&</sup>lt;sup>994</sup> Instruments Act 1958 (Vic) s 125ZA; Guardianship and Administration Act 1986 (Vic) s 35E.

<sup>&</sup>lt;sup>995</sup> Instruments Act 1958 (Vic) s 125Z; Guardianship and Administration Act 1986 (Vic) s 35E(2).

### 7.5.1 Should VCAT deal with general (non-enduring) powers of attorney?

At present VCAT has no powers in relation to general (non-enduring) powers of attorney. Any actions regarding these powers need to be brought in the Supreme Court. Both the Law Institute of Victoria and the Victorian Bar suggested that VCAT's jurisdiction should be expanded, with VCAT having the same powers in relation to both general (non-enduring) powers of attorney and enduring powers of attorney (financial). However, these participants did not specify which part of VCAT should be able to deal with general (non-enduring) powers of attorney.

In the Committee's view the powers that VCAT's Guardianship List has in relation to enduring powers of attorney (financial) and enduring powers of attorney (guardianship) are in keeping with that List's mandate to protect vulnerable people with impaired decision-making capacity. The Committee does not believe it is appropriate for general (non-enduring) powers of attorney, which are often used in commercial transactions, to be dealt with by the Guardianship List.

However, the Committee believes there may be scope for matters relating to general (non-enduring) powers of attorney to be brought in another part of VCAT, in particular the Civil Claims List. This would give principals the opportunity to resolve disputes about these powers without expensive and time-consuming litigation in the Supreme Court. The Committee suggests that the Victorian Government should give further consideration to expanding VCAT's jurisdiction to include general (non-enduring) powers of attorney.

### 7.5.2 When should VCAT be able to revoke a representative's appointment?

When a principal has impaired decision-making capacity, only VCAT can revoke the appointment of a representative under an enduring power of attorney (financial) or an enduring power of attorney (guardianship). <sup>997</sup>

VCAT may revoke the appointment of a representative under an enduring power of attorney (financial) if it is satisfied that 'it is in the best interests of the donor to do so'. <sup>998</sup> The appointment of a representative under an enduring power of attorney (guardianship) may be revoked if VCAT is satisfied that the representative has 'not acted in the best interests of the appointor or has acted in an incompetent or negligent manner'. <sup>999</sup>

Neither the Instruments Act nor the Guardianship and Administration Act gives any guidance about what constitutes 'not acting in the best interests' of the principal or when it is in the best interests of the principal to revoke the document. Mr Billings

The Victorian Bar, Submission 40, app, 17-18; Law Institute of Victoria, Submission 41, 17; Laura Helm, Transcript of evidence, above n 891, 4. See also Phillip Hamilton, Submission 7, 4.

<sup>&</sup>lt;sup>997</sup> Instruments Act 1958 (Vic) s 125M(2); Guardianship and Administration Act 1986 (Vic) s 35D(1).

<sup>&</sup>lt;sup>998</sup> Instruments Act 1958 (Vic) s 125X(1).

<sup>&</sup>lt;sup>999</sup> Guardianship and Administration Act 1986 (Vic) s 35D.

explained the type of evidence that VCAT may consider in determining the principal's best interests:

The determination of a donor's best interests normally involves the Tribunal hearing allegations of an attorney's misuse of an EPA [enduring power of attorney]. History, including that of the attorney's use of the EPA, evidenced especially by financial records, is relevant. Sometimes a demonstrable conflict of interest or a particular transaction (perhaps where a donor has assigned property to the attorney in circumstances that cannot reasonably be explained) will lead the Tribunal to conclude that it is not in the donor's best interests for the power to continue. <sup>1000</sup>

Participants in this Inquiry called for consistent requirements for the revocation of appointments of representatives under both enduring powers of attorney (financial) and enduring powers of attorney (guardianship). 1001

In addition, Seniors Rights Victoria submitted that the circumstances in which VCAT can currently revoke a representative's appointment are 'vague'. <sup>1002</sup> It suggested that legislation should clarify that the grounds for revoking a representative's appointment include:

- failure to consult with the principal
- failure to consult with family members or other interested parties
- failure to notify family members or the personal monitor of the activation of an enduring power of attorney document
- failure to properly document or record transactions
- failure to act in the best interests of the principal. 1003

Seniors Rights Victoria also suggested that, in determining whether to revoke an instrument or appointment, VCAT should have to consider matters such as:

- the principal's wishes
- the implications of the revocation for the principal
- the wishes of any interested persons. 1004

These mirror the 'best interests' requirements currently in the Guardianship and Administration Act for making decisions about medical treatment on behalf of people with impaired decision-making capacity, including by representatives under enduring powers of attorney (guardianship). 1005

John Billings, *Submission 37*, above n 893, att, 4.

Seniors Rights Victoria, Submission 38, 38; Ministerial Advisory Council of Senior Victorians, Submission 48, 4.

Seniors Rights Victoria, Submission 38, 37.

<sup>&</sup>lt;sup>1003</sup> Ibid, 39.

<sup>&</sup>lt;sup>1004</sup> Ibid.

Guardianship and Administration Act 1986 (Vic) ss 38, 42U.

The Committee agrees that the proposed new Powers of Attorney Act should provide a single set of grounds for revoking the appointments of representatives under both enduring powers of attorney (financial) and enduring powers of attorney (guardianship).

The Committee believes that the grounds on which the appointment of representatives under these documents should be able to be revoked should be as simple and flexible as possible to allow VCAT to apply these in a broad range of situations whenever it is considered appropriate. Therefore the Committee recommends that VCAT should be able to revoke the appointment of a representative who has not acted in the best interests of the principal. It believes that the clarification of the duties of representatives, as recommended in chapter six, will provide additional assistance in determining whether a representative has or has not acted in the principal's best interests.

#### Recommendation 63: VCAT's power to revoke a representative's appointment

The Committee recommends the Powers of Attorney Act provide that VCAT may revoke a representative's appointment under an enduring power of attorney (financial) or an enduring power of attorney (guardianship) if satisfied that the representative has not acted in the best interests of the principal.

### 7.5.3 Who should be able to apply to VCAT?

The Instruments Act currently lists the people who can apply to VCAT for an order about an enduring power of attorney (financial). These are: the Public Advocate, the principal, a representative, and any other person whom VCAT is satisfied has a special interest in the principal's affairs. These people are also entitled to be given notice of any application made to VCAT about an enduring power of attorney. These people are also entitled to be given notice of any application made to VCAT about an enduring power of attorney.

Similarly, the Guardianship and Administration Act sets out who can make an application in relation to an enduring power of attorney (guardianship). In relation to an application to revoke an appointment, applications may be made by the representative, the Public Advocate, an alternative representative, an administrator or any other person with a special interest. There are no requirements to give notice of any applications under the Act in relation to enduring powers of attorney (guardianship).

Mr Billings suggested that examples of persons with a special interest in the principal's affairs are friends, relatives or, in some situations, the principal's accommodation provider if accommodation fees are unpaid. 1009

Two participants in the Inquiry argued that the legislation should clarify who is a person with a special interest in the principal's affairs. Seniors Rights Victoria

<sup>&</sup>lt;sup>1006</sup> Instruments Act 1958 (Vic) s 125V(2).

<sup>&</sup>lt;sup>1007</sup> Instruments Act 1958 (Vic) ss 125W(2)-(3).

<sup>&</sup>lt;sup>1008</sup> Guardianship and Administration Act 1986 (Vic) s 35D(2).

John Billings, Submission 37, above n 893, att, 5.

suggested, 'This will assist concerned family members or other concerned persons in understanding the procedures (and ascertaining if they have legal standing) for seeking the revocation of an enduring instrument once the donor has lost capacity.' 1010

The Financial Planning Association of Australia suggested that, in addition to the people currently listed in the two sets of legislation, the Act should also give standing to a close relative of the principal, a person named as a beneficiary in a will, or trust, a government agency with authority to protect the principal's welfare, the principal's carer, a person asked to accept the document and anyone else who can demonstrate sufficient interest in the principal's welfare. <sup>1011</sup>

The Committee considers that it is important that the legislation provides clear guidance about the range of people who may apply to VCAT for an order regarding an enduring power of attorney. It believes that the current list should be expanded to include a close relative and any personal monitor appointed by the principal. However, there is a danger in attempting to exhaustively list the people who may have standing and therefore the Committee recommends that VCAT still have some discretion to allow applications by persons with a special interest in the principal's affairs.

The Committee also considers that it is important that all relevant people are informed of any applications to VCAT regarding both an enduring power of attorney (financial) and an enduring power of attorney (guardianship). Therefore it recommends that notice of any applications or hearings should be required to be provided to this same group.

#### Recommendation 64: Clarifying who has standing to apply to VCAT

The Committee recommends the Powers of Attorney Act provide:

- a) the following people may apply to VCAT for an order about an enduring power of attorney (financial) or an enduring power of attorney (guardianship): the Public Advocate, the principal, the representative, a close relative of the principal, the personal monitor or monitors appointed by the principal, or any other person who VCAT is satisfied has a special interest in the principal's affairs.
- b) the following people must be given notice of an application to VCAT and any hearing of an application in relation to an enduring power of attorney (financial) or an enduring power of attorney (guardianship): the Public Advocate; the principal; the representative; a close relative of the principal; the personal monitor or monitors appointed by the principal; or any other person who VCAT is satisfied has a special interest in the principal's affairs.

<sup>&</sup>lt;sup>1010</sup> Seniors Rights Victoria, Submission 38, 39.

Financial Planning Association of Australia Ltd, Submission 49, 4.

## 7.5.4 What powers should VCAT have in relation to enduring powers of attorney that have been revoked?

Seniors Rights Victoria submitted that it is unclear whether VCAT's powers in relation to enduring powers of attorney (financial) apply to powers of attorney that have been revoked. It commented:

This ambiguity has significant implications in elder abuse cases, where it can be vital that there is an ability to take action against attorneys after the POA [power of attorney] is revoked. Given the significant barriers to reporting abuse in relation to POAs and bringing applications before VCAT, it is imperative that the donor, family, friends or other interested parties have the ability to seek orders from VCAT in respect of a POA which has been revoked, rather than having to initiate separate legal proceedings in a court. <sup>1012</sup>

Seniors Rights Victoria recommended that legislation or VCAT guidelines or practice notes clarify that VCAT's powers extend to powers of attorney that have been revoked. No other participants commented on this issue.

### 7.5.5 Should VCAT be able to refer cases of abuse for prosecution?

Several submissions suggested VCAT should have the power to refer a finding of abuse by a representative under an enduring power of attorney to the Director of Public Prosecutions where the abuse was criminal in nature. An alternative suggestion was that VCAT could be empowered to refer appropriate cases to Victoria Police for investigation and a report back.

The Committee acknowledges that VCAT's jurisdiction in relation to enduring powers of attorney may mean that it comes across cases of criminal abuse. The Committee agrees that in such circumstances, VCAT should be able to refer these cases on for prosecution if appropriate.

#### Recommendation 65: Referral by VCAT to Director of Public Prosecutions

The Committee recommends the Powers of Attorney Act provide that VCAT may refer cases of suspected abuse of enduring powers of attorney to the Director of Public Prosecutions.

<sup>&</sup>lt;sup>1012</sup> Seniors Rights Victoria, Submission 38, 40.

<sup>&</sup>lt;sup>1013</sup> Ibid, 40-41

Murray Mallee Community Legal Service, *Submission 22*, 3; John Billings, *Submission 37*, above n 893, 8.

Phillip Hamilton, Submission 7, 4.

## 7.6 Should applications about enduring powers of attorney be able to be brought in other courts?

Participants in this Inquiry were generally satisfied with the VCAT being the primary body responsible for hearing and determining matters relating to enduring powers of attorney. VCAT was lauded as being cost-effective and efficient. <sup>1016</sup>

Only one Inquiry participant suggested that applications in relation to enduring powers of attorney could be brought in another part of the justice system. Barrister, Mr Shepherd from the Victorian Bar told the Committee that parties could be allowed to elect to bring applications about powers of attorney in the Magistrates' Court of Victoria rather than VCAT:

From a cost-effective point of view, powers could be given to Magistrates to deal with these questions. Magistrates, we find, are well able to consider complex matters, but they are also experienced in problems which are commonplace within our community — family problems. They deal with family violence, domestic violence, restraining orders and the like. Perhaps we could make it VCAT and/or the Magistrates Court, perhaps with a section which would limit the power of the person or limit the ability of the court to order costs if a person decides to choose the Magistrates' Court rather than VCAT. <sup>1017</sup>

Evidence to this Inquiry suggests that the Guardianship List at VCAT provides an accessible and efficient service in relation to enduring powers of attorney. The Committee believes that expanding the powers of VCAT and, in particular, allowing it to award compensation as recommended earlier in this chapter will enable VCAT to provide even better outcomes. On the basis of evidence received, the Committee does not consider it is necessary to allow applications about powers of attorney to be brought in the Magistrates' Court. The Committee notes that applicants will continue to have the option to bring actions in relation to powers of attorney in the Supreme Court of Victoria.

### 7.7 Alternative dispute resolution

A court or tribunal may not always be the appropriate venue to resolve disputes involving powers of attorney. The *Older people and the law* report noted that a purely legal remedy for issues involving financial abuse within a family may not always be acceptable. The report found:

Pursuing a close relative through the courts and seeking to identify their behaviour publicly as criminal is not something many would want. Alternatives need to be found that provide for investigation of possible financial abuse and intervention short of criminal proceedings. <sup>1018</sup>

The report recommended increased funding for mediation and dispute resolution services to assist older people.

Robert Shepherd, Transcript of evidence, above n 852, 7.

<sup>&</sup>lt;sup>1017</sup> Ibid

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 831, 30-31.

Mr Billings informed the Committee that VCAT has a range of dispute resolution mechanisms available, such as compulsory conferences and mediation. He stated:

Mediation can take place at any time and can be helpful especially in cases where there are issues that may continue to arise from time to time, such as questions of access to financial information held by an attorney or consultation by an attorney with other interested persons.

As an example, siblings in conflict about the proper use of an EPA [enduring power of attorney] given by a parent to one of them may decide that instead of the applicant pursuing an application for revocation of the EPA the attorney may consent to directions that the attorney consult with siblings and regularly provide financial information to them. <sup>1020</sup>

Ms L Richmond, a participant in the Seniors' Forum who has been involved in proceedings at VCAT, told the Committee: 'VCAT is informal but it is still relatively formal when people are facing it ... It is certainly not the best answer for people to have to go to VCAT every time there is a problem.' In chapter six, the Committee also noted the potential benefits of alternative dispute resolution in terms of resolving disputes between multiple representatives.

Evidence to this Inquiry suggests that VCAT is currently playing a very effective role in quickly, informally and cost effectively resolving disputes about powers of attorney. The Committee notes that many of the issues that are raised in relation to powers of attorney involve disputes between family members. Informal dispute resolution mechanisms such as mediation may be useful in these circumstances. There are a plethora of such services currently operating in Victoria. The Committee believes that these services should be promoted in information materials for principals, representatives and personal monitors as a quick, informal and cost-effective way of resolving disputes about powers of attorney.

L Richmond, *Transcript of evidence*, Melbourne, 30 March 2010, 16.

<sup>&</sup>lt;sup>1019</sup> See Victorian Civil and Administrative Tribunal Act 1998 (Vic) ss 83, 88.

<sup>&</sup>lt;sup>1020</sup> John Billings, *Submission 37*, above n 893, att, 8.

Victorian Parliament Law Reform Committee, Inquiry into alternative dispute resolution and restorative justice (2009), 26-33.

### Case study 18: Solicitor uses power of attorney to steal from 'vulnerable' client 1023

Mr Halfpenny, a solicitor, ran a legal practice in Eltham as a sole practitioner. He had been described as a 'workaholic', and was suffering from 'burnout'. Despite this, Mr Halfpenny spent a large proportion of his time engaged in unpaid pro bono work which resulted in considerable financial difficulties for him. He was described by the court as 'significantly depressed and dependent on alcohol'.

Mr Halfpenny held an enduring power of attorney (financial) for Miss Elsie Connelly, 'a very elderly and vulnerable' client 'who trusted [him] implicitly'. Acting under the enduring power of attorney (financial), in August 1997 he withdrew the entire contents of Miss Connelly's pensioner security bank account, totalling over \$54 000. Mr Halfpenny deposited those funds into an account for a gift shop run by him and his wife.

Between November 1996 and February 1999 Mr Halfpenny stole from Miss Connelly on four other occasions, stealing over \$210 000 in total. These were part of a wider series of thefts from clients and deficient record keeping by Mr Halfpenny, though the thefts from Miss Connelly were the only ones involving the abuse of a power of attorney. In total, Mr Halfpenny's actions involved the fraudulent use of over half a million dollars of his clients' money.

Following Miss Connelly's death in January 1999, a beneficiary under her will made a claim through the Law Institute's Fidelity Fund. After an investigation by Victoria Police a number of criminal charges were laid, including eight counts of theft from clients including Miss Connelly. Mr Halfpenny pleaded guilty to all charges. He had his licence to practice law suspended for eight years by the Legal Profession Practice Tribunal.

In sentencing, Justice Redlich of the Supreme Court found there was 'very definitely' a nexus between Mr Halfpenny's depressive state and his crimes. However, it held that his depression did not prevent him from being aware of what he was doing. The court also took into consideration the fact that Mr Halfpenny subsequently fully repaid all of the amounts he had stolen, including to Miss Connelly's beneficiary. He sold his holiday and family homes in order to make restitution.

Justice Redlich said that these actions showed Mr Halfpenny was 'not a greedy person' and that he had 'genuine remorse, shame and embarrassment' for his actions. Nevertheless, Mr Halfpenny did 'abuse the trust of the clients repeatedly' and his Honour thought his case should act as a general deterrent by 'making an example to others'. Mr Halfpenny was sentenced to three years' imprisonment.

-

<sup>&</sup>lt;sup>1023</sup> R v Halfpenny [2003] VSC 308.

### Case study 19: Administrator to investigate money withdrawn by representative 1024

The principal lived in an aged care facility and was in very poor health. He had suffered a stroke resulting in 'significant cognitive impairment' that left him incapable of making 'informed and well reasoned decisions pertaining to finances and lifestyle'. The principal had two children: a son, A, and a daughter, B. A held an enduring power of attorney (financial) for his father. A had run his father's affairs for many years, and claimed that the principal was 'very happy with him doing so'.

Conflict arose between A and B as to how their father's financial affairs were being managed. A opposed allowing B's daughter to continue living rent free in a property owned by the principal. B was concerned that A had withdrawn \$86 500 from their father's account without explanation. This meant there was no money available to pay the principal's accommodation expenses. B applied to VCAT for the appointment of an administrator.

In April 2008 VCAT found that the enduring power of attorney held by A was not valid because it was not in the approved form required by the *Instruments Act 1958* (Vic) and was not witnessed properly. VCAT appointed A and B joint administrators for their father. However, it became apparent that there was 'little prospect of the brother and sister being able to cooperate' and in May 2008 VCAT revoked the administration order and appointed State Trustees Limited as an independent administrator.

In September 2008 A requested that VCAT appoint him as sole administrator of his father's financial affairs, claiming it had 'always been his father's wish that he should be the administrator'. VCAT refused to appoint A as the principal's administrator for a range of reasons, including that as administrator A would need to investigate his own behaviour in withdrawing money from the principal's bank account while acting as his representative. VCAT therefore reappointed State Trustees as the principal's administrator. State Trustees indicated that it was seeking legal advice to determine if any action should be taken to recover from A the funds he had withdrawn from the principal's bank accounts.

-

<sup>&</sup>lt;sup>1024</sup> SA (Guardianship) [2008] VCAT 2345.

## Chapter 8: A register of power of attorney documents?

In this chapter the Committee considers the case for and against a registration system for power of attorney documents. Overall the Committee finds that the benefits of a registration system outweigh the risks and recommends the introduction of compulsory registration for enduring power of attorney documents in Victoria. The Committee's proposed registration system has been carefully designed to ensure that principals' privacy is protected and that the cost of registration does not discourage people from making power of attorney documents.

## 8.1 The benefits and risks of a register of power of attorney documents

The terms of reference for this Inquiry specifically ask the Committee to advise whether there is a need to implement either voluntary or mandatory registration of power of attorney documents. The Committee heard that registration potentially offers a number of benefits, but there are also some significant risks associated with a registration system.

### 8.1.1 The benefits of registration

The majority of participants in this Inquiry were strongly supportive of registering power of attorney documents. They cited a wide range of benefits of a registration system.

The most commonly emphasised benefit of registration is that it makes it easy to establish the existence of a power of attorney, particularly an enduring power. Ms Jill Linklater, Policy and Advocacy Officer with Alzheimer's Australia Victoria, told the Committee she has come across cases 'where people go into nursing homes and there are powers of attorney but nobody knows'. Similarly, Ms Jenny Chapman, Chief Social Worker at Werribee Mercy Hospital, who gave evidence on behalf of Palliative Care Victoria stated:

If someone has a power of attorney in place and they are incompetent and they have no other family, we have no way of working out whether they have anything in place ... So we have no option but to go to VCAT and go through that process,

Jill Linklater, Policy and Advocacy Officer, Alzheimer's Australia Vic, Transcript of evidence, Melbourne, 22 October 2009, 5. See also Latrobe Regional Hospital, Submission 44, 3; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2; Ministerial Advisory Council of Senior Victorians, Submission 48, 4; Kris Spark, Manager, Seniors Information Victoria, Council on the Ageing Victoria, Transcript of evidence, Melbourne, 22 October 2009, 4; Seniors Rights Victoria, Submission 38, 32; Lucy Cordone, General Counsel, St Vincent's Hospital, Transcript of evidence, Melbourne, 14 December 2009, 7; David Goldberg, Solicitor and Senior Adviser, Australian Medical Association Victoria, Transcript of evidence, Melbourne, 17 December 2009, 5-6; Stephen Taffe, Legal Counsel, Alfred Health, Transcript of evidence, Melbourne, 14 December 2009, 3; Office of the Public Advocate, Submission 9, 22; Paul Zanatta, Manager, Community Care, Policy and Small Rural Health, Aged & Community Care Victoria, Transcript of evidence, Melbourne, 14 December 2009, 3; State Trustees Limited, Submission 58, 11-12.

which in a palliative care setting is really not appropriate unless it is really urgent. It puts too much of an added burden on families at that time of life ... <sup>1026</sup>

Even if a principal's family and friends are aware that an enduring power of attorney document exists, they may not be able to locate a copy when it is needed, which is often in a crisis situation. Mr Lachlan Wraith, who represented the Trustee Corporations Association of Australia, told the Committee:

when people progress towards losing capacity their personal affairs can become quite chaotic. They may have made appropriate arrangements but those documents can quite readily be lost in the course of their decline. It [registration] means that where capacity is lost and that is identified, the identity of any attorneys appointed can be quickly and accurately identified, enabling a smooth transition in the administration of that person's affairs. <sup>1027</sup>

Some Inquiry participants emphasised that registration would ensure that these important documents are stored in a safe location and cannot be lost or destroyed. State Trustees informed the Committee that many original legal documents were destroyed in the 2009 Black Saturday bushfires, which has 'led to avoidable administrative difficulties arising in some cases, adding to the tragedy for those victims and their families'. <sup>1028</sup>

Another advantage of registration is that it promotes recognition and acceptance of these documents. Third parties such as banks and hospitals are able to check that a power of attorney document is the most recent version and has not been revoked. Southern Health's submission stated that it is currently 'difficult for staff to know for certain whether a valid POA [power of attorney] is in place. This is especially difficult when more than one POA is presented to staff by different members of the family. In Tasmania where there is compulsory registration of general (non-enduring) powers of attorney and enduring powers of attorney (financial), third

Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 4.

Lachlan Wraith, Senior Manager, Trusts & Estates, Equity Trustees Ltd, Trustee Corporations Association of Australia, *Transcript of evidence*, Melbourne, 22 October 2009, 4. See also Kate Varty, *Submission 1*, 1; Stephen Taffe, *Transcript of evidence*, above n 1025, 3-4; Joyce Jeffs, *Transcript of evidence*, Melbourne, 30 March 2010, 20.

State Trustees Limited, Submission 58, 12. See also Lachlan Wraith, Transcript of evidence, above n 1027, 4; Department of Lands, New South Wales, Powers of attorney in New South Wales (2005), 5.

Seniors Rights Victoria, Submission 38, 32-33; Stephen Taffe, Transcript of evidence, above n 1025, 3; Letter from Pauline Bagdonavicius, Public Advocate, Office of the Public Advocate, Western Australia, to Chair, Victorian Parliament Law Reform Committee, 19 August 2009; State Trustees Limited, Submission 58, 11-12; Ministerial Advisory Council of Senior Victorians, Submission 48, 4; Law Institute of Victoria, Submission 41, 15; Australian Bankers' Association Inc, Submission 55, 3; Alfred Health, Submission 66, 3; Peter MacCallum Cancer Centre, Submission 20, 2; Office of the Public Advocate, Submission 9, 22; Southern Health, Submission 30, 2; D Constantine, Transcript of evidence, Melbourne, 30 March 2010, 6.

Southern Health, Submission 30, 2. See also Royal District Nursing Service, Submission 25, 2; Trustee Corporations Association of Australia, Submission 27, 3; David Davis, Elder Law Committee, Law Institute of Victoria, Transcript of evidence, Melbourne, 1 October 2009, 6; Palliative Care Victoria, Submission 70, 2; Stelvio Vido, Executive General Manager, Strategic and Support Services, Royal District Nursing Service, Transcript of evidence, Melbourne, 17 December 2009, 2; Alfred Health, Submission 66, 3; Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 8.

parties generally require evidence of a current search of the register before relying on these documents. 1031

The Committee heard that other benefits of registration include:

- providing certainty about the powers conferred on a particular representative 1032
- increasing the recognition of Victorian powers of attorney in other jurisdictions 1033
- forming a basis for providing targeted education to representatives <sup>1034</sup>
- providing a platform for oversight of representatives' activities.

In addition, a registration scheme would enable information to be gathered about the number of powers of attorney created. This would fill a significant evidence gap because, as noted in chapter two, there is currently a dearth of data about the level of use of powers of attorney in Victoria.

### Will a register protect principals from abuse?

Some participants in this Inquiry argued that a register would safeguard against abuse of power of attorney documents. The Office of the Public Advocate submitted that registration would be 'another formal threshold requirement that would discourage attorneys from abusing' powers of attorney. For example, registration may discourage a representative from using a power of attorney that has been revoked or acting outside his or her powers. Mr Wraith told the Committee that 'a central register would provide warnings where someone may seek to register a later enduring power of attorney which is prepared when the donor's capacity is in doubt'. 1041

Letter from Richard Fielding, Principal Examiner, Powers of Attorney/General Law, Land Titles Office, Tasmania, to Executive Officer, Victorian Parliament Law Reform Committee, 19 August 2009, 4.

Peter MacCallum Cancer Centre, Submission 20, 2.

Association of Independent Retirees (AIR) Ltd (Victorian Division), Submission 12, 5-6; Australian Bankers' Association Inc, Submission 55, 7.

Dale Reddick, Transcript of evidence, above n 1030, 8, 9; Janet Wood, President, Council on the Ageing Victoria, Transcript of evidence, Melbourne, 22 October 2009, 4.

Council on the Ageing Victoria, Submission 39, 4; Law Institute of Victoria, Submission 41, 15; Dahni Houseman, Policy Officer, Seniors Rights Victoria, Transcript of evidence, Melbourne, 22 October 2009, 4; Lauren Adamson, Acting Manager and Principal Solicitor, Seniors Rights Legal Clinic, Seniors Rights Victoria, Transcript of evidence, Melbourne, 22 October 2009, 6.

David Davis, *Transcript of evidence*, above n 1030, 10.

John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 5; Peter MacCallum Cancer Centre, Submission 20, 2; Law Institute of Victoria, Submission 41, 15.

Office of the Public Advocate, *Submission 9*, 22. See also John Chesterman, *Transcript of evidence*, above n 1037, 5; Letter from Martin John, Public Guardian and Chief Executive, Office of the Public Guardian, United Kingdom, to Chair, Victorian Parliament Law Reform Committee, 27 May 2010, 5.

State Trustees Limited, Submission 58, 11-12.

National Seniors Australia, Submission 54, 2.

Lachlan Wraith, Transcript of evidence, above n 1027, 4.

Not all participants agreed that registering power of attorney documents would protect principals from abuse. Commercial law firm Hunt & Hunt submitted, 'The registration of powers of attorney will not assist those vulnerable elderly members of our community who are at threat of being intimidated, abused and exploited'. <sup>1042</sup> In fact, Ms Imelda Dodds, Acting Chief Executive Officer of the New South Wales (NSW) Trustee and Guardian warned the Committee that registration may in fact increase abuse, as a public register 'could provide fresh incentives for organised crime to discover susceptible victims (i.e. vulnerable principals) or seek out and intimidate an attorney'. <sup>1043</sup>

There is very limited evidence about the impact of registration on the abuse of power of attorney documents. A recent Victorian study found that 'registration and supervision of EPAs [enduring powers of attorney] may help avert financial abuse' but cited research indicating abuse in 10% to 15% of registered enduring power of attorney documents. 1044

The Committee wrote to the agencies that register powers of attorney in other Australian jurisdictions requesting information about any research that has been conducted into the effect of registration on abuse. No such research appears to have been conducted. <sup>1045</sup>

However, the Committee did receive some anecdotal evidence about this issue. Mr Peter Maloney, Chief Executive of Tasmania's Public Trustee, wrote:

We have had a number of cases where there has been financial abuse of an elderly person. The registration of the document has not prevented that, and  $\dots$  cannot do so  $\dots$  I do not believe that mandatory registration has any effect on financial abuse, or detection or deterrence of fraud  $\dots$  1046

#### Further, Ms Dodds stated:

It is unlikely that the act of registration would have much impact on the prevalence of fraud or financial abuse. A Power of Attorney which appears validly executed on its face will be registered without further enquiry. Registration may in fact suggest

 $<sup>^{1042}\,</sup>$  Hunt & Hunt, Submission 24, 3.

Letter from Imelda Dodds, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010, 3.

Georgia Lowndes et al, *Financial abuse of elders: A review of the evidence*, Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2009), 4, 30 citing Denzil Lush, 'Taking liberties: Enduring powers of attorney and financial abuse' (1998) *Solicitors Journal* 808.

Letter from Max Locke, Registrar of Titles and Registrar of Water Allocations, Department of Environment and Resource Management, Queensland, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010, 3; Letter from Warwick Watkins, Registrar General, Land and Property Management Authority, New South Wales, to Chair, Victorian Parliament Law Reform Committee, 8 September 2009, 2; Letter from Peter Shoyer, Registrar-General, Land Titles Office, Department of Justice, Northern Territory, to Chair, Victorian Parliament Law Reform Committee, 26 February 2010, 2; Letter from Brian Marshall, Senior Manager, Registration and Client Services, Office of Regulatory Services, Australian Capital Territory, to Chair, Victorian Parliament Law Reform Committee, 16 February 2010, 2; Letter from Bruce Roberts, Registrar of Titles, Landgate, Western Australia, to Chair, Victorian Parliament Law Reform Committee, 21 August 2009, 3; Letter from Richard Fielding, above n 1031, 7. See also Letter from Martin John, above n 1038, 5.

Letter from Peter Maloney, Chief Executive Officer, Public Trustee, Tasmania, to Chair, Victorian Parliament Law Reform Committee, 3 November 2009, 2.

that the document is "official" and encourage third parties to place unfounded reliance on the scrutiny of the registration process. 1047

### 8.1.2 The risks of registration

While the vast majority of participants in this Inquiry supported a registration system for powers of attorney, the Committee heard there are a number of risks associated with registration.

Many participants expressed concern that a registration scheme may discourage members of the community from making powers of attorney. This could result in more people using informal approaches instead, potentially increasing financial abuse. <sup>1048</sup>

Inquiry participants identified three main ways that registration may act as a disincentive for people to make powers of attorney.

Firstly, many participants felt registration fees may have an impact on the uptake of powers of attorney. The Office of the Public Advocate's submission commented that 'many vulnerable Victorians would be reluctant and unable to pay any significant registration charge'. Similarly, Ms Dodds stated, 'If the aim [of registration] is to encourage people to make a power of attorney, any increase in costs will act as a deterrent.' 1050

Secondly, the Committee heard that principals may be reluctant to make powers of attorney if they have to make these arrangements public. Hunt & Hunt told the Committee that 'the donor may want to keep his/her arrangements under an enduring power of attorney private'. The Victorian Privacy Commissioner, Ms Helen Versey, told the Committee:

I feel one of the problems with compulsory registration might be in relation to individuals who are compulsorily required to disclose their personal information if they want to choose to enter into a power of attorney. If they are forced to provide information about themselves and the person they are appointing, it could act as a deterrent to some individuals making powers of attorney which are in their interests, such as medical or enduring powers of attorney, at a time when they should be encouraged to do it because they are capable of making those sorts of decisions for themselves ... <sup>1052</sup>

<sup>&</sup>lt;sup>1047</sup> Letter from Imelda Dodds, above n 1043, 3.

Georgia Lowndes et al, above n 1044, 4.

Office of the Public Advocate, *Submission 9*, 22. See also National Seniors Australia, *Submission 54*, 2; Hunt & Hunt, *Submission 24*, 3; State Trustees Limited, *Submission 58*, 12; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Submission 29*, 2.

NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales, Review of the *Powers of Attorney Act 2003* (2009), 18 as attachment to letter from Imelda Dodds, above n 1043.

Hunt & Hunt, Submission 24, 3. See also Trustee Corporations Association of Australia, Submission 27, att, 3; Terry Carney, Professor of Law, The University of Sydney, Transcript of evidence, Melbourne, 1 October 2009, 5; Lucie O'Brien, Policy Officer, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 7.

Helen Versey, Privacy Commissioner, Office of the Victorian Privacy Commissioner, Transcript of evidence, Melbourne, 14 December 2009, 4.

Thirdly, several participants indicated that the bureaucratic process involved with registering a power of attorney may discourage use of these instruments. Mr Alistair Craig, Senior Corporate Lawyer at State Trustees, advised against a registration system based on the United Kingdom (UK) model because 'it is almost akin to the person having to make the equivalent of an application for administration in the degree of bureaucracy around it'. The New South Wales Trustee and Guardian has also observed 'the more cumbersome the system the more likely it is that some people will simply not bother making a power of attorney'. 1054

A final issue associated with a registration system is the cost to the state, as establishing and maintaining a register is an expensive proposition. Alzheimer's Australia's submission drew the Committee's attention to a recent report in South Australia which found that 'registration of documents was considered to be expensive and would be problematic to implement, especially at a state level'. <sup>1055</sup>

### 8.1.3 Weighing the benefits and risks: Should a registration system be implemented?

The overwhelming majority of participants in this Inquiry supported the establishment of a registration system for powers of attorney. While issues were identified with registration, particularly privacy and cost, participants generally felt that these could be addressed through the strategic design of the registration system. The exception was the Law Institute of Victoria which recommended undertaking a comprehensive cost-benefit analysis prior to the introduction of any registration system.

A number of recent reviews in other Australian jurisdictions have considered whether powers of attorney should be registered. The 2007 House of Representatives Standing Committee on Legal and Constitutional Affairs report on *Older people and the law* recommended a national registration system in the long term and an

Alistair Craig, Senior Corporate Lawyer, State Trustees Limited, *Transcript of evidence*, Melbourne, 22 October 2009, 8. See also Jill Linklater, *Transcript of evidence*, above n 1025, 5.

NSW Trustee & Guardian, above n 1050, 13.

Alzheimer's Australia Vic, Submission 32, 3 citing Advance Directives Review Committee, Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008), 40.

Australian & New Zealand Society for Geriatric Medicine, Victorian Division, Submission 29, 2; Australian Bankers' Association Inc, Submission 55, 6-7; Seniors Rights Victoria, Submission 38, 32-34; Council on the Ageing Victoria, Submission 39, 5; Victoria Legal Aid, Submission 42, 5; Toni Higgins, Specialist Mental Health, Human Rights and Civil Law, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 2; Australian Medical Association Victoria, Submission 69, 3; Trustee Corporations Association of Australia, Submission 27, att, 3; Stelvio Vido, Transcript of evidence, above n 1030, 2; St Vincent's Hospital, Submission 19, 5; Peter MacCallum Cancer Centre, Submission 20, 2; Ministerial Advisory Council of Senior Victorians, Submission 48, 4; Shirley Glover, Submission 6, 2; Southern Health, Submission 30, 2; St Kilda Legal Service Co-Op Ltd, Submission 50, 4; Lucie O'Brien, Transcript of evidence, above n 1051, 7. See also Federation of Community Legal Centres (Victoria) Inc, Submission 47, 5; D Constantine, Transcript of evidence, above n 1029, 9; Jean Thomas, Transcript of evidence, Melbourne, 30 March 2010, 9; Gary Haley, Transcript of evidence, Melbourne, 30 March 2010, 10; Val Johnstone, Transcript of evidence, Melbourne, 30 March 2010, 11.

Julian Gardner, Transcript of evidence, Melbourne, 1 October 2009, 7-8; Lucie O'Brien, Transcript of evidence, above n 1051, 7-8. See also Federation of Community Legal Centres (Victoria) Inc, Submission 47, 5.
 Law Institute of Victoria, Submission 41, 16.

integrated state- and territory-based system in the short term. <sup>1059</sup> The report highlighted that this would protect the interests of principals, allow the monitoring of representatives' activities and facilitate increased recognition of power of attorney documents by third parties. A 2008 study of financial issues affecting older consumers in central Victoria also recommended registration of powers of attorney as a safeguard against abuse. <sup>1060</sup>

However, a number of state-level reports have rejected registration. For example, a review in South Australia found the costs of a state-based register outweigh the benefits and that the outcomes could be achieved in other ways. Instead, a voluntary national repository for power of attorney documents was recommended. A recent review in New South Wales in relation to enduring powers of attorney (financial) also recommended against compulsory registration, although another report in that state recommended further consideration of registration for enduring powers of attorney (guardianship).

The Committee is persuaded by the overwhelming evidence provided by participants in this Inquiry that a registration system for powers of attorney is needed. The Committee believes that a register of power of attorney documents will have a number of significant benefits, in particular:

- making it easy to establish the existence of a power of attorney document
- making it easy to locate a power of attorney document, which will help provide certainty about the powers conferred on the representative
- facilitating the validation of power of attorney documents.

The Committee believes these benefits warrant the introduction of a registration system, even in the absence of evidence that registration reduces abuse of power of attorney documents.

The Committee notes that some of the advantages offered by registration could potentially be achieved in other ways. These alternative measures were discussed in chapter four and include distributing promotional material such as wallet cards with information about a principal's power of attorney arrangements, or including this

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 99-100.

Frances Gibson, Financial & consumer credit issues for older consumers in central Victoria, report for Loddon Campaspe Community Legal Centre (2008), 8.

Advance Directives Review Committee, above n 1055, 39-40.

Land and Property Management Authority, New South Wales, Review of the Powers of Attorney Act 2003, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009), 12. See also Australian Law Reform Commission, Community law reform for the Australian Capital Territory: Third report - Enduring powers of attorney Report no 47 (1988), 14; Alberta Law Reform Institute, Enduring powers of attorney: Safeguards against abuse Final report no. 88 (2003), 22; Law Commission of New Zealand, Misuse of enduring powers of attorney Report 71 (2001), 18; Queensland Law Reform Commission, Assisted and substituted decisions: Decision-making by and for people with a decision-making disability Report no 49, volume 1 (1996), 158-159.

Standing Committee on Social Issues, Substitute decision-making for people lacking capacity Report 43, New South Wales Legislative Council (2010), 123-124.

information on patient medical records. However, these ad hoc approaches will not provide the same level of certainty as a comprehensive registration system.

The Committee agrees with stakeholders that concerns about cost, bureaucracy and privacy associated with a registration system may potentially discourage some people from making powers of attorney. However, it believes these issues can be resolved through careful design of the registration system.

### 8.2 What would a registration system look like?

There are a number of possible models for a register of power of attorney documents. This section discusses the key issues the Committee believes need to be fully considered in creating a registration system.

### 8.2.1 Should registration be national or state-based?

Throughout this report the Committee has identified issues with the state-based regulation of powers of attorney in an increasingly national environment. The majority of participants in this Inquiry who supported registration called for this to be at a national level, with many supporting the *Older people and the law* report recommendation that a state-based system be established as a precursor to a national scheme. <sup>1065</sup>

The Committee notes that the registration of power of attorney documents at a national level would alleviate current issues with the recognition of documents across jurisdictions, and make dealing with these documents easier for businesses that operate nationally, such as banks. In addition, the Committee recognises there would be considerable cost savings and efficiencies in establishing a single national register.

In chapter three the Committee discussed the benefits of the national harmonisation of power of attorney laws and recommended the Victorian Government support a coordinated national approach to powers of attorney through the Standing Committee of Attorneys-General. The development of a national register is potentially an important component of this work. However, the Committee acknowledges that the development of a national registration system is a longer-term solution.

The evidence to this Inquiry shows there is currently considerable need and demand for a registration system to help identify that power of attorney documents exist, assist to locate documents and to verify and validate documents. The Committee therefore recommends that the Victorian Government implement a registration system for powers of attorney. This state-based registration scheme has the potential to provide a best practice model for a future national registration system.

Seniors Rights Victoria, Submission 38, 32-34; Council on the Ageing Victoria, Submission 39, 5.

232

Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Submission 29*, 2; Victoria Legal Aid, *Submission 42*, 5; Trustee Corporations Association of Australia, *Submission 27*, att, 4; Laura Helm, Policy Adviser, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 3.

#### Recommendation 66: A register for power of attorney documents

The Committee recommends the Victorian Government:

- a) develop and implement a register for power of attorney documents
- b) through the Standing Committee of Attorneys-General, actively promote and support the development and implementation of a national register for power of attorney documents.

### 8.2.2 Should registration be voluntary or mandatory?

One of the threshold questions in designing a registration system is whether registration should be optional or compulsory.

Two Australian jurisdictions have implemented mandatory registration for powers of attorney. In Tasmania general (non-enduring) powers of attorney and enduring powers of attorney (financial) must be registered with the Recorder of Titles. <sup>1066</sup> In the Northern Territory enduring powers of attorney (financial) must be registered with the Registrar-General and general (non-enduring) powers of attorney must be registered if they are used for dealings in land. <sup>1067</sup> In most other Australian jurisdictions, powers of attorney are required to be registered if the power is to be used for dealings with land, such as selling or mortgaging a property. <sup>1068</sup> Most of these jurisdictions permit other financial powers of attorney to be registered voluntarily.

There is no provision for the registration of enduring powers of attorney (guardianship) documents in most states and territories. However, in Queensland and the Australian Capital Territory (ACT), these documents may be registered voluntarily. <sup>1069</sup> In the UK enduring powers of attorney that relate to both financial and personal matters must be registered before they can be activated. <sup>1070</sup>

The majority of participants who commented on this issue supported mandatory registration of power of attorney documents. The Committee heard that voluntary registration would not offer the same benefits as a mandatory system. For example, Mr Wraith, who gave evidence on behalf of the Trustee Corporations Association of

<sup>&</sup>lt;sup>1066</sup> Powers of Attorney Act 2000 (Tas) ss 4, 9, 16, 18, 30.

<sup>1067</sup> Powers of Attorney Act 1980 (NT) ss 7, 8, 13. See also Powers of Attorney Regulations 1982 (NT) reg 3.

Powers of Attorney Act 2003 (NSW) ss 51, 52; Land Title Act 1994 (Qld) s 132; Land Titles Act 1925 (ACT)
 s 130(2); Real Property Act 1886 (SA) ss 155-156.

Letter from Max Locke, above n 1045, 1; Public Advocate of the ACT, *The power to choose: Your guide to completing an enduring power of attorney* (2009), 14.

<sup>&</sup>lt;sup>1070</sup> Mental Capacity Act 2005 (UK) c 9 s 9(2), sch 1.

<sup>Jeni Lee, Submission 57, 3; Audrey Cooke, Submission 5, 1; Aged & Community Care Victoria, Submission 53, 3; National Seniors Australia, Submission 54, 2; Australian Bankers' Association Inc, Submission 55, 6; State Trustees Limited, Submission 58, 11-12; David Fara, Deputy President, Association of Independent Retirees (AIR) Ltd (Victorian Division), Transcript of evidence, Melbourne, 22 October 2009, 4; Royal College of Nursing Australia, Submission 26, 4; Stephen Taffe, Transcript of evidence, above n 1025, 3; Dale Reddick, Transcript of evidence, above n 1030, 8; Lucy Cordone, Transcript of evidence, above n 1025, 7-8; Jean Thomas, Transcript of evidence, above n 1056, 9; Gary Haley, Transcript of evidence, above n 1056, 10; Val Johnstone, Transcript of evidence, above n 1056, 10.</sup> 

Australia, told the Committee, 'If it is voluntary, there is a potential for low uptake, which would risk the system itself because there may be doubt about whether any reliance could be placed on the fact of registration'. Likewise, Ms Lucy Cordone, St Vincent's Hospital's General Counsel, told the Committee that the problem with an opt-in system is:

it will not be a comprehensive register then, so it will not be given the level of comfort that we as lawyers like when we advise the doctors — 'Do they have one or don't they have one?'. ... I think the doctors would want certainty. <sup>1073</sup>

Evidence received by the Committee suggests that existing voluntary registration systems have low take-up rates. For instance, in both Queensland and the ACT most registrations of powers of attorney relate to dealings with land, where registration is a requirement under the legislation.<sup>1074</sup>

Dr John Chesterman, Manager of Policy and Education at the Office of the Public Advocate, suggested that mandatory registration could be introduced from a certain date, with registration of documents created prior to that date being voluntary. <sup>1075</sup>

While there was widespread support for mandatory registration, many Inquiry participants acknowledged that this potentially raises privacy issues. The Mental Health Legal Centre expressed strong reservations about compulsory registration warning it would dramatically decrease the number of enduring powers of attorney made 'as donors may not wish to have their private affairs subject to public scrutiny'. Ms Catherine Leslie, the Centre's Legal and Policy Officer, said the Centre would support a voluntary system where people are provided with information about the effect of registering, particularly in relation to their private information and can then make their own choice to register or not. 1077

There has not been any research into the impact of compulsory registration on the uptake of powers of attorney. However, Mr Maloney of the Public Trustee in Tasmania advised the Committee that compulsory registration has not had any effect on the uptake and use of enduring powers of attorney in that state. <sup>1078</sup>

The Committee believes that a registration system will only meet its objectives of assisting to locate, verify and validate power of attorney documents if registration is mandatory. The Committee notes compulsory registration raises privacy issues but believes that these can be addressed through well-considered design of the system, in particular limiting the information that can be accessed on the register and who can access that information. These issues are discussed further later in this chapter.

\_

Lachlan Wraith, Transcript of evidence, above n 1027, 4. See also Office of the Public Advocate, Submission 9, 22.

Lucy Cordone, Transcript of evidence, above n 1025, 7.

Letter from Max Locke, above n 1045, 3; Letter from Brian Marshall, above n 1045, 2.

<sup>&</sup>lt;sup>1075</sup> John Chesterman, *Transcript of evidence*, above n 1037, 10.

Mental Health Legal Centre Inc, *Submission 59*, 11. See also Trustee Corporations Association of Australia, *Submission 27*, att, 3; Helen Versey, *Transcript of evidence*, above n 1052, 4.

Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, *Transcript of evidence*, Melbourne, 14 December 2009, 5.

Letter from Peter Maloney, above n 1046, 2.

### 8.2.3 What documents should be registered?

In this section the Committee considers the types of documents that should be required to be registered under a mandatory registration system.

The first issue is whether all powers of attorney should be registered, or whether registration should be limited to enduring powers of attorney. Ms Versey suggested that requiring only enduring powers to be registered was one way of reducing the privacy impacts of a register:

You might therefore want to consider distinguishing between the different types of power of attorney and whether or not you want to make them compulsory. For example, if my husband was overseas and we were selling our house and he wanted to give me a very temporary power of attorney so I could deal with the legal necessities, would you really see any reason for that to be registered on a register? That is the sort of situation where a power of attorney is created compared to medical powers of attorney where others such as hospitals may need to know whether or not there is a valid power of attorney. 1079

Other participants observed that enduring powers of attorney are the most susceptible to abuse and supported mandatory registration of these powers. <sup>1080</sup>

In the Northern Territory it is compulsory to register enduring powers of attorney (financial) and general (non-enduring) powers of attorney may be registered voluntarily. However, the Territory's Registrar-General advised the Committee that most general (non-enduring) powers of attorney in that jurisdiction are still registered, as financial institutions usually request this. <sup>1081</sup>

There was also strong support among Inquiry participants for registering enduring powers of attorney (guardianship). These documents are often required in an emergency situation, when a principal has impaired decision-making capacity and is not able to tell others that he or she has a power of attorney or where the document is located.

Another approach taken in some other jurisdictions is to require registration only when a document is used for dealings with land. In New South Wales, for example, both general (non-enduring) and enduring powers of attorney (financial) must be registered if they are used to deal with land. Powers not used for dealing with land may be registered voluntarily. As the case studies set out throughout this report demonstrate, many of the concerns about abuse of powers of attorney relate to property, and in particular dealings with the principal's home, which is often his or her most valuable asset. However, only one participant in this Inquiry suggested that this middle-ground approach be adopted in Victoria.

Helen Versey, *Transcript of evidence*, above n 1052, 4.

See, for example, Office of the Public Advocate, Submission 9, 22-23; John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, 7; Ministerial Advisory Council of Senior Victorians, Submission 48, 4.

Letter from Peter Shoyer, above n 1045, 2.

Paul Zanatta, *Transcript of evidence*, above n 1025, 3; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 5; Office of the Public Advocate, *Submission 9*, 22-23.

Powers of Attorney Act 2003 (NSW) ss 51-52.

<sup>1084</sup> Credit Union Australia Ltd, Submission 18, 2.

A further issue that needs to be considered is whether the revocation of a power of attorney document would need to be registered, in addition to the document creating the power. Many participants in this Inquiry assumed that revocations would need to be registered because if they were not the register would be incomplete and potentially outdated. Participants suggested that requiring revocations to be registered addresses the issue of a principal creating multiple powers of attorney, with the registration of a subsequent document automatically revoking a previous one. 1086

However, not all registration systems require revocations to be registered. In New South Wales, registration of revocations is not compulsory, even where the registration of the power of attorney document is. 1087 A recent review in that state found concern that the costs associated with the compulsory registration of subsequent revocations could dissuade people from making powers of attorney in the first place. The review concluded that the risk that powers of attorney will continue to be used after they have been revoked is better addressed through improved education for representatives. 1088

The Committee agrees that requiring registration of enduring powers of attorney (financial) and enduring powers of attorney (guardianship) would provide considerable benefits. Compulsory registration of these documents would mean that these documents could be located and validated when a principal has impaired decision-making capacity. In order to ensure that the register is accurate and up-to-date the Committee also recommends that all revocations of these documents must be registered.

The Committee believes that registration is not required for general (non-enduring) powers of attorney. These documents operate while the principal still has capacity and is able to provide advice about the validity of the document. However, principals should be able to register these documents if they choose. The Committee also notes that there is the potential for other powers of attorney and related documents that are not being considered as part of this Inquiry to be included in the register, for example, enduring powers of attorney (medical treatment) and advance statements.

#### Recommendation 67: Mandatory registration of enduring powers of attorney

The Committee recommends the Powers of Attorney Act require all documents creating and revoking enduring powers of attorney (financial) and enduring powers of attorney (guardianship) to be registered.

### Recommendation 68: Voluntary registration of general (non-enduring) powers of attorney

The Committee recommends the Powers of Attorney Act permit documents creating and revoking general (non-enduring) powers of attorney to be registered.

Palliative Care Victoria, *Submission 70*, 1; Hunt & Hunt, *Submission 24*, 3; John Billings, *Submission 37*, above n 1080, 7; Julian Gardner, *Transcript of evidence*, above n 1057, 9.

St Vincent's Hospital, Submission 19, 6; Australian Bankers' Association Inc, Submission 55, 8-9.

Land and Property Management Authority, New South Wales, above n 1062, 7.

<sup>&</sup>lt;sup>1088</sup> Ibid, 13.

### 8.2.4 When should a power of attorney document be registered?

There are two possible points at which a power of attorney document could be registered. It could be required to be registered either at the time it is created or at the time it is activated.

In Tasmania the relevant legislation states that a representative's action under a power of attorney is not valid unless the document is registered. However, the Committee heard there is some disagreement in that state about when documents should be registered. Mr Maloney of the Public Trustee told the Committee that, in his view, enduring powers of attorney (financial) are not required to be registered until they are activated. However, he advised that the majority of Tasmanian lawyers take the view that the document must be registered when it is made. He suggested that if Victoria introduces a registration system, only active powers should be registered because 'otherwise you will have a register full of documents which are never going to be utilised and which of course is no use to anyone'. 1090

Ms Versey opined that registration at the time of activation of an enduring power of attorney document may better protect a principal's privacy. She stated:

The other issue if you are considering the difference between compulsory and voluntary registration is distinguishing between those who have the capacity to choose and those who do not. For example, you might say a power of attorney does not need to be registered unless the donor has become incapable of making decisions for themselves. <sup>1091</sup>

The Office of the Public Advocate, Seniors Rights Victoria and the Council on the Ageing all supported a requirement that a document must be registered before it can be used, but did not specifically mention at what point that should occur. 1092

Mr John Billings, Deputy President of the Guardianship List at the Victorian Civil and Administrative Tribunal (VCAT), observed that allowing registration at the activation stage can create problems, with representatives delaying registration for as long as possible to avoid the registration fee. He told the Committee about evidence that in Tasmania 'some attorneys have got cold feet and decided not to register at all and not to take up the power that the donor intended them to have'. <sup>1093</sup> State Trustees' submission also highlighted that delaying registration until the document is required 'may negate a number of advantages' of a registration system. <sup>1094</sup>

The Committee acknowledges that requiring registration at the time a power of attorney document is created may result in principals' personal information being available, and the principal incurring a cost, even though the document may never be

237

Powers of Attorney Act 2000 (Tas) s 16. See also Mental Capacity Act 2005 (UK) c 9 s 9, sch 1.

Letter from Peter Maloney, above n 1046, 2-3.

Helen Versey, Transcript of evidence, above n 1052, 4.

Office of the Public Advocate, Submission 9, 22-23; Seniors Rights Victoria, Submission 38, 32-34; Council on the Ageing Victoria, Submission 39, 5.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Transcript of evidence*, Melbourne, 1 October 2009, 5.

State Trustees Limited, Submission 58, 12.

used. However, the Committee believes that unless registration is required at the time the document is created, many of the benefits of its proposed registration system will be lost. In particular, such a system would not ensure that documents can be located or verified in an emergency situation. The Committee believes that the legislation should provide a reasonable period after a document is created within which registration must take place. The Committee suggests that a period of 60 days would be appropriate.

#### Recommendation 69: The time of registration

The Committee recommends the Powers of Attorney Act require all documents required to be registered under the Act to be registered at the time they are created.

### 8.2.5 What happens if an enduring power of attorney document is not registered?

Some participants, while generally supporting registration, were concerned that a failure to register could potentially invalidate a document, frustrating a principal's attempt to plan for his or her future. Ms Belinda Evans, Senior Advocate with Elder Rights Advocacy, told the Committee:

I would like to think my document was effective from the moment it is supposed to become effective, and if I happen to forget to register it, or I am not aware of the need to register it, I would hope that would not impact on its effectiveness. <sup>1095</sup>

However, some participants argued that documents creating and revoking powers of attorney should not be valid unless they are registered. They commented that this would reduce the potential for abuse, as well as providing greater certainty for organisations that act on these documents.

The Victorian Bar suggested that a court or VCAT could have the power to declare an unregistered document valid. 1097

In order for the registration system proposed in this chapter to be effective, the Committee considers that the legislation must make it clear that any act performed by a representative is not legally effective unless the enduring power of attorney document is registered.

The Committee notes that there is a risk that requiring all enduring power of attorney documents to be registered when they are executed may create injustice where a person fails to register a document. The Committee believes this can be addressed by empowering VCAT to extend the time for registration of a document, where it is of the view that it is valid. This is consistent with the Committee's recommendation in

<sup>1097</sup> The Victorian Bar, Submission 40, 3.

\_\_\_

Belinda Evans, Senior Advocate, Elder Rights Advocacy, *Transcript of evidence*, Melbourne, 22 October 2009, 5. See also Association of Independent Retirees (AIR) Ltd (Victorian Division), *Submission 12*, 6; David Fara, *Transcript of evidence*, above n 1071, 4.

St Vincent's Hospital, *Submission 19*, 5-6; Seniors Rights Victoria, *Submission 38*, 34; Robert Shepherd, Barrister, The Victorian Bar, *Transcript of evidence*, Melbourne, 22 October 2009, 4.

chapter four that VCAT have the power to declare valid power of attorney documents that do not meet formal requirements.

### Recommendation 70: The effect of non-registration

The Committee recommends the Powers of Attorney Act provide:

- a) any act performed under an enduring power of attorney (financial) or enduring power of attorney (guardianship) has no legal effect unless the document is registered
- b) VCAT may extend the time for a document creating or revoking an enduring power of attorney (financial) or enduring power of attorney (guardianship) to be registered if it believes the document is valid.

# 8.2.6 Should a representative have to notify the registration body when an enduring power of attorney is activated?

As noted in chapters five and seven there is currently no formal process for activating an enduring power of attorney that commences operation when the principal has impaired decision-making capacity. The exact point of activation is often difficult to determine, because a principal's capacity may fluctuate, and a principal may be able to make some decisions but not others.

In section 8.2.4 the Committee recommended that enduring powers of attorney documents should be registered at the time they are made. However, this creates a situation where unactivated enduring powers of attorney documents are on the register. Unless there is a system for notifying activation, persons checking the register will have no way of knowing whether a registered document is active or not.

Several participants in this Inquiry supported people searching the register being able to access information about whether a power of attorney document has been activated. Others suggested that a medical assessment showing impaired decision-making capacity should have to be lodged with the registration body in order to activate an enduring document.

In the UK, enduring powers of attorney are required to be registered before they can be activated. However, they can be registered at any time prior to activation. The representative is not required to give any notice of activation.

Stephen Taffe, *Transcript of evidence*, above n 1025, 3; Alfred Health, *Submission 66*, 3; David Davis, *Transcript of evidence*, above n 1030, 10; John Billings, *Transcript of evidence*, above n 1093, 5-6. See also Letter from Peter Maloney, above n 1046, 3.

St Kilda Legal Service Co-Op Ltd, *Submission 50*, 2-3; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 3; Jeni Lee, *Submission 57*, 4; Seniors Rights Victoria, *Submission 38*, 29.

In chapter seven of this report the Committee recommended that a representative be required to notify key people, including the personal monitor and the registration body, when an enduring power of attorney document is first used, as a safeguard against abuse. The Committee believes that requiring the registration body to be notified of a document's activation also ensures that, where appropriate, people accessing the register are able to see whether a particular enduring power is currently in use. The Committee considers that this notification should be in written form, but, as discussed in chapter five, should not require formal medical evidence of impaired decision-making capacity.

### 8.2.7 Accessing the register

A register of power of attorney documents potentially has significant privacy implications. There was considerable debate among participants in this Inquiry about what information should be included in the register and who should have access to that information. These issues are explored in this section.

### What information should be included in the register?

The registers of powers of attorney in other Australian jurisdictions are public registers. <sup>1100</sup> Most registers can be searched by registration numbers, the name of the principal and representative/s and, in some cases, the type of powers granted. Many registration bodies will provide a copy of the power of attorney document upon payment of a fee.

Participants in this Inquiry expressed a variety of views about the information that should be accessible through the register. Suggestions included:

- the principal's name <sup>1101</sup>
- the principal's date of birth 1102
- the representative's name 1103
- the registration number <sup>1104</sup>
- the type of power granted 1105
- any conditions on the power 1106

Letter from Brian Marshall, above n 1045, 2; Letter from Peter Shoyer, above n 1045, 1; Letter from Max Locke, above n 1045, 2; Letter from Bruce Roberts, above n 1045, 2; Letter from Richard Fielding, above n 1031, att, 2.

Stephen Taffe, *Transcript of evidence*, above n 1025, 3; John Chesterman, *Transcript of evidence*, above n 1037, 11.

<sup>&</sup>lt;sup>1102</sup> John Chesterman, *Transcript of evidence*, above n 1037, 11.

Carers Victoria, Submission 65, 15; Stephen Taffe, Transcript of evidence, above n 1025, 3; Lionel H Parrott, Submission 67, 1; Alfred Health, Submission 66, 3.

Stephen Taffe, *Transcript of evidence*, above n 1025, 4; Helen Versey, *Transcript of evidence*, above n 1052, 4. Lionel H Parrott, *Submission 67*, 1; Australian Bankers' Association Inc, *Submission 55*, 7.

- the date the power was activated 1107
- the names and contact details of people entitled to notice of significant decisions under the power. 1108

However, many participants were concerned about making private information available through a register. Ms Leslie of the Mental Health Legal Centre emphasised that privacy issues were particularly important to the Centre's clients: 'many people have had an experience where their personal information has been used in ways that could be used against them'. <sup>1109</sup> Similarly, Ms Dorothy Trezise, a participant in the Committee's Seniors' Forum told the Committee, 'I certainly would not want that [personal] information to be publicly available. I would very much like to choose who has it and where it is.' <sup>1110</sup>

Ms Versey raised concerns about personal information such as date of birth being made available on a register because of concerns about identity theft and fraud. She told the Committee, 'the more information you put about an individual on a publicly available register then the more problems you may create in terms of people being able to use that information for unlawful purposes'. 1111

In Ms Versey's view, 'if you are setting up a register, the fact that in order to register you might collect quite a lot of information does not mean to say it has to be available to third parties'. 1112

Ms Versey suggested that one way that privacy could be promoted is through a document verification system. Rather than disclosing personal information, verification systems only confirm information provided by the searcher. Such a scheme is currently used Australia-wide to validate certificates for births, deaths, marriages and changes of name. <sup>1113</sup> Ms Versey described how a verification system for powers of attorney could operate:

the power of attorney might be given a number, and if the person who wants to check whether a power of attorney is in existence provides a name, number, type of power of attorney and whether it is current, they can then receive a yes or no answer. 1114

Stephen Taffe, Transcript of evidence, above n 1025, 3; Alfred Health, Submission 66, 3; Australian Bankers' Association Inc. Submission 55, 7: National Seniors Australia, Submission 54, 2.

Stephen Taffe, *Transcript of evidence*, above n 1025, 3; Alfred Health, *Submission 66*, 3; David Davis, *Transcript of evidence*, above n 1030, 10; John Billings, *Transcript of evidence*, above n 1093, 5-6.

<sup>1108</sup> Carers Victoria, Submission 65, 15; Stephen Taffe, Transcript of evidence, above n 1025, 3; D Constantine and L Richmond, Submission 75, 1.

<sup>1109</sup> Catherine Leslie, *Transcript of evidence*, above n 1077, 4.

Dorothy Trezise, Director, United Way Geelong, *Transcript of evidence*, Melbourne, 30 March 2010, 20. Cf D Constantine, *Transcript of evidence*, above n 1029, 20.

Helen Versey, Transcript of evidence, above n 1052, 3. See also L Richmond, Transcript of evidence, Melbourne, 30 March 2010, 21.

Helen Versey, *Transcript of evidence*, above n 1052, 4. See also Julian Gardner, *Transcript of evidence*, above n 1057, 7-8.

Department of Justice & Attorney General, New South Wales, *CertValid user guide* Document version 1.0 (2010), 1.

Helen Versey, *Transcript of evidence*, above n 1052, 4-5. See also Julian Gardner, *Transcript of evidence*, above n 1057, 7-8.

Alternatively, Ms Versey suggested a layered approach whereby different people have access to different levels of information in the register. This is discussed further in the next section.

Mr Stephen Taffe, Legal Counsel at Alfred Health, also acknowledged privacy concerns and commented, 'We think that a public register for powers of attorney could list limited information, maybe even their names, and that might be sufficient to overcome a lot of the problems.' He suggested that the list of registered nurses, which has subsequently been replaced by a national register of health practitioners, could be used as a model. That register is searchable online by surname or registration number. 1116

Several other Inquiry participants raised the protection of personal information as an issue, but did not offer suggestions about how it could be addressed. 1117

### Who should have access to the register?

The power of attorney registers in other Australian jurisdictions do not restrict who may access information on the register. A different approach is taken in the UK. Anyone can conduct a basic search of the UK register upon payment of a fee. Information received after a 'first tier' application includes:

- the principal's name
- the principal's date of birth
- the representative's name
- the type of power granted ('property and affairs' or 'personal welfare')
- the date the power of attorney was made
- the date the power of attorney was registered
- the date the power of attorney was revoked (if applicable)
- the date the power of attorney expires (if applicable)
- whether representatives are appointed jointly or jointly and severally
- any conditions on the power. 1119

Stephen Taffe, *Transcript of evidence*, above n 1025, 4.

Australian Health Practitioner Regulation Agency, *Registers of practitioners*, <a href="http://www.ahpra.gov.au/en/Registration/Registers-of-Practitioners.aspx">http://www.ahpra.gov.au/en/Registration/Registers-of-Practitioners.aspx</a>, viewed 14 July 2010.

Latrobe Regional Hospital, *Submission 44*, 3; Australian Bankers' Association Inc, *Submission 55*, 6-7; State Trustees Limited, *Submission 58*, 11-12.

Letter from Richard Fielding, above n 1031, 2; Letter from Brian Marshall, above n 1045, 2; Letter from Max Locke, above n 1045, 2; Letter from Peter Shoyer, above n 1045, 1; Letter from Bruce Roberts, above n 1045, 2.

Office of the Public Guardian, United Kingdom, LPA109 - Office of the Public Guardian Registers (2009), 7-8.

However, if more information is required, the applicant must complete a 'second tier' search application form, detailing why he or she needs the information. There is no additional fee for a second tier search, but applications must set out the specific information required, why this information is needed and the steps taken to obtain the information from the principal or another source. The release of further information is subject to the Public Guardian's discretion. 1120

Many participants in this Inquiry advocated for restricting access to information contained in the register in order to protect principals' privacy. Ms Versey told the Committee, 'I cannot honestly see why the general public needs to have access to that information or why the world at large needs to have access to it.' She suggested a tiered approach, with different people having different levels of access to the register:

you could take a layered approach whereby only certain, very limited people would have access to more information. If you want to verify there is a power of attorney, you might get a very limited amount of access, but if, say, you are a doctor who needs to know rather more information, then you may be entitled to have more access. 1122

There was considerable support among participants for a layered approach for access to the register. The Royal District Nursing Service suggested that members of the public could have part access to the register, while other bodies such as VCAT, the police, the Office of the Public Advocate and health care providers could have full access. Similarly, Mr Taffe of Alfred Health told the Committee that the organ donor scheme may be a good model:

certain types of institutions such as hospitals could be given access for their purposes, possibly also financial institutions, and those organisations could be subject to confidentiality obligations, and other members of the public could apply to have access. 1124

Other participants suggested that access to the register could be limited to particular organisations such as financial institutions, <sup>1125</sup> aged care services <sup>1126</sup> and health care providers. <sup>1127</sup> However, Ms Sarnia Birch, Acting Deputy Manager of Regional Offices at Victoria Legal Aid, expressed concern about an approach that lists the

Letter from Martin John, above n 1038, att, 3; Office of the Public Guardian, United Kingdom, LPA109 - Office of the Public Guardian Registers (2009), 9-10; Directgov, How the Office of the Public Guardian works, <a href="http://www.direct.gov.uk/en/Governmentcitizensandrights/Mentalcapacityandthelaw/Makingarrangements">http://www.direct.gov.uk/en/Governmentcitizensandrights/Mentalcapacityandthelaw/Makingarrangements</a> incaseyoulosementalcapacity/DG\_185924>, viewed 14 July 2010.

Helen Versey, *Transcript of evidence*, above n 1052, 5. See also Julian Gardner, *Transcript of evidence*, above n 1057, 7.

Helen Versey, *Transcript of evidence*, above n 1052, 4.

Royal District Nursing Service, Submission 25, 5. See also Lachlan Wraith, Transcript of evidence, above n 1027, 4.

Stephen Taffe, *Transcript of evidence*, above n 1025, 3. See also Alfred Health, *Submission 66*, 3.

Dahni Houseman, *Transcript of evidence*, above n 1035, 8; Lachlan Wraith, *Transcript of evidence*, above n 1027, 4.

<sup>&</sup>lt;sup>1126</sup> Aged & Community Care Victoria, Submission 53, 3.

Stelvio Vido, Transcript of evidence, above n 1030, 2; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 5; Paula Chatfield, Legal Counsel, St Vincent's Hospital, Transcript of evidence, Melbourne, 14 December 2009, 7; Aged & Community Care Victoria, Submission 53, 3; Palliative Care Victoria, Submission 70, 1; Dahni Houseman, Transcript of evidence, above n 1035, 8; Lachlan Wraith, Transcript of evidence, above n 1027, 4; L Richmond, Transcript of evidence, above n 1111, 21.

classes of people who can access the register because 'once you start making a list of those sorts of things you are bound to exclude someone who should be on it. I would think that as long as there is some rule whereby people have to establish their bona fides, then that is the way they access it'. Such an approach is used by the Registry of Births, Deaths and Marriages, with members of the public required to show an 'adequate reason' for wanting the information.

Another suggestion was that, at the time the principal creates the power of attorney document, he or she could nominate the people and organisations who are permitted to access information about the document on the register. 1130

### How should the register be searchable?

Some participants in this Inquiry called for the registration system for power of attorney documents to be online to ensure that it is readily accessible when required. For example, St Vincent's Hospital submitted, 'Medical and Legal Practitioners will need access online to the register 24/7.' Mr Billings of VCAT suggested that another advantage of online registration is that it would enable automatic updates or alerts to be sent to relevant agencies such as banks if a power was varied or revoked. 1132

Ms Versey observed that online technology means that registers potentially pose much greater risks to people's privacy. She stated:

It is a recurrent theme in inquiries to our office that organisations when they decide to put public registers online, if they do it without notice to those whose personal information is contained in the registers, then they often do it without thought about what information is contained in the registers. Once information is available online without any restriction it is capable of being collected, collated with other publicly available information and misused by people who have no legitimate use for it. Any creation of a new searchable, online register needs to take these matters into account. 1133

Searches of power of attorney registers in several Australian jurisdictions are able to be conducted online, often through land titles databases. <sup>1134</sup> In some jurisdictions, online searches are only available to registered users of those databases, such as solicitors and banks, with members of the public having to make applications for access to the documents. <sup>1135</sup>

Sarnia Birch, Acting Deputy Manager, Regional Offices, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 5. See also David Davis, Transcript of evidence, above n 1030, 10; D Constantine and L Richmond, Submission 75, 1. Cf The Institute of Legal Executives (Victoria), Submission 33, 2.

Births, Deaths and Marriages Registration Act 1996 (Vic) s 45. See also Victorian Registry of Births, Deaths and Marriages, Access Policy, Department of Justice, Victoria (2007), 7.1, 7.2.

Toni Higgins, *Transcript of evidence*, above n 1056, 5.

St Vincent's Hospital, Submission 19, 5; John Billings, Submission 37, above n 1080, 7. See also Credit Union Australia Ltd, Submission 18, 3; Robert C Bolch, JP, Submission 3, 12.

John Billings, Submission 37, above n 1080, 7.

Helen Versey, *Transcript of evidence*, above n 1052, 2.

Letter from Richard Fielding, above n 1031, att, 2.

Letter from Bruce Roberts, above n 1045, 2; Letter from Peter Shoyer, above n 1045, 1.

### Accessing the register – The Committee's view

The Committee strongly believes that protecting principals' privacy must be the highest consideration in the development of a register of power of attorney documents. The Committee agrees that a layered approach to accessing the register would safeguard principals' privacy, while at the same time ensuring the register meets its aims of providing information about power of attorney documents when needed.

The Committee notes that different stakeholders need different levels of access to information about power of attorney documents. A verification system would meet the needs of many service providers, particularly banks, which need only to confirm the authenticity of a sighted document. However, this would not meet the requirements of organisations which need to establish the existence of a power of attorney document or identify a representative in an emergency situation. The Committee therefore suggests that health care providers could have a higher level of access to information on the register.

The Committee believes it is appropriate for other members of the community to have access to information on the register if they have a clearly demonstrated interest in that information. The Committee envisages that a clearly demonstrated interest arises in situations such as a child trying to establish whether a parent who has impaired decision-making capacity has made an enduring power of attorney document.

Finally, the Committee notes concerns about making information about powers of attorney available online. It believes that information about power of attorney documents should not be generally available online. However, there would be benefit in a secure online system for verifying power of attorney documents, as well as providing information to health care providers, as this information is often required in emergencies.

### Recommendation 71: Accessing information on the register

The Committee recommends the Victorian Government, in implementing the registration system for power of attorney documents, ensure that the privacy of principals' information is protected by providing:

- a) a document verification system for approved service providers such as banks
- b) greater access to information on the register for health care providers
- c) access to information on the register only to members of the community who are able to clearly demonstrate that they have an interest in that information.

### 8.2.8 Registration costs

As noted earlier in this chapter, there was considerable concern among participants in this Inquiry about the potential for registration fees to discourage people from making powers of attorney. 1136 Several participants emphasised the need for low fees and concession rates for people of limited financial means. 1137

Other participants suggested there should be no registration fees at all. <sup>1138</sup> Mr Billings proposed that registration could be free, with larger charges for searching the register instead. While noting that such costs would ultimately still be passed on to consumers, he felt this arrangement would not discourage people from making powers of attorney. <sup>1139</sup>

Some participants were of the view that members of the community would be prepared to pay reasonable registration costs in order to reap the benefits of having enduring powers of attorney registered. Mr David Fara, Deputy President of the Association of Independent Retirees, told the Committee, 'I believe that enduring powers is such an important document that people would be prepared to pay for the little bit of satisfaction and ease of mind that registration would give them.' Similarly, Ms Val Johnstone, a participant in the Committee's Seniors' Forum commented:

it is like with any measure: it is not just about selling the cost, it is about selling the benefit. It comes back to people being informed as to why they have a power of attorney. It is about the benefit of having registration. 1141

Fees for registering power of attorney documents in other Australian states and territories currently range from \$92.00 in the ACT to \$124.20 in Queensland. 1142

The Committee notes that registration fees may act as a disincentive for members of the community to make powers of attorney. Therefore, in implementing the registration scheme suggested in this report, the Committee recommends the Victorian Government ensure that registration fees are kept to a minimum, with concession rates or fee waivers available where appropriate.

### **Recommendation 72: Registration fees**

The Committee recommends the Victorian Government, in implementing the registration system for power of attorney documents, ensure that registration fees are kept to a minimum, with concession rates or fee waivers available.

Office of the Public Advocate, *Submission 9*, 22-23. See also National Seniors Australia, *Submission 54*, 2; Hunt & Hunt, *Submission 24*, 3; State Trustees Limited, *Submission 58*, 12; Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Submission 29*, 2.

Office of the Public Advocate, Submission 9, 22-23; The Institute of Legal Executives (Victoria), Submission 33, 2; John Billings, Submission 37, above n 1080, 7; Alistair Craig, Transcript of evidence, above n 1053, 10.

St Kilda Legal Service Co-Op Ltd, Submission 50, 4; Catherine Leslie, Transcript of evidence, above n 1077, 4.
 John Billings, Submission 37, above n 1080, 7. See also Robert Shepherd, Transcript of evidence, above n 1096, 5.

David Fara, *Transcript of evidence*, above n 1071, 3.

<sup>&</sup>lt;sup>1141</sup> Val Johnstone, *Transcript of evidence*, above n 1056, 10.

Letter from Brian Marshall, above n 1045, 2; Letter from Max Locke, above n 1045, 2.

### 8.2.9 Where should the register be located?

The Committee received a number of different suggestions about where a Victorian register of powers of attorney could be located. These all proposed giving this function to an existing agency, rather than creating a new registration body.

Several participants thought the Office of Public Advocate was an appropriate organisation to house a register of powers of attorney. <sup>1143</sup> Ms Lucie O'Brien, the Federation of Community Legal Centres' Policy Officer, explained why her organisation supported the Office of the Public Advocate (OPA) hosting a registry:

It seems to have the existing expertise and experience with some of the relevant issues. It just seemed that it would be a logical addition to the current powers it has. ... it would require a significant resource — an addition of resources. ... We certainly would not want the OPA to be saddled with all of these new responsibilities without being adequately resourced to carry them out. 1144

However, Dr Chesterman of the Office of the Public Advocate indicated that his organisation did not wish to take on this role. He suggested that the Land Titles Office or the Registry of Births, Deaths and Marriages were better suited to housing a register of powers of attorney as, 'They have achieved administrative efficiency and they are used to dealing with applications and registrations ...' 1145

Other participants suggested VCAT<sup>1146</sup> and the Probate Office<sup>1147</sup> as potential locations for the registry.

In other Australian jurisdictions, powers of attorney registers are generally attached to the agency administering the register of land titles. This is also the case in Queensland and the ACT where enduring powers of attorney (guardianship) may be registered. In the UK, the register is administered by the Public Guardian, an office established by the *Mental Capacity Act 2005* (UK).

The Committee considers that the most appropriate location for a register of powers of attorney in Victoria is the Registry of Births, Deaths and Marriages. This organisation has extensive experience in managing complex registration systems, including about sensitive matters. Housing the register within an existing agency will reduce the costs of establishing and maintaining a register for powers of attorney.

247

Peter MacCallum Cancer Centre, Submission 20, 2; St Kilda Legal Service Co-Op Ltd, Submission 50, 4; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 5.

Lucie O'Brien, *Transcript of evidence*, above n 1051, 8. See also Lauren Adamson, *Transcript of evidence*, above n 1035, 6-7.

John Chesterman, *Transcript of evidence*, above n 1037, 7. See also Robert Shepherd, *Transcript of evidence*, above n 1096, 5; Joyce Jeffs, *Transcript of evidence*, above n 1027, 5.

Ronald T Jones, JP, Submission 2, 3; Robert Shepherd, Transcript of evidence, above n 1096, 5; Alistair Craig, Transcript of evidence, above n 1053, 10. See also John Billings, Transcript of evidence, above n 1093, 7.

Jeni Lee, Submission 57, 3. Cf Robert Shepherd, Transcript of evidence, above n 1096, 5.

Letter from Brian Marshall, above n 1045; Letter from Max Locke, above n 1045; Registration of Deeds Act 1957 (ACT) s 3; Land Title Act 1994 (Qld) s 33.

<sup>&</sup>lt;sup>1149</sup> Mental Capacity Act 2005 (UK) c 9 s 58(1).

### Recommendation 73: Location of the register

The Committee recommends the Powers of Attorney Act provide that the Registry of Births, Deaths and Marriages maintain the register of power of attorney documents.

# 8.2.10 Should a registration body play a role in checking that documents are correctly executed?

A registration system could potentially play a role in ensuring that power of attorney documents are correctly executed. The registration bodies in other Australian jurisdictions informed the Committee that they perform checks to ensure that documents have been correctly executed, in particular, that formality requirements are met. A significant number of documents are rejected in these jurisdictions, ranging from 5% in Queensland to an estimated 30% in the Northern Territory. 1151

Several participants in this Inquiry supported the registration body performing a basic check of documents lodged for registration. Dr Chesterman of the Office of the Public Advocate acknowledged this would have resource implications but commented:

I think there needs to be a rudimentary check that there are two witnesses and it has been completed. I certainly do not think the registering body should then ring up the witness and say, 'Did you go through and check?'. But, yes, there should be a rudimentary check to see that on the face of it it has been completed. <sup>1152</sup>

Similarly, State Trustees submitted, 'If appropriately staffed and empowered, a registry could provide a mechanism through which the executed document's conformity with the prescribed formalities (and thus its prima-facie validity) could be reviewed and confirmed.' 1153

However, Mr Maloney of the Tasmanian Public Trustee noted that a registration body will not be in a position to check that the principal had the necessary capacity to create a power of attorney document, which is the main issue in relation to the proper creation of power of attorney documents. 1154

Given the potentially serious consequences of a power of attorney document not being in the correct form, the Committee believes the registration body should conduct a basic check of each document lodged for registration to ensure that all formality requirements are met. If an error is found, the principal should have an opportunity to correct and re-lodge the document. This process will ensure that, as far as possible, principals' plans for their future are upheld.

-

Letter from Brian Marshall, above n 1045, 2; Letter from Max Locke, above n 1045, 3; Letter from Warwick Watkins, above n 1045, 1; Letter from Bruce Roberts, above n 1045, 2; Letter from Peter Shoyer, above n 1045, 2; Letter from Richard Fielding, above n 1031, 4-5. See also *Mental Capacity Act 2005* (UK) c 9 sch 1 cl 11; Letter from Martin John, above n 1038, att, 4-5.

Letter from Peter Shoyer, above n 1045, 2; Letter from Max Locke, above n 1045, 3.

John Chesterman, *Transcript of evidence*, above n 1037, 11.

State Trustees Limited, Submission 58, 11.

Letter from Peter Maloney, above n 1046, 2.

The Committee notes a check that an enduring power of attorney document complies with formal requirements will not be able to identify fraud, such as if a principal was forced to sign the document or had impaired decision-making capacity at the time the document was made. The Committee's recommendations in chapter four to strengthen witnessing requirements aim to address these issues.

### Recommendation 74: Registration body to check power of attorney documents

The Committee recommends the Powers of Attorney Act require the registration body to conduct a basic check to ensure that a document creating or revoking a power of attorney meets formal requirements before registering the document.

### 8.2.11 Who should be entitled to notice of registration?

In the UK principals are able to nominate a person who must be notified of any application to register the enduring power of attorney document. This notification must be given by the person applying for registration (either the principal or the representative) in a prescribed form. The nominated person may then object to the registration either to the Public Guardian on prescribed grounds such as the representative is dead or bankrupt, or the Court of Protection if the objection relates to the validity of the document. The Public Guardian advised that from 1 April 2009 to 31 March 2010 it received 77 such objections, while the Court of Protection received 190 objections in the same period.

The Office of the Public Advocate recommended that a process modelled on the UK system should be implemented in Victoria, with VCAT empowered to hear a challenge to the registration of an enduring power of attorney. The Victorian Bar suggested an alternative approach, allowing a person to lodge a caveat to prevent registration on the basis that a principal does not have capacity to make a power of attorney. It argued that the onus should be on the person lodging the caveat to establish why the caveat should not be removed. 1159

The UK legislation also requires the registration body to notify the representative if an application to register an enduring power of attorney is lodged by the principal, or the principal if the application is made by the representative. This ensures that an application for registration is not lodged without parties to the arrangement being aware of it. The person receiving the notice is then able to object to the registration. The person receiving the notice is then able to object to the registration.

<sup>1155</sup> Mental Capacity Act 2005 (UK) c 9 sch 1 cl 2(1)(c), 6, 9. Note that the Court of Protection may dispense with these notification requirements: cl 10.

<sup>&</sup>lt;sup>1156</sup> Mental Capacity Act 2005 (UK) c 9 ss 13(3), 13(6), sch 1 cl 13.

Letter from Martin John, above n 1038, att, 5.

Office of the Public Advocate, Submission 9, 22-23.

The Victorian Bar, Submission 40, 3.

<sup>&</sup>lt;sup>1160</sup> Mental Capacity Act 2005 (UK) c 9 sch 1 cl 7, 8.

<sup>&</sup>lt;sup>1161</sup> Mental Capacity Act 2005 (UK) c 9 sch 1 cl 13, 14.

While no participants in this Inquiry raised this issue, a recent New South Wales review noted concerns about the ease with which a forged power of attorney document could be registered, as the principal's identity is currently not checked. The review recommended that the registration body in that state should further investigate ways of confirming the identity of the principal, including providing a notice of registration to the principal. 1162

The Committee agrees that requiring key people to be notified of an application for registration potentially safeguards against abuse. In the previous chapter the Committee recommended that a principal should be able to nominate one or more personal monitors to play an oversight role. The Committee agrees the personal monitor's role should include ensuring appropriate registration of the document. The Committee notes that the importance of this role is diluted somewhat in the scheme proposed by the Committee, whereby registration is required when a document is created, rather than when it is activated.

The Committee also agrees that providing notice to a principal when an application is made by another person to register a power of attorney document would be an additional protection against fraud.

The Committee believes that it is appropriate that the persons notified of an application for the registration of an enduring power of attorney should be able to object to the registration of the document. VCAT should be empowered to hear and determine these objections.

### Recommendation 75: Notice of registration

The Committee recommends the Powers of Attorney Act require the registration body to:

- a) notify any personal monitors nominated in an enduring power of attorney document of an application for registration
- b) notify the principal that an application for registration has been made if the application is made by a person other than the principal.

#### Recommendation 76: Objections to registration

The Committee recommends the Powers of Attorney Act:

- a) empower a principal, any personal monitors nominated in an enduring power of attorney document or any other person with a special interest in the affairs of the principal, to object to the registration of an enduring power of attorney document
- b) empower VCAT to hear and determine objections to registrations of enduring power of attorney documents.

Land and Property Management Authority, New South Wales, above n 1062, 20.

# 8.2.12 Should the registration body play a role in educating representatives and monitoring their activities?

Some participants in this Inquiry suggested that the registration body could play a role in educating representatives about their responsibilities and overseeing their activities.

### The role of the registration body in educating representatives

As was noted in chapter six, representatives currently receive very little guidance and support to help them perform this sometimes onerous role. In that chapter the Committee suggested more information, education and support for representatives.

Some community legal centres currently provide information in relation to powers of attorney. Mr Dale Reddick, Advocacy and Support Worker with the Rights Advocacy and Support Program run by Gippsland Community Legal Service, who gave evidence on behalf of the Federation of Community Legal Centres told the Committee that a register could play a valuable role in providing targeted information to people acting as representatives. In the UK, the Office of the Public Guardian maintains the register as well as provides some education and resources for representatives. Representatives are advised to keep their information with the Public Guardian up to date so they can be provided with any information about the rules governing enduring powers of attorney.

The Committee has recommended the Registry of Births, Deaths and Marriages administer the register of powers of attorney in Victoria. The Committee selected this agency because of its expertise administering complex registers. While the Committee considers that the Registry of Births, Deaths and Marriages should make information available to representatives, the Committee believes this is a secondary function. The Committee believes that support and advice to representatives is better provided by the Office of the Public Advocate, which has extensive experience and knowledge in the area of supported decision making, as well as a proven track record in providing education and advice in this area. This was the approach the Committee recommended in chapter six of this report.

### The role of the registration body in monitoring representatives' activities

Some participants in this Inquiry felt that a registration body could provide increased scrutiny of representatives' activities. For example, Alfred Health's submission stated that a register would 'help to prevent abuse of the powers of appointees by monitoring their activities and the number of powers of attorney that they hold'. 1166

<sup>&</sup>lt;sup>1163</sup> Dale Reddick, *Transcript of evidence*, above n 1030, 8.

See, for example, Office of the Public Guardian, United Kingdom, *Your role and duties as an attorney*, <a href="http://www.publicguardian.gov.uk/decisions/role-duties-attorney.htm">http://www.publicguardian.gov.uk/decisions/role-duties-attorney.htm</a>, viewed 02 July 2010.

Department for Constitutional Affairs, United Kingdom, *Mental Capacity Act 2005 Code of practice*, The Stationery Office (2007), 119.

Alfred Health, Submission 66, 3. See also Council on the Ageing Victoria, Submission 39, 4; Dahni Houseman, Transcript of evidence, above n 1035, 3-4; Lauren Adamson, Transcript of evidence, above n 1035, 6.

The Committee considered the need for increased scrutiny of representatives' activities in the previous chapter and concluded that this function is better performed by a personal monitor appointed by the principal, rather than an external agency such as a registration body.

# 8.2.13 What is the effect of registration in another jurisdiction?

Some Inquiry participants argued that one significant benefit of registration is that it promotes the inter-jurisdictional recognition of power of attorney documents. The Australian Bankers' Association advocated that registration of a power of attorney document in one Australian jurisdiction should result in the automatic recognition of the document throughout the nation. 1167

The Australian Guardianship and Administration Council, a national body comprising public advocates, public and adult guardians, guardianship boards and tribunals and public trustees, is currently giving consideration to proposing a national scheme whereby a certified copy of an enduring power registered in one Australian jurisdiction is accepted as satisfying the requirement to register an enduring power of attorney in another jurisdiction in order to deal with land in that jurisdiction. <sup>1168</sup>

In chapter four of this report the Committee recommended that the Powers of Attorney Act provide for maximum recognition in Victoria of all power of attorney documents validly executed in another Australian state or territory. The Committee believes that the proposed Victorian register of power of attorney documents should be designed to promote maximum cross-jurisdictional recognition of power of attorney documents in order to ensure that members of the community who travel interstate are not inconvenienced if they need to rely on their power of attorney document outside their home state or territory. In addition, the Committee encourages the Victorian Government to actively support an arrangement which would allow representatives under a registered Victorian power of attorney to deal with a principal's land in another state and territory without having to separately register the document in that jurisdiction.

\_

Australian Bankers' Association Inc, *Submission 55*, 6-7. See also Association of Independent Retirees (AIR) Ltd (Victorian Division), *Submission 12*, 4.

Email from Andrew Taylor, Public Trustee for the ACT, to Executive Officer, Victorian Parliament Law Reform Committee, 16 February 2010, 3.

### Recommendation 77: Effect of registration in another jurisdiction

The Committee recommends:

- a) the Victorian Government, through the Standing Committee of Attorneys-General, actively support and promote an arrangement whereby a certified copy of an enduring power of attorney registered in one Australian jurisdiction is sufficient to satisfy the requirement to register an enduring power of attorney in another jurisdiction in order to deal with land in that jurisdiction
- b) the Powers of Attorney Act provide, to the maximum extent possible, for recognition of an enduring power of attorney document registered in another Australian state or territory.

### 8.3 Promoting the registration system

In this chapter the Committee has recommended a compulsory registration system for all new enduring powers of attorney (financial) and enduring powers of attorney (guardianship). It has also suggested that people who have previously created enduring powers of attorney should have the option of registering their documents. Therefore, it is important that the new registration system is widely promoted so that those required to register their documents are aware of their obligations, and those who have an option to register are aware of their choice.

In the Northern Territory, where registration of enduring powers of attorney (financial) is compulsory, the registration requirements are included on the forms for creating these documents. In addition, there is information displayed at the Land Titles Office and on the office's webpage. Mr Peter Shoyer, the Territory's Registrar-General, informed the Committee that 'Banks, Conveyancers and Solicitors also play a major role in informing clients of the registration requirements'. 1169

The Committee believes it is essential that the proposed new registration requirements are well promoted. Information about registration should be included on the forms for creating powers of attorney, as well as in information and educational materials for both principals and representatives which the Committee recommended in chapters four and six. It is also important that professionals advising people about powers of attorney, in particular lawyers, are familiar with the need to register the documents. In the next chapter the Committee recommends an education campaign to inform key professionals about the new requirements of the Powers of Attorney Act, as well as increased ongoing education for lawyers about powers of attorney. This education should include information about the registration requirements.

-

Letter from Peter Shoyer, above n 1045, 3.

The Committee acknowledges that many members of the community who have already made enduring power of attorney documents may wish to take up the opportunity to voluntarily register these. Therefore, the Committee recommends a community education campaign to advise members of the community about their ability to register their existing power of attorney documents. This should be particularly targeted at groups in the community who are more likely to have made these documents, such as seniors.

### Recommendation 78: Promoting the registration system

The Committee recommends the Victorian Government:

- a) provide information about registration requirements on the forms for creating enduring powers of attorney (financial) and enduring powers of attorney (guardianship)
- b) conduct a public education campaign to inform members of the community who have already made an enduring power of attorney (financial) or enduring power of attorney (guardianship) about the option of registering the document.

### Case study 20: The representatives had not protected her interests 1170

Mrs Q was diagnosed with dementia in 1998 at age 74. Her husband had died two years earlier. Mrs Q owned her own home and a commercial property. She has two children, a daughter, A, and a son, B.

In late 1999 Mrs Q and her children attended Mrs Q's solicitor and an enduring power of attorney (financial) was given jointly to the two children. The solicitor was not told that Mrs Q had been diagnosed with dementia and 'did not look into issues regarding capacity'. There was little explanation to B and A about their obligations as representatives. The power of attorney was used immediately, with her son taking control of financial matters and her daughter co-signing documents at his request.

Mrs Q's health deteriorated. In 2002, after significant health issues, she was placed into a care hostel.

-

<sup>1170</sup> Case study provided to the Committee by a person who requested that his/her name be withheld.

In late 2002, Mrs Q's son, B, arranged for the commercial property to be sold to pay for an accommodation bond and costs to an aged care facility. 'There was no proper evaluation of the property and it was not put to an open market. Rather the EPOAs [representatives] accepted an offer from the existing tenants...' B then arranged for Mrs Q's home to be transferred into his name, and paid half the appraised value to Mrs Q's daughter so that the asset was split equally between Mrs Q's two children.

There appears to have been no benefit to Mrs Q in this occurring; it would not have entitled her to an aged care pension or have any other financial benefit for her. In fact, disposing of the residential home precluded Mrs Q from being able to receive a pension she would otherwise have been entitled to for two years, if the commercial property had been sold only.

Her grandson told the solicitor that he had concerns about how she would survive financially.

To avoid the appearance of conflict, the transfer documents were this time signed by Mrs Q herself. But it appears that at all times the solicitor was sending correspondence to B, and B was advising Mrs Q where to sign and what to do. The only document signed by Mrs Q in all her years with dementia was the property transfer to her son, signed at a time when Mrs Q was unable to care for herself and dependent on nursing home care.

Mrs Q stayed in the aged care facility until 2009, when her grandson, D, became concerned as Mrs Q's health deteriorated due to neglect at the facility. When D looked for a new aged care home for Mrs Q, he realised that her choice of facilities was limited because she would not have enough money to pay the accommodation bonds due to ownership of her home being transferred to B and his wife. A new facility was found and Mrs Q's health improved at the new facility.

D also became aware that the representatives had not undertaken many of their responsibilities, such as lodging tax returns and applying for the aged pension.

D felt that the representatives, in particular B, 'had not protected her interests, though there is no question that they love and care for her'. He lodged an application with VCAT in 2010 to be appointed guardian and administrator for Mrs Q; however, B opposed the application. VCAT agreed that the transfer of the home to the two children was inappropriate in their role as representatives but advised that unless the family agreed on who was to be appointed, VCAT would be forced to appoint the Public Advocate as guardian which could decrease the quality of Mrs Q's care, and the State Trustees as independent administrator which would be costly for Mrs Q. 'D believes that it would probably be safer to have the EPOA [representative], B, appointed as an administrator because at the very least there will be some oversight into actions of administrators'.

### **Chapter 9: Promoting powers of attorney**

Many Victorians have never even heard of powers of attorney, let alone understand the benefits these arrangements offer. This chapter considers strategies for increasing awareness, understanding and use of these documents. In particular, it explores ways of promoting powers of attorney to groups in the community that do not use powers of attorney to the same extent as other members of the community. Groups less likely to use powers of attorney include young people and members of culturally and linguistically diverse and Aboriginal communities. This chapter also considers mechanisms for increasing awareness and understanding of powers of attorney among key professional groups who may be in a position to help educate members of the community about powers of attorney, or who may be asked to act on the basis of a representative's power.

### 9.1 Current activities promoting powers of attorney

There are a number of organisations currently conducting activities aimed at raising awareness of powers of attorney in Victoria.

### 9.1.1 The Office of the Public Advocate

Chapter two of this report highlighted that the Office of the Public Advocate (OPA) plays a key role in providing the community with information, education and support in relation to powers of attorney.

In collaboration with Legal Aid Victoria, OPA has produced *Take control: a kit for making powers of attorney and guardianship*. This kit includes a DVD and a comprehensive booklet about powers of attorney. Legal Aid Victoria stated that this is their 'most popular publication', while Dr John Chesterman, Manager of Policy and Education at OPA, indicated it is 'highly regarded'. This kit was commended by many participants in this Inquiry. <sup>1172</sup>

In 2008-2009, approximately 10 000 copies of *Take control* were downloaded from the websites of OPA and Victoria Legal Aid, and a further 36 000 hard copies were distributed. OPA also provides information on powers of attorney through its website and through the 200 community presentations that it holds each year. OPA has also recently conducted a postcard campaign aimed at informing young people about powers of attorney. That campaign is discussed later in this chapter.

Victoria Legal Aid, Submission 42, 3; John Chesterman, Manager, Policy and Education, Office of the Public Advocate, Transcript of evidence, Melbourne, 22 October 2009, 2.

Ministerial Advisory Council of Senior Victorians, Submission 48, 3; Peter MacCallum Cancer Centre, Submission 20, 1; Southern Health, Submission 30, 2; Hume Riverina Community Legal Service, Submission 36, 3; Latrobe Regional Hospital, Submission 44, 2; Joyce Jeffs, Transcript of evidence, Melbourne, 30 March 2010, 9.

Office of the Public Advocate, Annual report 2008-09 (2009), 41; John Chesterman, Transcript of evidence, above n 1171, 6.

Office of the Public Advocate, Submission 9, 5; John Chesterman, Transcript of evidence, above n 1171, 2, 6.

In addition, OPA provides information about powers of attorney through its telephone advice service. In 2008-2009 the service received in excess of 14 000 calls. Although detailed data is not available, OPA advised that the majority of calls to the advice service are general enquiries about either enduring powers of attorney or guardianship and administration. 1176

### 9.1.2 Council on the Ageing

The Council on the Ageing (COTA) is a community-based organisation that represents older people in Victoria. One of COTA's core functions is to provide information and services to older people, including information about powers of attorney.

COTA provides an information outreach program, *Need to Know?*, which includes sessions facilitated by the OPA on powers of attorney. <sup>1177</sup> Ms Kris Spark, Manager of COTA's Seniors' Information Victoria Telephone Service, described these sessions as 'well attended'. <sup>1178</sup> In 2008, 19 such sessions were held. <sup>1179</sup>

The Seniors' Information Victoria Telephone Service, which is managed by COTA, received over 25 000 contacts in 2008. Ms Spark told the Committee that many of these related to powers of attorney, with frequent queries relating to the different types of documents and powers and how to go about making or revoking a power of attorney. 1181

Ms Spark informed the Committee that many people who contact the service are on low incomes and are seeking 'a low-cost or no-cost way of addressing issues in their life'. 1182

### 9.1.3 Seniors Rights Victoria

Seniors Rights Victoria was established in 2008 to provide support, education, advocacy and legal services to older Victorians experiencing abuse, as well as education for the general community and service providers. It is jointly managed by COTA, the Public Law Interest Clearing House, Eastern Community Legal Centre and Loddon Campaspe Community Legal Centre.

Seniors Rights Victoria advised the Committee that in its first year of operation, its telephone advice line received over 1300 inquiries which resulted in 95 referrals to

Office of the Public Advocate, above n 1173, 38. See also Office of the Public Advocate, *Submission 9*, 5; John Chesterman, *Transcript of evidence*, above n 1171, 5-6.

John Chesterman, Transcript of evidence, above n 1171, 5-6.

Kris Spark, Manager, Seniors Information Victoria, Council on the Ageing Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 3-4; Council on the Ageing Victoria, *Submission 39*, 1.

Kris Spark, *Transcript of evidence*, above n 1177, 3.

Council on the Ageing Victoria, Annual report 2009 (2009), 10.

<sup>&</sup>lt;sup>1180</sup> Ibid, 9.

<sup>&</sup>lt;sup>1181</sup> Kris Spark, *Transcript of evidence*, above n 1177, 3-4.

<sup>&</sup>lt;sup>1182</sup> Ibid, 4.

Seniors Rights Victoria, *Submission 38*, 9; Seniors Rights Victoria, *Services*, <a href="http://www.seniorsrights.org.au/index.php?option=com\_content&view=article&id=27&Itemid=7">http://www.seniorsrights.org.au/index.php?option=com\_content&view=article&id=27&Itemid=7</a>, viewed 7 June 2010.

the advocacy service, and the provision of advice in 439 cases. <sup>1184</sup> Ms Dahni Houseman, Policy Officer with Seniors Rights Victoria, told the Committee, 'We currently undertake education sessions around powers of attorney for whichever community organisation requests them.' <sup>1185</sup>

In addition, as outlined in chapter seven, Seniors Rights Victoria plays an important role providing education to people who have been victims of elder abuse, including through powers of attorney, as well as general community information about elder abuse. <sup>1186</sup>

# 9.1.4 Other organisations providing education about powers of attorney

There are many other organisations in the community that provide information and education about powers of attorney.

Some community legal centres provide information in relation to powers of attorney. Mr Dale Reddick, Advocacy and Support Worker with the Rights Advocacy and Support Program run by Gippsland Community Legal Service, told the Committee that between July 2009 and May 2010 he conducted 16 community education sessions about powers of attorney, with approximately 560 attendees. Mr Reddick described how he goes about educating members of the community about powers of attorney:

I start with the idea that life is unpredictable and someone will be found to make decisions for them one way or another ... I discuss the four powers of attorney but point out that the enduring one is the one that is the instrument to plan for the future.

I open up some ideas around thinking who they would trust with the power should they choose to donate it. I put forward the idea of having conversations with whomever they choose to create an ongoing relationship with them. I explain that they can revoke the power whenever they choose, that the signature on the form and the witnessing does not lock them into an arrangement. I suggest that they can make it part of their future planning if that is what they feel they need. <sup>1188</sup>

Ms Sarnia Birch, Acting Deputy Manager, Regional Offices with Victoria Legal Aid, told the Committee that Legal Aid receives many queries about powers of attorney, especially in its regional offices from 'people who are ordinary members of the public who just want general information about what a power of attorney is, how they make one'. <sup>1189</sup> As mentioned above, Legal Aid partnered with OPA to produce the *Take control* kit and distributes many of these kits.

<sup>&</sup>lt;sup>1184</sup> Seniors Rights Victoria, Submission 38, 9.

Dahni Houseman, Policy Officer, Seniors Rights Victoria, *Transcript of evidence*, Melbourne, 22 October 2009, 5.

Seniors Rights Victoria, Submission 38, 9.

Email from Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, to Administrative Officer, Victorian Parliament Law Reform Committee, 7 June 2010.

Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, Federation of Community Legal Centres (Victoria) Inc, *Transcript of evidence*, Melbourne, 1 October 2009, 7.

Sarnia Birch, Acting Deputy Manager, Regional Offices, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 3.

Questions about powers of attorney are frequently raised in other forums. For example Ms Belinda Evans, Elders Rights Advocacy's Senior Advocate, told the Committee that issues about powers of attorney are sometimes raised in general information sessions her organisation runs for residents of Commonwealth-funded aged-care services, their families and health professionals working in the aged-care sector. 1190

# 9.2 A coordinated approach to promoting powers of attorney

The previous section highlights that current education activities about powers of attorney are uncoordinated and ad hoc. 1191

In the absence of any government agency with formal responsibility for providing community education about powers of attorney in Victoria, OPA has become 'a de facto expert' on powers of attorney in this state, playing a key role in providing education and support. However, Dr Chesterman told the Committee that OPA does not have the resources to provide more broad based education. He stated:

When I joined the Office of the Public Advocate I knew OPA had a clear role with enduring powers of attorney, but it was not altogether clear to me why we did. In a sense the principal role of OPA is the guardian of last resort — people with no others to play that role. It is clearly related — enduring powers of attorney are something someone can sign for in the event they become incapacitated to one extent or another. I think the best way to go would be for the government to take it on ... I would think the Department of Justice would be the appropriate place for that, and possibly VLA — Victoria Legal Aid. 1193

The Committee acknowledges the breadth of activities currently being undertaken to raise awareness of powers of attorney in Victoria, particularly by community organisations. The Committee believes that the Victorian Government should be much more active in promoting and providing education and support about powers of attorney. It recommends that a coordinated whole-of-government approach should be adopted in relation to education about powers of attorney in Victoria. This should be auspiced by the Department of Justice and involve a range of other departments and agencies, including the Department of Health, the Department of Human Services, the Department of Planning and Community Development, OPA and Victoria Legal Aid.

<sup>1193</sup> Ibid, 9.

Belinda Evans, Senior Advocate, Elder Rights Advocacy, *Transcript of evidence*, Melbourne, 22 October 2009, 2-3.

See generally Margaret Brown and Suzanne Jarrad, 'Putting "the powers" in place: Barriers for people with memory loss in planning for the future' (2008) 15(4) *Journal of Law and Medicine* 530, 531.

John Chesterman, *Transcript of evidence*, above n 1171, 2.

### Recommendation 79: A coordinated approach to providing information and education about powers of attorney

The Committee recommends the Victorian Government adopt a coordinated, wholeof-government approach to providing information and education about powers of attorney.

# 9.3 Increasing community awareness and understanding of powers of attorney

This section considers mechanisms for promoting community awareness and understanding of powers of attorney.

As noted in chapter two of this report, powers of attorney are not as widely used as they could be, with around 11% of the Australian community having made a power of attorney. <sup>1194</sup> Lack of awareness and understanding of powers of attorney were identified in that chapter as key barriers to more people making powers of attorney.

While increased awareness and understanding of powers of attorney may increase their use, evidence to this Inquiry suggested that not everyone is going to make a power of attorney. Professor Terry Carney, Professor of Law at the University of Sydney, told the Committee:

But one has to ask oneself the question: how many people are going to fill out any of these instruments in the best of all possible worlds? The answer to that appears to be about one in five. All kinds of endeavours have been made to better publicise and educate, there have been lots of research studies and it still appears as though the upper limit is not much more than about 20 per cent, and is often a great deal lower than that. <sup>1195</sup>

However, Mr John Billings, Deputy President of the Guardianship List at the Victorian Civil and Administrative Tribunal (VCAT) expressed the view that 'if there really were a concerted education campaign, the likes of which we probably have not seen yet, then my sense of it is that we could get well beyond 20 per cent'. 1196

The Committee acknowledges that there will never be universal uptake of powers of attorney. However, it believes we can do better than the present estimated 11% uptake. In particular, at present many people do not make powers of attorney, not because they choose not to, but rather because they are not aware that these arrangements exist. The Committee believes that more can be done to make more

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007), 71 citing Office of the Public Advocate, Queensland Government, submission 76 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006), 7.

Terry Carney, Professor of Law, The University of Sydney, Transcript of evidence, Melbourne, 1 October 2009, 2.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), *Transcript of evidence*, Melbourne, 1 October 2009, 3.

people aware of powers of attorney and to encourage use of these arrangements. This section explores how this can be achieved.

### 9.3.1 General community education

An overwhelming number of participants in this Inquiry called for greater community education to raise awareness about powers of attorney, especially enduring powers. 1197

### What are the benefits of general community education about powers of attorney?

The Committee heard that providing more education about powers of attorney to the general community would have a number of positive outcomes including increasing the uptake of all types of powers of attorney, encouraging earlier uptake and promoting understanding of the benefits of these arrangements.

Some members of the community have never heard of powers of attorney. Southern Health's submission to the Inquiry stated, 'Many patients are not even aware of the existence of POA [powers of attorney], or what they are for'. Similarly, Latrobe Regional Hospital reported that, 'For some people it is a new concept and some common responses are "I already have made a will", or "I do not need to make a Power of Attorney as there is nothing wrong with me".'

Even people who are aware of powers of attorney may lack detailed understanding of matters such as the benefits of making a power of attorney and the powers that are

262

<sup>1197</sup> Ibid, 2-3; Claire Hausler, Manager, TCL Legal Services (Vic) Pty Ltd, Trustee Corporations Association of Australia, Transcript of evidence, Melbourne, 22 October 2009, 2; Paul Zanatta, Manager, Community Care, Policy and Small Rural Health, Aged & Community Care Victoria, Transcript of evidence, Melbourne, 14 December 2009, 2; Latrobe Regional Hospital, Submission 44, 1; Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 4; Palliative Care Victoria, Submission 70, 1; Catherine Leslie, Legal and Policy Officer, Mental Health Legal Centre Inc, Transcript of evidence, Melbourne, 14 December 2009, 2; Elder Rights Advocacy, Submission 63, 4; Southern Health, Submission 30, 1; Hume Riverina Community Legal Service, Submission 36, 3-4; Victoria Legal Aid, Submission 42, 4; Federation of Community Legal Centres (Victoria) Inc, Submission 47, 4; Ministerial Advisory Council of Senior Victorians, Submission 48, 3-4; Moreland Community Legal Centre Inc. Submission 51, 1-3; Carers Victoria, Submission 65, 13-14; Robert C Bolch, JP, Submission 3, 12; Office of the Public Advocate, Submission 9, 24-25; Aged & Community Care Victoria, Submission 53, 2; Australian Bankers' Association Inc, Submission 55, 10; Mental Health Legal Centre Inc, Submission 59; Robyn Mills, Acting Director of Civil Law Services, Victoria Legal Aid, Transcript of evidence, Melbourne, 17 December 2009, 6; Kevin Larkins, Chief Executive Officer, Palliative Care Victoria, Transcript of evidence, Melbourne, 17 December 2009, 3; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 3; Dahni Houseman, Transcript of evidence, above n 1185, 5; John Chesterman, Transcript of evidence, above n 1171, 2, 5; Stelvio Vido, Executive General Manager, Strategic and Support Services, Royal District Nursing Service, Transcript of evidence, Melbourne, 17 December 2009, 2; Nikki Isaks, Senior Social Worker, Royal District Nursing Service, Transcript of evidence, Melbourne, 17 December 2009, 8.

Southern Health, Submission 30, 1. See also John Chesterman, Transcript of evidence, above n 1171, 5-6; Julian Gardner, Transcript of evidence, Melbourne, 1 October 2009, 3; Catherine Leslie, Transcript of evidence, above n 1197, 2-3; Julian Gardner, Submission 21, 3; Ministerial Advisory Council of Senior Victorians, Submission 48, 1; Susan Edwards and Antonia Fontana, Legal information needs of older people, Law and Justice Foundation of New South Wales (2004), 5.

Latrobe Regional Hospital, *Submission 44*, 1. See also Deborah Setterlund, Cheryl Tilse and Jill Wilson, *Substitute decision making and older people* Trends & issues in crime and criminal justice, no. 139, Australian Institute of Criminology (1999), 3.

granted. 1200 Mr Julian Gardner told the Committee that in his former role as Victoria's Public Advocate, he often attended public meetings such as Probus or Rotary clubs:

where I asked for a show of hands from those people who have got a will. I then asked for a show of hands from people who have got an enduring power. Almost everybody had a will, but a very small percentage had an enduring power. Then I asked, 'Why is it you are worried about what happens after you are dead, but you are not worried about what happens to you when you are alive?' I can only conclude that the reason was that they had not understood what was open to them. <sup>1201</sup>

Participants in the Inquiry indicated that there are particularly low levels of awareness of enduring powers of attorney (guardianship). Ms Jenny Chapman, Chief Social Worker at Werribee Mercy Hospital, who gave evidence on behalf of Palliative Care Victoria, told the Committee 'people will often come in with powers of attorney for financial ... but guardianship they have never heard of: they do not know what it is'. <sup>1202</sup> This corresponds with the low take-up rate of this power compared with financial powers that was noted in chapter two.

Respecting Patient Choices' submission to the Inquiry argued that increased community education about powers of attorney would not only encourage use of these arrangements, but may promote uptake at an earlier stage, when there is more certainty about a person's capacity to make the document. 1203

### What kind of community education campaign is needed?

Inquiry participants suggested that members of the public need information about:

- What is a power of attorney (including the different types of powers of attorney and the powers they grant)?<sup>1204</sup>
- The benefits and disadvantages of having a power of attorney.
- How to make a power of attorney and where to get information, advice and support. 1206

Palliative Care Victoria, Submission 70, 2; Moreland Community Legal Centre Inc, Submission 51, 1-2; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation: Limits to informed choice' (2002) 21(3) Australasian Journal on Ageing 128, 130; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 1199, 3.

Julian Gardner, *Transcript of evidence*, above n 1198, 6. See also Australian Bankers' Association Inc, *Submission 55*, 10.

Jenny Chapman, *Transcript of evidence*, above n 1197, 4. See also David Fara, Deputy President, Association of Independent Retirees (AIR) Ltd (Victorian Division), *Transcript of evidence*, Melbourne, 22 October 2009, 2; Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 1200, 130.

Respecting Patient Choices – Advance Care Planning, Submission 13, 1. See also Nikki Isaks, Transcript of evidence, above n 1197, 8.

Hume Riverina Community Legal Service, *Submission 36*, 4; Federation of Community Legal Centres (Victoria) Inc, *Submission 47*, 4; Victorian Coalition of Acquired Brain Injury Service Providers, *Submission 71*, 3; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 3.

Southern Health, Submission 30, 1; Victorian Coalition of Acquired Brain Injury Service Providers, Submission 71, 3; Australian Bankers' Association Inc, Submission 55, 10.

The Committee heard that a general community education campaign needs to recognise that different members of the community access information in different ways. Participants suggested a multifaceted approach including community information sessions, television and radio advertising, newspaper advertising, brochures, fact sheets and websites. <sup>1207</sup> Participants also supported making materials about powers of attorney widely available, for example through health and aged care facilities, community legal centres, local council networks and post offices. <sup>1208</sup>

Several Inquiry participants praised the *Take control* kit currently produced by OPA and Victoria Legal Aid. The Ministerial Advisory Council of Senior Victorians stated that the kit 'provides a good basis for raising awareness about enduring powers ... these materials could be updated and used for future community-based information sessions'. However, other participants suggested that the kit, particularly the booklet, is overly complex and not easily understood by some members of the community, particularly those with poor literacy or limited English. For example, St Kilda Legal Service stated that the booklet 'is too difficult for most of our clients to understand.' 1210

The Committee believes it is important for all Victorians to be aware of powers of attorney and the benefits they offer. This empowers people to make an informed choice to make or not make these arrangements. Therefore the Committee recommends that the Victorian Government implement an ongoing state-wide community education campaign to increase awareness and understanding of powers of attorney. This campaign should use a wide range of media. All written materials should be in simple English and also be available in a variety of community languages. In addition, the campaign should be supported by an information mechanism such as a telephone advice service to enable people to quickly and easily obtain further information and support if they wish.

In the Committee's view, it is important that all components of this general campaign, and the targeted campaigns discussed in the next section, are linked to those resources for principals and representatives recommended in chapters four and six, as well as information about abuse discussed in chapter seven. In particular, all these campaigns should have a common 'look and feel'.

The Committee also recognises that there are some groups in the community that have significantly lower levels of use of powers of attorney. Specific education campaigns targeting these groups are discussed in the next section.

Palliative Care Victoria, Submission 70, 1. See generally Margaret Brown and Suzanne Jarrad, above n 1191, 537.

Victoria Legal Aid, Submission 42, 3; John Chesterman, Transcript of evidence, above n 1171, 5-6; David Fara, Transcript of evidence, above n 1202, 3; Ministerial Advisory Council of Senior Victorians, Submission 48, 3-4; Moreland Community Legal Centre Inc, Submission 51, 3; Lucie O'Brien, Policy Officer, Federation of Community Legal Centres (Victoria) Inc, Transcript of evidence, Melbourne, 1 October 2009, 6.

Victoria Legal Aid, Submission 42, 3; Moreland Community Legal Centre Inc, Submission 51, 3.

Ministerial Advisory Council of Senior Victorians, Submission 48, 3. See also Peter MacCallum Cancer Centre, Submission 20, 1; Southern Health, Submission 30, 2; Hume Riverina Community Legal Service, Submission 36, 3; Latrobe Regional Hospital, Submission 44, 1-2; Joyce Jeffs, Transcript of evidence, above n 1172, 9.

Hume Riverina Community Legal Service, *Submission 36*, 3. See also Ethnic Communities' Council of Victoria, *Submission 23*, 2; St Kilda Legal Service Co-Op Ltd, *Submission 50*, 2.

### Recommendation 80: Increasing general community awareness and understanding of powers of attorney

The Committee recommends the Victorian Government develop and implement an ongoing state-wide community education campaign to increase awareness and understanding of powers of attorney. The campaign should:

- a) provide simple easy-to-understand information about powers of attorney in plain English and in a variety of community languages
- b) use a wide variety of media, including websites, DVDs, TV, radio, newspapers, newsletters, pamphlets, fact sheets and posters
- c) include a community engagement component, with information sessions provided through a range of existing community forums
- d) be supported by advice and support mechanisms, including a telephone advice service.

# 9.3.2 Educating groups with low levels of awareness, understanding and use

Many participants in the Inquiry called for any general community education campaign to be supplemented by targeted campaigns to increase awareness, understanding and use of powers of attorney in groups that have low levels of use of power of attorney documents.

Evidence to the Inquiry suggested education campaigns should be specifically directed at seniors, people from culturally and linguistically diverse backgrounds, young people and members of the Aboriginal community.

#### **Seniors**

While people aged over 60 years have one of the highest rates of uptake of powers of attorney, the Committee heard that there is still room for improvement, with many seniors lacking a good understanding of power of attorney arrangements. Seniors are a diverse group and awareness and use of powers of attorney may vary significantly within different subgroups. For example, one Queensland study found that older people living in retirement villages were almost twice as likely to have a power of attorney as those living in the community.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 1200, 130.

Office of the Public Advocate, *Submission 9*, 24-25; Moreland Community Legal Centre Inc, *Submission 51*, 1-2; Aged & Community Care Victoria, *Submission 53*, 2. See generally Deborah Setterlund, Cheryl Tilse and Jill Wilson, *Substitute decision making and older people*, above n 1199, 5.

There was strong support from Inquiry participants for a community eduction campaign about powers of attorney targeting seniors. <sup>1213</sup> Ms Pat Whaley, who attended the Committee's Seniors' Forum, stated that she volunteers in the nursing home system where she delivers two information seminars each year on powers of attorney: 'This is our best-attended seminar. People are hungry for knowledge. I do not know how much education there is in the community, but I believe these seminars should be widespread ...' <sup>1214</sup>

Mr Kevin Larkins, Chief Executive Officer of Palliative Care Victoria, told the Committee that education should be targeted at those aged 70 and older and should occur within the community:

That age group is ageing healthily, in the main ... that is a perfect opportunity to engage people in the issues that they are facing, because they have the capacity to do so. The education needs to take place within the community. It is great that the lessons are learnt in the acute care setting because the families are, if you like, sensitised to these issues at that time, and in fact they can be the ones that can be advocates for it. The better option is to have it embedded in the community where those conversations are just a normal part of what we, from 70 up, can expect. We can prepare. Just as you have insurance for your house, insurance for your car, this is something you have had a lifetime of insurance for, so get this in place as well. <sup>1215</sup>

As with the general community education campaign, there was support for a multifaceted approach to providing information about powers of attorney for seniors. Media suggested for targeting older people include local newspapers, seminars, radio and television advertising and providing information through local councils and health care services. <sup>1216</sup> Mr Gary Haley, a participant in the Committee's Seniors' Forum, observed that older people may not have the same ability or willingness to access information over the internet as younger people. <sup>1217</sup>

The Committee recognises that many senior Victorians are not aware of or do not understand powers of attorney. This is concerning because increasing life expectancy and rates of cognitive impairment mean that powers of attorney have the potential to benefit a significant number of people in this age group.

The Committee suggests a targeted education campaign about powers of attorney should be developed for seniors, with input from seniors and seniors' organisations. The campaign should make use of a wide range of media and be mindful that not all seniors have ready access to the internet. The Committee notes evidence that face-to-face interactions such as community seminars, may be particularly effective in raising awareness in this group.

\_

For example, Ministerial Advisory Council of Senior Victorians, Submission 48, 4; Office of the Public Advocate, Submission 9, 24. See also House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 87.

Pat Whaley, *Transcript of evidence*, Melbourne, 30 March 2010, 7. See also David Fara, *Transcript of evidence*, above n 1202, 3; Julian Gardner, *Transcript of evidence*, above n 1198, 2.

Kevin Larkins, *Transcript of evidence*, above n 1197, 7. See generally Margaret Brown and Suzanne Jarrad, above n 1191, 537.

Patricia Williams, *Submission 73*, 1-2; Pat Whaley, *Transcript of evidence*, above n 1214, 13, 17; John Billings, *Transcript of evidence*, above n 1196, 9.

Gary Haley, *Transcript of evidence*, Melbourne, 30 March 2010, 4-5.

### Recommendation 81: Educating seniors about powers of attorney

The Committee recommends the Victorian Government, in consultation with seniors and seniors' organisations:

- develop targeted information and resources about powers of attorney for a) seniors in a range of formats and disseminate these widely through appropriate media
- provide information sessions about powers of attorney for seniors through a b) range of existing community forums.

### People from culturally and linguistically diverse backgrounds

A significant number of older Victorians are from culturally and linguistically diverse (CALD) backgrounds. In 1996 an estimated 23.1% of senior Victorians had a CALD background, and this is projected to increase to 31% by 2011. 1218

The Ethnic Communities' Council of Victoria informed the Committee that CALD seniors have higher rates of dementia than the average Australian population 'because multicultural families tend not to seek aged care services and as a result avoid early diagnosis'. 1219

People from CALD backgrounds have been found to have low levels of use of powers of attorney. 1220 Participants in this Inquiry identified a number of possible reasons for this.

Firstly, the concept of power of attorney does not exist in many cultures. Ms Eva Wakim of the Victorian Arabic Social Services told the Committee that in many cultures it is assumed that decision-making responsibility is passed automatically to sons. She stated that when her organisation held sessions for seniors:

It was quite interesting for them to hear and understand that ... they can make choices and they can appoint their daughter. That was very interesting for them; it was a totally new concept that in Australia they need to take the step of putting someone as a power of attorney when they come to a state where they possibly cannot make decisions. 1221

Anna L Howe, Cultural diversity, ageing and HACC: Trends in Victoria in the next 15 years, report for Aged Care Branch, Department of Human Services, Victoria (2006), 11.

Ethnic Communities' Council of Victoria, Submission 23, 2.

<sup>1220</sup> Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 1200, 130, 132; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 1199, 5.

Eva Wakim, Victorian Arabic Social Services, *Transcript of evidence*, Melbourne, 30 March 2010, 4. See also Virginia Landrito-Balanon, Community Aged-Care Packages Case Manager, Filipino Community Council of Victoria Inc, Transcript of evidence, Melbourne, 30 March 2010, 5-6; Rosemarie Draper, Aged Services Development Officer (Equity & Access), New Hope Foundation, Transcript of evidence, Melbourne, 30 March 2010, 8-9; Santosh Kumar, President, Northern Region Indian Seniors Association of Victoria, Transcript of evidence, Melbourne, 30 March 2010, 14; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 1199, 3.

Secondly, several participants in the Committee's CALD forum emphasised that there is a low level of awareness of powers of attorney in CALD communities. In particular, there is a lack of understanding of the implications of powers of attorney, the different powers and matters such as that the powers can be revoked. Research has suggested that low levels of understanding of powers of attorney in the CALD community 'appeared to be a result of language barriers and unfamiliarity with the Australian law'. 1223

The Committee heard that other barriers to people from CALD backgrounds making powers of attorney include a tendency to avoid matters 'associated with later life and death' 1224 and mistrust of governments. 1225

Participants in this Inquiry emphasised that the CALD community is not a homogenous group and that specific strategies may need to be developed for specific subgroups. <sup>1226</sup> In particular, involving the target community in developing the campaign, and utilising CALD organisations' connections with the community were viewed as vital. <sup>1227</sup>

Inquiry participants suggested that any written materials about powers of attorney should be available in community languages and that the translations should use simple language. The *Take control* kit was considered too complex for some members of the CALD community. Ms Robyn Mills, Acting Director of Civil Law Services at Victoria Legal Aid, conceded that the publication is 'probably aiming more at a white Anglo-Saxon community...' 1230

Other participants noted that many members of the CALD community do not have high levels of literacy. They recommended using audio or audiovisual formats such as DVDs and advertising on ethnic radio and television to target information at this group. <sup>1231</sup> Ms Rosemarie Draper, Aged Services Development Officer (Equity & Access) with the New Hope Foundation, recommended a DVD with dramatised case studies 'because I think people understand things if they are told in a storyline'. <sup>1232</sup>

\_\_

Rosemarie Draper, *Transcript of evidence*, above n 1221, 4; Poppy Hearn, Senior Social Worker, Australian Greek Welfare Society, *Transcript of evidence*, Melbourne, 30 March 2010, 5; Rosemarie Draper, Aged Services Development Officer (Equity & Access), New Hope Foundation, *Submission 72*, 1; Terry Carney, *Transcript of evidence*, above n 1195, 7-8.

Deborah Setterlund, Cheryl Tilse and Jill Wilson, *Substitute decision making and older people*, above n 1199, 5.

<sup>&</sup>lt;sup>1224</sup> Ministerial Advisory Council of Senior Victorians, Submission 48, 4.

Eva Wakim, *Transcript of evidence*, above n 1221, 10-11. See also Ethnic Communities' Council of Victoria, *Submission* 23, 3.

Marion Lau, Deputy Chair, Ethnic Communities' Council of Victoria, *Transcript of evidence*, Melbourne, 30 March 2010, 7-8. See also Poppy Hearn, *Transcript of evidence*, above n 1222, 9.

Poppy Hearn, *Transcript of evidence*, above n 1222, 9; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 4.

Australian Greek Welfare Society, Submission 74, 3; Action on Disability in Ethnic Communities, Submission 14, 1.

Ethnic Communities' Council of Victoria, Submission 23, 2.

Robyn Mills, Transcript of evidence, above n 1197, 6.

Elizabeth Ignys, Manager, Advocacy, Action on Disability in Ethnic Communities, *Transcript of evidence*, Melbourne, 30 March 2010, 7; Marion Lau, *Transcript of evidence*, above n 1226, 7-8; Ethnic Communities' Council of Victoria, *Submission 23*, 2; Australian Greek Welfare Society, *Submission 74*, 3; Ministerial Advisory Council of Senior Victorians, *Submission 48*, 4.

Rosemarie Draper, *Transcript of evidence*, above n 1221, 8. See also Ethnic Communities' Council of Victoria, *Submission 23*, 2; Rosemarie Draper, *Submission 72*, above n 1222, 1.

Ms Jaklina Michael, Cultural Liaison Officer with the Royal District Nursing Service, explained how her organisation goes about developing material for its clients, who speak 105 different languages:

We write all documents in English now at grade 4 to grade 6 English language literacy level. We have what is called 'translation standards' at the Royal District Nursing Service ... We apply the standards to the written translation. From there we then look at the diversity within that cultural group and look at what types of information on what media that information needs to be presented in. For instance, with a Macedonian diabetes education package, when we consulted with Macedonian welfare, we found that 24 per cent of that, over 100 people, were illiterate, so we had to develop not only written material at grade 4 to grade 6, we then had flip charts for people who look more at pictures and look at very short messages.

Then we had short plays and audio recordings for people who like to learn through listening. There are people who like to learn through listening, people who like to learn through reading and people who like to look at more visuals. In that way the nurse is then able to go in, assess the learning style and present the information in the appropriate format to that individual. <sup>1233</sup>

Several participants in the Committee's CALD Forum emphasised that information seminars delivered through community organisations such as CALD senior citizens' clubs can be effective for targeting this group. 1234 Ms Marion Lau, Deputy Chair of the Ethnic Communities' Council of Victoria, told the Committee that a face-to-face approach has been used effectively to address problem gambling in the CALD community, with information sessions given to groups that meet for social activities. 1235

There was also support for disseminating information through doctors and other health care providers <sup>1236</sup> as well as other service providers such as local government and Centrelink. <sup>1237</sup> This is discussed in more detail later in this chapter.

The Committee is concerned that people from CALD backgrounds are not using powers of attorney as widely as other groups in the community. The Committee believes that a targeted education campaign should be developed to educate members of CALD communities about powers of attorney and their benefits. This campaign should recognise that the concept of powers of attorney may be totally unfamiliar to members of this group and provide appropriate background and contextual information.

Jaklina Michael, Cultural Liaison Coordinator, Royal District Nursing Service, Transcript of evidence, Melbourne, 30 March 2010, 9-10.

Eva Wakim, *Transcript of evidence*, above n 1221, 11; Australian Greek Welfare Society, *Submission 74*, 3; Naderah Edwards, Spectrum Migrant Resource Centre, *Transcript of evidence*, Melbourne, 30 March 2010, 14; Manuel Asuncion, Vice-Chair, Filipino Community Council of Victoria Inc, *Transcript of evidence*, Melbourne, 30 March 2010, 14.

Marion Lau, *Transcript of evidence*, above n 1226, 11.

Elizabeth Ignys, *Transcript of evidence*, above n 1231, 12; Eva Wakim, *Transcript of evidence*, above n 1221, 13; Ross Barnett, Director, Ethnic Communities' Council of Victoria, *Transcript of evidence*, Melbourne, 30 March 2010, 13-14; Naderah Edwards, *Transcript of evidence*, above n 1234, 14; Toni Higgins, Specialist Mental Health, Human Rights and Civil Law, Victoria Legal Aid, *Transcript of evidence*, Melbourne, 17 December 2009, 6; Rosemarie Draper, *Transcript of evidence*, above n 1221, 15.

Ross Barnett, *Transcript of evidence*, above n 1236, 14.

The Committee believes that a wide range of resources should be developed for CALD communities. These need to be available in a variety of languages and formats, including audio and audiovisual formats for those with low levels of literacy. These resources should be developed in consultation with CALD organisations and members of CALD communities. Information about powers of attorney needs to be widely disseminated, including through community radio, community organisations, local government and health care providers.

These recommendations will support previous recommendations in this report that the forms for making powers of attorney and information and educational materials about powers of attorney for principals, witnesses and representatives be available in a variety of community languages.

### Recommendation 82: Increasing awareness of powers of attorney in CALD communities

The Committee recommends the Victorian Government:

- a) develop targeted information and resources about powers of attorney for people from CALD backgrounds. This information should be developed in consultation with CALD organisations and members of CALD communities, be available in a wide range of formats and be disseminated through appropriate media
- b) provide information sessions for members of CALD communities through a range of existing community forums.

### Young people

Young people have very low uptake of powers of attorney. The House of Representatives Standing Committee on Legal and Constitutional Affairs found that only 8% of those who had made an enduring power of attorney were under the age of 35.  $^{1238}$ 

Young people may not make powers of attorney for a number of reasons. Firstly, they may not be aware of powers of attorney, as many current educational activities are aimed at an older audience. <sup>1239</sup> Secondly, even if they are aware of powers of attorney, they may not understand that these potentially offer benefits to young people. Mr Tom Worsnop, Chair of the Victorian Coalition of Acquired Brain Injury (ABI) Service Providers, told the Committee:

Young people, and young men in particular, are the main group of people who get brain injuries. They are not traditionally a group of people who have even thought about the future. They are mainly in a framework of immortality at the age when

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 71 citing Office of the Public Advocate, Queensland Government, above n 1194, 7.

Robyn Mills, Transcript of evidence, above n 1197, 6.

they get the brain injury, so thoughts about powers of attorney  $\dots$  are the furthest thing from their mind.  $^{1240}$ 

In 2009 the Office of the Public Advocate ran a 'consciousness raising' campaign targeting young people. This involved distributing 20 000 postcards to venues frequented by young people, such as cafes, bars, cinemas and universities. The postcard encouraged young people to make powers of attorney and advised how to obtain a copy of *Take control* free of charge. Only a small number of the cards were mailed back to OPA requesting a hardcopy of *Take control*. However OPA was not able to track how many people accessed information on its website as a result of the campaign. <sup>1241</sup>

OPA's submission called for more targeted education for young people about powers of attorney. Several other participants also advocated more education for young Victorians. Ms Val Johnstone, a participant in the Committee's Seniors' Forum suggested that young people could be introduced to powers of attorney as part of their legal studies or civic education at school. Mr Robert Bolch, a Justice of the Peace who made a submission to the Inquiry, recommended that younger people be educated about powers of attorney at important stages in their lives such as when they marry. 1244

Mr Worsnop acknowledged that educating young people about powers of attorney is very difficult but suggested that messages should be kept 'matter-of-fact and straightforward'. 1245

The Committee believes that young Victorians should be encouraged to plan for the future by making powers of attorney. The Committee acknowledges that increasing the uptake of powers of attorney in this demographic will be extremely challenging, as many young people are reluctant to contemplate future death and disability. The Committee believes that a targeted education campaign about powers of attorney should be developed in conjunction with young people and youth organisations and be disseminated though media that appeal to young people.

#### Recommendation 83: Education about powers of attorney for young people

The Committee recommends the Victorian Government develop targeted information and resources about powers of attorney for young people. This information should be developed in consultation with young people and youth organisations and disseminated through youth-focused media.

Tom Worsnop, Chair, Victorian Coalition of Acquired Brain Injury Service Providers, *Transcript of evidence*, Melbourne, 30 March 2010, 2. See also ibid, 6; Victorian Coalition of Acquired Brain Injury Service Providers, *Submission 71*, 3; Robert C Bolch, JP, *Submission 3*, 12.

Office of the Public Advocate, *Submission 9*, 24-25; Email from John Chesterman, Manager, Policy and Education, Office of the Public Advocate, to Executive Officer, Victorian Parliament Law Reform Committee, 3 June 2010.

Office of the Public Advocate, Submission 9, 24-25.

Val Johnstone, Transcript of evidence, Melbourne, 30 March 2010, 8. See also Gary Haley, Transcript of evidence, above n 1217, 5-6.

<sup>&</sup>lt;sup>1244</sup> Robert C Bolch, JP, Submission 3, 12.

<sup>&</sup>lt;sup>1245</sup> Tom Worsnop, *Transcript of evidence*, above n 1240, 6.

### **Members of the Aboriginal community**

There is very limited information available about the use of powers of attorney by members of the Aboriginal community. However, the available data suggests that power of attorney arrangements are not often used by Aboriginal people. 1246

The Committee did invite Aboriginal organisations to participate in this Inquiry; however, no information was received.

Only one participant in this Inquiry commented on the level of use of powers of attorney by members of the Aboriginal community. Moreland Community Legal Centre suggested there should be specific strategies to increase awareness of powers of attorney in the Aboriginal community. 1247

The lack of evidence means that it is not possible for the Committee to report on how members of the Aboriginal community view power of attorney arrangements.

The Committee recommends that the Victorian Government should work with the Aboriginal community to investigate current levels of use of enduring powers of attorney and to identify any barriers to use, with a view to implementing strategies to assist members of the Aboriginal community to be appropriately supported in the event of impaired decision-making capacity.

### Recommendation 84: Research about use of powers of attorney by members of the Aboriginal community

The Committee recommends the Victorian Government, in consultation with Aboriginal people and organisations, conduct research into the level of use of enduring powers of attorney by members of the Aboriginal community. This research should identify ways in which the principles underpinning powers of attorney arrangements may be applied to better support members of the Aboriginal community in the event of impaired decision-making capacity.

### Other groups

Evidence also suggests that some other specific groups in the community may also lack awareness of powers of attorney. Ms Catherine Leslie, the Mental Health Legal Centre's Legal and Policy Officer, told the Committee that the Centre's clients:

often were not aware of the opportunities to put in place some of those, if you like, more supported decision-making documents like an enduring power of attorney which perhaps may well have avoided the necessity, or the deemed necessity, of say, a substitute decision-maker being appointed by the tribunal to make financial or other health-care decisions for that person. 1248

Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 1200, 130, 132; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 1199, 4.

Moreland Community Legal Centre Inc, Submission 51, 3.

<sup>&</sup>lt;sup>1248</sup> Catherine Leslie, *Transcript of evidence*, above n 1197, 2-3.

Research has identified that people on lower incomes, people with physical disabilities and gay, lesbian, bisexual or transgender people have limited understanding of powers of attorney and are less likely to make these arrangements. 1249

OPA suggested there should be a broad study of the uptake of powers of attorney, focusing on the demographic profile of principals to inform the development of a targeted education campaign. <sup>1250</sup>

The Committee does not feel it has received sufficient evidence about the need to target information and education about powers of attorney to groups other than those discussed earlier in this section. However, it recognises that there would be significant benefit in better understanding the demographic profile of those making powers of attorney. This would enable information and education about powers of attorney to be aimed at groups with low levels of use.

#### Recommendation 85: Research about who makes powers of attorney

The Committee recommends the Victorian Government conduct a study of the demographic profile of people making powers of attorney in Victoria to inform the development of information and education about powers of attorney.

### 9.4 Educating lawyers about powers of attorney

Many participants in the Inquiry called for more education for lawyers about powers of attorney. <sup>1251</sup>

The Committee heard that lawyers are in a good position to promote the use of powers of attorney. State Trustees informed the Committee that when it gives clients estate planning advice, it also advises them of the option of making a power of attorney 'as a matter of course'. Ms Pat Long, a participant in the Committee's Seniors' Forum, stated, 'I have an enduring power of attorney which was offered to me when I made my will with the solicitor and I thought that was a very good idea ...' 1253

Another reason for educating lawyers about powers of attorney is that they often draft these documents for clients. If lawyers have a thorough understanding of powers of attorney and the formal requirements, this will help reduce errors when the

Deborah Setterlund, Cheryl Tilse and Jill Wilson, 'Older people and substitute decision making legislation', above n 1200, 130-131; Deborah Setterlund, Cheryl Tilse and Jill Wilson, Substitute decision making and older people, above n 1199, 5; House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 71; Department of Planning and Community Development, Victoria, Ageing in Victoria Discussion paper (2008), 4.

Office of the Public Advocate, Submission 9, 24.

Seniors Rights Victoria, Submission 38, 36; Sue Field, NSW Trustee and Guardian Fellow in Elder Law, University of Western Sydney, Submission 61, 3; L Richmond, Transcript of evidence, Melbourne, 30 March 2010, 8; Jill Linklater, Policy and Advocacy Officer, Alzheimer's Australia Vic, Transcript of evidence, Melbourne, 22 October 2009, 2; D Constantine, Transcript of evidence, Melbourne, 30 March 2010, 7; Hume Riverina Community Legal Service, Submission 36, 4.

<sup>1252</sup> State Trustees Limited, Submission 58, 4-5.

Pat Long, Transcript of evidence, Melbourne, 30 March 2010, 4.

documents are made. It will also ensure that lawyers are in a good position to advise clients about the implications of powers of attorney. Mr Gary Haley, a participant in the Committee's Seniors' Forum suggested that all law firms should have a lawyer who is a specialist in powers of attorney. 1254

Unfortunately, some lawyers do not appear to have a comprehensive understanding of powers of attorney law. The Federation of Community Legal Centres submission observed, 'Even within the legal profession, there is some confusion regarding the various instruments and their effects.' Case study 21 provides an example of a lawyer who did not understand the difference between an enduring and non-enduring power of attorney.

### Case study 21: 'The solicitor should have advised them to create an enduring power of attorney, 1256

'A rural community legal centre was contacted by [a] woman seeking information and assistance on behalf of her father. The woman's father had decided to grant her power of attorney, prior to his admission to the local hospital for major surgery.

Several days previously, the father and daughter had seen a private solicitor. The solicitor had helped them create a general power of attorney.

The daughter advised the CLC [community legal centre] that they had requested a power of attorney that would take effect if the father lost capacity. In this case, the solicitor should have advised them to create an enduring power of attorney.

A Queensland study found that a lawyer had witnessed half of the enduring power of attorney documents that were subsequently found to be invalid by the Guardianship and Administration Tribunal in that state on the basis that the principal did not have the capacity to make a power of attorney. The authors observed, 'Legal practitioners, more than any other category of witness, should be familiar with and understand the test for capacity that is prescribed in the legislation.' 1257

Another concern raised by participants was that some lawyers do not appear to understand who they are acting for, and may act in situations where there is a conflict of interest. It is not uncommon for a principal to be taken to a lawyer's office by a family member or other person to make a power of attorney. 1258 Mr Peter Maloney, Chief Executive of Tasmania's Public Trustee, provided an example showing that some lawyers appear to be acting for the representative rather than the principal:

Gary Haley, *Transcript of evidence*, above n 1217, 7.

<sup>&</sup>lt;sup>1255</sup> Federation of Community Legal Centres (Victoria) Inc, Submission 47, 4.

<sup>&</sup>lt;sup>1257</sup> Lindy Willmott and Liza Windle, 'Witnessing EPAs: Empirical research' (2007) 27 Queensland Lawyer 238, 242. See also Margaret Brown and Suzanne Jarrad, above n 1191, 537.

John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), Submission 37, att, 9-10; L Richmond, Transcript of evidence, above n 1251, 18; Letter from Peter Maloney, Chief Executive Officer, Public Trustee, Tasmania, to Chair, Victorian Parliament Law Reform Committee, 3 November 2009, 2-3; Owen Ames and John Harley, 'Elder abuse: Being part of the solution, not part of the problem' (2001) 23(3) Law Society Bulletin 34, 35.

I know of one instance where a daughter came into a lawyer and asked the lawyer to prepare an EPA [enduring power of attorney] document for her mother, appointing herself as the Attorney. The daughter then took the prepared document to her mother for signature, which was returned to the lawyer. The lawyer in this instance never even saw his client. 1259

Ms Lucie O'Brien, a Policy Officer with the Federation of Community Legal Centres, told the Committee that lawyers' lack of understanding in this area is not surprising, as there is little education about powers of attorney provided to lawyers both at university and once they commence working. She stated, 'I did not study powers of attorney at university; I do not think I was the only one to know nothing at all about it on graduating from university.' 1260

Victorian lawyers are required to undertake continuing professional development each year in order to maintain their practicing certificate. The Law Institute of Victoria informed the Committee that it does currently offer educational units that are relevant to powers of attorney as part of its continuing professional development program. For example, in 2009 the Institute hosted a forum on *Incapacity and representing your client*. 1262

The Federation of Community Legal Centres suggested that all lawyers should be required to undertake training on substitute decision making as part of their continuing professional development. The Federation suggested that a compulsory unit on substitute decision making should include:

- determining capacity
- taking instructions from older clients and clients with a mental illness or disability
- substitute and supported decision making
- representatives' duties. 1263

The Federation suggested that this unit could be developed by the Victorian Government in conjunction with the Law Institute of Victoria and relevant specialists such as the Mental Health Legal Centre and Seniors Rights Victoria.

However, both the Law Institute and the Legal Services Board (the body which endorses the continuing professional development rules for lawyers) opposed mandatory education on this issue. Both organisations expressed the view that the rules currently requiring lawyers to undertake annual continuing professional

<sup>&</sup>lt;sup>1259</sup> Letter from Peter Maloney, above n 1258, 2-3.

Lucie O'Brien, *Transcript of evidence*, above n 1207, 6. See also Jeff Giddings and Jody Thomas, 'Teaching the law as it relates to older people' (2002) 27(2) *Alternative Law Journal* 78, 80-81.

Law Institute of Victoria, Submission 41S, 1; Federation of Community Legal Centres (Victoria) Inc, Submission 47S, 1; Legal Services Board, Continuing professional development rules 2008 (2008); Law Institute of Victoria, Law Institute continuing professional development rules 2008 (2008).

Law Institute of Victoria, Submission 41S, 2.

Federation of Community Legal Centres (Victoria) Inc, *Submission 47S*, 1-2. See also Lucie O'Brien, *Transcript of evidence*, above n 1207, 6.

development do not allow specific compulsory topics to be nominated. The Law Institute stated:

Lawyers are expected to identify matters which are relevant to their own practice of law and practice needs ... the topic of substitute decision making will not be relevant to all lawyers, particularly where they deal predominantly with corporate clients. <sup>1264</sup>

In chapter four of this report the Committee noted the value of legal advice to assist people to make powers of attorney and recommended that such advice be made more accessible. The Committee believes it is important that lawyers advising people about powers of attorney, and potentially also drafting and witnessing these documents, have a good understanding of the relevant law. The Committee also recognises that many lawyers are well placed to encourage more members of the community to make powers of attorney.

While recognising the value in educating lawyers about powers of attorney and substitute decision making more generally, the Committee does not believe that such education should be compulsory, as many lawyers do not provide services in this area. However, the Committee's view is that information and education about powers of attorney for lawyers should be available as broadly as possible, including on the curriculum of law schools and as part of continuing professional development. The Committee recognises the significant role that professional organisations such as the Law Institute will play in developing educational resources and training programs about powers of attorney for lawyers.

### Recommendation 86: Educating lawyers about powers of attorney

The Committee recommends the Victorian Government:

- a) develop targeted information and resources about powers of attorney for lawyers. This information should be developed in consultation with legal organisations and disseminated through appropriate media
- b) encourage law schools to incorporate substitute decision making into their curriculum, including education about powers of attorney
- c) encourage the Law Institute of Victoria to develop a training program on powers of attorney for lawyers and encourage lawyers to participate in this training as part of their continuing professional development.

-

Law Institute of Victoria, *Submission 41S*, 1. See also Letter from Michael McGarvie, Chief Executive Officer, Legal Services Board, to Chair, Victorian Parliament Law Reform Committee, 1 March 2010.

# 9.5 Educating the health and community sectors

Evidence to the Inquiry highlighted that it is important for people working in the health and community sectors to have a thorough understanding of powers of attorney.

Firstly, those working in the health and community sectors may be in a good position to encourage people to make powers of attorney and to refer them to services that can help make these arrangements. Older people, for example, may be far more likely to regularly see health care professionals than they are to consult with lawyers. Ms Leslie from the Mental Health Legal Centre told the Committee, 'We know from the experience of clients that often health care professionals can play a significant gate-keeping role in the amount of information that a person is receiving ...' Many participants in the two forums the Committee held for seniors and members of culturally and linguistically diverse communities also emphasised that health care providers can play an important role in educating these communities about powers of attorney. 1266

The Committee heard there is potential to improve the performance of health care professionals in educating patients about powers of attorney. Ms Chapman, who gave evidence on behalf of Palliative Care Victoria, stated, 'GPs and primary care providers can have much more of a role in saying, 'Look, this is a really good thing', whereas they wait until the crisis has occurred.' Similarly, one South Australian study involving medical professionals who diagnose people with early memory loss found that these professionals often do not provide education and referrals about powers of attorney, even though they are in a good position to do so. 1268

Secondly, members of the health and community sectors are often required to act on powers of attorney, such as when a principal who has impaired decision-making capacity is in hospital or is a resident in an aged care facility. Participants in the Inquiry emphasised that in these situations the service provider's employees need to thoroughly understand the power of attorney presented and the powers it confers. <sup>1269</sup>

As was noted in chapter six, many professionals in these areas do not have a good grasp of power of attorney laws. The acceptance by service providers of an enduring power of attorney (financial) to make personal and lifestyle decisions on behalf of

Catherine Leslie, *Transcript of evidence*, above n 1197, 6. See also Letter from Leonie Schween, Legal & Privacy Officer, Royal District Nursing Service, to Executive Officer, Victorian Parliament Law Reform Committee, 14 January 2010.

Elizabeth Ignys, *Transcript of evidence*, above n 1231, 12; Eva Wakim, *Transcript of evidence*, above n 1221, 13; Ross Barnett, *Transcript of evidence*, above n 1236, 13-14; Naderah Edwards, *Transcript of evidence*, above n 1234, 14; Toni Higgins, *Transcript of evidence*, above n 1236, 6; Rosemarie Draper, *Transcript of evidence*, above n 1221, 15.

Jenny Chapman, *Transcript of evidence*, above n 1197, 5. See also Kevin Larkins, *Transcript of evidence*, above n 1197, 5; Letter from Leonie Schween, above n 1265; Palliative Care Victoria, *Submission 70*, 1-2; Pat Long, *Transcript of evidence*, above n 1253, 4; Gary Haley, *Transcript of evidence*, above n 1217, 8; Elizabeth Ignys, *Transcript of evidence*, above n 1231, 12; Eva Wakim, *Transcript of evidence*, above n 1221, 4; Rosemarie Draper, *Transcript of evidence*, above n 1221, 4.

Margaret Brown and Suzanne Jarrad, above n 1191, 534-535.

St Vincent's Hospital, Submission 19, 6; Letter from Leonie Schween, above n 1265; Seniors Rights Victoria, Submission 38, 35-36; Aged & Community Care Victoria, Submission 53, 2. See also ibid, 537.

the principal, was of particular concern to Inquiry participants. The Federation of Community Legal Centres submitted that abuse of powers of attorney, such as refusing visitors to a principal in an aged care facility on the basis of an enduring power of attorney (financial), would be less prevalent if stakeholders such as nursing home staff had a better understanding of these documents and the powers they give representatives. 1270

St Vincent's Hospital described powers of attorney law as 'overly complex' and has produced a written policy for medical staff because, 'No clinician is going to be able to retain this knowledge'. Ms Lucy Cordone, the hospital's General Counsel, stated:

I think even though we have got policies and guidance for the doctors, it is quicker for them to pick up the phone and speak to us rather than read a document, because they are very time poor. They do not have much education and training about legal documents. They are very confused with the terminology. 1272

Dr Kristen Pearson, President of the Victorian Division of the Australian & New Zealand Society for Geriatric Medicine told the Committee that programs such as Respecting Patient Choices are providing education about powers of attorney (medical treatment) for medical professionals:

But increasingly there is more of a focus on educating people about the role of a medical power of attorney and the concept of the person responsible as understood in the Medical Treatment Act. I think there is a lot of education rolling out through the health services on that, and I think that hopefully that will include the doctors. I think increasingly health services will be asking people when they present 'Who is your medical power of attorney?' and looking for those documents, which will be prompting everyone. So instead of talking so much about next of kin or asking 'Who will we ring up if there is an issue?', they will increasingly be systemically asking for the medical power of attorney. 1273

The Committee did not receive any evidence that such education is being provided to health care professionals in relation to the types of powers of attorney under review in this Inquiry.

Participants in this Inquiry were generally strongly supportive of providing more education about powers of attorney for GPs and staff in hospitals and aged care facilities. 1274

Lucy Cordone, General Counsel, St Vincent's Hospital, Transcript of evidence, Melbourne,
 14 December 2009, 3. See also Julian Gardner, Transcript of evidence, above n 1198, 6; David Goldberg,
 Solicitor and Senior Adviser, Australian Medical Association Victoria, Transcript of evidence, Melbourne,
 17 December 2009, 6.

Federation of Community Legal Centres (Victoria) Inc, Submission 47, 4. See also Seniors Rights Victoria, Submission 38, 35.

St Vincent's Hospital, Submission 19, 4.

Kristen Pearson, President, Australian & New Zealand Society for Geriatric Medicine, Victorian Division, *Transcript of evidence*, Melbourne, 22 October 2009, 7.

Hume Riverina Community Legal Service, *Submission 36*, 4; Health Services Commissioner, *Submission 46*, 4; Seniors Rights Victoria, *Submission 38*, 36; Letter from Leonie Schween, above n 1265; Office of the Public Advocate, *Submission 9*, 24.

Ms Sally Costar, Manager, Patient and Family Services at Caulfield Hospital, Alfred Health, suggested that training about powers of attorney 'could be very nicely incorporated into professional development across various professional groups. Most of us have to do a certain amount of professional development in order to maintain registration and I feel this is an absolutely ideal area to be incorporated.' Ms Cordone suggested it would be opportune to conduct such training in February or March when hospitals receive an influx of new doctors. She opined that such training should be simple, practical and scenario-based. 1276

A recent review in South Australia recommended compulsory training about powers of attorney for those working in the health and community sectors. While generally endorsing increased education for people in these spheres, no participants in this Inquiry suggested that this training should be mandatory.

The Committee agrees that those working in the health and community sectors need to have a good working knowledge of powers of attorney. The Committee believes it would be useful for people employed in these sectors to receive education and information about:

- referring clients to services that can provide advice and support to make powers of attorney
- the different types of powers of attorney and the powers they grant
- principals' rights
- the duties of representatives
- how to assess capacity
- identifying and responding to abuse (discussed in detail in chapter seven).

This information and training should be developed in collaboration with community and health sector organisations. It should particularly be targeted at GPs and staff in hospitals and aged care facilities as these professionals may be most likely to encounter powers of attorney or be in a position to refer someone to make a power of attorney.

Sally Costar, Manager, Patient and Family Services, Caulfield Hospital, Alfred Health, *Transcript of evidence*, Melbourne, 14 December 2009, 7. See also Letter from Leonie Schween, above n 1265.

Lucy Cordone, Transcript of evidence, above n 1272, 8. See also Kristen Pearson, Transcript of evidence, above n 1273, 7.

Advance Directives Review Committee, Planning ahead: Your health, your money, your life. Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008), 56.

# Recommendation 87: Educating the health and community sectors about powers of attorney

The Committee recommends the Victorian Government:

- a) develop targeted information and resources about powers of attorney for those working in the health and community sectors. This information should be developed in consultation with community and health sector organisations and disseminated through appropriate media
- b) develop a training program on powers of attorney for workers in the community and health sectors in conjunction with relevant professional associations and encourage workers in these sectors to participate in this training
- c) encourage tertiary institutions to incorporate information about powers of attorney into relevant courses for the health and community sectors.

# 9.6 Increasing the recognition and acceptance of powers of attorney

The Committee heard that sometimes valid power of attorney documents are not recognised by third parties, particularly government agencies and financial institutions. This can give rise to significant distress and inconvenience. In this section the Committee explores ways of increasing the widespread recognition and acceptance of powers of attorney.

## 9.6.1 Government agencies

Some government agencies do not recognise power of attorney arrangements. In particular, there was concern among some Inquiry participants that Centrelink does not routinely accept powers of attorney.

Centrelink clients can nominate a person or organisation to manage their affairs with Centrelink on their behalf. Centrelink is not required to recognise powers of attorney, guardianship and administration orders made under state legislation, nor notify a representative that a principal intends to appoint another person as a nominee. If a client has impaired decision-making capacity and is unable to make a nomination, Centrelink will not accept a power of attorney for a nominee arrangement without examining the document first.

The House of Representatives Standing Committee on Legal and Constitutional Affairs report on *Older people and the law* found that enduring powers of attorney

<sup>&</sup>lt;sup>1278</sup> Social Security (Administration) Act 1999 (Cth) ss 123B, 123C, 123D, 123O.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 101, 103; Alzheimer's Australia Vic, *Submission 32*, 34.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 102.

(financial) should be recognised by Centrelink for the purposes of nominee arrangements and that representatives should be advised if a principal seeks to appoint another person as a nominee. 1281 However, evidence presented to the current Inquiry suggests that little has changed since that report was released in 2007.

Mr David Davis, a solicitor who gave evidence on behalf of the Law Institute of Victoria told the Committee, 'Centrelink will say, "We don't recognise the power of attorney because we have on record someone else, who is really responsible for the information we are seeking". 1282 Seniors Rights Victoria suggested that the implementation of a registration system for powers of attorney would assist government agencies such as Centrelink to recognise these documents more readily. 1283

Evidence to the Older people and the law inquiry suggested that other Australian Government Agencies such as the Department of Veterans Affairs and the Department of Health and Ageing are also reluctant to recognise power of attorney arrangements. 1284 Participants in the present Inquiry did not identify any Commonwealth Government agencies other than Centrelink that did not routinely recognise powers of attorney.

In the Committee's view it is unacceptable that validly executed documents that are specifically designed to help people to plan for future impaired decision-making capacity are not recognised by some Australian Government agencies. The nonrecognition of these documents by Centrelink is particularly of concern because many older Victorians who rely on representatives to manage their assets receive pensions and other benefits.

The Committee believes that the Victorian Government should advocate for the automatic acceptance of validly executed powers of attorney by Australian Government agencies. It acknowledges that the introduction of a registration system, as recommended in the previous chapter, should assist in the wider recognition of powers of attorney created in Victoria.

#### Recommendation 88: Recognition of powers of attorney by Australian **Government agencies**

The Committee recommends the Victorian Government, through the Standing Committee of Attorneys-General, advocate that all Australian Government agencies implement policies and practices to recognise powers of attorney that are validly created under state or territory law.

<sup>&</sup>lt;sup>1282</sup> David Davis, Elder Law Committee, Law Institute of Victoria, *Transcript of evidence*, Melbourne, 1 October 2009, 5. See also Seniors Rights Victoria, Submission 38, 33-34; Alzheimer's Australia Vic, Submission 32, 3-4.

Seniors Rights Victoria, Submission 38, 34.

House of Representatives Standing Committee on Legal and Constitutional Affairs, above n 1194, 103.

# Case study 22: A bank's failure to recognise a general power of attorney 1285

'In around 2004, Mr and Mrs T were planning a trip abroad for several months. They requested their eldest son, Sam (a lawyer), to attend to their financial matters until they returned from abroad. ... To facilitate this Sam, suggested that Mr and Mrs T execute a General Power of Attorney (GPOA) with Sam as the Donee to provide sufficient authority to the banks and financial institutions.

Sam made arrangements under the GPOA with Mr and Mrs T's bank (ABC Bank) to arrange for access to Mr and Mrs T's bank accounts. Initially, the bank refused to deal with Sam despite the existence of the GPOA but after numerous tiring and lengthy discussions with senior management, Sam was permitted to deal with Mr T's VISA account.

In around 2006 Mr and Mrs T again went abroad for several months and Sam was tasked with managing their financial affairs as a Donee of the pre-existing GPOA. Very soon after Mr and Mrs T's departure in 2006, bills and accounts started to come in. Sam contacted ABC Bank to notify it of his intention to access Mr and Mrs T's bank accounts for the purposes of making repayments etc. in accordance with the Donors' requests. However, ABC Bank again refused to recognise the GPOA ...

While ABC Bank acknowledged it had had the GPOA, listing Sam as a Donee, the bank erroneously took the view that this power only attached to Mr T's VISA account and to no other account Mr and Mrs T had in their names at the bank. ... The GPOA deed was general as the name suggested and not limited to a specific task or in any other way. It appears ... staff due to ignorance incorrectly limited the scope to which the GPOA could be used.

This matter was eventually escalated to ABC Bank's internal legal counsel who also refused to provide Sam with the required access. Legal counsel accepted that a GPOA was on foot, but the bank had since lost the copy that was provided to it ...

By this time, VISA statements requesting payments as high as \$12,000 per month were being received and required payment. These outstanding balances were incurring high interest rates ...

In desperation, Sam set up a phone banking account ... This in turn gave Sam access to his parent's accounts and enabled him to make the necessary payments ... Around one week later, this facility was closed down without notice ... The ABC Bank made every effort to frustrate a simple process that had been set up by Mr and Mrs T to enable their son to effectively and efficiently deal with their finances. Sam often spent hours either on the telephone or attending the bank to undertake the necessary functions he was tasked with while his parents were away.

Sam realised that staff at the ABC Bank refused him access because they had a limited understanding about the scope of an unlimited GPOA. The bank was ill equipped to ... address matters appropriately regarding POAs ... and this resulted in time delays, interest expenses and a high level of frustration.'

<sup>1285</sup> Case study provided to the Committee by a person who requested that his/her name be withheld.

#### 9.6.2 Financial institutions

Some participants in the Inquiry highlighted that service providers, particularly banks, may sometimes not recognise a valid power of attorney. Latrobe Regional Hospital's submission stated:

Patients' families have reported that Powers of Attorney drawn up using the **Take Control kit** have <u>not</u> been accepted by the major banks, telephone companies and other utilities and the family members have been told that the institution will only accept Powers of Attorney drawn up by a Solicitor. <sup>1286</sup>

The Trustee Corporations Association of Australia submitted that some financial institutions have 'over the top' and inconsistent processes for validating powers of attorney. These processes may cause significant delays in completing transactions. 1288

Mr D Constantine, a participant in the Committee's Seniors' Forum, told the Committee that his parents had appointed him as their representative when they went overseas. In his experience, financial institutions were reluctant to accept his authority:

what I found quite frustrating was talking to their banks and effectively transferring funds across to other accounts while they were overseas and to pay for things and having people on the end of the line or even within the bank itself saying, 'You only have power of attorney for one particular account'. I went to great pains to explain to them that it was a general power of attorney and that meant I could deal with their finances the way I wanted to. 1289

The Australian Bankers' Association acknowledged that banks need to develop standard processes for validating and acting on powers of attorney. It suggested these processes could include:

- compliance procedures to ensure the authenticity of the documents and legitimacy of the representatives
- clear acceptance policies about when the bank will or will not accept a power of attorney
- registration of power of attorney documents, so banks can rely on the permanent record rather than requiring certified copies to be produced for each dealing. 1290

The Association also recognised that it would be useful to develop a training program for bank staff which would include information about matters such as

<sup>&</sup>lt;sup>1286</sup> Latrobe Regional Hospital, Submission 44, 1. See also Ethnic Communities' Council of Victoria, Submission 23, 2.

<sup>&</sup>lt;sup>1287</sup> Trustee Corporations Association of Australia, Submission 27, att, 4.

<sup>1288</sup> State Trustees Limited, Submission 58, 16.

D Constantine, *Transcript of evidence*, above n 1251, 6.

Australian Bankers' Association Inc, Submission 55, 8. See also Diane Tate, Director, Financial Services, Corporations, Community Policy, Australian Bankers' Association Inc, Transcript of evidence, Melbourne, 17 December 2009, 8.

financial abuse, identification and verification processes and the various power of attorney documents and the powers they confer. [129]

Not all Inquiry participants had encountered problems having powers of attorney recognised. Ms Janice Reynolds, who attended the Committee's Seniors' Forum stated:

I used to take my mother to the bank and she would sign the form for her fees for the nursing home. When it got to the stage when she could no longer sign, I just produced the document and the bank accepted that I was the person to sign the withdrawal forms for the cheques for the nursing home. There were no problems at all. 1292

The Committee's view is that it is important that powers of attorney are recognised as widely as possible. While noting the difficulties experienced by organisations that operate nationally, which must deal with documents created under inconsistent state-based law, the Committee believes it is imperative that service providers, and in particular the financial services sector, develop and implement standard policies and procedures for accepting powers of attorney.

The Committee recognises that the implementation of a registration system as recommended in the last chapter will make it easier for third parties such as financial institutions to verify a power of attorney document created in Victoria.

# Recommendation 89: Enhancing recognition of powers of attorney by financial institutions

The Committee recommends the Victorian Government encourage the banking and finance sector to develop policies and procedures for accepting powers of attorney and provide appropriate training for staff.

## 9.7 Educating about changes to the law

In this report the Committee suggests a fresh approach to powers of attorney in Victoria. It has recommended a new consolidated legislative structure, simplified forms for creating powers of attorney, improved accountability requirements and a registration system for power of attorney documents.

In chapter three the Committee observed there are two factors critical to the successful introduction and ongoing operation of any new legislation: education about the legislative changes and monitoring and evaluation of the legislation. In that chapter the Committee recommended periodic review of the proposed new Powers of Attorney Act. The Committee now turns its attention to considering education about the legislative changes.

<sup>&</sup>lt;sup>1291</sup> Australian Bankers' Association Inc, Submission 55, 10. See also Diane Tate, Transcript of evidence, above n 1290, 4.

Janice Reynolds, *Transcript of evidence*, Melbourne, 30 March 2010, 19. See also David Fara, *Transcript of evidence*, above n 1202, 4; Jean Thomas, *Transcript of evidence*, Melbourne, 30 March 2010, 19.

The Committee believes it is essential that the introduction of the new legislation is accompanied by an education campaign to alert key stakeholders about the reforms. This is in addition to the ongoing education proposed elsewhere in this report for principals (chapter four), representatives (chapter six) and the general community and key professionals (this chapter).

The aim of the education campaign would be to support the new legislation by alerting those who regularly deal with powers of attorney about the altered requirements. This will help ensure there is a strong recognition of and compliance with the new requirements and prevent situations such as outdated forms being used or representatives not being informed of their obligations.

The target audience of this campaign would be those professionals involved in creating or advising on powers of attorney, for example lawyers, financial advisors and trustee corporations. The campaign should be developed by the Victorian Government in conjunction with the relevant professional associations. The ongoing education campaigns for lawyers and other professionals recommended in this chapter should build on and supplement this initial campaign.

# Recommendation 90: Educating key professionals about the new Powers of Attorney Act

The Committee recommends the Victorian Government, in conjunction with relevant professional bodies, conduct an education campaign to inform key professionals about the new Powers of Attorney Act. This should be targeted at those professionals who are likely to help people make powers of attorney or provide advice about powers of attorney.

# **Chapter 10: Conclusion**

Powers of attorney are valuable tools that empower people to plan for their future financial, health and lifestyle needs. The recommendations in this report aim to make these arrangements accessible to more Victorians and to provide safeguards to ensure that principals' rights are protected to the maximum extent possible.

The Committee proposes a new Powers of Attorney Act. This statute will provide a stand-alone framework for all laws in Victoria relating to general (non-enduring) powers of attorney, enduring powers of attorney (financial) and enduring powers of attorney (guardianship). The Act will provide the foundation for a streamlined system, including simple consistent terminology and simplified forms for creating powers of attorney. The new Act will be underpinned by general principles that will ensure that principals' human rights are at the core of the exercise of all powers and functions under the Act on behalf of a principal with impaired decision-making capacity.

Powers of attorney can be extremely empowering for members of the community, allowing them to plan for their future needs. While most power of attorney documents work well, the Committee was told that abuse is not rare. Abuse of these private documents is often perpetrated by a family member and is extremely difficult to detect.

There is no single strategy for preventing the abuse of powers of attorney. The Committee has recommended a multi-faceted approach, providing protection both at the time a power of attorney document is created, and when the powers are exercised. The Committee has also struck a delicate balance between protecting principals and ensuring that safeguards do not deter principals and representatives from entering into these arrangements.

Key elements of the Committee's strategy to protect principals from abuse include:

- strengthening witnessing requirements and limiting the class of people who may act as witnesses
- allowing the principal to appoint a 'personal monitor' to oversee the operation of an enduring power of attorney
- providing more clarity about gifts made by representatives using the principal's funds.

The Committee heard that many representatives and those working for service providers, such as banks, do not have a good understanding of the scope of representatives' powers. The recommendations in this report aim to provide more certainty for all parties about what powers a representative has under each type of power of attorney. A key part of the Committee's approach is to provide more support and guidance to persons taking on the role of representatives, including a set of principles to clarify how they should approach decision making.

The concept of capacity is central to powers of attorney, relevant both at the time a document is created and, in some cases, when a document is activated. However, this

Inquiry found significant confusion about what is capacity and how and by whom it should be assessed. The Committee has recommended providing greater certainty about capacity by setting out in legislation clear definitions of capacity and impaired decision-making capacity and a presumption of capacity and providing resources to assist all those involved in or affected by decisions about capacity.

Many participants in this Inquiry strongly supported the introduction of a registration system for powers of attorney. While the Committee heard that registration has no impact on the abuse of power of attorney documents, it believes registration will assist with the ready location, verification and validation of enduring power of attorney documents. Therefore, it has proposed a system of mandatory registration of all documents creating and revoking enduring powers of attorney.

The final component of the Committee's approach is a comprehensive education campaign to ensure that members of the community, representatives, principals, personal monitors and professionals such as lawyers and those working in the aged care, health and community sectors have a good understanding of powers of attorney. The proposed education campaign aims to increase awareness of the benefits of powers of attorney, increase understanding of the powers and duties of representatives, promote wider acceptance of power of attorney documents, as well as encouraging the greater detection and reporting of abuse.

With Australia's population rapidly ageing, and increasing rates of dementia and other disability, powers of attorney have the potential to provide benefits to even more Victorians in the future. The Committee believes that the framework for powers of attorney provided in this report will empower more Victorians to make plans for their future, in the knowledge that their rights will be protected and they will be safe from abuse.

**Adopted by the Law Reform Committee** 

12 August 2010

# **Appendix A: List of submissions**

	Name of individual or organisation	Date received
1	Ms Kate Varty	18 June 2009
2	Mr Ronald T Jones JP	29 June 2009
3	Mr Robert C Bolch JP	7 July 2009
4	Mr Glen Cooper	13 July 2009
5	Ms Audrey Cooke	16 July 2009
6	Ms Shirley Glover	16 July 2009
7	Professor Phillip Hamilton	20 July 2009
8	The Royal Women's Hospital	27 July 2009
9	Office of the Public Advocate	4 August 2009
10	Combined Pensioners and Superannuants Association of Victoria Inc	6 August 2009
11	Ms Alex Pottage	7 August 2009
12	Association of Independent Retirees (AIR) Ltd (Victorian Division)	10 August 2009
13	Respecting Patient Choices – Advance Care Planning	12 August 2009
14	Action on Disability in Ethnic Communities	18 August 2009
15	Australian Association of Gerontology Inc	19 August 2009
16	Dr Ian F Turnbull	20 August 2009
17	Ms Shirley Logan	20 August 2009
18	Credit Union Australia Ltd	20 August 2009
19	St Vincent's Hospital	20 August 2009
20	Peter MacCallum Cancer Centre	20 August 2009
21	Mr Julian Gardner	20 August 2009
22	Murray Mallee Community Legal Service	20 August 2009
23	Ethnic Communities' Council of Victoria	20 August 2009
24	Hunt & Hunt	20 August 2009
25	Royal District Nursing Service	20 August 2009
26	Royal College of Nursing Australia	20 August 2009
27	Trustee Corporations Association of Australia	21 August 2009
28	Royal Victorian Association of Honorary Justices	21 August 2009
29	Australian & New Zealand Society for Geriatric Medicine, Victorian Division	21 August 2009

	Name of individual or organisation	Date received
30	Southern Health	21 August 2009
31	Moores Legal Pty Ltd	21 August 2009
32	Alzheimer's Australia Vic	21 August 2009
33	The Institute of Legal Executives (Victoria)	21 August 2009
34	Mr George Madden	21 August 2009
35	Mr Grant Sturgeon	21 August 2009
36	Hume Riverina Community Legal Service	21 August 2009
37	Mr John Billings, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT) (Note Mr Billings commenced as Deputy President of the Occupational and Business Regulation List on 9 November 2009)	21 August 2009
38	Seniors Rights Victoria	24 August 2009
38S	Seniors Rights Victoria – supplementary submission	25 November 2009
39	Council on the Ageing Victoria	24 August 2009
40	The Victorian Bar	24 August 2009
41	Law Institute of Victoria	24 August 2009
41S	Law Institute of Victoria – supplementary submission	22 December 2009
42	Victoria Legal Aid	24 August 2009
43	SouthPort Community Legal Service Inc	24 August 2009
44	Latrobe Regional Hospital	24 August 2009
45	MF Spottiswood	24 August 2009
46	Health Services Commissioner	24 August 2009
47	Federation of Community Legal Centres (Victoria) Inc	25 August 2009
47S	Federation of Community Legal Centres (Victoria) Inc – supplementary submission	19 October 2009
48	Ministerial Advisory Council of Senior Victorians	25 August 2009
49	Financial Planning Association of Australia Ltd	25 August 2009
50	St Kilda Legal Service Co-Op Ltd	25 August 2009
51	Moreland Community Legal Centre Inc	26 August 2009
52	Victorian Equal Opportunity and Human Rights Commission	28 August 2009
53	Aged & Community Care Victoria	28 August 2009
54	National Seniors Australia	31 August 2009
55	Australian Bankers' Association Inc	31 August 2009

	Name of individual or organisation	Date received
56	Dr John Myers	3 September 2009
57	Ms Jeni Lee	3 September 2009
58	State Trustees Limited	7 September 2009
59	Mental Health Legal Centre Inc	8 September 2009
60	Mr James Roughley	8 September 2009
61	Ms Sue Field, NSW Trustee and Guardian Fellow in Elder Law, University of Western Sydney	11 September 2009
62	Mr Matthew Hughes	14 September 2009
63	Elder Rights Advocacy	16 September 2009
64	Mr Ange Kenos JP	25 September 2009
65	Carers Victoria	25 September 2009
66	Alfred Health	24 November 2009
67	Mr Lionel H Parrott	26 November 2009
68	Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society	11 December 2009
69	Australian Medical Association (Vic) Ltd	16 December 2009
70	Palliative Care Victoria	17 December 2009
71	Victorian Coalition of Acquired Brain Injury Service Providers	25 January 2010
72	Ms Rosemarie Draper, Aged Services, New Hope Foundation	30 March 2010
73	Ms Patricia Williams	9 April 2010
74	Australian Greek Welfare Society	19 April 2010
75	Mr D Constantine and Ms L Richmond	22 April 2010

# **Appendix B: List of witnesses**

# Public hearing, 1 October 2009 Room G2, 55 St Andrews Place, East Melbourne

Witness(es)	Organisation
Mr Julian Gardner	
Ms Jeni Lee, Elder Law Committee and Disability Law Committee	
Ms Laura Helm, Policy Adviser	Law Institute of Victoria
Mr David Davis, Elder Law Committee	
Ms Lynne Barratt, Elder Law Committee	
Professor Terry Carney, Professor of Law	The University of Sydney
Ms Lucie O'Brien, Policy Officer  Mr Dale Reddick, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service	Federation of Community Legal Centres (Victoria) Inc
Mr John Billings, Deputy President, Guardianship List (Note Mr Billings commenced as Deputy President of the Occupational and Business Regulation List on 9 November 2009)	Victorian Civil and Administrative Tribunal (VCAT)

# Public hearing, 22 October 2009 Room G2, 55 St Andrews Place, East Melbourne

Witness(es)	Organisation	
Dr John Chesterman, Manager, Policy and Education	Office of the Public Advocate	
Mr Tony Fitzgerald, Managing Director Ms Angela Burton, General Manager, Personal Financial Solutions Mr Alistair Craig, Senior Corporate Lawyer	State Trustees Limited	
Mr Lachlan Wraith, Senior Manager, Trusts & Estates, Equity Trustees Ltd Ms Claire Hausler, Manager, TCL Legal Services (Vic) Pty Ltd	Trustee Corporations Association of Australia	
Ms Dahni Houseman, Policy Officer  Ms Lauren Adamson, Acting Manager and Principal Solicitor, Seniors Rights Legal Clinic  Ms Elizabeth Samra, Lawyer	Seniors Rights Victoria	
Ms Debra Parnell, Policy Officer Ms Janet Wood, President Ms Kris Spark, Manager, Seniors Information Victoria	Council on the Ageing Victoria	
Ms Belinda Evans, Senior Advocate	Elder Rights Advocacy	
Mr David Fara, Deputy President	Association of Independent Retirees (AIR) Ltd (Victorian Division)	
Dr Kristen Pearson, President Dr Sue Connelly, Committee member	Australian & New Zealand Society for Geriatric Medicine, Victorian Division	
Ms Jill Linklater, Policy and Advocacy Officer Ms Pamela Hore, Member, Consumer Reference Group Ms Margaret Brown, Adjunct Research Fellow, Hawke Research Institute, University of South Australia	Alzheimer's Australia Vic	
Mr Robert Shepherd, Barrister Ms Jacqueline Stone, Manager, Legal and Government Relations	The Victorian Bar	

# Public hearing, 14 December 2009 Room G2, 55 St Andrews Place, East Melbourne

Witness(es)	Organisation	
Ms Lucy Cordone, General Counsel Ms Paula Chatfield, Legal Counsel	St Vincent's Hospital	
Ms Helen Versey, Privacy Commissioner	Office of the Victorian Privacy Commissioner	
Mr Stephen Taffe, Legal Counsel Ms Sally Costar, Manager, Patient and Family Services, Caulfield Hospital	Alfred Health	
Mr Matthew Hughes, Principal Neuropsychologist, The Alfred Hospital and Acting Chair		
Mr David Stokes, Senior Manager Professional Practice, Australian Psychological Society and Neuropsychologist		
Mr Luke Delaney, Chair, 'Guidelines for the Preparation of Neuropsychological Reports for the Guardianship List of the Victorian Civil and Administrative Tribunal' Working Party and Senior Clinical Neuropsychologist, Alcohol Related Brain Injury Australian Services	Victorian Section of the College of Clinical Neuropsychologists of the Australian Psychological Society	
Ms Elizabeth Mullaly, Member of College of Clinical Neuropsychologists Course Approval Sub-Committee and Manager of Psychology Services, Caulfield Hospital		
Mr Paul Zanatta, Manager, Community Care, Policy and Small Rural Health Ms Janice Hadgraft, Manager, Residential Services	Aged & Community Care Victoria	
Ms Catherine Leslie, Legal and Policy Officer	Mental Health Legal Centre Inc	

# Public hearing, 17 December 2009 Room G2, 55 St Andrews Place, East Melbourne

Witness(es)	Organisation	
Mr Kevin Larkins, Chief Executive Officer Ms Jenny Chapman, Chief Social Worker, Werribee Mercy Hospital	Palliative Care Victoria	
Ms Robyn Mills, Acting Director of Civil Law Services		
Ms Sarnia Birch, Acting Deputy Manager, Regional Offices	Victoria Legal Aid	
Ms Toni Higgins, Specialist Mental Health, Human Rights and Civil Law		
Mr Ian Gilbert, Director, Retail Policy Ms Diane Tate, Director, Financial Services, Corporations, Community Policy	Australian Bankers' Association Inc	
Mr Stelvio Vido, Executive General Manager, Strategic and Support Services Ms Nikki Isaks, Senior Social Worker Ms Leonie Schween, Manager, Legal and Privacy	Royal District Nursing Service	
Ms Gill Pierce, Manager, Policy and Research Ms Penny Paul, Coordinator, Carer Consultations Ms Marianne Dalton, Parent carer of an adult with a capacity disability	Carers Victoria	
Mr David Goldberg, Solicitor and Senior Adviser Ms Madelene Fox, Policy Assistant	Australian Medical Association (Vic) Ltd	

## Public hearing, 30 March 2010 Legislative Council Committee Room, Parliament House Spring Street, East Melbourne

Witness(es)	Organisation
Mr Adrian Cohen	
Mr Tom Worsnop, Chair Ms Kerry Stringer, Community Partnerships Manager, Summer Foundation	Victorian Coalition of Acquired Brain Injury Service Providers
Professor Peteris Darzins, Professor of Geriatric Medicine, Monash University, and Director of Geriatric Medicine, Eastern Health	
Ms Lillian Jeter, Executive Director	Elder Abuse Prevention Association

#### Seniors' Forum, 30 March 2010 Legislative Council Committee Room, Parliament House Spring Street, East Melbourne

Witness(es)	Organisation
Mr John Nicolaou	Reservoir Greek Elderly Citizens Club Inc and Northern Federation of Ethnic Senior Citizens
Ms Dorothy Trezise, Director	United Way Geelong
Ms Margaret Scott Simmons	
Ms Val Johnstone	
Ms Pat Long	
Ms Dorothy Davies	
Ms L Richmond	
Mr D Constantine	
Ms Suzanne Birch	
Ms Jean Thomas	
Ms Janice Reynolds	
Ms Janet Horn	
Mr John Hogan	
Ms Pat Whaley	
Mr Keith Whaley	
Mrs Joyce Jeffs	
Mr Gary Haley	
Ms Margot Fitzpatrick, Policy Officer	Council on the Ageing Victoria

# Culturally and Linguistically Diverse Communities Forum 30 March 2010 Legislative Council Committee Room, Parliament House Spring Street, East Melbourne

Witness(es)	Organisation
Ms Jaklina Michael, Cultural Liaison Coordinator	Royal District Nursing Service
Ms Eva Wakim	Victoria Arabic Social Services
Ms Rosemarie Draper	New Hope Foundation
Ms Marion Lau, Deputy Chair	Ethnic Communities' Council of Victoria
Mr Ross Barnett, Director	Ethnic Communities' Council of Victoria
Ms Eve Poludniak	Spectrum Migrant Resource Centre
Ms Naderah Edwards	Spectrum Migrant Resource Centre
Mr Santosh Kumar, President	Northern Region Indian Seniors Association of Victoria
Ms Elizabeth Ignys, Manager, Advocacy	Action on Disability in Ethnic Communities
Mr Manuel Asuncion, Vice-Chair	Filipino Community Council of Victoria Inc
Ms Virginia Landrito-Balanon, Community Aged-Care Packages Case Manager	Filipino Community Council of Victoria Inc
Ms Poppy Hearn, Senior Social Worker	Australian Greek Welfare Society

# **Appendix C: List of events attended**

Organisation	Event	Date
Victoria Legal Aid	Legal issues for an ageing population seminar	21 August 2009
Law Institute of Victoria, Administrative Law & Human Rights and Elder Law Sections	Capacity law reform roundtable	29 October 2009
International Federation on Ageing	10th global conference: Climate for change: Ageing into the future	3-6 May 2010

# Appendix D: Persons who may witness affidavits and statutory declarations

# List of persons who may witness affidavits 1293

- a) any judge or the associate to any judge
- b) an Associate Judge of the Supreme Court or the associate to such Associate Judge
- ba) an associate judge of the County Court or the associate to such associate judge
- c) a justice of the peace or a bail justice
- d) the prothonotary or a deputy prothonotary of the Supreme Court, the registrar or a deputy registrar of the County Court, the principal registrar of the Magistrates' Court or a registrar or deputy registrar of the Magistrates' Court
- da) the registrar of probates or an assistant registrar of probates
- db) a senior member or ordinary member of the Victorian Civil and Administrative Tribunal who, immediately before the commencement of section 8.2.1 of the *Legal Profession Act 2004* (Vic), was the registrar or a deputy registrar of the Legal Profession Tribunal
- e) a member or former member of either House of the Parliament of Victoria
- ea) a member or former member of either House of the Parliament of the Commonwealth
- f) a public notary
- g) a legal practitioner
- ga) a member of the police force of or above the rank of sergeant or for the time being in charge of a police station
- gb) a person employed under Part 3 of the *Public Administration Act 2004* (Vic) with a classification that is prescribed as a classification to which this section applies
- gc) a senior officer of a Council as defined in the *Local Government Act 1989* (Vic)
- gd) a person registered as a patent attorney under Chapter 20 of the *Patents Act* 1990 (Cth)
- ge) a fellow of the Institute of Legal Executives (Victoria)
- h) any officer or person empowered authorised or permitted by or under any Act of Parliament to take affidavits in relation to the matter in question or in the particular part of Victoria in which the affidavit is sworn and taken.

\_

<sup>&</sup>lt;sup>1293</sup> Evidence (Miscellaneous Provisions) Act 1958 (Vic) s 123C(1)

# List of persons who may witness statutory declarations 1294

- a) a justice of the peace or a bail justice
- b) a public notary
- c) an Australian lawyer (within the meaning of the *Legal Profession Act 2004* (Vic))
- d) a clerk to an Australian lawyer
- e) the prothonotary or a deputy prothonotary of the Supreme Court, the registrar or a deputy registrar of the County Court, the principal registrar of the Magistrates' Court or a registrar or deputy registrar of the Magistrates' Court
- f) the registrar of probates or an assistant registrar of probates
- g) the associate to a judge of the Supreme Court or of the County Court
- h) the associate of an Associate Judge of the Supreme Court or of an associate judge of the County Court
- i) a person registered as a patent attorney under Chapter 20 of the *Patents Act* 1990 (Cth)
- i) a member of the police force
- k) the sheriff or a deputy sheriff
- 1) a member or former member of either House of the Parliament of Victoria
- m) a member or former member of either House of the Parliament of the Commonwealth
- n) a councillor of a municipality
- o) a senior officer of a Council as defined in the *Local Government Act 1989* (Vic)
- p) a medical practitioner registered under the *Health Professions Registration Act* 2005 (Vic)
- q) a dentist registered under the *Health Professions Registration Act* 2005 (Vic)
- r) a veterinary practitioner
- s) a pharmacist
- t) a principal in the teaching service

Evidence (Miscellaneous Provisions) Act 1958 (Vic) s 107A(1)

- u) the manager of an authorised deposit-taking institution
- v) a member of the Institute of Chartered Accountants in Australia or the Australian Society of Accountants or the National Institute of Accountants
- w) the secretary of a building society
- x) a minister of religion authorised to celebrate marriages
- y) a person employed under Part 3 of the *Public Administration Act 2004* (Vic) with a classification that is prescribed as a classification to which this section applies or who holds office in a statutory authority with such a classification
- z) a fellow of the Institute of Legal Executives (Victoria).

# **Bibliography**

#### Articles, books and reports

Access Economics Pty Limited, *Dementia estimates and projections: Australian states and territories*, report for Alzheimer's Australia (2005).

Access Economics Pty Limited, *Making choices: Future dementia care: projections, problems and preferences*, report for Alzheimer's Australia (2009).

Advance Directives Review Committee, *Planning ahead: Your health, your money, your life.* First report of the Review of South Australia's Advance Directives: Proposed changes to law and policy (2008).

Advance Directives Review Committee, *Planning ahead: Your health, your money, your life.* Second report of the Review of South Australia's Advance Directives: Stage 2 - Proposals for implementation and communication strategies (2008).

Age Concern New Zealand, Elder abuse and enduring power of attorney: A special report from the Age Concern New Zealand Elder Abuse and Neglect Database covering the period 1 July 2002 to 31 December 2003 (2004).

Aitken, Ken, 'Enduring powers and wills' (1999) 73(12) Law Institute Journal 94.

Alberta Law Reform Institute, *Enduring powers of attorney: Safeguards against abuse* Final report no. 88 (2003).

American Bar Association Commission on Law and Ageing and American Psychological Association, Assessment of older adults with diminished capacity: A handbook for lawyers (2005).

Ames, Owen and Harley, John, 'Elder abuse: Being part of the solution, not part of the problem' (2001) 23(3) Law Society Bulletin 34.

Attorney General's Department, New South Wales, Are the rights of people whose capacity is in question being adequately promoted and protected? A discussion paper (2006).

Attorney General's Department, New South Wales, Capacity toolkit: Information for government and community workers, professionals, families and carers in New South Wales (2008).

Australian Institute of Health and Welfare, *Dementia in Australia: National data analysis and development* AIHW cat. no. AGE 53 (2007).

Australian Law Reform Commission, Community law reform for the Australian Capital Territory: Third report - Enduring powers of attorney Report no 47 (1988).

The Australian Psychological Society Ltd, Guidelines for the preparation of neuropsychological reports for the Guardianship List of the Victorian Civil and Administrative Tribunal (VCAT) (2009).

Bennett, H and Hallen, P, 'Guardianship and financial management legislation: What doctors in aged care need to know' (2005) 35(8) *Internal Medicine Journal* 482.

Brown, Margaret and Jarrad, Suzanne, 'Putting "the powers" in place: Barriers for people with memory loss in planning for the future' (2008) 15(4) *Journal of Law and Medicine* 530.

Carney, Terry, 'Aged capacity and substitute decision-making in Australia and Japan' (2003/2004) *LawASIA Journal* 1.

Carney, Terry, 'Judging the competence of older people: An alternative?' (1995) 15 Ageing & Society 515.

Carney, Terry and Keyzer, Patrick, 'Planning for the future: Arrangements for the assistance of people planning for the future of people with impaired capacity' (2007) 7(2) *Queensland University of Technology Law and Justice Journal* 255.

Collier, Berna, Coyne, Chris and Sullivan, Karen (eds), *Mental capacity: Powers of attorney and advance health directives* (2005) The Federation Press.

Collier, Berna and Lindsay, Shannon, *Powers of attorney in Australia and New Zealand* (1992) The Federation Press.

Commonwealth of Australia, *Australia to 2050: Future challenges* Intergenerational report 2010, circulated by the Hon. Wayne Swan MP, Treasurer of the Commonwealth of Australia (2010).

Council on the Ageing Victoria, *Annual report 2009* (2009).

Creyke, Robin, 'Enduring powers of attorney: Cinderella story of the 80s' (1991) 21 Western Australian Law Review 122.

Darzins, Peteris, Molloy, D William and Strang, David (eds), Who can decide? The six step capacity assessment process (2000) Memory Australia Press.

Davis, David, 'Protecting elderly victims of economic crime' (2007) 81(9) Law Institute Journal 46.

Dawson, John and Kämpf, Annegret, 'Incapacity principles in mental health laws in Europe' (2006) 12(3) *Psychology, Public Policy, and Law* 310.

Delany, Clare, 'Decision-making and consent' in Kennedy, Rosemary (ed), *Allied health professionals and the law* (2008) The Federation Press.

Department for Constitutional Affairs, United Kingdom, *Mental Capacity Act 2005 Code of practice*, The Stationery Office (2007).

Department of Human Services, Victoria, Review of the Mental Health Act 1986: Government response to the Community consultation report - July 2009 (2009).

Department of Human Services, Victoria, With respect to age - 2009: Victorian Government practice guidelines for health services and community agencies for the prevention of elder abuse (2009).

Department of Lands, New South Wales, Review of the Powers of Attorney Act 2003 Issue paper (2009).

Department of Planning and Community Development, Victoria, *Ageing in Victoria* Discussion paper (2008).

Devereux, John and Parker, Malcolm, 'Competency issues for young persons and older persons' in Freckelton, Ian and Peterson, Kerry (eds), *Disputes and dilemmas in health law* (2006) The Federation Press.

Edwards, Susan and Fontana, Antonia, *Legal information needs of older people*, Law and Justice Foundation of New South Wales (2004).

Etchells, Edward, Darzins, Peteris, Silberfeld, Michel, Singer, Peter A, McKenny, Julia, Naglie, Gary, Katz, Mark, Guyatt, Gordon H, Molloy, D William and Strang, David, 'Assessment of patient capacity to treatment' (1999) 14(1) *Journal of General Internal Medicine* 27.

Field, Sue, 'Assessing mental capacity' (2008) 86 Precedent 45.

Folstein, MF, Folstein, SE and McHugh, PR, "Mini-mental state": A practical method for grading the cognitive state of patients for the clinician (1975) 12(3) *Journal of Psychiatric Research* 189.

Gibson, Frances, Financial & consumer credit issues for older consumers in central Victoria, report for Loddon Campaspe Community Legal Centre (2008).

Giddings, Jeff and Thomas, Jody, 'Teaching the law as it relates to older people' (2002) 27(2) *Alternative Law Journal* 78.

Grisso, Thomas and Appelbaum, Paul S, Assessing competence to consent to treatment: A guide for physicians and other health professionals (1998) Oxford University Press.

Groves, Matthew, 'Adeptly avoiding ademption' (2010) 84(08) Law Institute Journal 36.

Hamilton, Barbara and Cockburn, Tina, 'Assessment of capacity: Disciplinary issues and potential liability' (2009) 29(2) *Proctor* 15.

House of Representatives Standing Committee on Legal and Constitutional Affairs, *Older people and the law*, The Parliament of the Commonwealth of Australia (2007).

Howe, Anna L, Cultural diversity, ageing and HACC: Trends in Victoria in the next 15 years, report for Aged Care Branch, Department of Human Services, Victoria (2006).

Kent, Rachel, 'Misuse of enduring powers of attorney' (2003) 34(3) Victoria University of Wellington Law Review 497.

Land and Property Management Authority, New South Wales, *Review of the Powers of Attorney Act 2003*, undertaken by the Land and Property Management Authority on behalf of the Hon. Tony Kelly, MLC, Minister for Lands, October 2009 (2009).

Law Commission of New Zealand, Misuse of enduring powers of attorney Report 71 (2001).

The Law Reform Commission of Hong Kong, Enduring powers of attorney (2008).

Law Reform Commission of Ireland, *Consultation paper on law and the elderly* LRC CP 23 - 2003 (2003).

The Law Society of New South Wales, When a client's capacity is in doubt: A practical guide for solicitors (2009).

Lowndes, Georgia, Darzins, Peteris, Wainer, Jo, Owada, Kei and Mihaljcic, Tijana, *Financial abuse of elders: A review of the evidence*, Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2009).

Lush, Denzil, 'Taking liberties: Enduring powers of attorney and financial abuse' (1998) *Solicitors Journal* 808.

MacNab, Jenna, 'Capacity: A practical guide for lawyers' (2008) 46(5) Law Society Journal 68

McCawley, Anne-Louise, Tilse, Cheryl, Wilson, Jill, Rosenman, Linda and Setterlund, Deborah, 'Access to assets: Older people with impaired capacity and financial abuse' (2006) 8(1) *The Journal of Adult Protection* 20.

McRae, Mathis E, 'Policing the guardians: Combating guardianship and power of attorney fraud' (1994) 63(2) *The FBI Law Enforcement Bulletin* 1.

Ministry of Justice, United Kingdom, Reviewing the Mental Capacity Act 2005: Forms, supervision and fees Response to consultation (2009).

Monro, Rosslyn, 'Elder abuse and legal remedies: Practical difficulties?' (2002) 81 *Reform* 42.

National Center on Elder Abuse, The American Public Human Services Association, *The National Elder Abuse Incidence Study* Final report, report for U.S. Department of Health and Human Services (1998).

O'Connor, Deborah and Purves, Barbara (eds), *Decision-making*, *personhood and dementia: Exploring the interface* (2009) Jessica Kingsley Publishers.

Office of Senior Victorians, Elder abuse prevention consultation paper (2005).

Office of Senior Victorians, The mismanagement & misappropriation of older people's assets: How can the financial service sector help? A 'roundtable discussion' Elder abuse prevention project (2005).

Office of Senior Victorians, *Rights. Respect. Trust.* Victorian Government Elder Abuse Prevention Strategy (2009).

Office of Senior Victorians, *Strengthening Victoria's response to elder abuse* Report of the Elder Abuse Prevention Project (2005).

Office of Senior Victorians, Victorian Government elder abuse prevention strategic implementation plan 2006-09 (2007).

Office of the Adult Guardian, Queensland, Capacity guidelines for witnesses of enduring powers of attorney (2005).

Office of the Public Advocate, Administration guide: A guide for people appointed as administrators under the Guardianship and Administration Act 1986 (2009).

Office of the Public Advocate, Annual report 2008-09 (2009).

Office of the Public Advocate, Community guardianship manual.

Office of the Public Advocate, Good guardianship: A guide for guardians appointed under the Guardianship and Administration Act (2008).

Office of the Public Advocate, *The role of the Public Advocate: Preliminary position paper*, report prepared by Carter, Barbara (2010).

Office of the Public Advocate, Supported decision-making: Options for legislative recognition, report prepared by Chesterman, John (2010).

Office of the Public Advocate, Queensland Government and Queensland Law Society, *Elder abuse: How well does the law in Queensland cope?* (2010).

Parker, Malcolm, 'Judging capacity: Paternalism and the risk-related standard' (2004) 11(4) *Journal of Law and Medicine* 482.

Purser, Kelly, Magner, Ellis S and Madison, Jeanne, 'Competency and capacity: The legal and medical interface' (2009) 16(5) *Journal of Law and Medicine* 789.

Queensland Law Reform Commission, Assisted and substituted decisions: Decision-making by and for people with a decision-making disability Report no 49, volume 1 (1996).

Queensland Law Reform Commission, *Shaping Queensland's guardianship legislation: Principles and capacity* Discussion paper (2008).

The Royal Australian College of General Practitioners, *Medical care of older persons in residential aged care facilities* (4th edition, 2006).

Schultz, JH, Rosenman, L and Rix, SE, 'International developments in social security privitization: What risk to women?' (1999) 14(1) *Journal of Cross-Cultural Gerontology* 25.

Setterlund, Deborah, Tilse, Cheryl and Wilson, Jill, 'Older people and substitute decision making legislation: Limits to informed choice' (2002) 21(3) *Australasian Journal on Ageing* 128.

Setterlund, Deborah, Tilse, Cheryl and Wilson, Jill, *Substitute decision making and older people* Trends & issues in crime and criminal justice, no. 139, Australian Institute of Criminology (1999).

Smith, Russell G, 'Fraud and financial abuse of older persons' (2000) 11(3) *Current Issues in Criminal Justice* 273.

Standing Committee on Health and Community Care, *Elder abuse in the ACT* Report number 11, Legislative Assembly for the Australian Capital Territory (2001).

Standing Committee on Social Issues, Substitute decision-making for people lacking capacity Report 43, New South Wales Legislative Council (2010).

Tilse, Cheryl, Setterlund, Deborah, Wilson, Jill and Rosenman, Linda, 'Minding the money: A growing responsibility for informal carers' (2005) 25 *Ageing & Society* 215.

Tilse, Cheryl, Wilson, Jill and Setterlund, Deborah, 'The mismanagement of the assets of older people: The concerns and actions of aged care practitioners in Queensland' (2003) 22(1) *Australasian Journal on Ageing* 9.

Tilse, Cheryl, Wilson, Jill, Setterlund, Deborah and Rosenman, Linda, 'Older people's assets: A contested site' (2005) 24(s1) *Australasian Journal on Ageing* s51.

Victorian Civil and Administrative Tribunal (VCAT), Annual report 2008/2009 (2009).

Victorian Government, 'The *Instruments (Enduring Powers of Attorney) Act* 2003 - Approved forms' (2004) G9 *Victoria Government Gazette* 437.

Victorian Law Reform Commission, Annual report 2008-09 (2009).

Victorian Law Reform Commission, *Guardianship* Information paper (2010).

Victorian Parliament Law Reform Committee, *Inquiry into alternative dispute resolution and restorative justice* (2009).

Wainer, Jo, Darzins, Peteris and Owada, Kei, *Prevalence of financial elder abuse in Victoria*. Protecting Elders' Assets Study, Monash University, report for State Trustees Limited (2010).

Willmott, Lindy and Windle, Liza, 'Witnessing EPAs: Empirical research' (2007) 27 Queensland Lawyer 238.

Western Canada Law Reform Agencies, Enduring powers of attorney: Areas for reform (2008).

#### Case law

Borthwick v Carruthers (1787) 99 ER 1300.

DM (Guardianship) [2009] VCAT 1705.

Gibbons v Wright (1954) 91 CLR 423.

Hospital Products Ltd v United States Surgical Corporation (1984) 156 CLR 41.

Malherbe v Silver [1999] VSC 176.

R v Carmel Tracy (Unreported, VMC, Magistrate Capell, 18 January 2010, 30 March 2010).

R v Halfpenny [2003] VSC 308.

Re T [1992] 4 All ER 649.

SA (Guardianship) [2008] VCAT 2345.

TQ (Guardianship) [2007] VCAT 1300.

## Legislation

#### **Victoria**

Charter of Human Rights and Responsibilities Act 2006 (Vic).

Crimes Act 1958 (Vic).

Disability Act 2006 (Vic).

Evidence (Miscellaneous Provisions) Act 1958 (Vic).

Fair Trading Act 1999 (Vic).

Guardianship and Administration Act 1986 (Vic).

Guardianship and Administration (Amendment) Act 1999 (Vic).

Health Services Act 1988 (Vic).

Information Privacy Act 2000 (Vic).

Instruments Act 1958 (Vic).

Instruments (Enduring Powers of Attorney) Act 1981 (Vic).

Instruments (Powers of Attorney) Act 1980 (Vic).

Medical Treatment Act 1988 (Vic).

Mental Health Act 1986 (Vic).

Sentencing Act 1991 (Vic).

Transfer of Land Act 1958 (Vic).

Trustee Act 1958 (Vic).

Trustee Companies Act 1984 (Vic).

Victorian Civil and Administrative Tribunal Act 1998 (Vic).

Whistleblowers Protection Act 2001 (Vic).

Wills Act 1997 (Vic).

#### Commonwealth

Commonwealth of Australia Constitution Act 1900.

Mutual Recognition Act 1992 (Cth).

Social Security (Administration) Act 1999 (Cth).

#### **Australian Capital Territory**

Land Titles Act 1925 (ACT).

Powers of Attorney Act 2006 (ACT).

Registration of Deeds Act 1957 (ACT).

#### **New South Wales**

Guardianship Act 1987 (NSW).

Powers of Attorney Act 2003 (NSW).

#### **Northern Territory**

Powers of Attorney Act (NT).

Powers of Attorney Regulations (NT).

#### Queensland

Guardianship and Administration Act 2000 (Qld).

Land Title Act 1994 (Qld).

Powers of Attorney Act 1998 (Qld).

#### South Australia

Powers of Attorney and Agency Act 1984 (SA).

Real Property Act 1886 (SA).

#### **Tasmania**

Powers of Attorney Act 2000 (Tas).

#### Western Australia

Guardianship and Administration Act 1990 (WA).

Property Law Act 1969 (WA).

#### Canada

Adult Guardianship and Trusteeship Act, SA 2008, c A-4.2.

Criminal Code, RSC 1985, c C-46.

Representation Agreement Act, RSBC 1996, c 405.

#### **Hong Kong**

Enduring Powers of Attorney Ordinance (Hong Kong) LN 365 of 1997.

#### Ireland

Powers of Attorney Act 1996 (Ireland).

#### **United Kingdom**

The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian (Amendment) Regulations 2009 (UK).

The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian Regulations 2007 (UK).

Mental Capacity Act 2005 (UK) c 9.

#### **United States of America**

NY General Obligations Law § 5-1509 (2009).

#### **Treaties**

Convention on the Rights of Persons with Disabilities, opened for signature 30 March 2007, 993 UNTS 3 (entered into force 3 May 2008).

*International Covenant on Civil and Political Rights*, opened for signature 16 December 1966, 999 UNTS 171 (entered into force 23 March 1976).

International Covenant on Economic, Social and Cultural Rights, opened for signature 16 December 1966, 993 UNTS 3 (entered into force 3 January 1976).

#### Other sources

ACT Government, Enduring power of attorney under the Powers of Attorney Act 2006 (2009).

Email from Adamson, Lauren, Manager and Principal Solicitor, Seniors Rights Legal Clinic, Seniors Rights Victoria, to Executive Officer, Victorian Parliament Law Reform Committee, 20 May 2010.

Aged Rights Advocacy Service Inc., submission 38 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Alzheimer's Association Queensland, *Common questions*, <a href="http://www.alzheimersonline.org/facts/faq.php">http://www.alzheimersonline.org/facts/faq.php</a> viewed 29 June 2010.

Alzheimer's Australia, submission 55 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Australian Bureau of Statistics, 1301.0 - Year Book Australia, 2009-10 (2010).

Australian Bureau of Statistics, 4102.0, Australian Social Trends, Data Cube - Population Table 1 Population, National Summary, 1998-2008 (2009).

Australian Bureau of Statistics, 4102.0, Australian Social Trends, Data Cube - Population Table 2.2 Population, Vic. Summary, 1998-2008 (2009).

Australian Bureau of Statistics, 4102.0 - Australian Social Trends, 2005 Housing arrangements: Housing for older Australians (2005).

Australian Bureau of Statistics, 4102.0 - Australian Social Trends, 2009 Household debt (2009).

Australian Bureau of Statistics, 4430.0.55.001 Disability, ageing and carers, Australia: Disability and long term health conditions Table 1: All persons, disability rates by age and sex (2004).

Australian Health Practitioner Regulation Agency, *Registers of practitioners*, <a href="http://www.ahpra.gov.au/en/Registration/Registers-of-Practitioners.aspx">http://www.ahpra.gov.au/en/Registration/Registers-of-Practitioners.aspx</a> viewed 14 July 2010

Australian Institute of Criminology, submission 40 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Letter from Bagdonavicius, Pauline, Public Advocate, Office of the Public Advocate, Western Australia, to Chair, Victorian Parliament Law Reform Committee, 19 August 2009.

Letter from Billings, John, Deputy President, Guardianship List, Victorian Civil and Administrative Tribunal (VCAT), to Executive Officer, Victorian Parliament Law Reform Committee, 29 October 2009.

Carers Australia, submission 120 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2007).

Carers Victoria, *Guardianship submission*, submission to the Victorian Law Reform Commission, Review of the *Guardianship and Administration Act* 1986 (2009).

Caxton Legal Centre Inc., submission 112 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2007).

Chell, Roy, Business Manager, Centrelink, *Transcript of evidence for the Inquiry into older people and the law*, House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Canberra, 17 August 2007.

Email from Chesterman, John, Manager, Policy and Education, Office of the Public Advocate, to Executive Officer, Victorian Parliament Law Reform Committee, 31 December 2009.

Email from Chesterman, John, Manager, Policy and Education, Office of the Public Advocate, to Executive Officer, Victorian Parliament Law Reform Committee, 3 June 2010.

Council on the Ageing SA, submission 77 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Email from Craig, Alistair, Senior Corporate Lawyer, State Trustees Limited, to Executive Officer, Victorian Parliament Law Reform Committee, 30 June 2010.

Darzins, Peteris, Assessment of decision-making capacity: Key concept summary, supplementary evidence received 30 March 2010.

Dementia Collaborative Research Centres, *Cognition assessment measures*, <a href="http://www.dementia-assessment.com.au/cognitive/index.html">http://www.dementia-assessment.com.au/cognitive/index.html</a> viewed 29 June 2010.

Department of Health, Victoria, Information sheet: Mental Health Act 1986 Review - February 2010 (2010).

Department of Human Services, Victoria, *With respect to age 2009 - New guide and information sessions*, <a href="http://www.dhs.vic.gov.au/rrhacs/ph\_bulletin/with-respect-to-age/nocache">http://www.dhs.vic.gov.au/rrhacs/ph\_bulletin/with-respect-to-age/nocache</a> viewed 5 July 2010.

Department of Justice, Victoria, *Information for attorneys*, on Office of the Public Advocate, <a href="http://www.publicadvocate.vic.gov.au/file/file/Powerofattorney/">http://www.publicadvocate.vic.gov.au/file/file/Powerofattorney/</a> Information\_for\_Attorneys\_from\_the\_Department\_of\_Justice.pdf> viewed 29 April 2010.

Department of Justice & Attorney General, New South Wales, *CertValid user guide* Document version 1.0 (2010).

Department of Justice and Attorney-General, Queensland Government, *Enduring power of attorney: Long form* Form 2, Queensland Powers of Attorney Act 1998 (Section 44(1)) (2002).

Department of Justice and Attorney-General, Queensland Government, *Enduring power of attorney: Short form* Form 3, Queensland *Powers of Attorney Act 1998* (Section 44(1)) (2003).

Department of Lands, New South Wales, Powers of attorney in New South Wales (2005).

Dickson, Glenn, 'The enduring power of attorney form: Should it be changed?' (Paper presented at the Queensland Law Society Symposium 2007, Brisbane, 2-3 March 2007).

Directgov, *How the Office of the Public Guardian works*, <a href="http://www.direct.gov.uk/en/Governmentcitizensandrights/Mentalcapacityandthelaw/Makingarrangementsincaseyoulosementalcapacity/DG\_185924">http://www.direct.gov.uk/en/Governmentcitizensandrights/Mentalcapacityandthelaw/Makingarrangementsincaseyoulosementalcapacity/DG\_185924</a> viewed 14 July 2010.

Letter from Dodds, Imelda, Acting Chief Executive Officer, NSW Trustee & Guardian, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010.

Explanatory memorandum to The Lasting Powers of Attorney, Enduring Powers of Attorney and Public Guardian (Amendment) Regulations 2009 (UK) No 1884.

Letter from Fielding, Richard, Principal Examiner, Powers of Attorney/General Law, Land Titles Office, Tasmania, to Executive Officer, Victorian Parliament Law Reform Committee, 19 August 2009.

Gardner, Julian, Public Advocate, Office of the Public Advocate, submission 70 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Graycar, Adam and James, Marianne, 'Crime and older Australians: Understanding and responding to crime and older people' (Paper presented at the Family Futures: Issues in Research and Policy - 7th Australian Institute of Family Studies Conference, Sydney, 24-26 July 2000).

Human Rights and Equal Opportunity Commission, submission 92 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Email from Huxley, Lynton, Health, Community and Disability Services Ministerial Council Secretariat, to Administration Officer, Victorian Parliament Law Reform Committee, 22 July 2010.

Letter from John, Martin, Public Guardian and Chief Executive, Office of the Public Guardian, United Kingdom, to Chair, Victorian Parliament Law Reform Committee, 27 May 2010

Law Institute of Victoria, Law Institute continuing professional development rules 2008 (2008).

Legal Services Board, Continuing professional development rules 2008 (2008).

Letter from Locke, Max, Registrar of Titles and Registrar of Water Allocations, Department of Environment and Resource Management, Queensland, to Chair, Victorian Parliament Law Reform Committee, 23 February 2010.

Letter from Macken, Mary, President, The Law Society of New South Wales, to Chair, Victorian Parliament Law Reform Committee, 25 May 2010.

Letter from Maloney, Peter, Chief Executive Officer, Public Trustee, Tasmania, to Chair, Victorian Parliament Law Reform Committee, 3 November 2009.

Letter from Marshall, Brian, Senior Manager, Registration and Client Services, Office of Regulatory Services, Australian Capital Territory, to Chair, Victorian Parliament Law Reform Committee, 16 February 2010.

Letter from McGarvie, Michael, Chief Executive Officer, Legal Services Board, to Chair, Victorian Parliament Law Reform Committee, 1 March 2010.

Mountjouris, Barbara, Director, Office of Senior Victorians, 'Empowerment, partnership and collaboration to prevent elder abuse' (Paper presented at the 10th International Federation on Ageing Conference, Melbourne, 3-6 May 2010).

NSW Trustee & Guardian, submission to the Land and Property Management Authority, New South Wales, Review of the Powers of Attorney Act 2003 (2009).

Office of the Public Advocate, submission 70 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Office of the Public Advocate, Queensland Government, submission 76 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Office of the Public Guardian, United Kingdom, Lasting power of attorney - property and financial affairs (2009).

Office of the Public Guardian, United Kingdom, Lasting power of attorney for health and welfare (2009).

Office of the Public Guardian, United Kingdom, LPA109 - Office of the Public Guardian Registers (2009).

Office of the Public Guardian, United Kingdom, *Your role and duties as an attorney*, <a href="http://www.publicguardian.gov.uk/decisions/role-duties-attorney.htm">http://www.publicguardian.gov.uk/decisions/role-duties-attorney.htm</a> viewed 02 July 2010.

Letter from Paul, Penny, Coordinator, Carer Consultations, Carers Victoria, to Executive Officer, Victorian Parliament Law Reform Committee, 18 January 2010.

Public Advocate of the ACT, The power to choose: Your guide to completing an enduring power of attorney (2009).

Letter from Reddick, Dale, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, to Executive Officer, Victorian Parliament Law Reform Committee, 1 October 2009.

Email from Reddick, Dale, Advocacy and Support Worker, Rights Advocacy and Support Program, Gippsland Community Legal Service, to Administrative Officer, Victorian Parliament Law Reform Committee, 7 June 2010.

Letter from Roberts, Bruce, Registrar of Titles, Landgate, Western Australia, to Chair, Victorian Parliament Law Reform Committee, 21 August 2009.

Letter from Schween, Leonie, Legal & Privacy Officer, Royal District Nursing Service, to Executive Officer, Victorian Parliament Law Reform Committee, 14 January 2010.

Seniors Rights Victoria, *Recognising the rights of older Australians*, submission to the National Human Rights Consultation (2009).

Seniors Rights Victoria, *Services*, <a href="http://www.seniorsrights.org.au/index.php?">http://www.seniorsrights.org.au/index.php?</a> option=com\_content&view=article&id=27&Itemid=7> viewed 7 June 2010.

Shepherd, Robert, The Victorian Bar, Further information on the legal authority in terms of the common law scope of a power of attorney, supplementary evidence received 12 November 2009.

Letter from Shoyer, Peter, Registrar-General, Land Titles Office, Department of Justice, Northern Territory, to Chair, Victorian Parliament Law Reform Committee, 26 February 2010.

Standards Australia, *Electronic data interchange for administration, commerce and transport (EDIFACT) - Forms design - Basic layout* AS 3805-1991.

Standing Committee of Attorneys-General, Communiqué - November 2008 (2008).

Standing Committee of Attorneys-General, *Current projects and achievements: Spent convictions*, <a href="http://www.scag.gov.au/lawlink/SCAG/ll\_scag.nsf/pages/scag\_achievements#Spent%20convictions">http://www.scag.gov.au/lawlink/SCAG/ll\_scag.nsf/pages/scag\_achievements#Spent%20convictions</a>> viewed 29 April 2010.

State Government of Victoria, submission 121 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2007).

State Trustees Limited, submission 88 to the House of Representatives Standing Committee on Legal and Constitutional Affairs, Parliament of Australia, Inquiry into older people and the law (2006).

Email from Taylor, Andrew, Public Trustee for the ACT, to Executive Officer, Victorian Parliament Law Reform Committee, 16 February 2010.

Hulls, Attorney-General).

Victoria, *Parliamentary debates*, Legislative Assembly, 28 August 2003, 215 (The Hon. Rob

Victoria Legal Aid and Office of the Public Advocate, *Take control: A kit for making powers of attorney and guardianship* 10th edition (2007).

Victoria Police, Code of practice for the investigation of family violence (2004).

Victorian Civil and Administrative Tribunal (VCAT), *Guardianship and Admin*, <a href="http://www.vcat.vic.gov.au/CA256DBB0022825D/page/Guardianship+and+Admin?OpenDocument&1=60-Guardianship+and+Admin~&2=~&3=~> viewed 31 May 2010.">viewed 31 May 2010.</a>

Victorian Civil and Administrative Tribunal (VCAT), VCAT information sessions for newly appointed guardians and administrators: Dates for 2010, <a href="http://www.vcat.vic.gov.au/CA256902000FE154/Lookup/guardianship/\$file/information\_sessions-for\_guardians\_and\_administrators\_2010.pdf">http://www.vcat.vic.gov.au/CA256902000FE154/Lookup/guardianship/\$file/information\_sessions-for\_guardians\_and\_administrators\_2010.pdf</a> viewed 29 April 2010.

Victorian Registry of Births, Deaths and Marriages, *Access Policy*, Department of Justice, Victoria (2007).

Walsh, David, 'Laws and lawyers: Should we be doing more for elders?' (Paper presented at the Is there justice for older Australians? legal seminar, Melbourne, 25 October 2006).

Letter from Watkins, Warwick, Registrar General, Land and Property Management Authority, New South Wales, to Chair, Victorian Parliament Law Reform Committee, 8 September 2009.

Zombor, Rachel, Clinical neuropsychologist, 'The role of neuropsychology in the assessment of decision making capacity' (Paper presented at the Australian Guardianship and Administration Council: Social inclusion: The future of ageing, disability and substituted decision-making conference, Brisbane, 19-20 March 2009).

# **Extract from the minutes of proceedings**

## **Thursday 12 August 2010**

The minutes of the proceedings of the Committee show the following division which took place during the consideration of the draft report.

#### Motion

That the text under the heading '6.4.5 Which representative's decisions take precedence?' and recommendation 47 stand part of the report.

Moved: Colin Brooks MP Seconded: Martin Foley MP

The Committee divided on the question:

Ayes: 4 Noes: 3

Colin Brook MPRobert Clark MPLuke Donnellan MPJan Kronberg MLCMartin Foley MPHeidi Victoria MP

Johan Scheffer MLC

Carried.